

Protection and conservation of Posidonia Oceanica meadows







Originally in French, written in the context of the RAMOGE Agreement between France, Italy and Monaco, funded by RAMOGE and the Provence-Alpes-Côte d'Azur Regional Council and coordinated by GIS Posidonie.

This is the English-language version, translated jointly by RAC/SPA and Okianos, and brought out by RAC/SPA. Note: The designations employed and the presentation of the material in this document do not imply the expression of any opinion whatsoever on the part of UNEP/MAP, RAC/SPA or RAMOGE concerning the legal status of any State, Territory, city or area, or of its authorities, or concerning the delimitation of their frontiers or boundaries.

THE RAMOGE AGREEMENT

THE RAMOGE AGREEMENT, signed in 1976, is the instrument endowed to the Governments of France, Monaco and Italy for the prevention and combat against pollution in the marine environment and the littoral of the PACA Region, the Principality of Monaco and the Liguria.

THE RAMOGE AGREEMENT facilitates scientific, technical, legal and administrative cooperation for the integrated management of the littoral as well as for public awareness for the respect of the environment.

RAC/SPA

RAC/SPA (Regional Activity Centre for Specially Protected Areas) was established in Tunis, in 1985, by decision of the Contracting Parties to the Barcelona Convention.

THE RAC/SPA's mission is to provide assistance to Mediterranean countries in the implementation of their commitments under the Protocol concerning Specially Protected Areas and Biological Diversity in the Mediterranean (SPA/BD Protocol, 1995), especially in regard to developing and promoting marine and coastal protected areas and reducing the loss of biodiversity in the Mediterranean region.

This publication should be cited as follows:

Boudouresque C. F., Bernard G., Bonhomme P., Charbonnel E., Diviacco G., Meinesz A., Pergent G., Pergent-Martini C., Ruitton S., Tunesi L., 2012. Protection and conservation of *Posidonia oceanica* meadows. RAMOGE and RAC/SPA publisher, Tunis: 1-202.

ISBN N° 2-905540-31-1 (RAC/SPA and GIS Posidonie publ., Marseille)

Chapters should be cited as follow, e.g.:

Pergent-Martini C., Coppo S., Pulcini M., Cinquepalmi F., 2012. Chapter 5. Policies applying to *Posidonia* oceanica meadow. In: Protection and conservation of *Posidonia oceanica* meadows. Boudouresque C.F., Bernard G., Bonhomme P., Charbonnel E., Diviacco G., Meinesz A., Pergent G., Pergent-Martini C., Ruitton S., Tunesi L. (eds.), RAMOGE and RAC/SPA publisher, Tunis: 48-60

Protection and conservation of **Posidonia Oceanica**

meadows

Boudouresque C.F., Bernard G., Bonhomme P., Charbonnel E., Diviacco G., Meinesz A., Pergent G., Pergent-Martini C., Ruitton S., Tunesi L.

2012

Authorized English translation of "Boudouresque *et al.*, 2006. Préservation et conservation des herbiers à *Posidonia oceanica*. Ramoge publisher, Monaco: 1-202 (ISBN N° 2-905540-30-3 – RAMOGE and GIS Posidonie publisher, Marseille)"

This book can be downloaded at : www.ramoge.org and www.rac-spa.org

The authors:

Charles François Boudouresque (1, 2), Guillaume Bernard (2), Patrick Bonhomme (2), Eric Charbonnel (3), Giovanni Diviacco (4), Alexandre Meinesz (5), Gérard Pergent (6), Christine Pergent-Martini (6), Sandrine Ruitton (2) and Leonardo Tunesi (7).

(1) Mediterranean Institute of Oceanography (MIO), UMR 7294, Aix-Marseille University, Campus de Luminy, 13288 Marseille cedex 9, France.

(2) GIS Posidonie, campus de Luminy, 13288 Marseille cedex 9, France.

(3) Syndicat mixte Parc marin de la Côte Bleue, Observatoire, plage du Rouet, 31 avenue Jean-Bart, BP 42, 13960 Sausset-les-Pins, France.

(4) Servizio Parchi e Aree Protette, Regione Liguria, Via D'Annunzio 113, 16121 Genova, Italia.

(5) Laboratoire ECOMERS, Université de Nice-Sophia Antipolis, 06108 Nice Cedex 2, France.

(6) Equipe Ecosystèmes Littoraux, Faculté des Sciences et techniques, Université de Corse Pasquale Paoli, 20250 Corte, France.

(7) Istituto Superiore per la Protezione e la Ricerca Ambientale (ISPRA), Via di Casalotti, 300, 00166 Roma, Italia.

Translated by Anne Murray, reviewed by Cecilia Lopez y Royo.

Scientific revision of the translation by: Habib Langar, Souha El Asmi, Cyrine Bouafif and Chedly Rais.

This book is a collective work. It was coordinated by Charles-François Boudouresque. All co-authors contributed to writing all the chapters. However, the wording has been specifically coordinated by the following authors:

- **Chapter 1.** Introduction : Charles-François Boudouresque.
- Chapter 2. Posidonia oceanica meadows: Charles-François Boudouresque.
- Chapter 3. The role of Posidonia oceanica meadows: Gérard Pergent.
- **Chapter 4.** The causes of *Posidonia oceanica* meadows regression: Leonardo Tunesi and Charles-François Boudouresque.
- Chapter 5. Policies applying to *Posidonia oceanica* meadows: Christine Pergent-Martini, Stefano Coppo, Marina Pulcini and Federico Cinquepalmi.
- **Chapter 6.** Dead *Posidonia oceanica* leaves, beaches and sand replenishment: Giovanni Diviacco, Leonardo Tunesi and Charles-François Boudouresque.
- **Chapter 7.** The *Posidonia oceanica* meadow and management of facilities on the maritime public domain: Charles-François Boudouresque, Guillaume Bernard, Patrick Bonhomme, Eric Charbonnel and Giovanni Diviacco.
- **Chapter 8.** The *Posidonia oceanica* meadow and mooring: Giovanni Diviacco and Charles-François Boudouresque.
- **Chapter 9.** The *Posidonia oceanica* meadow and the marking of the 300 m strip: Frédéric Bachet, Boris Daniel and Eric Charbonnel.
- Chapter 10. The *Posidonia oceanica* meadow and trawling: Eric Charbonnel and Leonardo Tunesi.
- Chapter 11. The Posidonia oceanica meadow and fish farms: Gérard Pergent.
- Chapter 12. The *Posidonia oceanica* meadow and discharge of effluents: Giovanni Diviacco.
- Chapter 13. The *Posidonia oceanica* meadow and solid waste: Charles-François Boudouresque and Patrick Bonhomme.
- **Chapter 14.** *Posidonia oceanica* meadows and the laying of cables and pipes on the seabed: Charles-François Boudouresque and Eric Charbonnel.
- Chapter 15. Can dead meadows be restored?: Charles-François Boudouresque and Alexandre Meinesz.
- Chapter 16. Methods of monitoring *Posidonia oceanica* meadows: Charles-François Boudouresque, Eric Charbonnel, Stefano Coppo, Laurence Le Direach and Sandrine Ruitton.
- **Chapter 17.** The *Posidonia oceanica* meadow and the Water Framework Directive: Pierre Boissery and Christine Pergent-Martini.
- Chapter 18. The *Posidonia oceanica* meadow. A summary: Charles-François Boudouresque.

Eight people were actively involved in writing a specific chapter and deserve special mention: Frédéric **Bachet** (Côte Bleue Marine Park – chapter 9), Pierre **Boissery** (Agence de l'Eau Rhône-Méditerranée-Corse – Chapter 17), Federico **Cinquepalmi** (Direzione Generale Protezione Natura, Ministero dell'Ambiente e della Tutela del Territorio e del Mare – Chapters 5 and 16), Stefano **Coppo** (Settore Ecosistema Costiero, Regione Liguria – Chapters 5 and 16), Boris **Daniel** (Côte Bleue Marine Park; chapter 9), Laurence **Le Direach** (GIS Posidonie; chapter 16), Marina **Pulcini** (Direzione Generale Protezione Natura, Ministero dell'Ambiente e della Tutela del Territorio e del Mare – Chapters 5 and 16).

The final layout of the work was done by Sandrine Ruitton.

Members of the **RAMOGE Agreement**'s **"Biodiversity Conservation"** work group actively contributed during a number of meetings to defining the format and contents of the work and the major lines of the message it conveys.

This book should be considered as a "work of authors" This means that, beyond the scientific data actually published, the authors have used their personal experiences and beliefs (what is known today the "expert opinion"). Regardless of the rules and regulations that may govern the management of *Posidonia oceanica* meadows in various countries, the recommendations that are made in the present work are the responsibility of the writers alone. They are not therefore the responsibility of neither the RAMOGE Agreement nor the RAC/SPA (see inserts), or the institutions that helped produce or fund this work (Provence-Alpes-Côte d'Azur Region), or those people who gave information to the writers, and have thus no statutory value.

THANKS

The writers would like to thank the following people for their help in producing this work (providing information and documents, re-reading certain chapters, critical corrections etc.): Xavier Archimbault (RAMOGE Agreement), Patrick Aubel (Port-Cros National Park), Marta Azzolin (RAMOGE Agreement), Richard Barety (Port-Cros National Park), Daniel Barbaroux (deputy mayor of Hyères), Mary-Christine Bertrandy (Bouche-du-Rhône Coastal Water Quality Department), Dominique Bresson (Préfecture Maritime de Méditerranée, France), Jacques Bruno (engineer in the Hyères Environment Department), Eric de Chavanes (DIREN PACA), Stefano Coppo (Settore Ecosistema Costiero, Regione Liguria), Eric Coulet (Camargue National Reserve), Daniel Coves (Ifremer), Gérard Feracci (Corsican Region Nature Park), Simone Fournier (GIS Posidonie), Laurence Gaglio (RAMOGE Agreement), Frédérique Lantéri-Gimon (Hyères Environment Department), Michel Leenhardt (French Federation of Regional Nature Parks), Corine Lochet (Provence-Alpes-Côte d'Azur Regional Council), Françoise Loques (Lérins Islands Scientific Council), Elodie Martin (RAMOGE Agreement), Virginie Michel (Provence-Alpes-Côte d'Azur Regional Council), Roger Miniconi, Alexandra Nardini (RAMOGE Agreement), Béatrice Pary (Cépralmar), Michèle Perret-Boudouresque (Mediterranean Institute of Oceanography, Marseille, France), Frédéric Platini (Secretary to the RAMOGE Agreement), Nathalie Quelin (DIREN PACA), Valérie Raimondino (Provence-Alpes-Côte d'Azur Regional Council), Giulio Relini (Genoa University), Eglantine Ricard (RAMOGE Agreement), Philippe Robert (Port-Cros National Park), Emmanuelle Roques (Ifremer), Nicolas Schmitt (RAMOGE Agreement), Didier Sauzade (Ifremer, La Seyne Centre, France), Christophe Serre (Conseil Général des Alpes-Maritimes), Eric Tambutté (Centre Scientifique de Monaco) and Jean de Vaugelas (ECOMERS Laboratory, University of Nice Sophia Antipolis).

CONTENT

1.	Introduction	7
2.	Posidonia oceanica meadows	10
	2.1. Geographical distribution	10
	2.2. The plant	11
	2.3. Ecology	14
	2.4. The structure of meadows	15
	2.5. Types of meadows	17
	2.6. How the ecosystem functions	21
3.	The role of <i>Posidonia oceanica</i> meadows	25
	3.1. General remarks	25
	3.2. Ecological role in coastal systems	25
	3.3. Physical role in coastal systems	27
	3.4. Economic role	28
	3.5. Bioindicator role	30
4.	The causes of Posidonia oceanica meadows regression	32
	4.1. Covering or direct inclusion in coastal development	
	and modification of sedimentary flow	36
	4.2. Changes due to river inputs	37
	4.3. Reduction in water transparency	39
	4.4. Presence of excessive amounts of nutrients and chemical contaminants	39
	4.5. Anchoring	41
	4.6. Trawling	42
	4.7. Explosives	43
	4.8. Coastal aquaculture	44
	4.9. Laying cables and pipes	44
	4.10. Dumping	45
	4.11. Competition with introduced species	45
	4.12. Overgrazing	46
	4.13. Synergy of different causes of regression	46
	4.14. Conclusions	47
5.	Policies applying to Posidonia oceanica meadows	48
	5.1. Direct protection measures	48
	5.1.1. International conventions and European Community texts	48
	5.1.2. Policies in the countries of the RAMOGE areas	50
	5.1.3. Other policies in the Mediterranean	54
	5.2. Indirect protection measures	55
	5.2.1. Protected areas	55
	5.2.2. Fishing gear	57
	5.2.3. Impact studies	57
	5.3. Legal enforcement of these policies: examples of case law	59
6.	Dead Posidonia oceanica leaves, beaches and sand replenishment	61
	6.1. The problem	61
	6.1.1. Dead Posidonia leaves	61
	6.1.2. Use of dead Posidonia leaves	62
	6.1.3. Erosion of beaches	64
	6.2. Case studies	65
	6.2.1. Managing banquettes in Malta	65
	6.2.2. Port-Cros and Porquerolles beaches	66
	6.2.3. Almanarre beach, Hyères	67
	6.3. Recommendations	68

7.	The Posidonia oceanica meadow and management of facilities	
	on the maritime public domain	70
	7.1 The Problem	70
	7.2. Case studies	72
	7.2.1. Developing Le Mourillon beaches in Toulon	72
	7.2.2. Building Pointe-Rouge port in Marseille	73
	7.2.3. Banyuls-sur-Mer port	74
	7.2.4. Ospedaletti	75
	7.2.5. Spotorno	77
	7.3. Recommendations	78
	7.3.1. Can a meadow remain in good health in a port?	78
	7.3.2. Minimum distance between artificial rip-rap and the meadow	79
	7.3.3. Necessary precautions for a building site	80
	7.3.4. Less "harmful" solutions	82
8.	The Posidonia oceanica meadow and mooring	83
	8.1. The problem	83
	8.2. Case studies	85
	8.2.1. The Italian coast under the RAMOGE Agreement	85
	8.2.2. The French coast under the RAMOGE Agreement	86
	8.2.3. Areas outside the RAMOGE Agreement	88
	8.3. Recommendations	89
9.	The <i>Posidonia oceanica</i> meadow and the marking of the 300 m strip	92
	9.1. The problem	92
	9.2. Case study: the Côte Bleue Marine Park	92
	9.3. Recommendations	93
10	. The <i>Posidonia oceanica</i> meadow and trawling	94
	10.1. The problem	94
	10.2. History of anti-trawl reefs	94
	10.3. Recommendations	97
11.	. The <i>Posidonia oceanica</i> meadow and fish farms	99
	11.1. The problem	99
	11.2. Case studies	100
	11.3. Summary and recommendations	105
12	. The Posidonia oceanica meadow and discharge of effluents	110
	12.1. The problem	110
	12.2. Case studies	110
	12.2.1. Meadows of Genoa Region	110
	12.2.2. The Marseille outfall	111
	12.3. Recommendations	113
13	. The <i>Posidonia oceanica</i> meadow and solid waste	115
	13.1. The problem	115
	13.2. Case studies	115
	13.3. The legislative framework	117
	13.4. Recommendations	118
14	. Posidonia oceanica meadows and the laying	
	of cables and pipes on the seabed	119
	14.1. The problem	119
	14.2. Case studies	119
	14.2.1. Drinking water pipe between Hyères and the island of Porquerolles	119
	14.2.2. Water pipes between Cannes and Sainte-Marguerite Island	122
	14.2.3. Telephone cable between the continent and the island of Port-Cros	123

14.2.4. Electric cable between Corsica and Sardinia	124
14.3. Recommendations	126
14.4. Conclusions	130
15. Can dead meadows be restored?	132
15.1. The problem	132
15.2. Transplanting techniques	133
15.3. Restoring the meadows	136
15.4. A code of good conduct	139
15.5. Conclusions	140
16. Methods of monitoring Posidonia oceanica meadows	141
16.1. Introduction	141
16.2. Monitoring tools	141
16.2.1. Markings	141
16.2.2. Acoustic positioning	143
16.2.3. Measuring meadow cover and shoot density	144
16.2.4. Permanent transects	146
16.2.5. Permanent quadrats	147
16.2.6. Aerial photographs	148
16.2.7. Microscale tools	148
16.3. The main monitoring systems	151
16.3.1. RSP (the Posidonia Monitoring Network)	151
16.3.2. The Prado Bay monitoring system	152
16.3.3. The Monaco monitoring system	153
16.3.4. The Liguria Region (Italy) monitoring system	154
16.3.5. Other monitoring systems 16.4. Conclusions	155 155
17. The Posidonia oceanica meadow and the Water Framework Directive	156
17. The <i>Posidonia oceanica</i> meadow and the Water Framework Directive 17.1. Some key elements of the Water Framework Directive	156 156
 17. The Posidonia oceanica meadow and the Water Framework Directive 17.1. Some key elements of the Water Framework Directive 17.1.1. A major innovation of the WFD: targets for all aquatic environments 	156 156 156
 17. The Posidonia oceanica meadow and the Water Framework Directive 17.1. Some key elements of the Water Framework Directive 17.1.1. A major innovation of the WFD: targets for all aquatic environments 17.1.2. The WFD's analysis unit: the waterbody 	156 156 156 157
 17. The Posidonia oceanica meadow and the Water Framework Directive 17.1. Some key elements of the Water Framework Directive 17.1.1. A major innovation of the WFD: targets for all aquatic environments 17.1.2. The WFD's analysis unit: the waterbody 17.1.3. Good ecological status and coastal waterbodies 	156 156 157 157
 17. The Posidonia oceanica meadow and the Water Framework Directive 17.1. Some key elements of the Water Framework Directive 17.1.1. A major innovation of the WFD: targets for all aquatic environments 17.1.2. The WFD's analysis unit: the waterbody 17.1.3. Good ecological status and coastal waterbodies 17.2. The Posidonia meadow as a "biological quality element" 	156 156 157 157 157
 17. The Posidonia oceanica meadow and the Water Framework Directive 17.1. Some key elements of the Water Framework Directive 17.1.1. A major innovation of the WFD: targets for all aquatic environments 17.1.2. The WFD's analysis unit: the waterbody 17.1.3. Good ecological status and coastal waterbodies 17.2. The Posidonia meadow as a "biological quality element" 18. The Posidonia oceanica meadow. A summary 	156 156 157 157 157 158 161
 17. The Posidonia oceanica meadow and the Water Framework Directive 17.1. Some key elements of the Water Framework Directive 17.1.1. A major innovation of the WFD: targets for all aquatic environments 17.1.2. The WFD's analysis unit: the waterbody 17.1.3. Good ecological status and coastal waterbodies 17.2. The Posidonia meadow as a "biological quality element" 18. The Posidonia oceanica meadow. A summary 18.1. Introduction: Why be interested in Posidonia? 	 156 156 157 157 158 161 162
 17. The Posidonia oceanica meadow and the Water Framework Directive 17.1. Some key elements of the Water Framework Directive 17.1.1. A major innovation of the WFD: targets for all aquatic environments 17.1.2. The WFD's analysis unit: the waterbody 17.1.3. Good ecological status and coastal waterbodies 17.2. The Posidonia meadow as a "biological quality element" 18. The Posidonia oceanica meadow. A summary 18.1. Introduction: Why be interested in Posidonia? 18.2. Posidonia and the meadows 	 156 156 157 157 158 161 162 162
 17. The Posidonia oceanica meadow and the Water Framework Directive 17.1. Some key elements of the Water Framework Directive 17.1.1. A major innovation of the WFD: targets for all aquatic environments 17.1.2. The WFD's analysis unit: the waterbody 17.1.3. Good ecological status and coastal waterbodies 17.2. The Posidonia meadow as a "biological quality element" 18. The Posidonia oceanica meadow. A summary 18.1. Introduction: Why be interested in Posidonia? 18.2. Posidonia and the meadows 18.3. The role of the Posidonia meadows 	 156 156 157 157 158 161 162 162 162 163
 17. The Posidonia oceanica meadow and the Water Framework Directive 17.1. Some key elements of the Water Framework Directive 17.1.1. A major innovation of the WFD: targets for all aquatic environments 17.1.2. The WFD's analysis unit: the waterbody 17.1.3. Good ecological status and coastal waterbodies 17.2. The Posidonia meadow as a "biological quality element" 18. The Posidonia oceanica meadow. A summary 18.1. Introduction: Why be interested in Posidonia? 18.2. Posidonia and the meadows 18.3. The role of the Posidonia meadows 18.4. Regression of the Posidonia meadows 	 156 156 157 157 158 161 162 162 163 165
 17. The Posidonia oceanica meadow and the Water Framework Directive 17.1. Some key elements of the Water Framework Directive 17.1.1. A major innovation of the WFD: targets for all aquatic environments 17.1.2. The WFD's analysis unit: the waterbody 17.1.3. Good ecological status and coastal waterbodies 17.2. The Posidonia meadow as a "biological quality element" 18. The Posidonia oceanica meadow. A summary 18.1. Introduction: Why be interested in Posidonia? 18.2. Posidonia and the meadows 18.3. The role of the Posidonia meadows 18.4. Regression of the Posidonia meadows 18.5. The regulatory texts that apply to the meadows 	 156 156 157 157 158 161 162 162 162 163
 17. The Posidonia oceanica meadow and the Water Framework Directive 17.1. Some key elements of the Water Framework Directive 17.1.1. A major innovation of the WFD: targets for all aquatic environments 17.1.2. The WFD's analysis unit: the waterbody 17.1.3. Good ecological status and coastal waterbodies 17.2. The Posidonia meadow as a "biological quality element" 18. The Posidonia oceanica meadow. A summary 18.1. Introduction: Why be interested in Posidonia? 18.2. Posidonia and the meadows 18.3. The role of the Posidonia meadows 18.4. Regression of the Posidonia meadows 18.5. The regulatory texts that apply to the meadows 18.6. Dead Posidonia leaves, beaches and beach nourishment 	 156 156 157 157 158 161 162 162 163 165 166 166
 17. The Posidonia oceanica meadow and the Water Framework Directive 17.1. Some key elements of the Water Framework Directive 17.1.1. A major innovation of the WFD: targets for all aquatic environments 17.1.2. The WFD's analysis unit: the waterbody 17.1.3. Good ecological status and coastal waterbodies 17.2. The Posidonia meadow as a "biological quality element" 18. The Posidonia oceanica meadow. A summary 18.1. Introduction: Why be interested in Posidonia? 18.2. Posidonia and the meadows 18.3. The role of the Posidonia meadows 18.4. Regression of the Posidonia meadows 18.5. The regulatory texts that apply to the meadows 18.6. Dead Posidonia leaves, beaches and beach nourishment 18.7. Coastal development in the maritime public domain 	 156 156 157 157 158 161 162 162 163 165 166 166
 17. The Posidonia oceanica meadow and the Water Framework Directive 17.1. Some key elements of the Water Framework Directive 17.1.1. A major innovation of the WFD: targets for all aquatic environments 17.1.2. The WFD's analysis unit: the waterbody 17.1.3. Good ecological status and coastal waterbodies 17.2. The Posidonia meadow as a "biological quality element" 18. The Posidonia oceanica meadow. A summary 18.1. Introduction: Why be interested in Posidonia? 18.2. Posidonia and the meadows 18.3. The role of the Posidonia meadows 18.4. Regression of the Posidonia meadows 18.5. The regulatory texts that apply to the meadows 18.6. Dead Posidonia leaves, beaches and beach nourishment 18.7. Coastal development in the maritime public domain 18.8. Mooring 	 156 156 157 157 158 161 162 162 163 165 166 166 167
 17. The Posidonia oceanica meadow and the Water Framework Directive 17.1. Some key elements of the Water Framework Directive 17.1. A major innovation of the WFD: targets for all aquatic environments 17.1.2. The WFD's analysis unit: the waterbody 17.1.3. Good ecological status and coastal waterbodies 17.2. The Posidonia meadow as a "biological quality element" 18. The Posidonia oceanica meadow. A summary 18.1. Introduction: Why be interested in Posidonia? 18.2. Posidonia and the meadows 18.3. The role of the Posidonia meadows 18.4. Regression of the Posidonia meadows 18.5. The regulatory texts that apply to the meadows 18.6. Dead Posidonia leaves, beaches and beach nourishment 18.7. Coastal development in the maritime public domain 18.8. Mooring 18.9. Marking the 300-metre zone 	 156 156 157 157 158 161 162 163 166 166 167 168
 17. The Posidonia oceanica meadow and the Water Framework Directive 17.1. Some key elements of the Water Framework Directive 17.1.1. A major innovation of the WFD: targets for all aquatic environments 17.1.2. The WFD's analysis unit: the waterbody 17.1.3. Good ecological status and coastal waterbodies 17.2. The Posidonia meadow as a "biological quality element" 18. The Posidonia oceanica meadow. A summary 18.1. Introduction: Why be interested in Posidonia? 18.2. Posidonia and the meadows 18.3. The role of the Posidonia meadows 18.4. Regression of the Posidonia meadows 18.5. The regulatory texts that apply to the meadows 18.6. Dead Posidonia leaves, beaches and beach nourishment 18.7. Coastal development in the maritime public domain 18.8. Mooring 	 156 156 157 157 158 161 162 162 163 165 166 167 168 169
 17. The Posidonia oceanica meadow and the Water Framework Directive 17.1. Some key elements of the Water Framework Directive 17.1.1. A major innovation of the WFD: targets for all aquatic environments 17.1.2. The WFD's analysis unit: the waterbody 17.1.3. Good ecological status and coastal waterbodies 17.2. The Posidonia meadow as a "biological quality element" 18. The Posidonia oceanica meadow. A summary 18.1. Introduction: Why be interested in Posidonia? 18.2. Posidonia and the meadows 18.3. The role of the Posidonia meadows 18.4. Regression of the Posidonia meadows 18.5. The regulatory texts that apply to the meadows 18.6. Dead Posidonia leaves, beaches and beach nourishment 18.7. Coastal development in the maritime public domain 18.8. Mooring 18.9. Marking the 300-metre zone 18.10. Trawling 	 156 156 157 157 158 161 162 162 163 165 166 167 168 169 169
 17. The Posidonia oceanica meadow and the Water Framework Directive 17.1. Some key elements of the Water Framework Directive 17.1.1. A major innovation of the WFD: targets for all aquatic environments 17.1.2. The WFD's analysis unit: the waterbody 17.1.3. Good ecological status and coastal waterbodies 17.2. The Posidonia meadow as a "biological quality element" 18. The Posidonia oceanica meadow. A summary 18.1. Introduction: Why be interested in Posidonia? 18.2. Posidonia and the meadows 18.3. The role of the Posidonia meadows 18.4. Regression of the Posidonia meadows 18.5. The regulatory texts that apply to the meadows 18.6. Dead Posidonia leaves, beaches and beach nourishment 18.7. Coastal development in the maritime public domain 18.8. Mooring 18.9. Marking the 300-metre zone 18.11. Fish farms 	 156 156 157 157 158 161 162 162 163 166 166 166 167 168 169 169 169
 17. The Posidonia oceanica meadow and the Water Framework Directive 17.1. Some key elements of the Water Framework Directive 17.1.1. A major innovation of the WFD: targets for all aquatic environments 17.1.2. The WFD's analysis unit: the waterbody 17.1.3. Good ecological status and coastal waterbodies 17.2. The Posidonia meadow as a "biological quality element" 18. The Posidonia oceanica meadow. A summary 18.1. Introduction: Why be interested in Posidonia? 18.2. Posidonia and the meadows 18.3. The role of the Posidonia meadows 18.4. Regression of the Posidonia meadows 18.5. The regulatory texts that apply to the meadows 18.6. Dead Posidonia leaves, beaches and beach nourishment 18.7. Coastal development in the maritime public domain 18.8. Mooring 18.9. Marking the 300-metre zone 18.10. Trawling 18.11. Fish farms 18.12. Discharge of effluents 	 156 156 157 157 158 161 162 162 163 165 166 167 168 169 169 169 169 170
 17. The Posidonia oceanica meadow and the Water Framework Directive 17.1. Some key elements of the Water Framework Directive 17.1.1. A major innovation of the WFD: targets for all aquatic environments 17.1.2. The WFD's analysis unit: the waterbody 17.1.3. Good ecological status and coastal waterbodies 17.2. The Posidonia meadow as a "biological quality element" 18. The Posidonia oceanica meadow. A summary 18.1. Introduction: Why be interested in Posidonia? 18.2. Posidonia and the meadows 18.3. The role of the Posidonia meadows 18.4. Regression of the Posidonia meadows 18.5. The regulatory texts that apply to the meadows 18.6. Dead Posidonia leaves, beaches and beach nourishment 18.7. Coastal development in the maritime public domain 18.8. Mooring 18.9. Marking the 300-metre zone 18.10. Trawling 18.11. Fish farms 18.12. Discharge of effluents 18.13. Solid waste 	 156 156 157 157 158 161 162 162 163 165 166 167 168 169 169 169 170 171
 17. The Posidonia oceanica meadow and the Water Framework Directive 17.1. Some key elements of the Water Framework Directive 17.1. A major innovation of the WFD: targets for all aquatic environments 17.1. The WFD's analysis unit: the waterbody 17.1. Good ecological status and coastal waterbodies 17.2. The Posidonia meadow as a "biological quality element" 18. The Posidonia oceanica meadow. A summary 18.1. Introduction: Why be interested in Posidonia? 18.2. Posidonia and the meadows 18.3. The role of the Posidonia meadows 18.4. Regression of the Posidonia meadows 18.5. The regulatory texts that apply to the meadows 18.6. Dead Posidonia leaves, beaches and beach nourishment 18.7. Coastal development in the maritime public domain 18.8. Mooring 18.9. Marking the 300-metre zone 18.11. Fish farms 18.12. Discharge of effluents 18.13. Solid waste 18.14. Laying cables and pipes on the seabed 	 156 156 157 157 158 161 162 162 163 166 166 167 168 169 169 169 170 171 171
 17. The Posidonia oceanica meadow and the Water Framework Directive 17.1. Some key elements of the Water Framework Directive 17.1. A major innovation of the WFD: targets for all aquatic environments 17.1. The WFD's analysis unit: the waterbody 17.1. The WFD's analysis unit: the waterbody 17.1. The Posidonia meadow as a "biological quality element" 18. The Posidonia oceanica meadow. A summary 18.1. Introduction: Why be interested in Posidonia? 18.2. Posidonia and the meadows 18.3. The role of the Posidonia meadows 18.4. Regression of the Posidonia meadows 18.5. The regulatory texts that apply to the meadows 18.6. Dead Posidonia leaves, beaches and beach nourishment 18.7. Coastal development in the maritime public domain 18.8. Mooring 18.9. Marking the 300-metre zone 18.1.1. Fish farms 18.1.2. Discharge of effluents 18.1.3. Solid waste 18.1.4. Laying cables and pipes on the seabed 18.1.6. Can we restore meadows that have been destroyed? 	 156 156 157 157 158 161 162 162 163 165 166 167 168 169 169 169 169 169 170 171 171 171

1. INTRODUCTION

The seagrass *Posidonia oceanica* and the meadows that it forms have recently become a major issue in marine environmental management and protection in the Mediterranean (Pergent, 1991a; Boudouresque *et al.*, 1995b; Anonyme, 2000; Pergent-Martini, 2000; Anonyme, 2001a; Boudouresque, 2003; Procaccini *et al.*, 2003).

Posidonia oceanica meadows constitute an element that is fundamental for the **quality of the coastal environment** (Boudouresque and Meinesz, 1982; Videau and Merceron, 1992), which is basic to artisanal fishing and tourism development. Its socio-economic importance and the contribution it makes to trade balances make tourism a key element which no Mediterranean country can do without (about 10% of the Mediterranean states' GDP (Gross Domestic Product), except for Algeria and Syria; UNEP, 1999). Artisanal fishing, whose economic importance is more modest, has a major socio-cultural dimension, with positive consequences for tourism (Boudouresque *et al.*, 2005).

The protection and conservation of *Posidonia oceanica* meadows is thus justified not only by their very great heritage value but also for economic reasons. It is thus an textbook illustration of the idea of sustainable development that derived from the 1992 Rio de Janeiro Summit (Boudouresque, 2002b).

SUSTAINABLE DEVELOPMENT

Sustainable development is **the union of the environment and development**. Protecting the environment not only has heritage value, it is not antinomical to economic development, but can be a development tool (Timoshenko, 1996).

The **definition** of sustainable development adopted by the Rio de Janeiro Summit in June 1992 is the following: human activities that enable the present generation of human beings and other species living on the Earth to satisfy their needs without endangering the earth's ability to satisfy the needs of future generations of either men or the other species that people the Earth (*That range of activities and development which enables the needs of the present generation of humans and all other species to be met without jeopardizing the ability of the biosphere to support and supply the reasonably foreseeable future needs of humans and all other species). The strengths of this concept are: (i) humans and all other species living on Earth are equal as regards their rights and requirements; (ii) the future is put on an equal footing with the present; (iii) there is a mutualistic symbiosis between protection of the environment and economic development.*

Furthermore, sustainable development brings together an inseparable trinity: environmental **protection**, **economic development** and **social justice**. There can be no sustainable economic development without environmental protection, no environmental protection without economic development and social justice, and no social justice without economic development and environmental protection.

It is regrettable that the idea of sustainable development is very often betrayed by ecologists, sociologists and politicians – (i) ecologists for whom nature takes precedence over man; (ii) sociologists who do not agree that animals should be protected while men are endangered; and (iii) certain politicians who can only tolerate protection of nature if this does not interfere with their naive, archaic and very (very) short-term ideas of economic development.

Protection and conservation are two similar ideas. But **protection** is a more static idea than **conservation**, for it is particularly based on regulations and their implementation. Now conservation is a dynamic concept, implying management/taking other species and other habitats into account in a context of human usage and conflicts of usage and on a scale that cannot be uniquely local.

A considerable amount of **scientific literature** (over a thousand publications) has been devoted to the protection and conservation of *Posidonia oceanica* meadows. This literature, usually written in English, Spanish, French or Italian, appears in dozens of scientific reviews that are sometimes local and thus not very accessible to non-specialists. Reference lists, classed by subject matter, have been brought out, but are now rather old (Boudouresque *et al.*, 1977, 1979, 1980b; Boudouresque, 1989). One can add to this a summary (Cinelli *et al.*, 1995a).

For the wider public, some **popular works** are available: Ballesteros *et al.* (1984), Flores-Moya and Conde-Poyales (1998), Luque and Templado (2004) and Romero (2004b) in Spain, Boudouresque and Meinesz (1982) and Thébaudin and Cadeau (1987) in France and Mazzella *et al.* (1986) in Italy.

In the field of **managing** *Posidonia oceanica* meadows, information is spread by almost a hundred reports with very small circulations, thus practically inaccessible, and in doctoral theses, also scarcely accessible. Generally speaking, the information contained in these reports is limited and relates to particular sites or situations. The lessons to be learned, likely to become widespread around the Mediterranean, have rarely been summarized, except regarding monitoring tools, legislation and restoration (Boudouresque *et al.*, 1990b, 1995b, 2000; Pergent-Martini, 2000; Boudouresque, 2002a).

All in all, a work that takes stock of the knowledge about *Posidonia oceanica* and the meadows it forms, and that summarizes the information that enables us to best respond to the various problems managers face, does not yet exist.

We have chosen to present **3 levels of reading**: (1) the scientific level, (2) the level of specialist managers and (3) the level of non-specialist managers and politicians.

(1) The scientific level corresponds to Chapters 2-5 and the first paragraphs ("the problem") of the other chapters. This information is necessary so that the recommendations contained in the two other levels of reading should not appear as simple, fairly arbitrary formulae. Those who so wish can find basic information there and even go further, referring back to the source (bibliographical references are systematically provided). This information is necessary, for managers increasingly have to justify their choices when these choices are not strictly framed by regulatory provisions. Moreover, this information will enable them to put the recommendations we provide today in their context, when faced with new data to which they may have access.

(2) The **specialist managers' level** corresponds to the "case studies" and "recommendations" paragraphs in Chapters 6-13 and to Chapter 14. In the case studies, managers will find similar situations to those which they are faced. The style of the recommendations, with very clear reference to scientific data, will enable them to justify the choices they will have to propose.

(3) The non-specialist decision-maker managers' and politicians' level (Chapter 18) enables quick reading with no scientific references or evidence. The recommendations that appear here are "robust," i.e. it is extremely unlikely that research being done today will lead to them being modified. Managers

or politicians who have to make decisions will not run any risks in following the recommendations. Our choice of on the one hand offering several levels of reading and on the other dividing the text into chapters that focus on a specific problem implies frequent cross-referencing between chapters. However, certain ideas **recur** in many chapters; the authors made this choice deliberately to make it easier to use the book.

Despite the care taken in drafting the book, and despite the considerable documentary base which it draws on, the authors are aware that certain documents have been left out, that certain particular cases can be found outside the margins they envisaged, and that their analysis is not above being criticized. They therefore thank in advance all those who wish to point out possible omissions, errors or lacunae.

The present work mainly concerns the coasts of Provence (France), Monaco and Liguria (Italy), i.e. the area covered by the RAMOGE Agreement. However, the writers wished whenever possible to locate it on **a pan-Mediterranean** level. The result is that in any one given country their recommendations may seem unnecessary (according to the existing regulatory texts), naive or possibly unrealistic. They wish to apologize for this in advance..

2. POSIDONIA OCEANICA MEADOWS

Some 120-100 million years (Ma) ago, in the Cretaceous period, some continental Magnoliophytes¹ (Plantae) returned to the marine environment. Further back in time, about 475 Ma ago, in the Ordovician (Primary era), their distant ancestors had left this marine environment to go in conquest of the continents (Boudouresque and Meinesz, 1982; Wellman *et al.*, 2003).

By number of their species in today's natural environment, marine Magnoliophytes represent rather a small group: 13 genera² and 60 species (Kuo and Hartog, 2001). For purposes of comparison, there are 234 000 Magnoliophyte species (almost all continental) and about 200 000 marine species of fauna and flora together (Fredj *et al.*, 1992; Heip, 1998; Lecointre and Le Guyader, 2001). Why was the **diversification** of marine Magnoliophytes so meagre, while since that time (100 Ma) the continental Magnoliophytes diversified to such an extent? 3 hypotheses exist: (i) The predominance of **vegetative reproduction** compared to sexual reproduction for marine Magnoliophytes; sexual reproduction, through the genetic recombination it involves, is a powerful evolutionary motor; moreover, there is often self-fecundation (Romero, 2004a). (ii) The absence of **mutualistic** symbiosis with insects (absent in the marine environment) for pollination; in the continental environment this often extremely specific symbiosis has been a powerful motor of speciation (Romero, 2004a). (iii) Lastly, **the competitive advantage** the marine Magnoliophytes enjoy over other marine primary producers³ is so great that competition has not driven evolution.

However, although the marine Magnoliophytes are not numerous, their ecological weight is considerable in coastal environments: many of them are ecosystem engineers⁴, or at least key

- ² The following are genera of marine Magnoliophytes Amphibolis, Cymodocea, Enhalus, Halodule, Halophila, Heterozostera, Nanozostera, Phyllospadix, Posidonia, Syringodium, Thalassia, Thalassodendron and Zostera. Magnoliophytes present in brackish water (e.g. the genera Ruppia and Potamogeton) are not taken into account here.
- ³ Here we are talking about the polyphyletic group formerly known as 'algae'. This group is made up of (i) Chromobionta ('brown algae') which belong (with part of what were called 'muchrooms') to the Kingdom Stramenopiles, (ii) Rhodobionta ('red algae') which belong to the Kingdom Plantae, and (iii) Chlorobionta ('green algae') close to the Magnoliophytes, which also belong to the Kingdom Plantae.
- ⁴ An ecosystem engineer (engineering species) is an organism which directly or indirectly modulates the availability of resources (other than the resource it itself may constitute) for other species by provoking physical changes in the biotic or abiotic material (Lawton, 1994).
- ⁵ A key species is a species whose impact on the functioning of the ecosystem in which it participates is greater than its abundance might lead one to believe (Bond, 2001).
- 6 Nanozostera noltii = Zostera noltii.

species⁵. The ecosystems they build up or in which they are the major actors play an essential role in many parts of the world. This is so for the Mediterranean.

In the Mediterranean, there are 5 species of Magnoliophytes. As well as *Posidonia oceanica*, there are *Cymodocea nodosa*, *Nanozostera nolti*[®], *Zostera marina*, and a Red Sea species that entered the Mediterranean via the Suez Canal, *Halophila stipulacea* (Hartog, 1970; Por, 1978). Australia seems comparatively much richer, with 30 species, including 8 *Posidonia* species: *P. angustifolia, P. australis, P. coriacea, P. denhartogii, P. kirkemanii, P. ostenfeldii, P. robertsonae* and *P. sinuosa* (Kuo and Hartog, 2001). The fact that the genetic differences (DNA) between the Mediterranean species (*P. oceanica*) and the Australian Posidonia species are relatively great suggests that the separation of the 2 groups happened long ago, certainly in the late Eocene (Waycott and Les, 2000).

2.1. GEOGRAPHICAL DISTRIBUTION

Posidonia oceanica is present almost throughout the Mediterranean. In the west, it disappears just before the Strait of Gibraltar, near Calaburros in the north and Melilla in the south (Conde Poyales, 1989). In the east, it is absent from the Egyptian coast (east of the Nile Delta), Palestine, Israel and Lebanon (Por, 1978). It does not penetrate the Marmara Sea or the Black Sea. And it

Magnoliophytes are what used to be called Phanerogams.

is rare or absent in the far north of the Adriatic (Zalokar, 1942; Gamulin-Brida *et al.*, 1973; Gamulin-Brida *et al.*, 1974) and along the Languedoc coast (France) between the Camargue and Port-la-Nouvelle (Boudouresque and Meinesz, 1982).

In literature there is some mention of *Posidonia oceanica* outside the Mediterranean: the Bay of Biscay (Sauvageau, 1890; Flahault, 1908; Sauvageau, 1927; Fernandez-Casas *et al.*, 1992), Portugal (Daveau, 1896; Flahault, 1908), Canary Islands (Viera *in* Carrillo and Gil-Rodriguez, 1980), the Red Sea (Makkaveeva, 1968), the Indian Ocean (Saporta and Marion, 1878) and even Texas, USA (Correl and Johnston, 1970). These are uncritical citations from very old books, or surprising mistakes due to confusion with *Zostera marina, Thalassia testudinum* or *T. hemprichii* (Hartog, 1970; Correl and Correl, 1975; McMillan *et al.*, 1975; Carrillo and Gil-Rodriguez, 1980).

All in all, *Posidonia oceanica* is an endemic Mediterranean species, i.e. strictly confined to this Sea. We know that most of this Sea dried up in the Messinian period, 5.6 to 5.3 million years ago, because of the closing of the Strait of Gibraltar (Krijgsman *et al.*, 1999; McKenzie, 1999). We do not know how *P. oceanica* survived this crisis. One or many refuge areas certainly existed in the Mediterranean or the Atlantic nearby, from which it could recolonize the Mediterranean after the Gibraltar Strait reopened.

2.2. THE PLANT

Posidonia oceanica is made of creeping or erect stems usually buried in the sediment, called rhizomes. Creeping **rhizomes** are called **plagiotropic**, and erect rhizomes **orthotropic**. There is no determined differentiation between plagiotropic and orthotropic rhizomes; according to the available space an orthotropic rhizome may become plagiotropic and vice versa (Caye, 1980). The rhizomes end in groups of 4-8 leaves (shoots) that are 8-11 mm wide and 20-80 cm long. The length may reach 156 cm (⁷). Rhizomes also have roots that can grow to 70 cm beneath the surface of sediment (Fig. 1; Giraud *et al.*, 1979; Boudouresque and Meinesz, 1982).

New leaves form all year round. They live for between 5 and 8 months, and more rarely up to 13 months. The leaf's growth zone (meristem) lies at the base. Leaves less than 5 cm long are called **juveniles** and those longer than 5 cm without basal sheath (=petiole) are called **intermediate**; when the growth is over, a petiole is formed and the leaf is said to be **adult** (Fig.2; Giraud, 1979; Ott, 1980; Thélin and Boudouresque, 1983).

When they die, the leaves do



Fig. 1. A plagiotropic rhizome of *Posidonia oceanica*, from which leave upward a half-dozen of orthotropic rhizomes and, downward, roots. Each rhizome bears a shoot of leaves. Frayed scales cover the rhizomes. The scale bar measures 2 cm. From Boudouresque and Meinesz (1982).

⁷ The island of Ischia (Gulf of Naples, Italy), 10 metres depth, in June (Gérard Pergent and Christine Pegent-Martini, unpublished data).





Fig. 3. The lepidochronology. At the top Arrangement of scales along a *Posidonia* oceanica rhizome. Below scale thickness (in µm). M maximum thickness. m minimum thickness. fs remains of a floral stalk. p prophyll (= preleaf) accompanying the floral stalk. I.I. the oldest living leaf. Year -1 previous year. From Pergent *et al.* (1989b).

not fall off: only the limb⁸ is deciduous, while the basal sheath (petiole), several centimetres long, remains attached to the rhizome. It is then called a scale or sheath (Fig. 1 and 2). The leaves drop, as they are formed, all year round (Pergent and Pergent-Martini, 1991). The **sheaths** (like the rhizomes) are not very putrescible and so can last for several centuries or even thousands of years. An entire set of parameters of the sheaths (length, thickness, anatomy) varies cyclically during a yearly cycle (Fig. 3). The analysis of these cycles is known as **lepidochronology** (Crouzet, 1981; Crouzet *et al.*, 1983; Pergent *et al.*, 1983; Pergent, 1990a).

Lepidochronology is a powerful tool for measuring the speed of growth of the rhizomes, the number of leaves formed each year, the dynamics of building up meadows, past primary production, old pollutant levels, etc. (Pergent, 1990b; Pergent and Pergent-Martini, 1990, 1991; Pergent *et al.*, 1992; Pergent-Martini and Pergent, 1994; Pergent-Martini, 1998). Much of the data on *Posidonia oceanica* meadows, presented throughout this book, comes from use of the lepidochronological tool.

Posidonia oceanica flowers in the autumn (September-November). The flowers are hermaphrodite, i.e. both male and female at the same time; 4-10 flowers are grouped in an inflorescence at the tip of a 10-30 cmlong stalk (Fig. 4). It does not flower every year, especially in the relatively cold waters of the northwestern Mediterranean. Some years there is a particularly intense flowering, throughout the whole

Mediterranean, for example in 1971, 1982, 1993, 1997 and 2003 (Giraud, 1977c; Boudouresque and Meinesz, 1982; Mazzella *et al.*, 1983, 1984; Caye and Meinesz, 1984; Pergent, 1985; Thélin and



⁸ A leaf has a petiole (base or sheath) by which it is attached to the stem or to the rhizome, and a limb, the part where photosynthesis happens.



Boudouresque, 1985; Pergent *et al.*, 1989a; Acunto *et al.*, 1996; Piazzi *et al.*, 1999; Gobert *et al.*, 2005). Flowering seems to be caused by high summer temperatures and by a temperature of 20°C in October (Caye and Meinesz, 1984; Thélin and Boudouresque, 1985; Pergent *et al.*, 1989a; Stoppelli and Peirano, 1996).

The **fruits** of *Posidonia oceanica* require 6-9 months to ripen. Between May and July, they drop off and float for a certain time. According to the direction of the currents, they may be washed up on beaches in great quantities. The fruits have the shape and size of an olive; they are dark green, dark brown to black (Fig. 5). They have just one **seed** (Hartog, 1970; Boudouresque and Meinesz, 1982). Germination of the seeds has been observed *in situ* on several occasions

(Acunto et al., 1996; Piazzi et al., 1996; Balestri et al., 1998; Gambi and Guidetti, 1998; Piazzi et al., 1999; Eric Charbonnel, unpublished observations). However, P. oceanica reproduces mainly in a vegetative way, through cuttings (Molinier and Picard, 1952). In the Bay of Villefranche-sur-Mer (Alpes-Maritimes, France), Meinesz and Lefèvre (1984) estimated that in a favourable site (P. oceanica "dead matte") the number of cuttings that attach successfully is on average 3/ha/year. Cymodocea nodosa and Caulerpa prolifera settlements are equally favourable to the attachment of cuttings, and possibly the germination of seeds (Cinelli et al., 1995b). Another form of vegetative reproduction by pseudo-viviparity^a was recently observed in May 2004 in the Balearic Islands (Ballesteros et al., 2005). Vegetative plantlets form directly on the inflorescences and replace the organs of

Fig. 5. Fruits of *Posidonia oceanica* ("sea olives"; arrows), in a meadow. They are 1.5-2.0 cm long and about 1 cm wide.



sexual reproduction. This strategy contributes to short-distance dispersal. For the time being we do not know if this is a very local mode of reproduction or whether it concerns other parts of the Mediterranean.

Posidonia oceanica's **low genetic variability** could be a weakening factor for this species (Raniello and Procaccini, 2002). Indeed, Capiomont *et al.* (1996) highlighted the fact that enzymatic polymorphism between populations of the western Mediterranean (Italy, continental France, Corsica and Algeria) is very low, particularly in the areas of Port-Cros (Var, France) and Nice (Alpes-Maritimes, France). The rarity of the flowering and, especially, seed production, as well as self-pollination, and inversely the frequency of vegetative reproduction (by cuttings) could explain this low variability. It should however be noticed that, based on anatomical, morphological (leaf width) and karyological features, a population of *P. oceanica* presenting original characteristics was found in the area of Algiers (Semroud *et al.*, 1992). Moreover, genetic markers (RAPD, microsatellites) do not confirm the low genetic variability based on enzymatic polymorphism (Reusch, 2001).

⁹ There are two kinds of viviparity (i) strict viviparity, where the seeds resulting from sexual reproduction germinate on the inflorescence before dropping off the plant and (ii) pseudo-viviparity, when the vegetative propagules (bulbils, plantules) replace the organs of sexual reproduction on the inflorescence. Like cuttings, these plantules give birth to clones of the mother plant.

2.3. ECOLOGY

In calm conditions, *Posidonia oceanica* can develop very near average sea level: its leaves then spread out on the surface. Maximum **depth** depends on water transparency: 15-23 m in the Pyrénées-Orientales, France (Ben, 1971; Pergent *et al.*, 1985; Ballesta *et al.*, 2000), 21-28 m in Liguria, Italy (Bianchi and Peirano, 1995), 20-30 m in Latium, Italy (Diviacco *et al.*, 2001), 28-32 m in the Bouches-du-Rhône, France (Bourcier, 1979; Cristiani, 1980), 28-38 m in the Var, France (Bourcier, 1979; Harmelin and Laborel, 1976), 22-35 m in the Alpes-Maritimes, France (Meinesz and Laurent, 1977, 1978), 30-39 m in Corsica (Molinier, 1960; Bay, 1979; Meinesz *et al.*, 1987; Pasqualini, 1997), 30-38 m in Ischia (Gulf of Naples, Italy; Giraud *et al.*, 1979) and 43-44 m in Malta (Schembri, 1995). In the Var and in Corsica, isolated clumps of *P. oceanica* have been observed up to 45-48 m depth (Augier and Boudouresque, 1979; Boudouresque *et al.*, 1990c). **Light** is one of the most important factors for the distribution and density of *P. oceanica* (Elkalay *et al.*, 2003).

Posidonia oceanica dislikes **low salinity**. It dies off immediately below 33‰ (Ben Alaya, 1972). It is the low salinity that keeps it out of the Marmara Sea (21-27‰), the brackish lagoons of the Languedoc coast and around the coastal river mouths. The absence of *P. oceanica* from the central part of many beaches could correspond to the area where ground water reappears (Leriche, 2004). The species appears to resist high salinity levels more successfully, although Ben Alaya (1972) has shown that 41‰ constitutes its upper tolerance limit. In fact it is present in the hypersaline lagoons of Tunisia (Bahiret el Biban; 46‰ average in August) and Libya (Farwa: 39-44‰ according to the season); in these lagoons its vitality (number of leaves produced per year, growth of rhizomes) seems identical to or above what is observed out at sea (Pergent and Zaouali, 1992; Pergent and Pergent-Martini, 2000; Pergent *et al.*, 2002a).

The extremes of **temperature** measured in a *Posidonia oceanica* meadow are 9.0 and 29.2°C (barrier reef in Port-Cros Bay, Var, France; Augier *et al.*, 1980; Robert, 1988). It is however possible that low (under 10°C) and high (over 28°C) temperatures are only exceptionally borne. *P. oceanica*'s absence from the Levantine coast (eastern Mediterranean) and its scarcity in the northern Adriatic and along the Languedoc coast could be due respectively to the summer and winter temperatures (Boudouresque and Meinesz, 1982). Also, in deep meadows, Mayot *et al.* (2005) suggest that the increase in seawater temperature that is currently observed (Salat and Pascual, 2002) could have a harmful effect on *P. oceanica*.

Posidonia oceanica dislikes too intense **hydrodynamism**. Storms tear off shoots, some of which will constitute cuttings. They can also erode the "matte¹⁰", either directly, or by leaching the sediment, which weakens the meadows (see below). That is why in rough sea conditions the meadow grows no nearer the surface than 1 or 2 metres. "Dead matte" can thus be a natural phenomenon, as for example in the La Palud Bay in Port-Cros, Var, France (Augier and Boudouresque, 1967). In literature, the presence of "dead matte" has not infrequently been wrongly interpreted as an unequivocal sign of human impact (Moreno *et al.*, 2001).

10 See page 15 for the definition of the "matte".



2.4. THE STRUCTURE OF MEADOWS

The leaves and rhizomes of Posidonia oceanica support many flora and fauna, some of which are calcified. When they die, their remains fall off and form an autochthonous sediment (debris of sea urchin spines or tests, mollusc shells, corallinales", etc.). Moreover, because of their density (up to 5 000/m²) and distribution the leaves of P. oceanica reduce the speed of the current; this reduces the kinetic energy of the sedimentary particles carried by the water, which then deposit to the seabed (allochthonous sediment).

Posidonia oceanica rhizomes grow in height, even in the absence of sedimentation. To resist being buried, they are capable of speeding up their growth (Molinier and Picard, 1952; Caye, 1980; Boudouresque and Jeudy de Grissac, 1983; Boudouresque et al., 1984; Jeudy de Grissac and Boudouresque, 1985; Boudouresque et al., 1994b).

The "matte" is the whole mass composed of rhizomes, sheaths, roots and the sediment that fills the interstices. The rhizomes, sheaths and roots are not very putrescible and thus are conserved within the "matte" for several centuries or even thousands of years (Boudouresque et al., 1980d; Boudouresque and Jeudy de Grissac, 1983).

Over the course of time, the "matte" rises to the surface. Comparing bathymetric maps of 1839 and 1950 for the Levant-Port-Cros and Bagaud-Port-Cros passes (Var, France), Molinier and Picard (1952) measured a seabed elevation of 1 metre per century. Certainly, we must probe the precision of these maps. In Catalonia, Spain, over a period of 3 000 years, average growth was 18 cm per century (Mateo et al., 1997). In the Gulf of Giens (Var, France), a Roman wreck which sank in about 50 or 60 BC is covered with 2 metres of "matte" (Fig. 6; Tchernia et al., 1978; Boudouresque and Meinesz, 1982). In cape Moulin (Port-Cros, Var, France) radiocarbon dating (14C) of the remains of rhizomes indicates an average growth of 10 cm per century (Boudouresque and Jeudy de Grissac, 1983). A slower average speed was measured in the Bay of Calvi, Corsica (Boudouresque et al., 1980d).

The rising of the "matte" can bring the meadow near the surface. In exposed conditions, this rise stops 1 or 2 metres below the sea surface. Hydrodynamism prevents the rise continuing and determines the forming of a peneplain of "dead matte" (Fig. 7; Molinier and Picard, 1952;

Fig. 6. The Madrague de Giens (Var, France) Roman wreck. There is the "matte" (arrow) under which the wreck and the amphorae were buried. From Tchernia et al. (1978).



Sea level

Fig. 7. Dynamics of the Posidonia oceanica meadow in exposed conditions. The rise of "matte", over time, stops at 1-2 m deep. The erosion by sea surface hydrodynamism determines formation of a peneplain of "dead matte". After Boudouresque (unpublished).



¹¹ Corallinales are calcified photosynthetic organisms which belong to the Rhodobionts (Plantae).



Fig. 8. In a sheltered bay, the *Posidonia oceanica* meadow approaches the sea surface. Photo E. Charbonnel.



Fig. 9. The barrier reef of Port-Cros Bay (Var, France). Posidonia oceanica leaves are spread on the surface of the water. Photo S. Ruitton.

Boudouresque and Meinesz, 1982). But in **sheltered conditions**, especially at the innermost part of bays, the rise of the "matte" can continue right up to the sea surface (Fig. 8). The leaves spread out on the surface (Fig. 9). At a first stage, the emersion of the tips of the leaves happens parallel to the coast. This formation is known as a **fringing reef** (Fig. 11). Then the continuing rise of the "matte" widens the fringing reef. Within the fringing reef the leaves and the shallowness hamper water circulation; the temperature may go below (in winter) or above (in summer) the limits of *Posidonia oceanica*'s tolerance. The same holds good for salinity, in times of rain. Between the coast and *P. oceanica*'s emersion front, the shoots die, and a **lagoon** is formed (Fig. 10) (Molinier and Picard, 1952; Boudouresque and Meinesz, 1982).

Posidonia oceanica's emersion front thus constitutes a **barrier reef** (Fig. 10). With time, the barrier reef moves out to sea and the lagoon grows (Fig. 11) (Molinier and Picard, 1952; Augier and Boudouresque, 1970a; Boudouresque and Meinesz, 1982). The barrier reef's movement out to sea is estimated to be 8-10 metres a century (Boudouresque, unpublished data). In the lagoon, the bottom of which is muddy, 2 Magnoliophytes with narrower, shorter leaves than those of *P. oceanica* can establish: *Cymodocea nodosa* and *Nanozostera noltii*.

Many barrier reefs have been destroyed because they were located in bays that have been made into ports. The most spectacular barrier reefs still in existence are those of Port-Cros, Le Brusc and La Madrague de Giens (Var, France) (Molinier and Picard, 1952; Augier and Boudouresque, 1970a; Boudouresque *et al.*, 1975; Boudouresque and Meinesz, 1982; Bernard *et al.*, 2002; Charbonnel *et al.*, 2002). Less typical or less well-known barrier reefs also exist (or did exist¹²) south of Port-Bou (Catalonia, France), in Puerto de Sanitja (Minorca, Balearic Islands, Spain), in Bajos de Roquetas (Almeria, Spain), in Toulon



12 Some of these barrier reefs, cited below from fairly old sources, have in fact disappeared since then, without this disappearance being recorded by a scientific publication.







2.5. TYPES OF MEADOWS

Posidonia oceanica meadows can be of different morphostructural types linked to hydrodynamism, currents and/or water temperature. However, the type of meadow does not seem to influence density of shoots, length of leaves, number of leaves per shoot or biomass (Borg *et al.*, 2005).

The **plain meadow** is the most usual kind of meadow in the Mediterranean, especially the western Mediterranean. It consists of a fairly continuous, horizontal or gently sloping meadow, broken by erosive structures (erosion scarp, erosive intermatte, shifting intermatte, "return river") and

non-erosive "dead matte" (structural intermatte) (Fig. 12; Boudouresque et al., 1980d, 1985a). All these structures are of natural origin (Blanc and Jeudy de Grissac, 1984). Erosive intermattes are a sort of circular or ovoid "potholes" dug into the "matte"; when they are deep, Posidonia oceanica can start growing again at the bottom of the intermatte (Molinier and Picard, 1952). Shifting intermattes are furrows several dozen metres long and several metres wide, lying parallel to the shore. The side of the shifting intermatte that is closest to the shore is made up of an erosion scarp; which is actively eroded. The central part of the shifting intermatte is made up of "dead matte" possibly covered by sand. The side that is furthest away from the shore is made up of a meadow front with plagiotropic rhizomes that tends to recolonize the intermatte. Over time, the shifting intermatte moves along in a parallel direction towards the coast (Fig. 12, 13) (Boudouresque et al., 1980d; Leriche et al., 2004). Typical shifting intermattes have been observed



Fig. 12. The plain meadow. At the top, a cross-section perpendicular to the coastline showing a shifting intermatte, an erosive intermatte and a structural intermatte. Below, a top view of the same structures. From Boudouresque (unpublished).



Fig. 13. A shifting Intermatte in the Bay of Calvi (Corsica). On the left, the erosive scarp. On the right, *Posidonia oceanica* plagiotropic rhizomes recolonizing the intermatte. Photo A. Meinesz.



Fig. 14. A return river in a bay. The wind pushes surface waters toward the coast (arrows), and they return to seaward from the bottom (arrows), This circulation movement of water is called undertow. From Boudouresque and Meinesz (1982).



Fig. 15. Cross-section in a hill meadow (at the top) young hills (to the right and left) and an old hill, the destruction of which began (in the center). Top view of a hill meadow (below), with hills of various age, including a hill almost completely destroyed (below on the left). After Boudouresque (unpublished).

Fig. 16. History of a hill (Hill meadow) its birth (1), its growth in height and width (2-4), the formation of an erosive intermatte at its top (4) and its destruction (5-7). A new Hill may arise from the remains of a hill. Light grey sand. Green leaves. Brown rhizomes. After Boudouresque (unpublished).

in Calvi Bay (Corsica), in the Giens Gulf (Var, France) and in Torre Astura, Circeo and Terracina (southern Latium. Italy) (Boudouresque et al., 1980d; Paillard et al., 1993; Diviacco et al., 1999, 2001). "Return rivers", or sagittal channels, are channels cut into the meadow that run perpendicular to the coast, whereby surface water pushed by the wind towards the coast returns to the open sea at the level of the seabed. The return riverbed can be followed up to 10-15 metres depth, more rarely 20 metres; it is 100-300 m wide, sometimes less, and is often bordered by erosion scarps (Fig. 14) (Blanc, 1974, 1975; Boudouresque and Meinesz, 1982; Blanc and Jeudy de Grissac, 1984). The speed of the return current (undertow) can be important during storms, as Blanc (1974) observed rocks weighing 50 kilos carried several hundred metres. Lastly, structural intermattes are little stretches of "dead matte" (0.2-0.5 m²) whose origin, which remains to be explained, seems to be natural (Boudouresque, unpublished data).

The **hill meadow** is less frequent. It is encountered between 15 and 30 metres depth, in sectors where there is great

hydrodynamism (Boudouresque *et al.*, 1985a). In a hill meadow, *Posidonia oceanica* cuttings give birth to "hills" that grow wider and higher. The hills are usually surrounded by sand (Fig. 15).

When the hills grow higher they are exposed to hydrodynamism: at the summit, the sediment of the "matte" is retained badly and the rhizomes lose their hold. The exposed rhizomes are vulnerable and a kind of intermatte forms. Over time, this intermatte widens until the hill is entirely, or mostly, destroyed (Fig. 15 and 16). The lifetime of a hill between its birth and its destruction is usually about one century (Boudouresque *et al.*, 1985a, 1986a). It seems that the destruction is not always complete and that a new hill can be born from the remains of a former hill. The hill meadow has been described in Corsica (Boudouresque *et al.*, 1985a); it has also been observed in the Var and in Tuscany; it is probably much more widespread in the Mediterranean than these rare mentions suggest, at least in the central-western Mediterranean.

Fig. 17. A striped meadow, seen from top meadow strips, more or less parallel, surrounded by "dead mattes" usually occupied by a *Cymodocea nodosa* and/or *Caulerpa prolifera* settlements. From Boudouresque et al. (1985b), redrawn (Boudouresque, unpublished).



Fig. 18. Cross section in a strip of striped meadow. The progress is made towards the right, whereas the strip is eroded on the left by the hydrodynamism (generated by the dominant current). From Boudouresque *et al.* (1990a).



The **striped meadow** consists of 1-2 m-wide *Posidonia oceanica* meadow strips, which are several dozen metres long and separated by "dead matte" occupied by a *Cymodocea nodosa* and/or *Caulerpa prolifera* settlements (Chlorobionta, Plantae) (Fig. 17). Each meadow strip shifts, parallel to itself, against the dominant current, at an average speed of 10 cm/year. A cross section of a meadow strip shows on one side a front of plagiotropic rhizomes that progresses onto the "dead matte", a gentle slope behind the front, and a small erosive scarp where the strip disintegrates (Fig. 18) (Boudouresque *et al.*, 1985b; Boudouresque *et al.*, 1990a; Boudouresque and Ben Maïz, unpublished data).

The striped meadow, which develops in shallow water (less than 10 m depth), is especially present in the Gulf of Gabès, Tunisia, mainly around the Kerkennah Islands (Fig. 19). In a less typical form it is encountered in the Bouches de Bonifacio (southern Corsica) and Marsala (western Sicily) (Blanpied *et al.*, 1979; Calvo and Fradà-Orestano, 1984; Boudouresque *et al.*, 1990a; Boudouresque and Ben Maïz, unpublished).

Posidonia oceanica **micro-atolls** are often associated with the striped meadow. A micro-atoll is originally a more or less circular spot of *P. oceanica* in very shallow water. The Posidonia dies at the heart of the spot while the spot itself grows, thanks to the plagiotropic rhizomes on its periphery, thus giving birth to a Posidonia crown (Fig. 20; Boudouresque *et al.*, 1990a). Micro-atolls have been described in Turkey, in Marsala (western Sicily), and in Saint-Florent, Corsica (Calvo and Fradà-Orestano, 1984; Boudouresque *et al.*, 1990a; Pasqualini *et al.*, 1995).

The **sugar loaf meadow** was described by Molinier and Picard (1954) in Tunisia. At the beginning it is a rather shallow plain meadow. Certainly because of the high water temperature, the meadow dies, except for some more or less circular spots. These spots continue to rise to the surface and simultaneously their diameter decreases, thus forming characteristic "sugar loafs" (Fig. 21). Outside the Tunisian shores, this kind of meadow has been observed in the Gulf of Giens (Var, France).







Fig. 23. The tiered meadow. A "step" of Posidonia shifts up the slope and is eroded back by the hydrodynamism. From Boudouresque (unpublished).



The tiered meadow (or staircase meadow) develops on hard substrate with a relatively steep slope and descending bottom currents (Boudouresque, unpublished data). Its origin is similar to that of the striped meadow. Parallel strips of meadow, 0.5-3 metres wide, shift up the slope against the current. Upstream of each "stair", plagiotropic rhizomes advance at an average speed of 10 cm/year, whereas downstream the current erodes the vertical part of the "stair" (Fig. 22, 23). In Punta Ciuttone (western Corsica, Parc naturel régional de Corse), where this kind of meadow was discovered, on average several centuries are needed for a "stair" to start from the base of the slope, move up it completely, and finally be destroyed there by hydrodynamism (Boudouresque, unpublished data). The tiered meadow is also present in Port-Cros (Var, France). It is probably present in other parts of the north-western Mediterranean, where it should be sought.

Lastly, the **undulating meadow** develops at the lower limit of *Posidonia oceanica* between 25 and 40 metres depth, on a sub-horizontal substratum (Clairefond and Jeudy de Grissac, 1979). It has, however, also been observed at shallow depths (El Asmi-Djellouli *et al.*, 2000). It is characterized by wide, anastomotic, parallel strips of meadow (up to ten metres), of very slight rise, separated by parallel strips of sand (possibly) covered with "dead matte" (Fig. 24). It has been described between Port-Cros and Bagaud Islands (Var, France) (Clairefond and Jeudy de Grissac, 1979) and found

at the entrance to the Galeria Bay (Corsica; Bianconi and Boudouresque, unpublished data), in La Ciotat Bay (Bouches-du-Rhône, France; Charbonnel and Francour, 1994; Bonhomme *et al.*, 1999) and in Tunisia in shallow water (El Asmi-Djellouli *et al.*, 2000). It is probably not very rare. So far its genesis has not been explained.



2.6. HOW THE ECOSYSTEM FUNCTIONS

A basic feature of the *Posidonia oceanica* ecosystem is the juxtaposition of **2 kinds of primary production**. At global level, only marine Magnoliophyte (=Phanerogam) ecosystems present this special feature (Boudouresque, 1996). (i) The primary production from *P. oceanica* is rich in cellulose and in lignin, compounds that are not used much by herbivores, and in phenolic compounds, one of whose roles is to dissuade potential consumers (Piovetti *et al.*, 1984). The net primary production of *P. oceanica* is on average 420 g DM/m²/year(¹³) and may reach 1 300 g DM/m²/year; it drops in relation to depth (Mazzella and Ott, 1984; Romero-Martinengo, 1985; Pergent *et al.*, 1994; Pergent-Martini *et al.*, 1994; Pergent *et al.*, 1997). (ii) The primary production from the Multicellular Photosynthetic Organisms (MPOs) leaf epibiota (Fig. 25) is



Fig. 25. *Posidonia oceanica* leaves covered with MPOs leaf epibiota, in the Bay of Port-Cros (Var, France). From Boudouresque and Meinesz (1982).

composed of very palatable Rhodobionta and Chromobionta, thus easily usable by herbivores; it is between 100 and 500 g DM/m²/year (Giorgi and Thélin, 1983; Mazzella and Ott, 1984). All in all, the *P. oceanica* ecosystem is one of **the planet's most productive** ecosystems.

Vegetal biomass is very high: up to 900 g DM/m² (¹⁴) for the leaves, up to 470 g DM/m² for the MPOs leaf epibiota, up to 50 g DM/m² for the MPOs rhizome epibiota, and up to 5 500 g DM/m² for the rhizomes, sheaths and roots; it decreases with depth (Thélin and Bedhomme, 1983; Pirc, 1983; Libes, 1984; Boudouresque and Jeudy de Grissac, 1986; Ballesteros, 1987; Francour, 1990; Mazzella *et al.*, 1992). No other marine ecosystem, except mangroves, presents such high vegetal biomass. This is due to the storage of the biomass over a long period of time in the "matte". High vegetal biomass and storage are attributes usually associated with continental forest ecosystems.

Animal biomass is considerably lower than vegetal biomass. This is also a feature that the *Posidonia oceanica* ecosystem shares with the continental forest ecosystems. For each taxonomical group or trophic compartment, the values vary considerably from one station to the next and according to depth. For example, the following values (or value intervals) can be seen as common (in g DM/m²): 2-180 g for the leaf epizoans, 2-25 g for fish, 0.2 g for the starfish *Marthasterias glacialis*, 3-6 g for vagile crustaceans and molluscs, 2-33 g for sea urchins (mainly *Paracentrotus lividus* and *Psammechinus microtuberculatus*), 6-9 g for the holothurians (*Holothuria polii* and *H. tubulosa*) and 50-70 g for the "matte" endofauna (Harmelin, 1964; Vadon, 1981; Harmelin-Vivien, 1982, 1983, 1984; Francour, 1984; Ballesteros, 1987; Francour and Paul, 1987; Francour, 1990; Harmelin-Vivien and Francour, 1992; Jiménez *et al.*, 1997). All in all, the fauna usually represents 100-200 g DM/m².

Little (less than **10%**) of the *Posidonia oceanica* primary production is used by herbivores. These are mainly the fish *Sarpa salpa* (Fig. 26, 28),



Fig. 26. The fish *Sarpa salpa* (salema) is one of the main consumers of *Posidonia oceanica*. Photo S. Ruitton.

- ¹³ DM = dry mass. Higher values have been mentioned in the literature (e.g. Ott, 1980); their value is local unless due to methodological artefacts.
- ¹⁴ Higher values have been mentioned by several authors (e.g. Drew and Jupp, 1976; Ott and Maurer, 1977; Boumaza and Semroud, 1995; Romero et al., 1998).



Fig. 27. A "banquette" of *Posidonia oceanica* dead leaves on a beach in Corsica. Photo S. Ruitton.

the sea urchin Paracentrotus lividus, isopod crustaceans Idotea hectica, spider crabs Pisa mucosa and P. nodipes (Issel, 1918a; Vadon, 1981; Verlague, 1981; Boudouresque and Meinesz, 1982; Wittmann and Ott, 1982; Chessa et al., 1983; Lorenti and Fresi, 1983; Nédélec and Verlaque, 1984; Velimirov, 1984; Verlague, 1987, 1990; Pergent et al., 1994, 1997; Rico-Raimondino, 1995; Boudouresque and Verlaque, 2001). The modest grazing role of Sarpa salpa could constitute an artefact linked to human action; indeed, in a certain number of Marine Protected Areas (Tabarca and Mèdes Islands in Spain, Port-Cros in France, El Kala in Algeria) overgrazing by S. salpa was observed (Laborel-Deguen and Laborel, 1977; Pergent et al., 1993; Sanchez-Lizaso and Ramos-Espla, 1994; Tomàs-Nash, 2004). On the other hand, the leaf epibiota are widely used, in particular by the gastropods Bittium reticulatum,

Calliostoma langieri, Cerithium vulgatum, Columbella rustica, Gibbula umbilicaris, Rissoa sp. plur. and *Jujubinus* sp. plur. (Boudouresque and Meinesz, 1982; Templado-Gonzalez, 1982; Chessa *et al.*, 1983; Templado, 1984; Mazzella *et al.*, 1986). Furthermore, macro-herbivores which graze on leaves simultaneously eat the leaf epibiota borne on these leaves; *Paracentrotus lividus* even prefers the leaves covered with epibiota to leaves without epibiota (Traer, 1979).

Much of the primary production (24 to 85%) is **exported** in the form of dead leaves (Fig. 28; Ott and Maurer, 1977; Francour, 1990; Boudouresque *et al.*, 1994b; Pergent *et al.*, 1994; Mateo-Minguez, 1995; Cebrian and Duarte, 2001). In other ecosystems, these dead leaves represent a sizeable food source: they may constitute up to 40% of the digestive contents of the sea urchin *Paracentrotus lividus* in a hard substratum community some hundreds of metres away from the nearest meadow (Verlaque and Nédélec, 1983; Cebrian and Duarte, 2001). They may also pile up temporarily on beaches, forming "**banquettes**": particularly big "banquettes," up to 2.5 m high, can be seen in Sardinia (Alghero), in Corsica (Cap Corse), in Sicily (Marsala) and in Libya (Fig. 27; Boudouresque and Meinesz, 1982; Bellan-Santini and Picard, 1984; Farghaly and Denizot, 1984).

Some of the dead leaves of *Posidonia oceanica* remain inside the meadow, where they form the **litter**. The litter is most abundant in summer and autumn (shallow depth) and in winter (great depth). Its mass increases with depth, and represents between 25 and 200% compared with the biomass of live leaves (Pergent-Martini *et al.*, 1992b; Romero *et al.*, 1992; Mateo-Minguez, 1995). Leaf litter decomposition (by micro-organisms and detritus feeders) is fairly slow: after one month, at 20 m depth, 11% (Ischia, Italy; Pergent *et al.*, 1994) to 35% (Marseille, France; Rico-Raimondino, 1995) only of its mass had disappeared. After 6 months (in Ischia, Italy) the percentage of degradation reached 64% (5 m) and 44% (20 m) (Pergent *et al.*, 1994). **Consumption by detritus feeders** is the main way primary production of *P. oceanica* leaves is transferred to higher trophic levels (Chessa *et al.*, 1983; Boudouresque *et al.*, 1994b).

At the basis of the detritus food web are detritus feeders, such as sea urchins *Psammechinus microtuberculatus* and *Spaerechinus granularis*, amphipod crustaceans *Atylus guttatus*, *Melita palmata* and *Gammarella fucicola*, the isopod *Zenobiana prismaticus* and the Brachyura *Sirpus zariquieyi*, which dilacerate the dead leaves (Wittmann *et al.*, 1981; Campos-Villaça, 1984; Paul *et al.*, 1984; Vadon, 1984; Mazzella *et al.*, 1986). Further down the food web are found the



holothurian *Holothuria tubulosa* and the brittle stars *Ophiura texturata* and *Ophioderma longicauda* (Verlaque, 1981; Zupi and Fresi, 1984; Coulon and Jangoux, 1992). In reality, the effective degradation is carried out especially by bacteria and fungi¹⁵, for example *Corollospora maritima*, the other organisms actually only have a role in fragmenting the leaf debris (Cuomo *et al.*, 1982; Mazzella *et al.*, 1995).

Many predators feed off small invertebrates, whether these live on the leaves, in the litter or within the "matte" (Fig. 28): the starfish *Echinaster sepositus* and *Asterina panceri*, the crustacean *Palaemon xiphias*, the molluscs *Chauvetia minima* and *Sepia officinalis* and the fishes *Coris julis*, *Diplodus annularis*, *Hippocampus guttulatus*, *Symphodus cinereus*, *S. doderleini*, *S. ocellatus*, *S. roissali* and *S. rostratus*, etc. (Galan *et al.*, 1982; Templado-Gonzalez, 1982; Chessa *et al.*, 1983; Ballesteros *et al.*, 1984; Fresi *et al.*, 1984; Lejeune, 1985; Harmelin-Vivien and Francour, 1992). The sea urchin *Paracentrotus lividus* is eaten by the starfish *Marthasterias glacialis*, the spider crab *Maja squinado* and the fishes *Coris julis*, *Diplodus vulgaris*, *D. sargus*, *Sparus aurata*, *Symphodus mediterraneus*, *S. roissali* and *S. tinca* (Dance and Savy, 1987; Savy, 1987; Boudouresque and Verlaque, 2001). The bivalve noble pen shell *Pinna nobilis* is eaten by the octopus *Octopus vulgaris* (De Gaulejac, 1989). Fish predators are the scorpion fish *Scorpaena notata*, *S. porcus*, *S. scrofa*¹⁶, the combers *Serranus cabrilla* and *S. scriba* and the conger *Conger conger* (Harmelin-Vivien, 1984; Lejeune, 1985; Harmelin-Vivien *et al.*, 1989).

The **plankton-eating** fish, such as *Chromis chromis*, *Spicara smaris* and *S. maena*, by day exploit the water column, and at night sleep in the meadow (Harmelin-Vivien, 1984). There they are likely to be eaten by fish-eating predators, some of which are active by night, such as the congers and the scorpion fish (Harmelin-Vivien, 1982, 1984), this represents an **input** of organic carbon into the ecosystem (Fig. 28; Boudouresque *et al.*, 1994b).

Another energy **input** into the ecosystem is constituted by **filter** and **suspension feeders**, such as the leaf epibiota (hydroids, bryozoans) or of rhizomes (e.g. the polychaeta *Spirographis spallanzani*, the bivalve *Pinna nobilis* and the ascidian *Halocinthia papillosa* (Fig. 28; Mazzella *et al.*, 1986; Boudouresque *et al.*, 1994b).

Some features of the *Posidonia* oceanica ecosystems are unusual in the marine environment, and resemble the **continental forest ecosystems** (Boudouresque, 1996): (i) the accumulation of



Fig. 28. Trophic relationships and functional compartments in the *Posidonia oceanica* ecosystem. From Boudouresque et al. (1994b), modified.

15 Fungi (=Eumycetes) are one of the groups that used to be called 'mushrooms'. In the popular sense of the term, 'mushrooms' are a polyphyletic, i.e. heterogeneous and artifical, group (Lecointre and Leguyader, 2001).

¹⁶ Of the 3 species of scorpion fish, Scorpaena scrofa is the greatest consumer of fish (Harmelin-Vivien et al., 1989).

vegetal biomass over many decades, resulting in exceptionally high vegetal biomass. (ii) relatively modest animal biomass, concentrated in the "matte". (iii) low consumption of the primary production by herbivores, the main route for its transfer to upper trophic levels being the detritus fedders. However, the system is more open than the forest ecosystems, with organic carbon inputs (filter feeders, suspension feeders, plankton-eating fish), consistent organic carbon outputs (dead leaves, adult fish) and, especially, a poor retention of nutrients derived from organic matter recycling. Lastly, it is an original¹⁷ system, considering the juxtaposition of two kinds of primary production (slow recycling and rapid recycling) and the presence of the "matte" that acts as a **sink**¹⁸ (Fig. 28) for organic carbon and for nutrients (Pergent *et al.*, 1994; Boudouresque, 1996).

17 This originality is shared with other marine Magnoliophyte ecosystems.

18 A sink organic carbon and nutrients are buried too deeply in the "matte" to participate in recycling. They are sequestrated for a very long period of time. In certain conditions, and on a geological time scale, the organic carbon thus sequestrated can participate in the genesis of hydrocarbons (Burollet and Oudin, 1979).

3. THE ROLE OF POSIDONIA OCEANICA MEADOWS

3.1. GENERAL REMARKS

The role of *Posidonia oceanica* meadows in marine coastal environments is often correctly compared to that of the forest in terrestrial environments. These meadows constitute the basis of the richness of coastal waters in the Mediterranean, given the surface area they occupy (20-50% of the seabeds between 0 and 50 metres depth), and, in particular, given their essential biological role in maintaining the equilibrium of coastal waters and their concomitant economic activities.

3.2. ECOLOGICAL ROLE IN COASTAL SYSTEMS

Like most of the settlements of marine Magnoliophyte species, *Posidonia oceanica* meadows have a vital ecological role.

Darwin (1859) was the first to say that marine Magnoliophytes can constitute a food base for many species of macro-herbivores (marine turtles), but research done over the past few decades allows us today to better determine, and, especially, quantify, the true part these species play in the functioning of the coastal food web.

These Magnoliophytes produce enormous quantities of vegetal biomass that forms the basis of many food webs (McRoy and McMillan, 1977; Mazzella *et al.*, 1992; Pergent-Martini *et al.*, 1994; Romero, 2004b). This primary production is comparable to or greater than that of other high-production environments, whether terrestrial (temperate and tropical forests, cereal crops) or ocean (upwelling areas¹⁹, mangroves, coral reefs and estuaries) (summary in Fergusson *et al.*, 1980). Marine Magnoliophyte meadows, which only occupy 0.15% of the world's ocean surface, produce **1%** of the ocean's net primary production, i.e. 6 Gt C/year (*in* Duarte and Chiscano, 1999: Templado, 2004). This is so for the *Posidonia oceanica* meadows, one of **the planet's most productive** ecosystems (see § 2.6).

However, as in most of the ecosystems based on marine Magnoliophytes, very little of the primary production is directly consumed by herbivores (Fig. 29). Most of this production is either **(i) stored** (in the "matte") or degraded (by detritus feeders) in the meadow's litter; or **(ii) exported** to other ecosystems as dead leaves (Pergent *et al.*, 1994). The exporting of big amounts of dead *Posidonia oceanica* leaves (Fig. 30) is a boon for the deeper areas (with little or no light) and the beaches that benefit from this allochthonous production (Wolff, 1976; Walker *et al.*, 2001).

Most of *Posidonia oceanica*'s production is thus assimilated by detritus feeders (micro-organisms, crustaceans, gastropods, echinoderms) which will then themselves be eaten and integrated within the food web. The few macro-herbivores that are present (the edible sea urchin *Paracentrotus lividus*, the isopod crustacean *Idotea baltica* and the fish *Sarpa salpa*) can nevertheless play a major part locally according to their numbers (Pergent *et al.*, 1993; Alcoverro *et al.*, 1997; Havelange *et al.*, 1997; Romero, 2004b).

¹⁹ Upwelling rise of deep waters, generally rich in nutrient salts. Upwellings are sites of an important production of plant and animal plankton. It results a large abundance of fish, and therefore very active fisheries.



Fig. 29. Future of the primary production (in percentage of carbon) of *Posidonia oceanica*. The primary production of MPOs leaf epibiota is not considered here. After G. Pergent (unpublished).

It should also be noted that the Magnoliophytes offer support for many species of Multicellular Photosynthetic Organisms (MPOs) leaf epibiota that have large primary production, which is added to that of the Magnoliophyte itself (Modigh *et al.*, 1998), especially specific food for many animal species (Borowitzka and Lethbridge, 1989; Mazzella *et al.*, 1992; Havelange *et al.*, 1997). With *Posidonia oceanica*, the biomass of leaf epibiota varies between 6 and 34% of the above-ground biomass (Mazzella and Ott., 1984; Lepoint *et al.*, 1999). The bacteria present on the plant and in the substratum, as well as the high phytoplanktonic production measured at canopy make a significant contribution to this production (Velimirov and Walenta-Simon, 1992; Elkalay, 2002; Gobert, 2002).

One of the results of plant photosynthesis is also the **production of oxygen** (Fig. 30). Even if at death of the leaves part of this oxygen is consumed in their degradation (Mateo and Romero, 1996), the production of oxygen can be considerable as regards the shoots and the associated MPOs leaf epibiota,

particularly at shallow depth (Alcoverro *et al.*, 1998). The quantities produced are greatly in excess of requirements, and the *Posidonia oceanica* meadows therefore constitute a major element for water oxygenation. For example, at a depth of 10 m, in Corsica, 1 m² of meadow produces up to 14 litres of oxygen per day (Bay, 1978).

Lastly, *Posidonia oceanica* meadows constitute a **spawning ground**²⁰, a **nursery** or a permanent habitat for very many species (Fig. 30); over 400 different plant species and several thousand animal species populate the *P. oceanica* meadows, making these underwater meadows a unique **biodiversity hotspot** (Boudouresque and Meinesz, 1982; Bell and Harmelin-Vivien, 1982; Bellan-Santini *et al.*, 1994; Francour, 1997; Boudouresque, 2004). As basis of the food web, the meadows are an essential factor in the organisation of animal communities and control the complexity of habitats, the diversity of species and the abundance of associated invertebrates (Heck and Wetstone, 1977; Stoner, 1980; Mazzella *et al.*, 1992). All these species live on the surface of leaves (attached or vagile²¹), the sediment, around the leaves and also within the "matte" that shelters a particularly rich and varied fauna (Bellan-Santini *et al.*, 1986; Francour, 1990; Somaschini *et al.*, 1994).



²⁰ Spawning ground= place of laying.

21 A vagile species is a mobile species, by opposition to fixed species.



3.3. PHYSICAL ROLE IN COASTAL SYSTEMS

On the coastal seabed, *Posidonia oceanica* meadows are real plant barriers that encourage the decantation and sedimentation of particles in suspension in the water column (trapping sediment) (Boudouresque and Meinesz, 1982; Boudouresque and Jeudy de Grissac, 1983; Jeudy de Grissac and Boudouresque, 1985; Gacia and Duarte, 2001; SDAGE, 2003; Romero, 2004b). This sediment is then retained under the canopy²², between the rhizomes and the roots, thus forming a unique structure, the "matte", where it represents between 20 and 60% of the volume (Fig. 31, 32; Jeudy de Grissac, 1984a). This sediment of allochthonous origin, associated with the autochthonous sedimentation (debris of organisms that had lived on the leaves and at the base of the rhizomes), generates the vertical growth of rhizomes (and thus the "matte"), enabling it to fight against burial

(Molinier and Picard, 1952). This speed of growth is very variable according to the site and the time scale; the elongation of rhizomes, measured via lepidochronology, varies between 3.2 and 20.9 mm/year (7.4 mm/year on average in Pergent et al., 1995); higher values have been observed (Boudouresque et al., 1984). The vertical growth, measured over several decades, corresponds to the



Fig. 31. Trapping sediment and reduction of hydrodynamism in a *Posidonia oceanica* meadow. From Boudouresque and Meinesz (1982, modified).

balance between the accretion and erosion phenomena (Blanc and Jeudy de Grissac, 1978; Mateo *et al.*, 1997); it can reach 1 m/century (Molinier and Picard, 1952). The meadow plays a similar role to European beachgrass and pine trees that stabilize the coastal sand dunes (trapping sediment, and **stabilization**). It should also be noticed that sediment settling (mainly fine particles) and trapping in the "matte" help increase the **transparency** of the coastal water (Boudouresque and Meinesz, 1982; Jeudy de Grissac and Boudouresque, 1985).

The *Posidonia oceanica* meadow's considerable vegetal biomass also acts as a kind of barrier which slows and **effectively absorbs hydrodynamism** (swell, currents) at the seabed (Fig. 31). This reduction in hydrodynamism was



Fig. 32. A view of the thickness of the "matte" at an erosion scarp. Photo G. Pergent.

measured in the laboratory (ICI company, Delft laboratory, Holland; unpublished data) and *in situ* in extensive meadows (Jeudy de Grissac and Boudouresque, 1985; Gambi *et al.*, 1989; Gacia and Duarte, 2001; Duarte, 2004). Hydrodynamism was reduced by 10 to 75% under the cover of leaves (Jeudy de Grissac, 1984a; Gambi *et al.*, 1989; Gacia *et al.*, 1999), which reduces sediment re-suspension during storms (Gacia *et al.*, 1999; Terrados and Duarte, 2000; Gacia and Duarte,

2001; Duarte, 2004). Hydrodynamism was also reduced above the meadow. Several dozen centimetres above the canopy, the reduction in the speed of the current is 20% (Gacia and Duarte, 2001).

22 The canopy is constituted by all the leaves.



Fig. 33. "Banquette" of *Posidonia oceanica* dead leaves. From Boudouresque and Meinesz (1982).

For other marine Magnoliophyte species of similar size and structure (for example, *Zostera marina* and *Amphibolis griffithsii*), hydrodynamism reduction values both above and beneath the leaves are of the same order (Fonseca *et al.*, 1982a; Gambi *et al.*, 1990; Komatsu, 1996; Van Keulen and Borowitzka, 2002).

This hydrodynamism reduction is particularly visible behind the *Posidonia oceanica* barrier reefs, where

the presence of these plant barriers enables sheltered lagoons to be established (Molinier and Picard, 1952; Boudouresque and Meinesz, 1982; Bellan-Santini *et al.*, 1994; Boudouresque *et al.*, 1994b). More generally, the reduction in waves and currents is likely **to protect from coastal erosion** and helps stabilize the shoreline (Blanc and Jeudy de Grissac, 1978; Jeudy de Grissac and Boudouresque, 1985; Short *et al.*, 1989; Gacia and Duarte, 2001; Duarte, 2004). There are many instructive examples of coastal erosion after the regression of marine Magnoliophyte meadows (Larkum and West, 1990; Pergent and Kempf, 1993; Pasqualini *et al.*, 1999).

In the autumn, the increase in the mass of dead leaves (rate of leaf loss, size of leaves) plus the meteorological conditions (increased hydrodynamism, storms) drive large amounts of this dead matter towards the beaches (Boudouresque and Meinesz, 1982; Pergent *et al.*, 1997; Walker *et al.*, 2001). The leaves pile up on the shore as the currents take them, forming real banks (he eafte **banquettes**) likely **to protect the beaches** from erosion, particularly during the winter storms (Fig. 33; also see Chap. 2, Fig. 27) (Boudouresque and Meinesz, 1982; Jeudy de Grissac and Audoly, 1985; Chessa *et al.*, 2000; SDAGE, 2003). Despite the initially inhospitable²³ appearance of these banquettes, maintaining them on the beaches seems essential for the protection of the coast (see Chap. 6); in many communes (administrative districts), their removal (in the context of "cleaning up" the beaches) has often gone hand in hand with a significant regression of the shoreline (Pergent and Kempf, 1993; Pasqualini, 1997b).

3.4. ECONOMIC ROLE

Generally speaking, the economic role of *Posidonia oceanica* meadows is the result of its abovementioned ecological and physical importance in coastal system.

(1) It obviously affects the management of **living resources** via its high biological production, the protection from predators it offers to young fish and young organisms (nurseries), and it also constitutes a much sought-after spawning ground for many species of commercial interest (crustaceans, cephalopods, fishes) (Jimenez *et al.*, 1996; Francour, 1997; Romero, 1999; Le Direach and Francour, 2001). This role is seen throughout the world: in Australia, for example, *Zostera* and *Posidonia*²⁴ species are the feeding grounds of a commercially exploited species of fish (Connolly *et al.*, 2005).

(2) It also affects the development of tourism and seaside activities, by helping maintain **water quality** (transparency), and especially by stabilizing the shoreline (beaches) that it defends by

 23 "Where to put the towel?" Wonders, legitimately, the bather. 24 It is *Posidonia australis* and *P. sinuosa*.

protecting it from erosion (reduction of water movement,



banquettes of dead leaves). Furthermore, even if the meadows are not always the "spots" sought by divers, they are at the origin of significant exported biological richness (in terms of species and food) to other more sought-after habitats (rocky bottoms) (see § 2.6).

On the other hand, the economic importance of *Posidonia oceanica* meadows is often highlighted by the negative results of its regression or disappearance.

Thus the Gulf of Gabès, in Tunisia, was in the second half of the 20th century the site of a dramatic regression of the meadows caused by the joint action of very destructive fishing practices ("gangave" trawl net25), large-scale industrial waste (chemical, phosphogypsum, cement and agro-alimentary industry), unprecedented demographic development and a general silting of the area (Burollet et al., 1979; Darmoul et al., 1980; Pergent and Kempf, 1993; Ben Mustapha et al., 1999; anonymous, 2002a). The Gulf of Gabès, considered to be an extremely sought-after fishing area, stagnated over the 1980s and then experienced a significant decline (28 to 34%) in fish catch in the 1990s (Ben Mustapha, 1995), forcing the Tunisian Government to introduce a policy of redirecting the fishing fleet to less popular geographical areas (Pergent and Kempf, 1993). Furthermore, the Island of Jerba, lying to the east of the Gulf, whose economy is essentially dependent on seaside tourism (intake capacity 15 000 beds in 1990), witnessed the erosion of its shoreline, in some places up to several dozen metres, thus depriving several hotels of their beaches, even of part of their infrastructure, and requiring the construction of costly protection facilities whose medium-term efficacy has not been proven (Pergent and Kempf, 1993; anonymous, 2002a). This regression of the shore could be due to that of the Posidonia oceanica meadows in the Gulf of Gabès (see Chap. 4).

Table I. Average annual value of services provided by a few major types of terrestrial and marine ecosystems. Mha= million hectares. G\$= billion US dollars. From Costanza *et al.* (1997).

ECOSYSTEMS	Surface area (in Mha)	Value/ha/year	Total value/year
TERRESTRIAL			
Temperate and boreal forests	2 955	\$302	G\$894
Tropical forests	1 900	\$2 007	G\$3 813
Meadows	3 898	\$232	G\$906
Wetlands	330	\$14 785	G\$4 879
Lakes and rivers	200	\$8 498	G\$1 700
Other (deserts, tundra, glaciers, etc.)	6 040	< \$100	< G\$130
Total	15 323	\$804	G\$12 319
MARINE			
Oceans (high seas)	33 200	\$252	G\$8 381
Oceans (coastal areas)	2 660	\$1 610	G\$4 283
Estuaries	180	\$22 832	G\$4 110
Macrophyte habitats (meadows, etc.)	200	\$19 004	G\$3 801
Coral reefs	62	\$6 075	G\$375
Total	36 302	\$577	G\$20 949

²⁵ The gangave is a kind of trawl constituted by a beam, on which are arranged hooks and nets, that is towed in shallow depths, in particular in Tunisia, for harvesting sponges.

More generally, although the importance of the natural ecosystems is universally recognised, particularly as regards maintaining the natural equilibrium (ecological role), their overall economic value is harder to assess (Costanza *et al.*, 1997; Ami and Boudouresque, 2002). This economic assessment must take into account direct profits (for example, fishing and diving), indirect profits (ecosystem services, for example protecting the shore from erosion, water oxygenation), and optional values (for example, future uses). Beyond the figures advanced it is interesting to notice that marine meadows are, on a world scale, one of the ecosystems whose economic value (US\$19 000 per hectare per year) is **the highest**: 10 times greater than the tropical forests and 3 times more than the coral reefs (Table I: Costanza *et al.*, 1997).

3.5. BIOINDICATOR ROLE

For several years now, the use of marine Magnoliophytes for environmental monitoring, to assess the evolution of impact, or more generally to manage coastal ecosystems has been envisaged (Brix *et al.*, 1983; Augier, 1985; Ward, 1987; Maserti *et al.*, 1988; Pergent, 1991b; Phillips, 1994; Abal and Dennison, 1996; Fourqurean *et al.*, 1997). The use of these species, known as "bioindicators," seems to be a quick and effective way of assessing the **quality of the environment** (Bellan, 1993). Their wide geographical distribution (Hartog, 1970), their longevity, their permanence over the seasons, the ease with which they can be sampled, their abundance, and their ability to concentrate a vast range of xenobiotics²⁶ (McRoy and Helferich, 1980; Ward, 1989), make marine Magnoliophytes potentially interesting organisms.

In the Mediterranean, Posidonia oceanica meadow constitutes a powerful integrator of overall marine water quality (Augier, 1985; Pergent, 1991b; Pergent et al., 1995). Very widely distributed all along the coast, particularly sensitive to pollution (Augier et al., 1984a; Bourcier, 1989) and other pressures linked to human activities (Ardizzone and Pelusi, 1984; Meinesz and Laurent, 1978; Boudouresque and Meinesz, 1982; see Chap. 4), with benthic characteristics, it demonstrates by its presence and its vitality (or its regression, betrayed by "dead mattes") the quality of the water above it. The footprint of water quality on the P. oceanica meadows is permanent: it does not therefore depend on wind or current direction at the time of observation. Many parameters are thus likely to be recorded by the meadow: (i) the average turbidity of the water (revealed by the position of the lower limit of the meadow and its shoot density); (ii) currents and hydrodynamism (revealed by the erosive structures that affect the "matte"); (iii) rate of sedimentation (revealed by the speed of growth of the rhizomes and, by their loss of hold where there is a deficit); (iv) stable pollutants (concentration and memorization of concentration over time); (v) desalination at the mouths of coastal rivers or underground water (revealed by the disappearance of the meadow); (vi) stress (revealed by the plant's level of phenolic acids and detoxication enzymes); and (vii) organic matter and nutrients (revealed by the leaf epibiota and the plant's chemical²⁷ composition). However, although several of these descriptors are today well understood (standardization of measuring, quality grids) and provide reliable information that can be reproduced, the decoding of other descriptors is still ongoing (Pergent et al., 1995).

26 Xénobiotics chemical elements introduced by man into the environment and having a negative impact on the organisms and/or the ecosystems.

27 Content of the various organs of the plant in carbon and nitrogen. Among the validated information that has been routinely used for years we should mention the assessment of the water's average **turbidity**. Like all plants, *Posidonia oceanica* needs light to carry out photosynthesis; thus, its maximum bathymetric extension (lower limit) is a function of the amount of light received, and thus of the average transparency of the



Moreover, the **type of limit** observed is also able to inform us about temporal changes in water transparency (Meinesz and Laurent, 1978; Boudouresque and Meinesz, 1982): (i) progressive limits (Fig. 34), characterized by the presence of horizontal rhizomes growing parallel to the slope and that colonise substrata located further down, theoretically demonstrate an overall improvement in water transparency; (ii) sharp limits, characterized by the presence of partially vertical rhizomes without a deeper "dead matte" often indicate stable water transparency; and (iii) regressive limits, characterized by the presence of "dead matte" and a few deeper sample shoots, indicate a withdrawing meadow linked to an increase in the average turbidity of the water (Fig. 38).

Another descriptor that is particularly interesting to take into account, even if it is still being standardized, is stable **pollutant** concentration (Pergent-Martini and Pergent, 2000). Like many Magnoliophytes, *Posidonia oceanica* presents both **(i)** great power of "trace metal" concentration, proportional to the levels present in the environment (Augier, 1985; Capiomont *et al.*, 2000; Pergent-Martini and Pergent, 2000; Baroli *et al.*, 2001), and **(ii)** good resistance to metallic contamination (the species persists near major sources). Additionally, its ability to be kept in an aquarium for artificial contamination experiments (Ferrat *et al.*, 2002a), and especially its ability to memorize former levels within its tissues, allied to the possibilities of dating offered by lepidochronology, open up unique prospects for the temporal monitoring of pollution (Calmet *et al.*, 1988, 1991; Carlotti *et al.*, 1992; Pergent-Martini, 1998; Pergent and Pergent-Martini, 1999) and allow us to keep true biological archives that can inform us of the temporal development of a particular pollution (Fig. 35).









Fig. 35. Evolution over time of the mercury concentration (g.g-1) in the petioles of *Posidonia oceanica* in three localities of the Western Mediterranean. Dating by lepidochronology. From Pergent and Pergent-Martini (1999).

4. THE CAUSES OF POSIDONIA OCEANICA MEADOWS REGRESSION

During the 20th century, and certainly more especially since the 1950s, Posidonia oceanica meadows has considerably regressed, particularly around major urban centres and ports: Barcelona (Spain), Marseille, Toulon, Nice-Villefranche-sur-Mer (France), Genoa, Naples, Trieste (Italy), Athens (Greece), Alexandria (Egypt), Gabès (Tunisia) etc. (Pérès and Picard, 1975; Boudouresque and Meinesz, 1982; Pérès, 1984; Boudouresque, 1996, 2003; Romero, 2004b; Solis-Weiss et al., 2004; Figs. 36 and 37). Meadow is regressing in deep areas (withdrawal of the lower limit because of the reduction in water transparency) (Fig. 38), at intermediate depths, and also at its upper limits. For Liguria (Italian region) as a whole, meadows have lost between 10 and 30% of their surface area compared to the beginning of the 20th century (Bianchi and Peirano, 1995; Peirano and Bianchi, 1995). In Genoa, the P. oceanica meadow has become extremely scattered, and has even disappeared from many kilometres of the coast (Balduzzi et al., 1984; Bianchi and Peirano, 1995). In Latium (Italian region), the regression is general, and P. oceanica has been replaced in some areas by another Magnoliophyte, Cymodocea nodosa (Diviacco et al., 2001). In the Alicante region (Spain), Ramos-Esplà et al. (1984) believe that 52% of the surface area of meadows has been destroyed. In Marseille, almost 90% of the meadow mapped in the late 19th century by Marion (1883) has disappeared today (Boudouresque, 1996). In the Hérault (France), the meadow which used to stretch from Carnon to Agde over several dozen kilometres has mostly disappeared (Foulquié and Dupuy de la Granrive, 2003). This is so also in true in the Toulon Gulf (Var, France; Bourcier et al., 1979), and in the Gulf of Gabès (Tunisia; CNT in Pergent and Kempf, 1993). But it should be noted that the decline is not generalised; in some regions, the limits of the P. oceanica meadows have remained stable. This is so around the island of Ischia (Gulf of Naples, Italy), where Colantoni et al. (1982) noted a fairly stable stretch between the late 1920s, the 1950s and the 1970s.

In their seminal work on *Posidonia oceanica* meadows, Molinier and Picard (1952) had already noticed this regressive trend. They put forward the hypothesis that it was at least partially due to a **lack of adaptation** of the plant to the Mediterranean's present hydrological and climate conditions, especially in the north-western coasts. Molinier and Picard's (1952) speculation was based on 2 things: **(1)** the rarity of the plant's flowering and fruit formation, especially in the western Mediterranean, and **(2)** the ageing of individual plants, deduced from the thickness of the "mattes," which seemed to imply a lifespan of several thousand years.

In fact, flowering and fruit formation are not as rare as had been supposed, except for the Gulf of Lion (see §2.2). Furthermore, successful reproduction of a plant with a very long life, a K²⁸ strategist (as defined by MacArthur and Wilson, 1967), does not require annual reproduction. Moreover, although the present warming of the Mediterranean (Béthoux and Gentili, 1998; Salat and Pascual, 2002) can disadvantage *Posidonia oceanica* in the eastern Mediterranean, this should rather encourage it²⁹ in the north-western Mediterranean, where the low winter temperatures are a limiting factor (see §2.3). In fact, if *P. oceanica* does have a fragility factor, this is rather its low genetic variability (Capiomont *et al.*, 1996; Raniello and Procaccini, 2002).

28 "K" strategists are species that invest most of their energy in their vegetative system and in (chemical or other) defence; for reproduction, they gamble on quality rather than quantity the chances of success of reproductive elements are maximized; these are often long-living species which reproduce discontinuously. Conversely, "r" strategists invest most of their energy in reproduction, preferring quantity to quality; these are often short-lived species which reproduce continuously.

29 Though Mayot et al. (2005) present data that do not confirm this hypothesis.




ded

It should be said that in the area it occupies today, *P. oceanica* (and the species that preceded it) has for 100 million years successfully come through very severe geological and climatic events. In particular, it survived the Messinian crises³⁰ some 5.3 to 5.6 million years ago (Krigsman *et al.*, 1999; McKenzie, 1999; Taviani, 2002), then the ice ages (that occur every 100 000 years), and then the alternately cold and hot climate cycles of 1 500 years that characterize the Quaternary climate (Clark *et al.*, 1999; Bradley, 2000; DeMenocal *et al.*, 2000; McDermott *et al.*, 2001; Crowley, 2002; Esper *et al.*, 2002). The recent decline of *P. oceanica* throughout most of the Mediterranean can thus hardly be solely attribute to a recent "lack of adaptation" or to the warming of the water (Béthoux and Gentili, 1998; Salat and Pascual, 2002) which has been seen for about thirty years now.

It is clear that **human activities** are the main factor in the regression of the *Posidonia oceanica* meadows. Although the causes usually act in synergy (see §4.13), and it is not always easy to isolate them (Fig. 39), we shall first consider them separately.

It is, however, important to notice that in a *Posidonia oceanica* meadow the presence of

Fig. 38. The regression of the *Posidonia oceanica* meadow is often shown in the rising of its lower limit. This results in 'dead matte' downstream from the present limit. From Meinesz and Laurent (1978), redrawn by C.F. Boudouresque.

sandy **intermattes**, or "dead matte" intermattes, can be quite normal, resulting from the meadow's natural dynamics, with the alternate shoot death and recolonization (see §2.5; Boudouresque *et al.*, 1986a, 1986b; Meinesz *et al.*, 1988). Using the percentage of "dead matte" compared to living

meadow as an index of the meadow's degradation (Moreno *et al.*, 2001) must thus be done with extreme caution. Similarly, some years, when the light is weak, the carbon budget (photosynthesis/loss) can show a loss, as happened in 1993 at 5 metres depth in the Mèdes Islands (Catalonia); then the plant draws on its reserves (stored in the rhizomes) and reduces shoot density (Alcoverro *et al.*, 2001). Man's impact is revealed in the multiplying of intermattes, which may come together, and in a regular decrease (over several years) of shoot density. In extreme cases, the meadow is completely replaced by vast stretches of "**dead matte**"; the very long persistence of the little putrescible rhizomes is a



Fig. 39. A residual clump of *Posidonia oceanica*. The causes of its very poor condition are certainly multiple pollution, leaf epibiota overload, turbidity etc. Photo by GIS Posidonie.

useful tool for determining the former extent of meadows and possibly for dating their disappearance³¹ (Meinesz and Laurent, 1978; Boudouresque *et al.*, 1980c).

The sensitivity of *Posidonia oceanica* meadows to human impacts makes this ecosystem the **biological indicator** *par excellence* of such impacts on the coastal environment (see §3.5;

Pergent *et al.*, 1995; Boudouresque *et al.*, 2000; Guidetti, 2001; Charbonnel *et al.*, 2003). The effectiveness of choosing *P. oceanica* for impact studies is increased by the role the meadows play in the coastal equilibrium in the Mediterranean (see §3.2 and 3.3). Furthermore, the very wide distribution of *P. oceanica* around the Mediterranean (see §2.1) enables comparative studies to be performed on the most diverse scales, from one particular sector to the entire Mediterranean basin.

- ³⁰ During the Messinian crises, the Strait of Gibraltar closed and most of the Mediterranean dried up. We do not know what allowed *Posidonia oceanica* to survive.
- ³¹ In the Prado Bay in Marseille, carbon 14 dating of "dead matte" located at depth (30 metres) has shown that the death of *Posidonia oceanica* probably happened in the 13th century, contrary to what one might have imagined, and thus cannot have been caused by recent human activity (Gravez et al., 1992).

4.1. COVERING OR DIRECT INCLUSION IN COASTAL DEVELOPMENT AND MODIFICATION OF SEDIMENTARY FLOW

The building of facilities such as sea walls, platforms on land claimed from the sea and ports represents a major threat to coastal environments, especially to *Posidonia oceanica* meadows. In Liguria (Italy) there used also to be urban solid waste discharge at sea (Balduzzi *et al.*, 1984), a practice that fortunately no longer exists today. To such direct impacts must be added indirect impacts (pollution, turbidity, modification of sedimentary flow, etc.).

In the Provence-Alpes-Côte d'Azur region (France), the **direct** impact of coastal development on shallow depths, and thus on the potential habitat of *Posidonia oceanica*, affects 16% of the coastline and 15% of the surface area of the seabed at less than 10 metres depth (Table II page 29). In Liguria, these percentages are even higher (Regione Liguria, 2000). These shallow depths, where light is not a restricting factor, are among the most productive of the marine environment; moreover, it is there that we find the nurseries of many species of fish of commercial interest (Boudouresque, 1996).

As well as direct impacts, building coastal facilities changes swells and currents locally, and consequently the processes of coastal **sediment transport** that determine their distribution: erosion or accumulation (Astier, 1984). Such changes alter the balance between the rate of sedimentation and the vertical growth of the rhizomes which responds to this. Excessive accumulation of sediment determines the covering of the vegetative tips of *Posidonia oceanica*; if the rate of sedimentation is greater than 5-7 cm/year, the vegetative tips will die (Fig. 40; Boudouresque *et al.*, 1984); conversely, if this rate is zero or negative (the sediment is disappearing), the rhizomes are bared (Fig. 41); they are then extremely easily broken off (hydrodynamics, anchors, trawling, etc.) (Boudouresque and Jeudy de Grissac, 1983).

Fig. 40. A *Posidonia oceanica* shoot buried under 8-10 cm of sediment for less than a year. The leaves, which have been unable to pierce the sediment, have folded in accordion pleats. If the burial continues, the vegetative tip (and thus the shoot) dies. From Boudouresque *et al.*, 1984).



The building of any facility that projects out into the sea (port, seawall running straight out to sea) displaces the current out towards the open sea, with **hypersedimentation** upstream (sediment in transit deposited) and **erosion** (deficit of sediment) downstream (Blanc and Jeudy de Grissac, 1989). Seawalls running parallel to the shore also modify currents and deflect swells, which can have a similar effect. The larger is the facility, the greater is the effect. In Monterosso (Liguria, Italy), Gongora-Gonzales *et al.* (1996) observed the burying of an area of *Posidonia oceanica* meadow under fine sediment, linked with coastal development on land claimed from the sea. Similar observations concern southern Spain (Ruiz-Fernández, 2000).

As well as changes in current and sedimentary flow, ports and platforms can, when being built, generate **turbidity**. Hydrodynamic activity sweeps part of the terrigenous material deposited out at sea. This turbid cloud acts in 3 ways: it reduces water transparency (and thus photosynthesis); it is deposited on the meadow (hypersedimentation); and lastly, the finest particles of sediment are re-suspended during storms, reducing in the long term water



Table II. Percentage of surface area of infralittoral seabed and percentage of coastline occupied by coastal development in the Provence-Alpes-Côte d'Azur region (France) from Martigues to Menton. From Meinesz et al., 1981a, 1982, 1990a, 1991b).

Sector	Seabed 0-10 metres	Seabed 0-20 metres	Coastline
East of Bouches-du-Rhône	27%	19%	21%
Var	11 %	7%	12%
Alpes-Maritimes and Monaco	20%	12%	24%
The region, as a whole	15%	10%	16%

transparency (Charbonnel, 1993). In Le Mourillon (Toulon, Var, France), Astier (1984) has shown the direct destruction (covering up) of **22 hectares** of *Posidonia oceanica* meadow, followed by the indirect destruction of **10 hectares** and the silting up of **27 additional hectares**. Similarly, in the Prado Gulf (Marseille, France), the building of the Pointe Rouge port directly destroyed **11 hectares** and indirectly **68 hectares** (Gravez *et al.*, 1992; Charbonnel *et al.*, 1995d).

Lastly, the port basins are often very polluted sites because of anti-fouling paint and discharge of waste water from boats. This pollution then spreads to the area surrounding the ports. Fig. 41. An exposed *Posidonia oceanica* meadow. At the bottom one can clearly see the rhizomes that are bared to over 10 cm. The sediment that would usually fill the interstices has been carried away. Such a meadow is extremely fragile (hydrodynamics, anchors, trawling, etc.). Photo by A. Meinesz.



4.2. CHANGES DUE TO RIVER INPUTS

Coastal rivers can have an impact on *Posidonia oceanica* meadows by (i) desalination (to which the plant is very sensitive; also see §2.3), (ii) nutrient inputs ³² (also see §4.4), and (iii) sediment input.

The flow of Mediterranean coastal rivers is subject to very strong seasonal (floods) and inter-annual variations (in particular as a result of the NAO³³). These fluctuations are natural and it is possible to suppose that the meadows were never able to settle sustainably in an area influenced by a **plume of fresh water** during ten-year floods. Human correction of rivers has acted in 2 opposite ways: (i) channeling the course and reducing the surface area of the main bed (accessible to water during spates) accentuates peak outflow during floods. (ii) Building dams and reservoirs absorbs the outflow peaks during floods, at least in the initial phase. Furthermore, holding back water (reservoirs) and using it for farming increase water loss (evaporation, evapo-transpiration) and reduce the quantity of water that reaches the sea. In Spain, the course of the Ebro River (and its tributaries) is interrupted by over 100 dams (Prat, 1993). The case of the Nile, in the eastern Mediterranean, is particularly spectacular – its average flow at the mouth was 100 Mm³/year in the 19th century; this dropped to 84-86 Mm³/year after the dam downstream from Aswan was built in 1902, and then to

3 Mm³/year after the high Aswan Dam (Nasser Dam) was built in 1964 (Abu-Zeid, 1991; Abu-Zeid and El Moatassem, 1993). Arguing from the hypothesis that desalination (as well as high temperature) is a major factor of the absence of *Posidonia oceanica* east of the Nile delta, it is possible that this species will settle there over the coming decades.

³² Nutrients (= nutritive salts, mineral salts) are salts of nitrogen (nitrates, nitrites, ammonium), of phosphorus (phosphates) and of silicum (silica). Nutrients are essential to photosynthetic organisms.

³³ NAO=North Atlantic Oscillation. When the NAO is positive, the climate is dry in southern Europe and the Mediterranean basin; when it is negative, it is, on the contrary, damp (Hurrell et al., 2001; Tourre, 2002).

The **nutrients** naturally brought by coastal rivers do not seem to play an important role in *Posidonia oceanica*; its distribution around the Mediterranean shows that it can have a high level of vitality in very oligotrophic water³⁴ as well as in mesotrophic water³⁵. Man has caused a sometimes considerable increase in the nutrient input by rivers. This is particularly so for the Tiber (Italy; Izzo and Nicolai, 1993). The negative effect on *P. oceanica* is exerted by the proliferation of leaf epibiota and not by the eutrophication³⁶ as such (Pergent-Martini *et al.*, 1996; Pirc and Wollenweber, 1988; Ruiz-Fernández, 2000). In the Gulf of Fos (Bouches-du-Rhône, France) the almost total disappearance of *P. oceanica* (Pergent and Pergent, 1988) could be due to the nutrients and to other pollutants that were introduced by the Rhône and the Arc (via the Etang de Berre) during the second half of the 20th century, before waste water treatment reduced this form of pollution.

Building dams on watercourses leads to the reduction of their sedimentary load in the river mouths and the accumulation of sediment in the lakes of the dam. For example, the Rhône is interrupted by 19 dams between Geneva and the sea. Exploiting the sand and gravel (for the building industry) in the beds of watercourses has also helped reduce their sedimentary load. Today the Rhône carries 20% of the solid elements it carried in the late 19th century (Pont, 1993), and the Ebro 5% of the sediment it carried in the 1930s (Pearce, 1996). Almost all of the 124 Mt/year of sediment that the Nile used to carry is now trapped in the Aswan lake; in the Rosetta branch (one of the two branches of its delta), while the annual load of sediment in suspension used to be 68 Mt in 1958 this has now fallen to 0.5 Mt in 1990; as to the Damiette branch, while the annual load used to be 25 Mt/year it is now nothing at all, for the branch has dried up (Abu-Zeid, 1991; Abu-Zeid and El Moatassem, 1993; Stanley, 1993; Lamy, 1999). The result has been that the dam lakes have filled up. In Tunisia, the dam on the Mellègue river (a tributary of the Medjerda), built in the 1950s, has kept back 48 Mm³ of sediment in 20 years and is now 1/5 full; the capacity loss of Tunisian dams is on average 1-2.5% per year. As for Lake Nasser, behind the Aswan Dam (Egypt), it should have filled up in 2 centuries' time (Pearce, 1996; Lamy, 1999). The low sediment input by the coastal rivers is partly responsible (see §4.1) for the Posidonia oceanica rhizome baring, that weakens them (hydrodynamism, anchors, trawling) (Boudouresque and Jeudy de Grissac, 1983; Jeudy de Grissac and Boudouresque, 1985). In Giens Gulf (Var, France), the very common baring of P. oceanica has been attributed to the deficit in sediment of the coastal rivers and to the change of a coastal river's mouth (Gravez et al., 1988; Paillard et al., 1993).

Correcting watercourses can also modify the **granulometric** features of the sediment carried down in favour of the finest particles. When hydrodynamism stirs up these fine sediment particles in suspension, this generates great turbidity in the water column (see §4.3) which restricts the photosynthetic action of *Posidonia oceanica* and also influences deeper assemblages (Tunesi *et al.*, 2001).

35 Mesotrophic water fairly rich in nutrients (=nutritive salts).

³⁴ Oligotrophic water poor in nutrients (=nutritive salts) especially nitrates and phosphates.

³⁶ Eutrophic water richness in nutrients.

4.3. REDUCTION IN WATER TRANSPARENCY

Urban water and waters from other human activities can increase the coastal water load in suspended particles, nutrients and dissolved or solid organic matter. In their turn, nutrients and organic matter provoke the proliferation of planktonic organisms. All in all, water transparency is reduced.

When water transparency decreases this has a direct effect on Posidonia oceanica meadows. Compensation depth (the depth at which the losses due to respiration balance out with photosynthesis production) becomes shallower, and with it becomes shallower the lower limit of the meadow (Fig. 38). Ruiz and Romero (2001) have demonstrated experimentally, placing screens above a meadow at 8-10 metres depth, that a 30% reduction in light reduces by 30 days both growth rate, shoot biomass and storage of starch in the rhizomes; shoot density drops by 30% in 3 months; 1 year after normal light intensity has been reestablished there is still no sign of



recuperation. A 70% drop in light intensity provokes within 3 months the death of almost 90% of the shoots. Overall, the *P. oceanica* meadow's response to a lessening of light is first shown in a reduction of covering and shoot density, and then in the death of the meadow, which may be rapid – a few weeks (Fig. 42; Ruiz-Fernández, 2000; Ruiz and Romero, 2001, 2003).

The **rising of the lower limit** of the *Posidonia oceanica* meadows is a general phenomenon in most of the areas around the big urban centres and ports. In much of the Alpes-Maritimes (France), for example, the limit rose from 35 to 25 metres depth between the 1950s and 1970s (Meinesz and Laurent, 1978, 1980). Similarly, in the Prado Gulf (Marseille, France), the lower limit rose from 35 metres in the 19th century³⁷ (Marion, 1883) to 30 metres depth in the early 1960s (Harmelin and True, 1964), and then to between 20 and 25 metres depth starting since the 1970s (Foucher, 1975; Niéri *et al.*, 1986; Gravez *et al.*, 1992). In Latium (Italian region), the lower limit rose from 35 metres to 25-30 metres depth (Diviacco *et al.*, 2001). Despite the improved quality of the coastal waters due to the opening of many waste water treatment plants between the 1970s and 1990s, the lower edge of the *P. oceanica* meadows is still rising in the Provence-Alpes-Côte d'Azur region (France) (Boudouresque *et al.*, 2000; Charbonnel *et al.*, 2003; Mayot *et al.*, 2005).

4.4. PRESENCE OF EXCESSIVE AMOUNTS OF NUTRIENTS AND CHEMICAL CONTAMINANTS

Anthropogenic waste, as well as greatly changing the sedimentary balance of the coastal

water, also implies a wide range of contaminants, including detergents, hydrocarbons, pesticides, herbicides,

³⁷ However, turn to Note 31 at the bottom of Page 35 for a careful interpetation of this data. "heavy metals" and elements of "anti-fouling" paint that affect the vitality of the *Posidonia* oceanica meadows (Pérès and Picard, 1975; Augier *et al.*, 1987). The effect of soluble substances is rapid, but insoluble substances can also have a very negative impact; being relatively stable, they can accumulate to reach concentrations that are toxic to flora and fauna. The presence of a high contaminant concentration can determine the alteration of the biosynthesis of photosynthetic pigments: along an increasing pollution gradient, the photosynthetic pigment content in *P. oceanica* leaves decreases (Augier and Maudinas, 1979).

Detergents (tensio-actives) accumulate in the sediment, contaminating the roots, rhizomes and leaves (Pérès and Picard, 1975). Augier *et al.* (1984b) have shown *in vitro* that dobane (an anionic tensio-active widely used in manufacturing commercial detergents) slightly encourages *Posidonia oceanica*'s photosynthesis at very low doses (50 ppb, parts per billion) but greatly lessens it between 100 and 500 ppb, concentrations which are found at the inputs of non-purified water discharges. As well as the decrease in photosynthesis, Augier *et al.* (1984b) describe morphological and histological alterations in the leaves of *P. oceanica*.

Mercury is absorbed by the root system, and laboratory studies confirm that the mercury content of *Posidonia oceanica* is correlated to its concentration in the sediment (Cristiani, 1980). An accumulation of mercury in the foliar tissue causes the plant severe physiological disorder that can lead to cellular necrosis and the stop of leaf growth.

Posidonia oceanica's ability to accumulate large amounts of "heavy metals", particularly in its rhizomes, and the possibility of dating rhizome segments by using lepidochronology, makes this species a useful bioindicator, able not only to integrate the average concentration but also to reconstitute its evolution over previous years (see §3.5) (Giaccone *et al.*, 1988; Catsiki *et al.*, 1987; Pergent-Martini *et al.*, 1998; Pergent and Pergent-Martini, 1999).

In the case of nutrients, these determine the proliferation of the *Posidonia oceanica* leaf epibiota, thus reducing leaf growth, and then reducing shoot density (Pergent and Pergent-Martini, 1995; Pergent-Martini *et al.*, 1996). In Australia, for *Posidonia australis*, the proliferation of leaf epibiota in a polluted site reduces by 65% the light that reaches the leaves (as against 15% in a non-polluted site); the resulting drop in primary production of the leaves is assessed as about 30% (Silberstein *et al.*, 1986). As to the plant itself, although *P. oceanica* in oligotrophic water is perhaps restricted by low-level nutrients, the experimental addition of nitrogen and phosphorus (N and P) does not increase its production (Romero *et al.*, 1998).

Studies intended to individualize the specific role of various pollutants in the context of short-term (*in situ*) or middle-term (in the laboratory) experiments do however usually conclude that the effects are only observed with doses that are rarely found *in situ*. For example, in the case of copper "anti-fouling" paint, reduced leaf growth is only observed for doses higher than those measured in the Mediterranean, including in port basins (Giglio, 1985). It is thus possible that in the past the direct role of pollution has been overestimated, by attributing to it those indirect effects (development of leaf epibiota or herbivores, for example), synergic inter-pollutant effects, or effects of other disturbance (e.g. turbidity) that are frequently associated with it (Balduzzi *et al.*, 1984).



4.5. ANCHORING

Posidonia oceanica meadows are particularly sensitive to human activites that cause direct impact by mechanical action. Among those mechanical impacts, the one most frequently mentioned as harming the meadows is the action of anchors (Fig. 43).

We can speak of anchoring (=mooring in the strict sense of the term) when a boat uses an anchor, and of organised mooring when boats moor alongside deadweight moorings that are placed legally (in France) in the context of a Temporary Occupation Permit (TOP) delivered by the Maritime Service of the *Direction Départementale de l'Equipement* (DDE). But we call it unauthorized mooring when boats moor alongside deadweight moorings that are placed illegally (without a TOP) (Ganteaume *et al.*, 2004, 2005).

The impact of anchors has become worrying because of the considerable increase in leisure boating over the past few decades. Not only during the touristic season but all Fig. 43. Impact of an anchor of leisure boat on *Posidonia oceanica* meadow, with pulling out shoots of leaves and ploughing of the matte. Photo E. Charbonnel.



year round at weekends, many sites, some of them of great ecological and landscape value, have become much frequented moorings. As well as the direct impact of the anchors (leaves and rhizomes torn out), it should be stressed that these mooring sites are places where there is significant pollution: "anti-fouling" paint, hydrocarbons, detergents, discharge of organic matter (see §4.4) and of macrowaste (Augier and Boudouresque, 1970a; Robert, 1983; Boudouresque *et al.*, 1995a; Francour *et al.*, 1997, 1998, 1999; Milazzo *et al.*, 2004).

An anchor can affect the *Posidonia oceanica* meadow in various ways: (i) at the moment of anchoring – breaking the rhizomes on which it drops or over which it drifts before catching hold; (ii) while it is on the seabed – the chain in front of the anchor slips on the seabed because of hydrodynamism and the current and tears out the leaves; (iii) when it is raised – the anchor breaks the rhizomes to which it has become attached; in some cases it can tear out a whole block of "matte". When a leisure boat anchors (the anchor is dropped, stays down and is raised) an average 16 to 34 shoots of *P. oceanica* are torn out³⁸; this amount is made worse by the fact that the rhizomes are bared and the "matte" becomes less cohesive (Boudouresque *et al.*, 1995a; Poulain, 1996; Francour *et al.*, 1997, 1998, 1999). In the Elbu Cove (Scandola, Corsica), Boudouresque *et al.* (1995a) have estimated that 68 000 shoots of *P. oceanica* have been torn out in one year by anchors over a surface area of 1.4 hectares. In the Monasterio Cove (Riou, Marseille), for an identical surface area the estimated number was 88 000 shoots/year (Charbonnel, 1996). In Porquerolles (Var), Porcher (1984) observed the tearing out of whole sections of "matte" with all their shoots.

The direct action of anchors, by tearing out *Posidonia oceanica* shoots or sections of "matte," reduces the cover of the meadow, and encourages the forming of erosive "intermattes" that can later spread (because of hydrodynamism) and join together, thus fragmenting the meadow (Porcher, 1984; Francour *et al.*, 1997; Pasqualini *et al.*, 2000). The anchoring of big ships (cruise ships, warships)

provokes particularly spectacular ploughing of the "matte" (Ganteaume *et al.*, 2005; see Fig. 77). Recent studies have shown that the kind of anchor is likely to

³⁸ Values of under 10 shoots torn out on average per anchorage cycle were measured in Ustica, Italy (Milazzo et al., 2004). influence the size of the impact on the meadow. Anchors of the "Hall" kind are those which have the least impact (Milazzo *et al.*, 2002, 2004). Furthermore, incorrect ways of raising the anchor (i.e. when the boat is not directly above its anchor before starting to raise it) worsen the impact on the meadow.

4.6. TRAWLING

Fishing activities that use gear pulled on the seafloor³⁹, which scrape and plough the meadow, are prohibited between 0 and 50 metres depth and/or within 5 556 m (3 nautical miles) of the coast in almost all Mediterranean countries. However, this legislation is infrequently, or even never (in certain countries) respected (Ardizzone and Pelusi, 1984).

The damage caused by trawling (trawls) is linked to the features of the fishing gear (Kaiser, 1998).



Fig. 44. Traces of a trawl (arrows) in a *Posidonia oceanica* meadow in Giens Gulf (Var, France). The furrows are dug by the "otter boards". Image obtained using side-scan sonar. From Paillard *et al.* (1993).

The trawl is dragged along by one boat (trawler) and the gear must be kept open both vertically and horizontally. The first function is achieved using floats (wood, or air-filled buoys) attached to the upper part of the net's opening, which lift it above the seabed, and by lead weights fixed to the lower part of the net's opening, which keep it in contact with the seabed. The second function (horizontal opening) is achieved using heavy metal or wooden "otter boards"⁴⁰ fixed to the net where the cables are inserted that allow it to be dragged by the trawler, in a way that widens the opening of the trawl due to the divergent pressure that the water pressure exerts on the "otter board".

Because of its structural features, the trawl has a big effect on the *Posidonia oceanica* meadow. This is not only because of the lead line, which tears

out the shoots (Ardizzone and Pelusi, 1984) but also because of its "otter board", likely to dig deep furrows into the "matte" (Fig. 44; Paillard *et al.*, 1993; also see Fig. 77). These deep scars encourage the start of erosive phenomena, due to currents, and worsen by the sedimentary disbalance provoked by the re-suspension of sedimentary matter formerly trapped by the "matte".

Trawling (i) opens up corridors (intermattes) in the meadow; in a non-degraded meadow, the trawl's "otter board". are responsible for 94% of the torn out shoots; in southern Spain, a trawl probably tears out a total 100 000-360 000 shoots per hour, i.e. 240 to 1 100 kg DM/hour (Ramos-Esplá, 1984; Paillard *et al.*, 1993; Chessa and Fresi, 1994; Martin *et al.*, 1997; Pasqualini *et al.*, 1999, 2000); (ii) reduces the average cover of the meadow (nearly 40%; Ardizzone and Pelusi, 1984); (iii) brings up to the surface vast quantities of leaves and shoots (100 to 1 000 kg, per trawl done;

³⁹ Fishing gear pulled on the seafloor e.g. trawls and "ganguis".

40 The "otter boards" are often called "trawl doors". Ardizzone and Pelusi, 1984); (iv) increases the mass of litter (over 80%); (v) allows fish and invertebrate species from the sandy or sandy-muddy bottoms to settle in the meadow and increases the abundance of filtering or detritus-feeders animals (Jiménez *et al.*, 1997; Ramos-Esplá *et al.*, 1997) and (vi) significantly reduces the biomass and the density of the ichthyofauna



In south-eastern Spain, 40-50% of the surface area potentially occupied by *Posidonia oceanica* is illegally trawled (Sánchez-Lizaso *et al.*, 1990). Illegal trawling has caused marked regressions of *P. oceanica* meadows in Italy (Ardizzone and Migliuolo, 1982) and Spain (Martin *et al.*, 1997). In Corsica, the rate of deterioration of all the *P. oceanica* meadows together due to trawling is believed to be 12% (Pasqualini *et al.*, 2000); locally it can reach 23% (Pergent-Martini, 2000). In the Alicante region (Spain), trawling is responsible for almost half the meadow's regression, according to Ramos-Esplá *et al.* (1994), representing the destruction of almost 2 400 hectares of meadow. In Latium (Italy), trawling is probably the main cause of its regression at depth (Diviacco *et al.*, 2001). Bared meadows (Boudouresque *et al.*, 1988; Paillard *et al.*, 1993).

4.7. EXPLOSIVES

Almost everywhere along the north-western Mediterranean coasts one can find circular spots of dead meadow that correspond to

underwater explosions (Fig. 45): bombs that fell in the 1939-1945 war, mines exploding during or after the war, or fishing with dynamite (Paillard *et al.*, 1993; Pergent-Martini, 1994; Charbonnel, 1996; Harmelin *et al.*, 1996; Pasqualini *et al.*, 1999, 2000).

Posidonia oceanica's **sensitivity** to explosives is certainly due to the presence of an aerarium inside its leaves: gas-filled channels (oxygen and/or carbon dioxide, according to the time of day). When there is an explosion, the aerarium bursts the leaves.



Fig. 45. Areas of dead matte (arrows) in a *Posidonia oceanica* meadow due to explosives in Giens Gulf (Var, France). Side scan sonar image. From Paillard *et al.*(1993).

Recolonization of surface areas by *Posidonia oceanica* is an extremely slow process since sometimes 50 years after the event that caused its loss, there is still only partial recolonization (Fig. 46; Meinesz and Lefèvre, 1984; Pergent-Martini, 1994; Pergent-Martini and Pasqualini, 2000). Similarly, in south-eastern Australia, explosions set off during geological research (seismic shots) explain the existence of circular spots with no *Posidonia australis*: after 20 years they have practically not been recolonized (West *et al.*, 1989).



Fig. 46. Partial recolonization in 1983 of an area of "dead matte" created by the explosion of a bomb in 1943 between 6 and 15 m depth in Villefranche-sur-mer Gulf (Alpes-Maritimes, France). From Meinesz and Lefèvre (1984)

4.8. COASTAL AQUACULTURE

The introduction of fish farming in coastal water has accelerated over recent years. Studies show clearly that fish farms, when located near a Posidonia oceanica meadow, have a great effect on it: the sediment's organic matter and nitrogen content, and the sediment's interstitial water's phosphorus and total phosphorus content go up as one approaches the fish farming site, with as a consequence the decline in the meadow's vitality: reduced shoot density and drop in the plant's primary production (Cancemi et al., 2000).

In El Hornillo Bay (Spain), mapping monitoring has shown that a fish farm led in 10 years to the destruction of 11 hectares of meadow and the degradation of another 10 hectares (Fig. 47; Ruiz-Fernández, 2000; Ruiz *et al.*, 2001).

It seems that the main cause of the impact of fish farming is the introduction of **organic matter**, the oxidation of which gives rise to anoxic conditions in the sediment lying under and near the farms (Delgado *et al.*, 1999), and the synthesis of reduced compounds that may be toxic to *Posidonia oceanica* (Hemminga, 1998). Moreover, nutrient enrichment of the water can cause the increase of leaf epiphytes, resulting in the reduction of *P. oceanica*'s photosynthesis (by restricting its access to light) and increasing the grazing of leaves by herbivores (Pergent *et al.*, 1999; Ruiz-Fernández, 2000). Lastly, the shade of the cages, also by restricting access to the light, significantly reduces the shoot density (Ruiz-Fernández, 2000; Ruiz and Romero, 2001)).

4.9. LAYING CABLES AND PIPES

Laying (water, gas, oil) pipes and underwater cables sometimes involves crossing a *Posidonia oceanica* meadow, when leaving or reaching the coast. For various reasons, not always relevant, even from a technical point of view (see Chapter 12), trenches have often been dug across the meadow.

These trenches, usually running straight out from the coast, can be a serious problem for the meadow: if sediment has been put down to close the trench, it is quickly borne off by hydrodynamism; hydrodynamics tend to widen the trench; and during the work, the meadow is usually harmed over a much greater width than the trench itself (see Chapter 12).



4.10. DUMPING

Dumping involves discharging into the sea dissolved or solid material, especially products of dredging. Its negative impact on a *Posidonia oceanica* meadow is direct (burial, silting) or indirect (fine particles re-suspension and increased turbidity; see §4.3). It has been highlighted particularly in Liguria (Italy; Peirano and Bianchi, 1995) and in Corsica, in the Gulf of Porti-Vechju (Pasqualini *et al.*, 1999).

Dumping permits usually clearly locate discharge sites far from the coast, thus not located over *Posidonia oceanica* meadows. But it has often been noticed that the public works enterprises responsible for such dumping, where there is no active monitoring by the authorities, **shorten**, sometimes considerably, the distance to the dumping site. Slabs of rock or products of dredging have thus been dumped directly onto *P. oceanica* meadows.

4.11. COMPETITION WITH INTRODUCED SPECIES

The problem of competition between the *Posidonia oceanica* meadow and introduced species has become topical, with the introduction into the Mediterranean of 2 Chlorobionta (Plantae) – *Caulerpa taxifolia* and *C. racemosa* var. *cylindracea* – and 2 Rhodobionta, *Womersleyella setacea* and *Acrothamnion preissii*.

Caulerpa taxifolia (Fig. 48) is native of Australia and was accidentally introduced into the north-western Mediterranean in 1984 (Meinesz and Hesse, 1991). Its geographical expansion



Fig. 47. Regression of the *Posidonia oceanica* meadow in El Hornillo Bay (south-eastern Spain) due to a fish farm (farming *Seriola dumerlii, Dicentrarchus labrax* and *Sparus aurata* fish) between 1988 (top) and 1998 (bottom). Scale in metres. From Ruiz-Fernández (2000).

was relatively rapid and in late 2000 it was present in 103 stations distributed throughout 6 countries (Croatia, France, Italy, Monaco, Spain and Tunisia) and colonized a total 131 km²(⁴¹) (Meinesz *et al.*, 2001a). *Caulerpa taxifolia* is able to colonize almost all kinds of substrata, particularly *Posidonia oceanica* meadows and "dead matte" (Boudouresque *et al.*, 1995c). Even if *Caulerpa taxifolia*'s ability to wipe out a *P. oceanica* meadow in good condition has not been demonstrated in the short term, stressed and degraded meadows are an extremely favourable environment for this species, and it can speed their decline (Villèle and Verlaque, 1995; Torchia *et al.*, 2000). In the long term, *C. taxifolia*'s ability to replace non-degraded *P. oceanica* meadows, or some of them, remains an open question (Ceccherelli and Cinelli, 1997, 1998; Chisholm *et al.*, 1997; Molenaar, 2001). Anyway, the presence of *Caulerpa taxifolia* in a *P.oceanica* meadow profoundly changes the way the ecosystem functions (Ruitton and Boudouresque, 1994; Gélin *et al.*, 1998; Harmelin-Vivien *et al.*, 1999).

Caulerpa racemosa var. cylindracea is also a Chlorobionta, introduced by 1990 in the Mediterranean,

from the southwest of Australia (Verlaque *et al.*, 2000, Durand *et al.* 2002; Verlaque *et al.*, 2003). Its expansion has been extraordinarily fast, since

41 This surface corresponds to the "concerned surface", i.e. occupied either continuously or discontinuously (isolated spots) by the species. (Meinesz et al., 2001).



Fig. 48. *Caulerpa taxifolia* (Chlorobionta, Plantae) general view of the tip of a rhizome on which some leaves are set. The longest leaf is about 15 cm. long. Photo A. Meinesz. it is now present in most of the Mediterranean and even in the Canary Islands (Verlaque *et al.*, 2004, Piazzi *et al.*, 2005). When *C. racemosa* colonizes bottoms bordering *Posidonia oceanica* meadow, its growth in height is a function of the *P. oceanica* shoot density and the orientation of the meadow (Ceccherelli *et al.*, 2000).

As far as the Rhodobionta *Womersleyella setacea* and *Acrothamnion preissii* are concerned, they can form a very dense cover on *Posidonia oceanica* rhizomes. However, their possible impact on the *P. oceanica* meadow is not clearly known.

4.12. OVERGRAZING

Populations of the sea urchin *Paracentrotus lividus* are usually controlled by predators, the first of which are the Sparidae seabreams (*Diplodus* particularly). Overfishing the seabream is one of the reasons why the sea urchin population has exploded; another reason is urban

pollution, which favours sea urchins (Harmelin *et al.*, 1981; Ruiz-Fernández (2000). Lastly, in nutrientrich water, the nitrogen content of *Posidonia oceanica* leaves and that of its leaf epibiota rises significantly, attracting the macro-herbivores *Paracentrotus lividus* and *Sarpa salpa* (Ruiz-Fernández, 2000). The result is thus the overgrazing of benthic primary producers.

Nédélec and Verlaque (1984) assess at between 24 and 51 (according to the season) the number of individual *Paracentrotus lividus* sea urchins of 50 mm diameter starting from which the consumption is greater than the production of *Posidonia oceanica*, determining its overgrazing. For example, in Fontagne Cove (Riou, Marseille), Charbonnel (1996) observed great overgrazing of the *P. oceanica* meadow, and he partly attributed its regression to this. Around a fish farm in Spain (El Hornillo, Murcia), overgrazing by sea urchins, encouraged by pollution, is probably the direct cause of the meadow's regression or disappearance (Ruiz-Fernández, 2000).

4.13. SYNERGY OF DIFFERENT CAUSES OF REGRESSION

It is probable that none of the above-mentioned causes of regression is by itself able to degrade or destroy the *Posidonia oceanica* meadow over vast stretches, if one excepts local effects (cover by some kind of development, immediate proximity of an untreated waste water discharge, etc.) It is more probably the combination of various types of disturbance along certain sectors of the coast, and the synergy between them, that can explain the serious, spatially extensive existing damage: the total disappearance of the meadow or its decreased vitality (cover, shoot density) (see start of Chapter 4).

In general, the meadows that are now the most degraded are those located near the major urban centres and great ports. But to establish a true cause and effect relationship, it is necessary to set up specific monitoring systems, designed to take into account the natural spatio-temporal variability that is a characteristic of the *Posidonia oceanica* ecosystem.

Indeed, the ecological processes likely to influence Posidonia oceanica shoot density are numerous

and are not all of human origin. Meadows exist which present a low shoot density (bearing in mind their depth), and/or low cover, and even vast stretches of "dead matte", while they are far from any known human impact. The reasons for this situation may be diverse and completely natural: exposure to hydrodynamism, the nature of the substratum, the rate of sedimentation, fresh water input etc. These natural causes, which are overlaid by the effects of human activity, make it sometimes hard to assess the health of a meadow in a given site and, especially, to make a local interpretation of the different causes of degradation. Furthermore, "dead mattes" may correspond to an old (natural or human) impact, possibly occured several centuries ago (Augier and Boudouresque, 1970a; Boudouresque *et al.*, 1980c; Gravez *et al.*, 1992).

The difficulty is made worse by the fact that a "natural" meadow (i.e. one in good health) does not present itself in a homogeneous way in different temporal and spatial scales, but presents a high degree of variability in shoot density and cover, even at a given depth. Indeed, a meadow can be seen as a mosaic where shoot density varies not only horizontally (from one point to the next) but in time, interrupted by spots without any living shoots (structural intermattes) and erosive structures whose position changes over time (see Chapter 2 and the start of Chapter 4). All these parameters must be carefully considered, in a site and on a given scale, before concluding that there is regression and setting up specific procedures.

One may legitimately wonder that after several decades of research on the *Posidonia oceanica* meadow, with hundreds of scientific publications, it is still so hard to decide between the different causes of regression. As well as the fact that *in situ* the causes can be difficult to isolate, this can be explained by *P. oceanica*'s morphology: one individual of this plant is made up of hundreds of shoots linked by rhizomes and stretching over several square metres; laboratory experiments on cut shoots (thus highly stressed), separated from the rest of the plant with which they usually exchange substances, cannot account for the *in situ* situation; in *in situ* experiments it is technically impossible to put under cloche any more than a few shoots, thus a very small part of the plant.

4.14. CONCLUSIONS

Since the early 1990s, in the north-western Mediterranean, the policy protecting *Posidonia* oceanica meadows (see Chapter 5) and to improve coastal water quality (setting up water treatment stations) have slowed the decline of the meadows and even, locally, led to a modest recovery (Gravez *et al.*, 1992; Charbonnel *et al.*, 1995d; Boudouresque *et al.*, 2000). This recovery must however be considered with caution: it is certainly very slow (a few centimetres per year), while the regression can be 10 to 100 times quicker. Moreover, if we exclude the north-western Mediterranean, in most of the Mediterranean the meadow's regression is continuing apace, and is probably going ahead in vast areas for which no data is available.

Furthermore, the fact that it is not always easy to distinguish between natural and human factors, or between the different kinds of human-induced factors, which almost always act simultaneously and certainly have synergic effects, should not hide a robust scientific certainty: **man really has been responsible** for most of the regressions observed in the second half of the 20th century.

5. POLICIES APPLYING TO POSIDONIA OCEANICA MEADOWS

Few policies directly aim at the protection of marine species other than turtles, birds and mammals, even if marked progress has been made – usually on the initiative of NGOs. Thus, in 1993 five marine invertebrate species were added to the list of species that are protected in France, and then in 1999 about twenty species of invertebrates and a dozen of macrophytes⁴² (Boudouresque *et al.*, 1991; Boudouresque, 1996; Boudouresque *et al.*, 1996; Boudouresque, 2002c). Plant formations, in particular *Posidonia oceanica* meadows, have benefited from the new awareness, and a growing number of national provisions, Community directives (European Union) and international conventions refer to this. But we should differentiate between **direct** legal protection measures either concerning the species *P. oceanica* or the habitats it constitutes, and regulatory measures which, without directly aiming to protect the meadows, can **indirectly** encourage their conservation.

5.1. DIRECT PROTECTION MEASURES

5.1.1. International conventions and European Community texts

In policy terms, the ecosystem approach is a relatively recent one (e.g. the 1992 Rio de Janeiro Summit) and only international conventions signed after 1990, or those that were produced before this date but were updated, possibly take *Posidonia oceanica* meadows into account.

This is the case of the **Bern Convention** (Convention on the conservation of European wildlife and natural habitats), signed in 1979 under the auspices of the European Council by several Mediterranean countries (Table III). Indeed, while it did not initially mention any marine plant species, its Annexes were modified (in 1996) to add 3 of the 5 Magnoliophyte⁴³ species of the Mediterranean Sea (*Cymodocea nodosa, Posidonia oceanica, Zostera marina*). These species are mentioned as requiring protection (Boudouresque *et al.*, 1996; Platini, 2000). As well as protection of the species itself, the Convention provides (Chapter II, Article 4) that:

" 1 - Each Contracting Party shall take appropriate and necessary legislative and administrative measures to ensure the conserva-tion of the habitats of the wild flora and fauna species, especially those specified in Appendices I and II, and the conservation of endangered natural habitats.

2 - The Contracting Parties in their planning and development policies shall have regard to the conservation requirements of the areas protected under the preceding paragraph, so as to avoid or minimize as far as possible any deterioration of such areas.

4 - The Contracting Parties undertake to co ordinate as appropriate their efforts for the protection of the natural habitats referred to in this article when these are situated in frontier areas. "

- ⁴² Macrophytes a polyphyletic (=artificial) set of usually photosynthetic pluricellular organisms, belonging to the Chlorobionta, Viridiplantae and Rhodobionta (Kingdom Plantae) and to the Chromobionta (Kingdom Stramenopiles).
- ⁴³ Magnoliophytes = Phanerorogams (flowering plants). Magnoliophytes belong to the Viridiplantae.
- 44 UNEP United Nations Environment Programme

The same holds good for the **Barcelona Convention**, adopted in 1976, which was the key convention for the protection of areas and species in the Mediterranean. A legal tool of the Mediterranean Action Plan (MAP), launched by the UNEP⁴⁴ for the protection of regional seas, the Convention initially focused on the fight against marine pollution (Tavoso, 1997). But from 1982 on, with the adopting of the Protocol on Mediterranean Specially Protected Areas, the 20 signatory countries (Albania, Algeria, Bosnia-Herzegovina, Croatia, Cyprus, Egypt, France, Greece, Israel, Italy, Lebanon, Libya, Malta, Monaco, Morocco, Slovenia, Spain, Syria, Tunisia and Turkey) and the European Community showed their interest in protecting marine habitats.

COUNTRY	RAMSAR	BERN	BARCELONA
Albania	1995 A*	1998 R*	1990 A
Algeria	1984 A	(1981 A
Bosnia-Herzegovina	2001 R*	6	1994 A
Croatia	1993 R*	2000 R	1993 A
Cyprus	2001 R*	1988 R	1979 R
Egypt	1988 R*	-	1978 R
European Union		1	1978 R
France	1986 A	1990 R	1978 R
Greece	1975 A	1983 R	1979 R
Israel	1997 R*	-	1978 R
Italy	1977 A	1982 R	1979 R
Lebanon	1999 R*	-	1977 A
Libya	2000 R*		1979 R
Malta	1988 A*	1993 A*	1977 R
Monaco	1997 A	1994 R	1977 R
Morocco	1980 R*	2001 R*	1980 R
Slovenia	1992 A*	1999 R*	1992 A*
Spain	1982 A	1986 R	1977 R
Syria	1998 R*	-	1978 A
Tunisia	1981 A	1996 R	1977 R
Turkey	1994 A	1999 R	1981 R
Yougoslavia ^a	1991 A	1999 R	3.45

Table III. Mediterranean countries which had adopted (A) or ratified (R) international conventions by 3.4.2003. - missing data; * data updated according to RAC/SPA (2003).

^aToday Serbia and Montenegro.

However, it was only in 1995, on the 20th anniversary of the MAP, that the Convention was amended and took the name of the Convention for the Protection of the Marine Environment and the Coastal Region of the Mediterranean. A new MAP, the Action Plan for the protection of the marine environment and the sustainable development of the coastal areas of the Mediterranean (MAP Phase 2) was then adopted, and came into force in December 1999, accompanied by a new Protocol on Specially Protected Areas and Biological Diversity in the Mediterranean (Platini, 2000). This Protocol was accompanied by 3 Annexes, respectively concerning: (i) common criteria for the choice of protected marine and coastal areas that could be included in the SPAMI (Specially Protected Areas of Mediterranean Importance) list, (ii) a list of endangered or threatened species, and (iii) a list of species whose exploitation is regulated (Anonymous, 1998). It is in Annex II that the marine Magnoliophytes (*Posidonia oceanica, Zostera marina* and *Nanozostera noltii*⁴⁵) are specifically mentioned. Also, as for the Bern Convention, the Contracting Parties to the Convention (Table III) are invited to draw up a list of SPAMIs; this list may include sites which: "*are of importance for conserving the components of biological diversity in the Mediterranean, contain ecosystems specific to the*

45 Nanozostera noltii = Zostera noltii, Z. nana

Mediterranean area or the habitats of endangered species, [or] are of special interest at the scientific, aesthetic, cultural or educational levels". This is expressed in *de facto* protection for *P. oceanica meadows* as a habitat. These provisions were enhanced when in October 1999 an Action Plan for the conservation of marine vegetation in the Mediterranean was adopted. Exclusively focusing on the protection of Mediterranean marine vegetation, this Action Plan identifies priority actions at national and regional level, such as (i) ensuring the conservation of species and plant formations by developing legal measures of protection and levels of knowledge about them; (ii) avoiding the loss and degradation of marine Magnoliophytes meadows and of other plant formations as habitats for marine species and maintaining them in a satisfactory state of conservation; and (iii) ensuring the conservation of formations that can be considered to be natural monuments, such as *P. oceanica* barrier reefs, bioconstructions (vermetid platforms) and belts of Chromobionta of the *Cystoseira* genus (Anonymous, 2000).

To these international conventions one should add the European Community Directives which first only concerned 4 Mediterranean states (France, Greece, Italy and Spain) but were then extended to countries like Cyprus, Malta and Slovenia as these entered the European Union.

Thus the Habitats Directive of 21 May 1992 (92/43 EEC/Natural Habitats) constituted the legal basis of conserving the natural habitats of wild fauna and flora and maintaining biodiversity throughout the European Union (Platini, 2000). The Directive has 6 Annexes; Annex 1 identifies types of natural habitat of Community interest whose conservation requires the designation of Special Conservation Zones (SCZs). In this Annex, in the context of coastal habitats and halophytic vegetation, appear the *Posidonia oceanica* meadows, which are, moreover, classified as a priority habitat (no. 1120). Similarly, through the Natura 2000 procedure, the states have identified meadow habitats as deserving to enjoy adapted conservation measures (Platini, 2000).

5.1.2. Policies in the countries of the RAMOGE areas

Regulations in France

In France, legal protection of the marine Magnoliophyte *Posidonia oceanica* comes under the Law of 10 July 1976 on nature protection, and its Implementing Decree of 25 November 1977 concerning the protection of the wild flora and fauna of the French natural heritage. This protection was made official by the Interministerial Order of 19 July 1988 (*Journal Officiel* of 9 August 1988, pp.10-128) on the list of protected marine plant species, which specifies: *"in order to prevent the disappearance of threatened plant species and to permit the conservation of the corresponding biotopes, the following are prohibited at all times and throughout Metropolitan France: the destruction, cutting, tearing out, mutilation, picking or removal, hawking, use, offering for sale, sale or purchase of all or part of the wild specimens of the species enumerated below (...) <i>P. oceanica and Cymodocea nodosa" (in* Pergent, 1991a). The text adds: *"However, the ban on destruction does not apply to current exploitation operations of coastal farming establishments on plots of land that are usually cultivated"* (t ans ated f om the F ench).

The Decree of 7 July 1999 of the Ministry of Foreign Affairs (J.O. of 18 July 1999, pp. 10741-10758) publishing the amendments to the Annexes of the Convention on the conservation of European wildlife and natural habitats (Bern Convention) also mentions *Posidonia oceanica*. As well as the *Posidonia oceanica* species itself, the meadows can enjoy protection under the terms of the Law of 3 January 1986, which states the principles relating to the development, protection and valorization of the coast. This Law, called in French *Loi littorale*, can enable the preservation of a meadow or part of a meadow that presents an ecological interest or that is deemed indispensable to the maintenance of the biological equilibrium (Platini, 2000). This is expressed by the Decree of 20 September 1989 (*Code de l'urbanisme* - Town Planning Code, provisions peculiar to the coast; Boudouresque *et al.*, 1995b) which stipulates that *"the following are protected, the moment that they constitute a site or landscape that is remarkable or characteristic of the coast's natural and cultural heritage, are necessary to the maintaining of biological equilibria or present an ecological interest: (...) environments sheltering natural concentrations of animal or plant species such as meadows (...)" (t ans ated f om the F ench),*

Lastly, some particular meadows, like the "reef" formations of *Posidonia oceanica* (e.g. the barrier reefs of Port-Cros and of Le Brusc in Provence, and the reef platform of Saint-Florent in Corsica), in the light of their remarkable landscape characteristics (Boudouresque *et al.*, 1991) are the subject of increased protection. Thus the barrier reef of Port-Cros, which is included in the Port-Cros National Park, and is specifically monitored (Augier and Boudouresque, 1975; Augier and Niéri, 1988) or the Saint-Florent platform (Boudouresque *et al.*, 1983), which has since 1999 been protected under a French Order called "*Arrêté préfectoral de protection de biotope*" (Anonymous, 2001b).

Regulations in the Principality of Monaco

In the Principality of Monaco, a bill is being prepared for those taxa which appear in the annexes to the international conventions that have been adopted to also benefit from specific protection under Monacan law (Platini, 2000).

Regulations in Italy

In Italy, the national competent authorities for protecting the biodiversity of the marine and coastal environment, protected marine species and the marine environment as a whole come under the Nature Protection Department of the *Ministero dell'Ambiente e della Difesa del Territorio e del Mare*.

	MARINE PROTECTED AREAS CREATED IN ITALY				
	Marine Protected Area Decree creating it (year)		Present managing body	Surface area	
1	The island of Ustica	1986	Palermo Port Authority (provisional management)	15 951 ha	
2	Gulf of Trieste Miramare	1986	Italian Association for WWF for Nature - ONLUS	127 ha	
3	The Tremiti Islands	1989	Ente Parco Nazionale del Gargano	1 466 ha	
4	Torre Guaceto	1991	Management consortium of the districts of Brindisi, Carovigno and WWF Italy	2 227 ha	
5	Cyclops Islands	1989	Management consortium of the district of Aci Castello and Catania University - Cutgana	902 ha	
6	Egadi Islands	1991	District of Favignana	53 810 ha	
7	Capo Rizzuto	1991	Province of Crotone	13 500 ha	
8	Ventotene and S. Stefano Islands	1997	District of Ventotene	2 799 ha	

Protecting marine biodiversity is a priority aspect of national strategies, and much energy has been devoted to selecting and setting up Marine Protected Areas, of which there are now 23.

able p 51 (ollowing)

9	Punta Campanella	1997	Management consortium of the districts of Massa Lubrense, Sorrente, S. Agnello, Piano di Sorrento, Vico Equense and Positano	1 128 ha
10	Sinis Peninsula- Mal di Ventre Island	1997	District of Cabras	30 000 ha
11	Porto Cesareo	1997	Management consortium of Lecce Province, the districts of Porto Cesareo and Nardò	17 156 ha
12	Tavolara-Punta Coda Cavallo	1997	Management consortium of the districts of Loiri Porto S. Paolo, S. Teodoro et Olbia	15 091 ha
13	Cinque Terre	1997	Ente Parco Nazionale delle Cinque Terre	2 784 ha
14	Golfe de Portofino	1998	Management consortium of the districts of Sta. Margherita Ligure, Portofino and Camogli, Genoa Province and University	372 ha
15	Capo Carbonara	1998	District of Villasimius	8 598 ha
16	Tor Patemo shallows	2000	Ente Roma Natura	1 387 ha
17	Capo Gallo-Isola delle Femmine	2002	Palermo Port Authority (provisional management)	2 173 ha
18	Asinara Island	2002	Ente Parco Nazionale dell'Asinara	10 732 ha
19	Capo Caccia-Isola Piana	2002	Alghero district (provisional management)	2 631 ha
20	Pelagi Islands	2002	Districts of Lampedusa and Linosa (provisional management)	4 136 ha
21	Baia underwater park	2002	Superintendency for archaeological property of the Provinces of Naples and Caserte (provisional management)	177 ha
22	Gaiola underwater park	2002	Superintendency for archaeological property of the Provinces of Naples and Caserte (provisional management)	41 ha
23	Plemmirio	2004	Management consortium of the Province of Syracuse and the District of Syracuse (provisional management)	2 429 ha

Italy adopted the Barcelona Convention's Protocol on Specially Protected Areas (SPAs) and on Mediterranean Biodiversity, which provides for the creating of Specially Protected Areas of Mediterranean Importance (SPAMIs) according to criteria that take into account the degree of biodiversity itself, the specificity of the habitats and the presence of rare, threatened or endemic species. Today, in addition to the Marine Mammal Sanctuary, the Marine Protected Areas of Portofino, Miramare, Plemmirio, Tavolara-Punta Coda Cavallo and Torre Guaceto are to be added to the list.

The activities currently carried out, coordinated by the *Ministerio dell'Ambiente e della Difesa del Territorio e del Mare*, involve implementing the Action Plans for cetaceans, marine turtles, *Posidonia oceanica* meadows and invasive non-native species.

The Nature Protection Department also organises, within a convention with 14 Coastal Regions, an activity for monitoring the marine and coastal environment which extends over about 6 000 km of coast.

The Ministry of the Environment and Territory pays special attention to *Posidonia oceanica*, devoting many projects to it over the past few years. Thus the distribution, and state of conservation, of *P. oceanica* meadows have been studied in many campaigns.

Furthermore, the Nature Protection Department of the Ministry of the Environment has set up a specific plan for mapping meadows on the Mediterranean coasts, in compliance with the "National Programme for locating and valorizing *Posidonia oceanica* and studying measures to protect it against phenomena likely to cause its degradation and destruction", provided for by Law no. 426/98.

In the 1990s, the first *Posidonia oceanica* meadow mapping programme in five Italian regions – Liguria, Tuscany, Latium, Basilicata and Apulia – was completed.

64 meadows were identified over a total surface area of 90 913 hectares. In Liguria, 25 meadows

were identified, only 2 of which seem to be in a good state of conservation, and only representing 2.5% of the total surface area covered by the meadows along the Ligurian coast, the state of the other meadows was evaluated mediocre, weak or bad. In Tuscany there were 7 meadows, 3 of them in good health, representing 44% of the total surface area; the others show mediocre or low vitality. In Latium, there were 15 meadows, 4 of these in good health, corresponding to about 20% of the total surface area; the others are classified in mediocre, weak or bad vitality. For Puglia, 16 meadows were mentioned, 9 of which are in good health, corresponding to 65% of the total surface area; the other meadows' conservation levels are mediocre, weak or bad. The only meadow present in Basilicata stretches over 646 hectares and its state of conservation is mediocre.

The studies described above are associated with a number of monitoring and mapping measures carried out on different scales (Bianchi and Peirano, 1995; Bianchi *et al.*, 1995; Diviacco *et al.*, 2001).

The "Bioltaly" project, launched in 1994 by the Ministry of the Environment, and aimed at identifying Sites of Community Importance (SCI, Habitats Directive), led to the selection of sites mainly composed of *Posidonia*

Posidonia oceanica

Programmes carr ed out thanks to the contr but on made by the M n ster o dell'Amb ente e della D fesa del Terr tor o e del Mare

• 1989 - 1991 <u>Mapping meadows</u> along the coast of Liguria, Tuscany and Elba Island, Latium and the Pontine Islands, Apuglia and the Tremiti Islands (Snamprogetti Project S.p.A. Ecologia – environmental studies)

• Since **1993** Research projects in every <u>Marine Protected Area</u> (a first study done on Multicellular Photosynthetic Organisms (MPOs) leaf epibiota, vagile fauna and mucilage in *P ocean ca* meadows in the Ustica Island Marine Protected Area)

• **1998-2002** <u>National programme</u> for locating and valorizing *P* ocean ca and study of measures to protect it against all phenomena likely to cause its degradation and destruction (Mare Vivo Association)

• **1999-2002** <u>Mapping *P ocean ca* meadows</u> along the coast of Sicily and the minor islands (CEOM Company)

• **1999-2002** <u>Mapping meadows</u> along the coast of Sardinia and the minor neighbouring islands (Nautilus Cooperative Society)

• 2001-2003 <u>Checking the state of conservation</u> of some *P* ocean ca meadows along the coasts mapped in the years 1989-1991 (Conisma)

• 2001-2003 Doing <u>surveys on banquettes</u> of washed up dead *P* ocean ca leaves as an additional element in monitoring and assessing the quality of *P* ocean ca meadows (Mare Vivo Association)

• 2002 <u>Checklist/database of Italian vascular flora</u> (Plant Biology Department, Rome University, "La Sapienza")

• 2003 As part of the "Furthering basic naturalist knowledge" project, basic description on a 1:250 000 scale of coastal marine biocenoses (Marine Biology and Animal Ecology Laboratory, Institute of Zoology, Genoa University)

 2002-2004 <u>Mapping P ocean ca meadows</u> along the coast of Campania and Calabria (Fugro Oceansismica Company)

oceanica meadows (Mariotti et al., 2002). Thus, *P. oceanica* meadows are now protected within protected areas, whether these are Marine Protected Areas or Natura 2000 zones.

Since 1998, Italy has set up a legal procedure that aims to ensure the protection of *Posidonia* oceanica meadows. This is the "*Nuovi interventi in campo ambientale*" Law (no. 426 – 9/12/98) and, more recently, the Law on "*Disposizioni in campo ambientale*" (no. 93 – 23/3/2001; Procaccini *et al.*, 2003). These texts, although very general, devote specific paragraphs to the meadows, with financial provision for carrying out studies and programmes to protect and map *P. oceanica*.

In 2001 Liguria adopted regulations to assess the impact of development projects on sites of Community importance (Habitats Directive)⁴⁶, in which *P. oceanica* meadows were included (Deliberazione di Giunta Regionale no. 646 of 8 June 2001). Liguria also adopted a document identifying technical normative requirements for determining both the state of conservation of *P. oceanica* meadows (*Deliberazione della Giunta Regionale* no. 773 of 2003) and the impact of coastal facilities on the meadows (*Deliberazione della Giunta Regionale* no. 1533 of 2005). These

normative requirements constitute an important instrument, both for developers (who must present projects that are compatible with the conservation of the habitat) and for administrations (who have to make an objective impact

⁴⁶ "Valutazione di Incidenza sui Piani e Progetti che possono avere effetti sui Siti di Importanza Comunitaria (Habitats Directive)" assessment on how the project affects the environment). Furthermore, in Liguria, the introduction of a Geographical Information System (GIS) for managing environmental knowledge of seabeds (Tunesi *et al.*, 2002) has permitted the detailed mapping of the main coastal biocenotic settlements (Coppo and Diviacco, in preparation). Using this instrument, the Ligurian Region has defined on a 1:10 000 scale all the marine SCIs (Sites of Communitary importance) (*Deliberazione della Giunta Regionale* no. 1561 of 2005).

5.1.3. Other policies in the Mediterranean

Outside the RAMOGE area, several countries now have specific laws about meadows or envisage such procedures, if only to make the provisions mentioned in the international conventions they have signed or ratified.

In Algeria, the Law on protecting and valorizing the coast (no. 02-02 of 22 Dhou El Kaada 1422 corresponding to 5 February 2002, which appeared in the *Journal Officiel* no. 10 of 12 February 2002) indicates that *"the occupation and use of coastal soil must protect those terrestrial and marine spaces that are remarkable or necessary to the maintaining of the natural equilibrium. The following are concerned by the present arrangement: rocky coasts of ecological interest, coastal dunes and moors, beaches and lidos, forests and coastal Woodland areas, coastal manmade lakes and land around these, islets and islands and all sites of ecological interest or scientific value on the coast, such as coral reefs, underwater meadows and underwater coastal formations." (t ans ated f om F ench) The ministerial circular on implementing this Law, in the context of the coastal development plan (no. 380/SPM of 19 October 2002) states clearly that as regards underwater meadows and underwater meadows and underwater mork must be undertaken in these natural spaces, except, however, for light facilities intended for their management or valorization (Rachid Semroud, verbal comm.).*

In Croatia, regulations are being introduced to protect the 4 species of marine Magnoliophytes – *Posidonia oceanica, Cymodocea nodosa, Zostera marina* and *Nanozostera noltii* (Platini, 2000).

In Spain, the autonomous governments of Catalonia (Catalunya) and the Comunitat Valenciana (out of the 5 which have authority over the Mediterranean coast) have effective protection for marine Magnoliophyte species. In Catalonia, the Order of 31 July 1991 enables the protection of *Posidonia oceanica, Cymodocea nodosa* and *Nanozostera noltii*. In the Comunitat Valenciana, the Order of 23 January 1992 forbids *"la destrucción de las praderas de Fanerógamas marinas, por ser zonas de interés pesquero"* (Boudouresque *et al.*, 1995b). As additional measures, in 1992 the *Direcció General de Pesca Marítima de la Generalitat di Catalunya* funded a detailed mapping of the meadows in the Catalan coasts and started a programme, *"Xarxa de vigilància de la qualitat biològica dels herbassars de Fanerógamas marinas"* that aimed at gathering data on how the marine Magnoliophyte meadows functioned in order to obtain information that could be useful for their protection and management (Javier Romero, verbal comm.).

In Libya, a law is being prepared to enable the protection of a large number of marine species. These species will be those adapted in the context of the Bern Convention and those identified by the Protocol on Specially Protected Areas (Platini, 2000).



In Malta, *Posidonia oceanica* is not legally protected, but since Malta is a signatory of the Bern Convention and the Barcelona Convention it must act to be in a position to put these Conventions into effect. Furthermore, Malta has joined the European Community and should, in the light of the Habitats Directive, consider *P. oceanica* meadows as priority habitats. Also, the Ministry of the Environment and of Development has set up a commission to discover the geographical distribution of the meadows and their state of health. The aim is to set up protection of the meadows that are in good health and/or the most sizeable meadows (Patrick Schembri, verbal comm.).

In Slovenia, *Posidonia oceanica* is included by a Ministerial Ruling, of 24 September 2002, in the red list of threatened flora and fauna, along with *Zostera marina* and *Nanozostera noltii*, which means that the state has to introduce conservation measures for these species. Currently, a Ministerial Decree is being prepared on the conservation of the natural heritage. It is anticipated that *P. oceanica* will be included in this Decree. Finally, Slovenia's *P. oceanica* meadows are classified as a Natura 2000 site and as such enjoy conservation measures (Robert Turk, verbal comm.).

In Turkey, strict protection steps are stated in the Ministry of Agriculture's Fishing Regulations for the marine Magnoliophytes *Posidonia oceanica* and *Nanozostera noltii. P. oceanica* is included in the Law on "aquatic products" (Ref. no. 1380) and its annual circular (Ebru Coskun, verbal comm.).

5.2. INDIRECT PROTECTION MEASURES

Indirect measures likely to help protect Posidonia oceanica meadows are extremely varied, since one can integrate them both within actions to conserve a given geographical territory (if it contains meadows) and also within all the approaches that aim at restricting, or compensating for, degradation of the coastal environment caused directly or indirectly by human beings. Many reasons have been advanced to explain the regression of the meadows (e.g. urban discharge, anchoring, use of trawls and explosives, coastal development and/or competition from introduced species; Boudouresque, 1996; see Chapter 4) and all the policy measures intended to reduce these can constitute a mode of protection. One can thus recall all the policy measures that aim at (i) restricting pollutant waste (e.g. the Protocols of the Barcelona Convention), (ii) ensuring the treatment of urban waste (e.g. European Directive 91/271/EEC), (iii) fighting against water eutrophication (e.g. European Directive 91/676/EEC), (iv) banning certain fishing techniques (e.g. European Community Regulation no. 1626/94 of the Council of 27 June 1994, modified, providing for certain technical measures to conserve fishing resources in the Mediterranean) and (v) fighting against the introduction of invasive species (e.g. European Directive 92/43/EEC). We are not aiming to offer an exhaustive inventory but to illustrate these approaches by giving some examples.

5.2.1. Protected areas

Setting up Marine Protected Areas (MPAs) can be a way of protecting *Posidonia oceanica* meadows, as Platini (2000) and Boudouresque *et al.* (2004) argue. Many Mediterranean MPAs include sizeable *P. oceanica* meadows (Augier and Boudouresque, 1970a; Ramos-Esplá and McNeill, 1994; Boudouresque, 1996; Platini, 2000; Francour *et al.*, 2001). Even if the conventions that aim at protecting and conserving spaces do not specifically refer to the *P. oceanica* meadows, or even

to marine plants, they can de facto concern these and constitute effective protection measures.

This is the case for the **Ramsar Convention**, which came into force in 1975, and which aims at permitting wetlands as defined in Article 1.1, *"areas of marsh, fen, peatland or water, whether natural or artificial, permanent or temporary, with water that is static or flowing, fresh, brackish or salt, including areas of marine water the depth of which at low tide does not exceed six metres." to be conserved and managed; it is a good example of such measures. Indeed, meadows are concerned by this Convention, since it foresees the setting up of a network of protected areas, including the upper parts of <i>Posidonia oceanica* meadows. And this is what was done by the Principality of Monaco when it ratified the Protocol (Table III), by including among the Monacan wetlands of international importance

Table IV. Mediterranean Specially Protected Areas (SPAs) where Magnoliophyte meadows (*Posidonia oceanica*, *Cymodocea nodosa* and *Nanozostera noltii*) have been identified (from Platini, 2000, modified). - missing data. Yougoslavia = Serbia and Montenegro.

Country	Number	Name	Concerned species
Albania	0		
Algeria	1	El Kala National Park	P oceanica
Bosnia-Herzegovina	0		
Croatia	5	Lokrum Nature Reserve Malostonski Zaljev Nature Reserve Kornati Islands National Park Brijuni Islands National Park Mljet National Park	C nodosa P oceanica P oceanica P oceanica and C nodosa P oceanica
Cyprus	1	Lara-Toxeftra Nature Reserve	P oceanica and C nodosa
Egypt	0		
France	4	Cerbère-Banyuls Marine Nature Reserve Bouches de Bonifacio Nature Reserve Port-Cros National Park Scandola Marine and Coastal Nature Reserve	P oceanica P oceanica, C nodosa and N noltii P oceanica P oceanica and N noltii
Greece	1	Sporades National Marine Park	P oceanica
Israel	0		
Italy	11	Miramare Marine Reserve Tuscan Archipelago Nature Reserve Orbetello and Feniglia Nature Reserve Portoferraio Fishing Reserve Castellabate Fishing Reserve Ustica Marine Reserve Ciclopi Islands Marine Nature Reserve Egadi Islands Marine Nature Reserve Tremit Islands Marine Nature Reserve Torre Guaceto Marine Nature Reserve Capo Rizzuto Marine Nature Reserve	C nodosa P oceanica, C nodosa and N noltii C nodosa and N noltii P oceanica P oceanica P oceanica P oceanica and C nodosa P oceanica P oceanica P oceanica P oceanica P oceanica P oceanica P oceanica
Lebanon	0		
Libya	1	Garabulli National Park	P oceanica
Malta	1	Fungus Rock Nature Reserve	P oceanica
Monaco	1	Larvotto Reserve	P oceanica
Morocco	0		
Slovenia	0		
Spain	9	Columbretes Islands Marine Reserve Medas Marine Reserve Punta N'Amer Nature Reserve Cabo de Palos Marine Reserve Cabrera National Park Salinas de San Pedro del Pinatar Marine Reserve Mar Menor Managed Zone Tabarca Marine Reserve Cabo de Gata Marine Reserve	C. nodosa P oceanica P oceanica P oceanica P oceanica P oceanica and C nodosa P oceanica and C nodosa P oceanica and C nodosa P oceanica and C nodosa P oceanica and C nodosa
Syria	-		
Tunisia	2	Kneiss Islands Nature Reserve Zembra National Park	P oceanica and C nodosa P oceanica
Turkey	-		
Yougoslavia	-		

an area that shelters a *P. oceanica* meadow (Platini, 2000), the only one present in the water of the Principality of Monaco (Vaugelas and Trastour, 2002). However, apart from some areas with particular topographical features (e.g. the Gulf of Gabès, the lagoons on the Libyan coast), the Ramsar Convention is still not very much used for the marine environment (Platini, 2000).

Setting up networks of special areas for conservation (e.g. the Emerald network, implemented in the context of the Berne Convention; the Natura 2000 network, implemented by the member states of the European Union) may also help protect meadow habitats (Platini, 2000). A directory⁴⁷ of Mediterranean protected areas in which marine Magnoliophyte meadows have been identified has been established by RAC/SPA (Platini, 2000; Table IV).

5.2.2. Fishing gear

Restricting or banning certain fishing techniques that are particularly destructive to *Posidonia oceanica* meadows (Ramos-Esplá *et al.*, 1993) can offer them physical protection. This is so for the regulations concerning trawling in Spain and Italy (banned above the 50 metre isobath) and in Tunisia (banned in a 5.6-km wide zone or above a given isobath between 20, 30, or 50 metres according to sector and fishing type; Anonymous, 2002a). The same holds good for Cyprus, where the stationing of trawls in seabeds deeper than 55 metres (Fisheries Law (CAP 135) and Regulations, 1990 to 2002) results in effective protection to meadows as habitats (Myroula Hadjichristophorou, verbal comm.). In France, trawling is prohibited above the 100 metre isobath) (except for some little ships of under 75 kW power which are permitted up to the 12 metre isobath).

5.2.3. Impact studies

Coastal development is one of the major causes of the regression of the *Posidonia oceanica* meadows as regards both the surface area concerned and the irreversible nature of the degradation. Also, the procedures that aim at assessing what impact a development will have before it is carried out, so as to determine whether the project should indeed be completed, constitute a tool for their conservation that could be more generally used in the Mediterranean. Indeed, although common to many States, the impact assessment procedure is still, for many Mediterranean countries, an innovative but essentially theoretical approach (Pergent-Martini, 2000). Although the idea of an impact assessment is known in all the Mediterranean countries, it does not appear systematically in national legislation, and meadows are never specifically mentioned in this (Pergent-Martini, 2000). On principle, according to the threats to meadows, any development being carried out in the maritime domain can justify an impact assessment procedure, and here one should mention the UNEP's approach, with its adopting of guidelines (Anonymous, 2001a).

As said above, in 2003 Liguria (Italy) adopted a technical standard as part of the environmental impact assessment in order to establish the state of conservation of *Posidonia oceanica* meadows, taking the following elements into consideration:

- Regional Law no. 38 of 1998 (*Valutazione di Impatto Ambientale*, VIA) provides that projects for new leisure ports should be subject to the VIA (environmental impact assessment) procedure, and removal of marine sediment, facilities built against coastal erosion, sea walls and any other facility that modifies the coastline should be subjected to the "verifica-screening" procedure.

⁴⁷ This directory does not include all the Mediterranean Marine Protected Areas but only those that have requested and obtained SPA (Specially Protected Area) status.

- The surveys done in Liguria have revealed how inexact are the borders of the Sites of Community Importance (SCIs) concerning the real extent of *Posidonia oceanica* meadows.

It is necessary to consider the true situation of the habitats, and not their theoretical extent appearing in the Sites of Community Importance, in order to avoid a negative judgement on projected facilities according to their impact on habitats that are supposed to be in good condition, independently of possible compensations.

The technical standard, based on the available bibliographical data on the specificity of Liguria's meadows, considers 3 kinds of meadow. The first corresponds to meadows whose state of conservation is unsatisfactory, i.e. does not seem to present the characteristics required by the Habitats Directive for Sites of Community Importance (Table V).

Table V. Determining the state of conservation of *Posidonia oceanica* meadows according to shoot density and depth in Liguria (Italy). To use this Table, one should not look at the absolute density measured in the meadow areas, but the relative density, taking R cover (as a %) into account. From Regione Ligura (2003) in Diviacco and Coppo (2006).

Depth			
(m)	Unsatisfactory	Intermediary	Satisfactory
0-3	< 550	550-900	> 900
>3-5	< 420	420-700	> 700
>5-7	< 330	330-600	> 600
>7-10	< 240	240-500	> 500
>10-14	< 160	160-400	> 400
>14-18	< 90	90-350	> 350
>18-23	< 30	30-280	> 280
> 23	< 10	10-200	> 200

In the upper sectors (<10 metres depth) of the *Posidonia oceanica* meadow in Liguria (Italy), a corrective factor was introduced in relation to the values of Table V. This involved taking into account the fact that the meadows in this region have regressed greatly over the past few decades and thus today present low R cover:

 $R_{corrected} = R + R(100 - R)/100$

The Ligurian technical standard also takes into account, when assessing *Posidonia oceanica* meadows and for impact assessment studies, meadows that reach (or almost reach) the surface of the sea and those that present a particular typology (see §2.4) as well as signs of flowering and fruit formation.

In France, too (Nathalie Quelin, personal comm.), prior to any request for permission for a project that could harm the environment, an assessment of its environmental consequences must be made (Art. L122-1 of the Environment Code). According to the size of the project, it is subject to an **impact assessment** or impact notice, and this document is an integral part of the dossier containing the request for a permit (since the request for a permit might come under various laws such as, for example, the Water Law). The contents are defined in Decree no. 77-1141 of 12/10/1977 modified to implement the Law of 10 July 1976 on nature protection, for projects for work and for development, and in Decree no. 77-1133 of 21/09/1977 for listed facilities, for protection of the environment. In its minimal form, as regards marine habitats, it includes:

- an analysis of the initial state of the site and of its environment,
- a study of the modifications the project would cause,
- measures envisaged to eliminate, reduce and where possible compensate for the harmful consequences on the environment.

Another indirect protection measure for *Posidonia oceanica* meadows in France (Nathalie Quelin, personal comm.) consists in submitting the dossier to a **public enquiry** under the Law of 12 July 1983. This is an indirect protection measure for meadows since this procedure aims at informing the public about development operations likely to affect the environment and collecting the public's remarks, suggestions and counter-proposals prior to certain decisions being made and certain operations set in train, in order to enable the competent authorities to have to hand all the elements needed to inform them. Many citizens have directly or through their representatives (users' associations, environmental protection associations, etc.) expressed their views to the investigating officer for these projects, in this context. Indeed, most of the dossiers on development and/or projects at sea are submitted to public inquiry.

It is interesting to notice that now, in France, a change is happening in awareness of the role of *Posidonia oceanica* banquettes in the beach concessions. Indeed, increasingly frequently there is a paragraph in the deed of concession specifically saying that *P oceanica* debris washed up on beaches will not be cleared away except during the periods of summer visiting (Nathalie Quelin, personal comm.).

5.3. LEGAL ENFORCEMENT OF THESE POLICIES: EXAMPLES OF CASE LAW

In France, legal protection of *Posidonia oceanica* has led to the refusing of certain developments that risked harming the meadow or modifying others. Since the 1988 Protection Order was signed, no development involving the destruction of a *P. oceanica* meadow has gone ahead. The possible destruction of some isolated shoots or residual patches of *P. oceanica* that do not form a meadow has often been envisaged, for example during the Corbière, Marseille, beach developments (Crebassa, 1992). Confronted with the danger of opening a sort of Pandora's box⁴⁸, a loophole in which developers driven by very short-term interests⁴⁹ can take refuge, the only response must be the strict application of the law.

During the project to enlarge the port of Pointe-Rouge (Marseille, France), the town of Marseille first required precise mapping of the sector (Francour and Marchadour, 1989), by which means the areas occupied by *Posidonia oceanica* were effectively avoided. This

was an exemplary case.

After building a private port on the island of Cavallo (Corsica) without a permit, its owner was sentenced by the Ajaccio Departmental Court. Among the offences he was charged with "*mutilating protected plants, and degrading species*" (Meinesz, 1989; Pergent, 1991a). The sentence (Ruling no. 90 of 2 February 1990) consisted of both a heavy fine and a request that the mooring post be destroyed and the coastline be reconstituted as it was originally (by closing an access channel at a depth

- ⁴⁸ Indeed, how does one define the threshold at which isolated shoots or patches of *P. oceanica* do not constitute a meadow that deserves protection? The example of Prado Bay (Marseille, France), where isolated shoots and little patches of *P. oceanica* are rapidly expanding after a waste water treatment plant opened (Gravez et al., 1999), illustrates the danger of such an interpretation.
- ⁴⁹ By very short-term interests, the authors of this work mean the developers' interests, which may be very different from the short-, middleor long-term interests of the people who live in the region.

of 2 metres) within 6 months (Pergent, 1991a). But it should be noted that by 1994 the channel was simply closed off by a chain and the coastline had still not been reconstituted (Boudouresque *et al.*, 1995b).

The existence of legal protection measures also allows us to envisage the implementation of environmental compensation measures for developments that harm a meadow. This is particularly so when work of collective interest such as the laying of underwater pipes or cables is carried out (Meinesz and Bellone, 1989). The replanting of *Posidonia oceanica* as cuttings or seeds has sometimes been envisaged as a compensatory step. However, as Boudouresque (2001) says, its effectiveness has not been fully demonstrated, and replanting must be considered with great caution; anyway, it should only be envisaged within an extremely precise and restrictive regulatory framework (see Chapter 13).

6. DEAD POSIDONIA OCEANICA LEAVES, BEACHES AND SAND REPLENISHMENT

6.1. THE PROBLEM

6.1.1. Dead Posidonia leaves

In the autumn, dead *Posidonia oceanica* leaves which have fallen off the rhizomes both accumulate in the meadow (litter: see §2.6) and are exported to areas of decantation, like sandy and intermatte

areas. Then the autumn and winter storms drive them into other benthic ecosystems, from the infralittoral to the bathyal⁵⁰ stage, or onto the beaches (Fig. 49).

On the beaches, the dead leaves, and sometimes the rhizomes, can pile up locally in considerable quantities until they are perhaps 1-2 metres thick. These piles of dead leaves are called "**banquettes**" in French (banchetti in Italian) (Fig. 27, Fig. 33) (Molinier and



Fig. 49. Dead *Posidonia oceanica* leaves and aegagropiles on a Spanish beach. Isn't it lovely? Photo by J. Corbera in Romero (2004b).

Picard, 1953; Picard, 1965a; Blanc, 1971). These banquettes are made up of *Posidonia oceanica* leaves and rhizomes in different stages of fragmentation and degradation (until they reach fibre stage), sediment and water. This ensemble forms a structure that is at the same time rigid and elastic.

The structure of the banquettes was studied in the Marseille (France) region by Jeudy de Grissac and Audoly (1985). They present a water content of between 30 and 90%, which increases from the upper part (exposed to the sun) down to the lower part of the banquette. The sand content varies from 0.5 to 85% according to the aspect of the site, the hydrodynamics, the texture of the beach matter (defined by granulometry) and the morphology of the beach. The plant matter is basically composed of leaves, fragments of leaves and fibres; the rhizomes are negligible in quantity. The plant debris which form the banquettes can be put into three categories:

- Type 1: Debris that have evolved very little, still green and still keep the 2 edges of the leaf. This represents under 1% of the banquettes.
- -Type 2: Debris that presents much the same features as those of Type 1, but are brown. This represents 1-26% of the banquettes.
- -Type 3: Debris that are very degraded, are brown, and present at most one of the 2 edges of the leaf. This represents 1-99% of the banquettes.

The beaches where *Posidonia oceanica* debris piles up, whether as a simple carpet (Fig. 49) or in banquettes, are usually disliked by bathers because of the smell they can give off, and especially because they interpret these piles as pollution. In reality, such banquettes are a **sign of good** water **quality** – they indicate the nearby presence of vast meadows.

⁵⁰ The main marine stages are, from top to bottom, the Supralittoral (spray area), the Mediolittoral (area of waves and tidal sway), the Infralittoral (between the surface and 25-45 metres deph, a well-lit area), the Circalittoral (up to 70-150 metres depth, an ill-lit area), and the Bathyal (down to the great depths, where there is no light or insufficient light to ensure the presence of photosynthetic organisms). Moreover, they are very **useful**. The base of the banquette is subjected to marine erosion and the leaf debris claimed by the water forms a dense suspension whose viscosity breaks the waves some metres ahead of the banquettes (Fig. 33) (Boudouresque and Meinesz, 1982). Thus the banquettes help **protect the beaches** from being eroded, especially during the winter storms.

Lastly, after the physico-chemical and biological processes of degradation of the dead leaves has finished, the banquettes constitute a source of carbon and nutrients that are used by various organisms along the **food chain**. Eliminating the banquettes should therefore not be done indiscriminately but carefully assessed, because of its negative consequences for the stabilisation of the beaches and the productivity of the coastal environment.

Elimination of banquettes is widely done by the territorial authorities on beaches of touristic interest. The dead leaves are carried off to (and possibly buried in) dumping sites, or piled up in areas near to the beaches, or pushed back into the sea. But on beaches of no resort interest, the banquettes are usually left alone.

Such practice means that in most of the western Mediterranean, where tourism is of great economic interest, the banquettes have become rare. Their scarcity can locally be due also to the regression of *Posidonia oceanica* meadows (see Chapter 4), and thus a decrease of the supply of dead leaves to the beaches.

6.1.2. Use of dead Posidonia leaves

Dead *Posidonia oceanica* leaves have been used by human beings since classical times (and even since prehistoric times) all around the Mediterranean (Boudouresque and Meinesz, 1982). Over 100,000 years ago, at about the end of the Riss ice age, men in the Lazaret cave (Alpes-Maritimes, France) certainly slept on litter composed of *P. oceanica* leaves (De Lumley *et al.*, 1969). Use of the leaves inside **mattresses**, or as bedding for animals, went on for a long time; indeed, "vermin" never entered the mass, certainly because of the phenolic acid contained in the leaves (Font-



Fig. 50. *Posidonia oceanica* aegagropiles on a beach. Photo by A. Meinesz. An aegagropile of record size (17x12 cm.) was observed in Mourillon (Toulon, France) by Jean-Marie Astier (personal comm.)

⁵² This is why botanists prior to Linnaean nomenclature referred to P. oceanica as "alga marina virtariorum" (glaziers' marine "alga"). (Bauhin, 1623, in Grenier, 1860).

51 Aegagropile = "sea ball". The term aegagropile was first used to refer to balls of hair that formed in the stomachs of animals that lick themselves, such as cats, and were later regurgitated; by analogy, balls of *Posidonia* oceanica fibres are so called (Weddel, 1877). Quer, 1990). In ancient Egypt it appears that people made **shoes** with felted aegagropiles⁵¹ (Täckholm and Drar, 1954); aegagropiles (*P. oceanica* balls) are fairly spherical agglomerates of fibre from dead *P. oceanica* leaves, formed by hydrodynamics in shallow waters and then thrown back onto the beaches (Weddel, 1877; Cannon, 1979, 1985).

For centuries, when there was not yet any bubble wrap or expanded polystyrene, *Posidonia oceanica* leaves were used by the Venetians to wrap up and transport their famous, delicate glasswork, to the extent that these leaves were known as "Venetian straw"⁵² (Boudouresque and Meinesz, 1982).

In North Africa (Egypt, Libya, Tunisia), in the early 20th century the coastal populations were still using dried *Posidonia oceanica* leaves to build roofs (Le Floch, 1983). In Corsica, under the roof of a classical sheep

barn, a coating of Posidonia leaves was discovered (Gérard Feracci, personal comm.), probably intended for **thermic insulation** (Boudouresque and Meinesz, 1982). Also for heat and sound insulation, conclusive tests using *P. oceanica* leaves were done in Sicily and Greece several

decades ago (Sordina, 1951). In the early 1980s, the roof of the *Casa communa* in Pigna (Corsica) was heat insulated using dead leaves (Jean-Marcel Vuillamer, personal comm.).

Dead Posidonia oceanica leaves were also used for a long time as **compost** by farmers on the Mediterranean coast. But it seems that they do not constitute a true compost but rather help maintain a certain rate of dampness in the surface soil or aerate overcompact soil if they are buried (Germain de Saint-Pierre, 1857; Sauvageau, 1892; Knoche, 1923; Braun-Blanquet et al., 1952; Astier, 1972). In Corsica, dead leaves were burned over cropland to improve it (Conrad, 1982; Fig. 51). Today, these methods have gone out of favour, but tests have been done in Italy, Tunisia and Greece to produce a P. oceanica leaf-based compost, with interesting results (Sordina, 1951; Saidane et al., 1979; Seri et al., 2004). In Spain, the Denya town council (Comunitat Valenciana) with European funding (Life Environment project, 1996) and jointly with the University of Valencia, set up a composting facility capable of handling 15,000 cubic metres/year of plant waste. The compost obtained by mixing dead P. oceanica leaves with other plant Fig. 51. Late 19th century or early 20th century Corsican girl with a basket full of dead *Posidonia oceanica* leaves. The leaves were burned over the cropland to fertilize it. Reproduction by J.J. Allegrini, from an anonymous photo given by Mme. Conrad.



detritus (in a ratio of about 1/3) has good agronomical features, is rich in oligo-elements and can be used for re-forestation and in other actions to restore the environment.

Freshly-picked dead *Posidonia oceanica* leaves have great nutritive value, like hay and alfalfa (Molinier and Pellegrini, 1966). By adding the powdered leaves to feed for hens in Italy the laying and the weight of the eggs was improved (Baldiserra-Nordio *et al.*, 1967, 1968; Gallarati-Scotti, 1968). In Tunisia, in the 1920s, attempts to feed cattle with the leaves mixed with fodder had a mixed success; donkeys and sheep refused to eat it, but 2 horses accepted it⁵³ (Pottier, 1929; Boudouresque and Meinesz, 1982). *P. oceanica* fruits washed up on the beach were eaten by cattle (Tunisia), pigs (Corsica) and even human beings in times of famine (Cuénod, 1954; Boudouresque and Meinesz, 1982; Roger Miniconi and Gérard Feracci, personal comm.).

Among other uses of *Posidonia oceanica* we can mention the production of paper at the end of the 19th century (Sauvageau, 1890; Lami, 1941). Lastly, the Egyptians attributed curative properties to it, especially for sore throats and skin problems, and an old botanics handbook (Cazzuola, 1880) mentions it as a product in the popular pharmacopeia.

All in all, although in the past dead *Posidonia oceanica* leaves were used by people living on the shore (Fig. 51), although this is usually anecdotal, modern tests have demonstrated the feasibility of its use but have usually run up against economic realities. Moreover, even if changing techniques

should one day make use of the dead leaves profitable, this valorization would not solve the problems raised when the leaves are removed from the beaches (see §6.1.1 and 6.1.3): erosion of the beaches and impact on the food webs of the coastal

⁵³ The Latin historian Hirtius told how in the African war horses and beasts of burden in Caesar's legions were saved by eating dead *P.oceanica* leaves (of course he did not call them this) since there was no other form of fodder (Pellissier, 1853; Brulard, 1885). ecosystems. Also, it would, in the countries where the species is protected (see §5.1.2), run against the ban on its use in any way whatsoever.

6.1.3. Erosion of beaches

A beach is made up of deposits of detritic sediment carried by water currents or produced by marine erosion of the rocky coasts. The existence of a beach is the result of a delicate **balance** between the amount of sediment arriving and that washed away by coastal currents. When the balance is positive, the beach grows. When it is negative, the beach is eroded (Paskoff, 1993; RAMOGE, 2002; SDAGE, 2003).

Beach dynamics is a product of natural parameters such as hydrodynamics, swell, currents, wind and soil erosion, and it can be greatly modified by human activities such as (Boudouresque and Meinesz, 1982; Paskoff, 1993; Boudouresque, 1996; RAMOGE, 2002):

- (1) Reduction of provision of solid matter from the watercourses that flow into the sea after they have been developed (dams, reservoirs) and/or material extracted from their beds.
- (2) Coastal development, with the construction of coastal buildings or structures. Such construction prevents the sand from moving between the beach and the back of the beach (dune); during storms, therefore, it is driven out to sea (Paskoff, 1993).
- (3) The building of port facilities that may constitute barriers to the carrying of sediment parallel to the coast. These bring about a sedimentary deficit for beaches located downstream of the facilities in relation to the dominant current (Cortemiglia, 1979).
- (4) The creating of coastal defence facilities (groynes, breakwaters) that modify the coastal transport of sediment as in point 3.
- (5) The degradation of *Posidonia oceanica* (or other marine Magnoliophytes) meadows. Such meadows stabilize the sediment and reduce hydrodynamism by dispersing the energy above them and on the coast where they grow (see §3.3). Their regression or disappearance thus leads to an increase in coastal hydrodynamism.
- (6) As mentioned above (§6.1.1), uninformed bathers think that dead *Posidonia oceanica* leaves on a beach are a sign that the beach is badly looked after. To make beaches look clean, or because cleaning machines cannot tell a dead leaf from a plastic bottle, the coastal district authorities make sure everything is cleaned up. Banquettes of dead leaves reduce the available space on a beach and smell like "the sea". On the Atlantic beaches, people are used to this smell and actually like it they associate the smell of kelp⁵⁴ with that of untamed nature. But people on Mediterranean beaches, whose ecological culture is often rather smaller, associate the smell with pollution. Now, dead leaves and banquettes help protect beaches, especially during winter storms (Boudouresque, 1996; SDAGE, 2003).
- (7) Over-visiting of beaches. On some beaches there can be 500,000 visitors a day and per 100 linear km.
- (8) Extraction of fresh water from underground reserves, which can cause subsidence as well as making the ground water irreversibly salty.

The frequent concomitance of these activities on the same stretch of coast, added to natural factors

54 "Varech" is the Breton word for "algae", especially seaweed on the beaches (foreshores). (Paskoff, 1993) brings about a very delicate situation: a considerable number of beaches are decreasing, sometimes dramatically so. To attempt to compensate for this, very many coastal districts have started operations of **beach replenishment**. The amounts of sediment added can be considerable: in 1994 and 1995, 14,600 t of gravel and 3,800t of sand were spread over a single beach, Almanarre (Gulf of Giens, Var, France). Between Capo Noli and Capo Vado (Liguria, Italy), a 9km-long beach has since 1970 been totally covered every two years with sand (Relini, 1992; Boudouresque, 1996).

Adding sedimentary material to beaches in an attempt to give them a greater extent of sand can be a serious ecological problem for the benthic ecosystems because of the unsuitable nature of the material (silt, clay, "earthy" material generally) which changes the granulometry of the sediment. The nature of the soft substratum settlements is closely linked to the granulometry (Pérès and Picard, 1964). The use of unsuitable material also has negative effects on the *Posidonia oceanica* meadow (Relini, 1992). The worst aspects linked to the use of unsuitable material (for it contains fine sediment) are:

- (1) The increase of water **turbidity**, which reduces the interval of depth compatible with photosynthesis of *Posidonia oceanica* and thus causes its lower limit to rise ("compensation depth") (see §4.3).
- (2) The silting of the meadow, a phenomenon that means that fine sediment is deposited on the leaves (reducing their photosynthetic capacity) and an increased sedimentation rate. It should be remembered that the meadow traps sediment, that the growth of orthotropic (vertical) rhizomes usually compensates for the entry of sediment, but that if this entry is greater than a thickness of 5-7 cm. a year it is no longer compensated for by the growth of the rhizomes: the vegetative tips are then buried and *Posidonia oceanica* dies (Boudouresque and Jeudy de Grissac, 1983; Boudouresque *et al.*, 1984; Jeudy de Grissac and Boudouresque, 1985) (see §4.1).

6.2. CASE STUDIES

6.2.1. Managing banquettes in Malta

In Malta, a coastal vegetation management programme has been undertaken in collaboration with the university. There is a double aim: eliminating dead *Posidonia oceanica* leaves from the beaches for touristic purposes, and restoring the coastal terrestrial vegetation in sectors where it is degraded.

Dead *Posidonia oceanica* leaves taken from the beaches are heaped up, with other organic matter, in piles about 1.5 metres thick. After 2 years, plants such as *Atriplex halimus* are planted along the perimeter of the piles, and *Tamarix* sp. and Acacias are planted inside. After 15 years of this experiment, the results seem to be positive, in that green barriers of about 2 metres have been created, barriers which hinder access to the beach (particularly for motor vehicles) and encourage the recolonization of the back of the beach by pioneer growth.

We notice that although this experiment constitutes one of the rare examples of true valorization of dead *Posidonia oceanica* leaves (see §6.1.2), it fails to solve the problem of erosion of beaches and the impact on the marine ecosystems that removal of the banquettes causes. The praiseworthy reconstitution of vegetation at the back of the beaches can be achieved by other methods (see §6.2.3).

6.2.2. Port-Cros and Porquerolles beaches

In the islands of Port-Cros and Porquerolles (Hyères, Var, France), the Hyères district and the Port-Cros National Park have carried out an interesting experiment. According to the location and visiting of the beaches, 5 cleaning levels were defined (Table VI; Auby, 1998). Generally speaking, 3 principles were adopted: (i) non-natural or dangerous waste is removed as far as possible; (ii) removal and cleaning is done exclusively by hand; (iii) dead *Posidonia oceanica* leaves and banquettes (and these are fairly low) are left where they are.

Level	Material removed (+) or not removed (-)					
of intervention	Artificial material®	Trunks (diameter >30cm)	Thick timber (diameter 5-30cm, length > 50cm)	Branches (diameter < 5cm, length < 50cm)	Dead Posidonia oceanica leaves	
0	2	2	<u></u>	<u>.</u>	79252	
1	+	÷.		-	1 1 15	
2	+	+	+	-	79 <u>2</u> 98	
3	÷	÷.	÷	÷	set.	
4	+	2	+	+	경달성	

Table VI. Cleaning levels on Port-Cros and Porquerolles beaches (Hyères, Var, France). + = removal, - = no removal. From Auby (1998)...

* Artificial material oil residue, plastics, rope, metal, glass and worked wood (=wood worked by humans).

For every beach and every season (touristic season, low season) a level of intervention has been defined (Table VII; Auby, 1998). Frequency of cleaning varies between daily and monthly according to the beach and the season (Patrick Auby, personal comm.).

Bathers are informed about this beach management strategy by explanatory boards (Fig. 52). They can read on the board: "A natural beach. The presence on the beach of dried leaves is a sign of the good health of the nearby marine environment, where a true underwater meadow of flowering plants, Posidonia, is developing. Parts of their leaves drop off in the autumn and lie on the beach

over the winter. This natural carpet is very clean and protects the beach sand against sea action."



It is interesting to state that this policy (the non-removal of dead Posidonia oceanica leaves) associated with information, has not stopped visitors: (i) number of visitors is no different for beaches with few dead leaves and those well covered with dead leaves; (ii) on beaches with a relatively high cover of dead leaves, the number of visitors did not decrease after the strategy of nonremoval of dead leaves was introduced. Simply, the bathers who arrive first choose the places without dead leaves (Philippe Robert, personal comm.). This experiment of non-removal of dead leaves plus information on boards has been taken up by the districts of the Marine Observatory of the des Maures Coast (Cavalaire, Rayol-Canadel, La Croix-Valmer and Ramatuelle, Var, France.

Fig. 52. An information board on a Porquerolles beach. Photo by P. Robert.



Table VII. Examples of levels of intervention (see Table VI) for cleaning beaches, in the islands of Port-Cros and Porquerolles (Hyères, France) according to site and season. Touristic season from 1 May to 30 September. Low season from 1 October to 30 April. From Auby (1998).

Beach	Season	Level of intervention
South beach (Port-Cros)	Low season	1
	Touristic season	4
Port-Man bay (Port-Cros)	All year round	1
Port-Cros beach (Port-Cros)	Low season	1
	Touristic season	2
Brégançonnet (Porquerolles)	Low season	0
	Touristic season	2
Middle of Plage d'Argent (Porquerolles)	All year round	3
Plage de la Galère (Porquerolles)	All year round	0

6.2.3. Almanarre beach, Hyères

Almanarre beach lies at the base of the western Giens tombolo (Hyères, Var, France). It is a muchvisited beach, 5 km. long. On several occasions, the most recent being 1992 and 1994, the spit of the western tombolo has been broken by storms of exceptional force (Frédérique Lantéri-Gimon, personal comm.). There are certainly many reasons for this breaking; the main one is probably the sedimentary deficit of the Giens harbour, into which (today) no major watercourse, likely to facilitate sediment input, flows. More generally speaking, in Provence, the 20th century has been marked by a drop in farming, reconstitution of the forest cover, reconstitution of the soil and thus reduction of terrigenous sedimentary inputs into the sea (Gravez et al., 1988). However, the destruction of the dunes behind the beach, the building of a road too near to the sea, and the rip-rap intended to fight against erosion of the beaches – but whose effect has been to accentuate this erosion – have also played a negative part. In any case, the breaking of the western tombolo in 1992 and 1994 was the spark that led managers to question a certain number of practices and to set up integrated management of the area as a whole (Daniel Barbaroux and Frédérique Lantéri-Gimon, personal comm.).

Almanarre beach (Fig. 53) is managed today by the Environmental Service of the Hyères town council, with technical advice from the Port-Cros National Park and the Botanical Conservatory of Porquerolles. Its **management measures** are as follows (R. Barety,

personal comm.): (i) non-removal of dead *Posidonia oceanica* leaves and banquettes over a 1 km stretch (since 1996) and then over the southern 3 km (since 2001); on the rest of the beach, these dead leaves are relatively rare; (ii) artificial macrowaste (plastics, glass, worked wood etc.) is removed by hand every day during the touristic season (15 June to 15 September); the natural wood (trunks, branches) is left where it is; (iii) since 1996, the dunes at the back of the beach have been protected by wooden barriers (ganivelles); people can only cross the dunes where there are signposted crossings (about one passage every 100 metres). Some of the crossings are wide enough to allow windsurfers to pass. This protection has enabled some degraded dunes to increase in height by 2-5 cm a year; (iv) in some sectors where the dunes had been completely destroyed by trampling they have been artificially reconstituted, and replanted with the sea



Fig. 53. The northern part of Almanarre beach in May 1997. The dead *Posidonia oceanica* leaves are visible on the beach. Photo by R. Barety and F. Lantéri-Gimon.

daffodil *Pancratium maritimum*, the maritime daisy *Anthemis maritima*, the cotton weed *Otanthus maritimus*, and the sea-holly *Eryngium maritimum*; all these plants are species that are characteristic of the dunes at the back of the beaches in the region; this operation was supervised by the Botanical Conservatory of Porquerolles, given the decrease of the northern part (over 2 km) of Almanarre beach, and in 2 points beach **replenishment** operations are continuing. At first, this involved gravel (15-20 mm in diameter) from the Durance. Then finer, marine origin, sand was chosen from a quarry in Signes (diameter <5 mm); and sand dredged from the access channels to certain Var ports was added. The total amount is 10,000 cubic metres a year. Rip-rap, thought to protect the beach but unfortunately eroding it instead, was removed (R. Barety and Frédérique Lantéri-Gimon, personal comm.).

Informing the **public** about why these management measures are being taken is done verbally by a dozen seasonal agents, through the local press, and, to a lesser extent, by a leaflet produced by the Hyères town council and by posters distributed to shopkeepers (Frédérique Lantéri-Gimon, personal comm.). Visitors to the beach did not decrease after these management measures were put into effect. They have remained stable from year to year; the restrictive factor is the number of parking spaces. Generally speaking, the public's reaction to these management measures has been extremely favourable; people are respecting the obligation to only cross the dunes behind the beach using the signposted crossings (R. Barety, personal comm.).

6.3. RECOMMENDATIONS

Beach replenishment aims at countering a situation (erosion of the beach, deficit of sediment) which has usually arisen because of an alteration in the former balance. To really solve a problem of this kind sustainably, and thus often more economically, the **cause** should be addressed, i.e. the factor leading to the disbalance should be corrected.

(1) Avoiding the reduction of beaches, means first of all permitting the free movement of sand between the beach and the back of the beach (dune): (i) protecting the dune against trampling by means of obstacles to crossing. Users must be informed why it is wished that they only cross the dune at certain duly signposted crossings. When the dune has been degraded by trampling, introducing protection allows it to reconstitute itself naturally⁵⁵; (ii) banning all construction (roads, buildings) on the beach and at the back of the beach - it must only start beyond the dune; (iii) keeping dead Posidonia oceanica leaves and banquettes of dead leaves on the beach. Users should be told why they have not been removed: to protect the beach from erosion and to maintain underwater life; moreover, the public should be made aware that leaves washed up on the beach indicate the presence of Posidonia meadows nearby and thus the good overall quality of the water. Use of phrases like "ecological beach" or "bio beach" is recommended⁵⁶; (iv) if a choice is made to remove (or reduce the amount of) banquettes of dead leaves, this should be done as late as possible before the touristic season to allow them to act as protection during the greater part of the year; (v) when the state of an artificial groyne (or any other facility intended to "protect" the beach) deteriorates, before it is repaired or rebuilt a study should be done on a correct scale (the hydro-sedimentary unit which it is part of) to make sure that such repair or rebuilding is really appropriate.

⁵⁵ As well as the examples mentioned above (particularly Hyères, Var, France), the management of Pampelonne beach (Var, France), with the reconstitution of the dune environment, deserves to be highlighted.

⁵⁶ Non-removal of dead *Posidonia oceanica* leaves is a practice that is becoming more widespread. As well as the examples described above (§6.2.2 and 6.2.3) one can mention Gigaro beach (Croix-Valmer, Var, France) and an Antibes beach (Alpes-Maritimes, France) (Corine Lochet, personal comm.).

(2) When there really is no alternative to **beach replenishment**, the following rules should be borne in mind: (i) the material used should be composed of sediment whose granulometry is sufficiently coarse to effectively absorb the force of the waves. And gravel gives the shore a more sloping profile and thus increases the surface area of the beach more than sandy material would do (RAMOGE, 2002); (ii) no sedimentary material must be discharged directly over Posidonia oceanica meadows; (iii) if P. oceanica meadows are present within 300 metres of any part of the beach (including parts of the beach where the material is not directly discharged; drifting currents disperse it fairly quickly over the entire beach), the materials that can be used for this spreading of sand should have certain specific features. In Liguria (Italy), the Normativa regionale della Liguria (L.R. no. 13/1999)⁵⁷ concerning sand spreading projects indicates, as well as respect for a number of conditions, the granulometric nature of the material to be used, provided according to Wentworth's scale, indicating the main granulometric fractions (as a percentage of the weight: gravel, sand, silt and clay); a prior study is needed when the addition is over 10 cubic metres per linear metre of beach. In particular, the Normativa regionale della Liguria provides that the material to be used on the coasts of Marine Protected Areas, Natura 2000 sites (Habitats Directive) and areas that host hard substratum settlements of great heritage value, must satisfy the 2 following conditions: a maximal quantity of pelite^{ss} (2%), and a maximal quantity of pelite per linear metre of beach (and per period of five years) (0.8 cubic metres). The Normativa regionale della Liguria could be used as a model in other Mediterranean regions.

(3) So far, **valorization** of dead *Posidonia oceanica* leaves taken from the beaches has remained, despite many attempts, more an issue of traditional ethno-sociology than of economic future. Even if in future the leaves could really be made good use of, their removal should be ruled out because of the role they play in protecting the beaches against erosion and in the food webs of the coastal ecosystems. In France, where *P. oceanica* is a protected species, removal is in fact **illegal** (see §5.1.2 and protection text) and the valorization of these dead leaves is also illegal. It is only in the Mediterranean regions where the species is not protected and where it is not threatened (regions so far unidentified) that use of the dead leaves of *P. oceanica* could legally be envisaged, although we strongly advise against it.

⁵⁷ http //www.comuneloano.it/comune/documenti/110lr13_99%20lr01-02.pdf
⁵⁸ Pelite sediment whose grain diameter is under 0.063 mm.

7. THE POSIDONIA OCEANICA MEADOW AND MANAGEMENT OF FACILITIES ON THE MARITIME PUBLIC DOMAIN

7.1. THE PROBLEM

The Maritime Public Domain is the space between the low water line and the high tide line or the limit to where the waves reach the greatest known temporary range. It includes the beaches and areas of loose material such as sand, gravel or pebbles, land reclaimed from the sea and ports.

Coastal development mainly involves building on the shore (like marinas), building ports, breakwaters (sea walls running parallel to the coast) or groynes (sea walls running straight out from the coast), creating artificial beaches and pushing the shoreline further out to sea using sea walls (reclamation⁵⁹). One could add seaside urbanization (especially along the beaches). "**Balearization**" means that such developments pretty much run into each other along the beach in the context of touristic activity: the word is taken from "Balearic Islands", which were among the first Mediterranean regions (starting from the 1950s) to undergo such massive, chaotic development (Ros, 1994).

In the Mediterranean, **the continental shelf is usually narrow**, so that seabeds of less than 50 metres only represent about 5% of the surface area. With a few rare exceptions, all coastal development is concentrated on seabeds of less than 20 metres; now, most of the **primary** production⁶⁰, of the plant biomass, and thus of the animal biomass, as well as a great deal of the **biodiversity** (species diversity and ecosystem diversity) is concentrated in these depths (Meinesz *et al.*, 1985; Boudouresque, 1996). Moreover, habitats defined as priority⁶¹ are found on seabeds of less than 50 m (*Posidonia oceanica* meadows, coralligenous bioconstructions). And fish **nurseries** are often located in very shallow water (less than 10 m depth) (Vigliola, 1998; Vigliola *et al.*, 1998).

The percentage of the surface area of infralittoral⁶² seabeds occupied (and whose settlements are irreversibly destroyed) by coastal development is already considerable in some Mediterranean regions, like the PACA Region (Provence-Alpes-Côte d'Azur, France; Table VIII; Fig. 54) and Liguria (Italy). The development of Le Mourillon artificial beaches in Toulon, where 22 hectares have been covered by direct discharge into the sea of earth, rubble and waste, is particularly representative (Astier *et al.*, 1980).

Table VIII. Percentage of the surface area of the infralittoral seabed and percentage of the shoreline occupied by coastal development in the Provence-Alpes-Côte d'Azur Region (France) from Martigues to Menton (656 km of coastline). From Meinesz and Lefèvre (1976a, 1976b, 1978), Meinesz et al. (1981a, 1982, 1985, 1990c, 1991b).

Sector	Seabed 0-10 m	Seabed 0-20 m	Shoreline
Eastern Bouches-du-Rhône	27%	19%	21%
Var	11 %	7%	12%
Alpes-Maritimes and Monaco	20%	12%	24%
Region as a whole	15%	10%	16%

⁵⁹ "Reclamation" means claiming areas from the sea and using these for coastal development (generally urbanization).

⁶⁰ Primary production production of living matter from carbon dioxide and nutrients by using light (photosynthesis) or chemical (chemosynthesis) energy.

⁶¹ Habitats of Community interest in the context of the "Habitats Directive" of the European Community and the network of Natura 2000 sites.

⁶² The infralittoral stage (=sublittoral stage) corresponds to the bathymetric area occupied by *Posidonia oceanica* meadows. It starts a few centimetres below average sea level and extends (according to the transparency of the water) down to 23-40 metres depth (Pérès and Picard, 1964).


In all, in the PACA Region (from Martigues to the Italian border, including Monaco), 3 059 hectares of shallow depths are covered by development, i.e. 15% of shallow depths located between 0 and –10 metres, and **10% of seabeds** between 0 and –20 metres (Table VIII; Meinesz *et al.*, 1985, 1991b). Out of 656 km of coast, **106 km** is now man-made, i.e. 16% of the coastline. There

are 134 ports and shelters (i.e. on average one port or shelter every 5 km); in the Alpes-Maritimes alone there are 38 ports (or shelters) for 119 km of coast (i.e. one port every 3.1 km). In the Camargue, 145 rocky groynes (about 200 metres long) have been built along the sandy coast, so that half the coastline is now manmade (Eric Coulet, personal comm.). In the Languedoc-Roussillon Region (France), there are 32 ports for 214 km of coast, i.e. on average one port every 6.7 km (Boudouresque, 1996). Liguria in Italy, the Balearic Islands (Mallorca, Ibiza), part of Sardinia, and southern Cyprus are also badly affected regions. In Genoa province (Liguria, Italy), 33% of the coastline is man-made (Giuliano Fierro⁶³). In Corsica⁶⁴, however, less than 1% of the seabeds between 0 to -10 metres are built areas (Meinesz et al., 1990c).



Fig. 54. Extent of the Lacydon (the Greek name for the bay) of Marseille (France) (outer contour) and isobaths (inner contours) in about 600BC, compared with the "Vieux port" and urbanization today. From Millet *et al*, (2000).

The impact of coastal development is a major cause of the reduction of the *Posidonia oceanica* meadows. This impact can be **direct**, by covering up – this is the case, for example, of the port of Beaulieu-sur-Mer (Alpes-Maritimes) (Fig. 55; Meinesz and Lefèvre, 1978) and the port of Bandol, where a car-park area was reclaimed from the sea over one of the last *P. oceanica* barrier reefs in the north-western Mediterranean (Fig. 56; Pérès and Picard, 1963; Boudouresque and Meinesz, 1982). It can also be **indirect**. A port is often a major source of pollution that escapes from waste water treatment: "anti-fouling" paint ⁶⁵ of ships' hulls, waste water discharged from boats⁶⁶ when the boats and/or ports are not fitted with waste water recuperation systems. Such discharges



Fig. 55. An example of the disappearance of a *Cymodocea nodosa* meadow and the reduction of a *Posidonia oceanica* meadow as a result of the building of a port cross section at the western quay of the port of Beaulieu-sur-Mer (Alpes-Maritimes). From Meinesz and Lefèvre (1978).

⁶³ Professor of Marine Geology at Genoa University, Italy. Verbal communication at the Hydrotop Scientific Colloquium, Marseille, April 1996.

64 The coastline of Corsica, measured on 1 25 000 maps, is 941 km long (Meinesz et al., 1990a).

65 "Anti-fouling" paint is intended to prevent living organisms (=fouling) from developing on ships' hulls. National and European laws have led since the 1980s to a reduction in the toxicity of the biocides used

⁶⁶ The discharging of waste water by boats is usually forbidden within the port, at least in countries in the RAMOGE area, although this does still happen there.



Fig. 56. The port of Bandol (Var, France). The two areas that surround the harbour (above and below) were reclaimed from the sea. The bottom one covered one of the last *Posidonia oceanica* barrier reefs in the north-western Mediterranean. Anonymous photo.

are frequent when leisure boats are used as secondary residences, as is often the case in the Mediterranean (Meinesz et al., 1985; Boudouresque, 1996). P. oceanica is known to be sensitive to pollution (Boudouresque and Meinesz, 1982; Augier et al., 1984b; Pergent et al., 1995; Boudouresque, 1996; Argyrou et al., 1999; Delgado et al., 1999; Pergent et al., 1999). Furthermore, as a photosynthetic organism, P. oceanica is very sensitive to even shorttermed water turbidity, for example turbidity generated during coastal development work (Meinesz and Laurent, 1978; Boudouresque, 1996; Ruiz-Fernández, 2000; Ruiz and Romero, 2001).

Rocky groynes running straight out to sea have often been built in an attempt to fight erosion of beaches (see Chapter 6). These groynes hinder the drift of currents and sediment along the coast. Ports and reclamations have a similar effect. The result is that upstream of the groyne sediment accumulates, and downstream there is a deficit of sediment. If the entry of sediment is over 6-7 cm/year, the orthotropic rhizomes of *Posidonia oceanica* are unable to compensate by their vertical growth for being buried (Boudouresque and Jeudy de Grissac, 1983; Boudouresque *et al.*, 1984; Jeudy de Grissac and Boudouresque, 1985; Romero, 2004b). The vegetative tips are then buried and die; the meadow is destroyed. Conversely, downstream from the development, the departure of sediment causes baring of the rhizomes (see Fig. 41). The bared meadow is then extremely vulnerable to hydrodynamism (swell, storms), trawling (at depth) and the anchoring of boats; in the long term it is also destroyed (Boudouresque, 1996).

7.2. CASE STUDIES

7.2.1. Developing Le Mourillon beaches in Toulon

The Vignette Bay (Le Mourillon, Toulon, Var, France) was until the 1960s occupied by a set of small



Fig. 57. The artificial beaches of Mourillon (Toulon, Var, France) were built over a *Posidonia oceanica* meadow, in particular a barrier reef. The most recent development (a nautical club), on the right, does not appear in this photograph. From Astier *et al.* (1980).

beaches much visited by the people of Toulon, and by a *Posidonia* oceanica barrier reef which extended out to sea in a vast plain meadow.

This was, moreover, a site much used by artisanal fishermen (Astier *et al.*, 1980; Astier, 1984). Between 1964 and 1979, reclamation took place (artificial beaches and a nautical club) in this site (Fig. 57). No precautions were taken (for example, prior rip-rap) before replenishing the area: companies and individuals discharged clayey earth, rubble and miscellaneous material into the sea. When the east wind blew strongly, the finer materials were carried off by hydrodynamism (Fig. 58 and 59; Astier, 1984). Overall, **22 hectares** (lagoon, barrier reef and *Posidonia oceanica* plain

meadow) were directly covered. Moreover, hypersedimentation and the increased turbidity of the water resulted in the destruction of **10 hectares** of meadow (rise of the lower limit) in front

of the facilities built. Lastly, 37 hectares of meadow were seriously degraded by silting (Nodot et al., 1978; Astier et al., 1980). Without a map of the meadows before the start of the development, the figures are probably much higher⁶⁷. At the exit to the most eastern cove of Le Mourillon, a bit of a P. oceanica barrier reef has escaped being covered (Charbonnel et al., 1996). Almost two decades later, the situation of the meadow seems to have stabilized in front of the development as regards both its upper limit, where it reaches the foot of the rip-rap, and its lower limit; 78 hectares of meadow remain, the lower limit of which is between 10 and 14 metres depth; beyond this, the seabed is made up of nearly 200 hectares of fairly silted up "dead matte"; as hydrodynamism stirs this sediment up into suspension, great turbidity results, which seems to slow down seriously any recolonization by P. oceanica (Charbonnel et al., 1996; Bernard et al., 2001).

If the economic and social benefits (increased visiting of the beaches, several years of activity for maritime industries) and the ecological and economic costs, which will increase in the long term, i.e. for centuries, are weighed in the balance, the final outcome will be strongly negative. Today, such an error in coastal development would probably finish in legal proceedings, but in fact would perhaps not even have occurred.



Fig. 58. Developing Le Mourillon's artificial beaches (Toulon, Var, France). A lorry discharging earth directly into the sea.



Fig. 59. Developing Le Mourillon's artificial beaches (Toulon, Var, France). Fine sediment is carried off by hydrodynamism. From Astier et al. (1980).

7.2.2. Building Pointe-Rouge port in Marseille

Until the end of the 19th century, Prado Gulf (Marseille, France) was occupied by a vast *Posidonia oceanica* meadow (Marion, 1883). In the 1960s, the meadow was still well represented there, although its lower limit was only 24-29 metres deep (Massé, 1962; Harmelin and True, 1964). Vast stretches of "dead matte" extended further in depth, down to about 32 metres, but it is possible that these were old (several centuries old; Gravez *et al.*, 1992) and their presence not due to contemporary human activity. Later, the Prado Gulf meadow regressed seriously; not only did its lower limit rise (to 22-24 metres depth) but there were splits between its upper limit (located at about 9-10 metres depth) and its lower limit; it was interrupted by stretches of "dead matte" (Gravez *et al.*, 1995, 1997).

There were many reasons for this regression. (i) Pollution and turbidity introduced by the inputs of a coastal river, the Huveaune (before it was diverted towards the Cortiou sewage discharge in 1977);

(ii) pollution due to the town of Marseille's waste water discharges into the sea (equal to 1.5 million population equivalents) into

⁶⁷ Vast stretches of "dead matte" located beyond the lower limit of the meadow, which occupies almost 200 hectares, could be at least partly linked to development.



From Francour and Marchadour (1989).

Fig. 60. Map of types of seabed and of the Posidonia oceanica meadow around

Pointe-Rouge port (Marseille, France). Only part of the maps made appears here.

the Cortiou calangue, waste water brought by the dominant east-west current into the Prado Gulf (before a treatment plant started operating in 1987) (Bellan-Santini, 1966; Bellan et al., 1999); (iii) turbidity brought about by the construction of the Prado artificial beaches and, especially, by the building of Pointe-Rouge port in 1968, which directly covered 11 hectares of Posidonia oceanica meadow and indirectly destroyed 68 other hectares, i.e. six times the surface area of the development itself (see §4.1) (Gravez et al., 1992). From the moment the town of Marseille's waste water treatment plant started operating (1987) the trend was reversed: monitoring of permanent quadrats and permanent transects has demonstrated the progression of

clumps of *P. oceanica* (Gravez *et al.*, 1995, 1997); this progression is of course very slow, given the plant's biological characteristics (see §4.7), and it will certainly take several centuries to get back to the situation of the late 19th century. In these conditions it became vital to manage coastal development (former or future) so that no impact, direct or indirect, could undermine the natural reconstitution of the Prado Gulf *P. oceanica* meadow.

In the late 1980s, the town of Marseille envisaged enlarging Pointe-Rouge port. So as not to repeat past errors and to take into account the legal protection of *Posidonia oceanica* and its ecological and economic importance, an exemplary procedure was followed. Unlike what was usually done – first making development plans and then doing an impact study, whose conclusions might run awkwardly counter to projects that were already far advanced – the town of Marseille undertook a preliminary study of the seabed (in particular the *P. oceanica meadows*; Fig. 60); this study was not restricted to the immediate surroundings of the port but took in a relatively vast area (Francour and Marchadour, 1989; Francour and Gravez, 1990). Only after this did the town of Marseille ask promoters to come up with projects for enlarging the port; all the ecological information, in particular the map of *P. oceanica*'s distribution, was by then available (Fig. 60). Naturally, this procedure does not exempt it from making a later impact study, which is compulsory.

7.2.3. Banyuls-sur-Mer port

The port of Banyuls-sur-Mer (Pyrénées-Orientales, France) is a little recreational port (350 boats in all) that also has some fishing boats. Between 2 and 5 metres depth, near the entrance to the port, there is a *Posidonia oceanica* meadow. It occupies about 10% of the port area and presents signs

68 Plagiotropic rhizomes are shoots situated at the tip of creeping rhizomes that tend to colonize free areas and thus extend the surface area of the meadow.

⁶⁹ Leaf Area Index (LAI) the total surface area of the leaves per m of soil surface. Only one side of the leaf is counted when calculating the LAI. of fairly good vitality: shoot density (350-500/m²), presence of many plagiotropic shoots⁶⁸, length of leaves and leaf area index⁶⁹ as high as,





or higher than, in the open sea in the region, scarcity of "dead matte" (Fig. 61; Pergent *et al.*, 1991). This meadow was present before the port was built (Pruvot, 1894) and nothing allows us to argue that it has regressed after the building, except at the entrance to the port, where it was destroyed by the dredging of an access channel (Pergent *et al.*, 1991).

The persistence of a *Posidonia oceanica* meadow in a port basin is atypical; in almost all the ports built over a meadow, the meadow has disappeared. It is therefore interesting to try to understand the reasons for this situation (Pergent *et al.*, 1991). (i) This is a little port; (ii) the port's rate of occupancy outside the touristic season is fairly modest – under 60%; (iii) no waste water is discharged, and there is no boat maintenance companies, in the port. Furthermore, leisure boats are not used as floating campers in the summer, which restricts the discharge of pollutants; (iv) there is no storm overspill in the port. Thus the salinity remains permanently high there. And *P. oceanica* is very sensitive to desalination (Ben Alaya, 1972); (v) the port is exposed to hydrodynamism in a region with a great deal of wind and probably rapid renewal of water; also, it is near the entrance to the port that the *P. oceanica* meadow is located (Fig. 61); (vi) the landing stages for the boats are all built on stilts, which facilitates the movement of water in the port.

7.2.4. Ospedaletti

The coast in front of Ospedaletti (Liguria, Imperia province, Italy) is characterized by the presence of a vast *Posidonia oceanica* meadow. Between the railway line and the sea, sizeable discharges, dating from about thirty years ago, have changed both the landscape above the sea and the seabed (Fig. 63 and 64). The *P. oceanica* meadow (Fig. 62) that used probably to be continuous along this coast, today shows signs of regression there, with in particular areas of "dead matte" that break it up and reduced width. Its length is 2.5 km and its surface area about 45 hectares.

This part of the coast was the subject of a project for a recreational port accompanied by various coastal facilities. Something may be said about this project and its implications for the marine environment.



On one hand, a **priority habitat** (the *Posidonia oceanica* meadow) is present, a habitat for which a Site of Community Interest (SCI) has been defined. Thus any intervention that could have consequences for this site must be subjected to an impact assessment.

On the other hand, there is **local demand** for a port to be built, which would correct the area's lack of mooring places (berths) and provide jobs. Furthermore, the project would solve the problem of the big waste dump that over the past decades has had a very negative impact on the natural environment both above and below the sea surface.

Analysis of the available information enables the following considerations to be formulated. (i) There is in the area a *Posidonia oceanica* meadow whose vitality is good or acceptable; part of this meadow is included in the SCI. In front of this meadow, but rather at more shallow depth, the meadow is much more damaged and has partially disappeared. Finally, there is an area with practically no *P. oceanica* but which is occupied by *Cymodocea nodosa* (another Magnoliophyte), "dead matte" and sand. (ii) As required by the European Union's Habitats Directive, the area of the meadow in good or acceptable health should be protected as a priority habitat. For the more



Fig. 63. Cervo (Imperia province) facility for defence against erosion of the coast and the railway line (Settore Ecosistema Costiero archives – Liguria region).

degraded parts, like those located at the upper limit, one can envisage carrying out recuperation work, after first making the dump safe and then building the port facilities.

The company which established the project has studied and mapped in detail the marine biocenoses and its conclusions agree with the map drawn by the Liguria region (Diviacco, 2000; Fig. 64). According to the regional regulations and indications (see §5.2.3), the original project – which would have caused the certain destruction of a fairly sizeable part of the meadow – has been modified to only affect the most superficial, fragmented parts of the meadow that are already degraded.

Also, as well as what was provided for by the VIA (Valutazione di Impatto Ambiantale: impact

assessment) regulation, the Liguria Region considers that it is necessary, if the project is accepted, to comply with a certain number of prescriptions intended to minimize the possible damage these coastal facilities could have on the marine environment, for example: (i) using materials that reduce to the utmost the re-suspension of fine particles that is likely to make the surrounding water turbid, (ii) using building techniques that restrict re-suspension of fine particles (turbidity), (iii) paying special attention to how materials are moved about on the seabed, (iv) monitoring the health of the surrounding meadows both during and after the work, and (v) undertaking restoration work, payable by the Project Manager, by replanting *Posidonia oceanica* in the surrounding areas (including areas where the plant was already scattered for other reasons than the building work).



Fig. 64. Cervo (Imperia province) building a coastal defence facility (Settore Ecosistema Costiero archives – Liguria region).

7.2.5. Spotorno

The coast at the limit of the Noli and Spotorno districts (Savona province, Liguria, Italy) was also selected for a project to build a recreational port although the area is occupied by a *Posidonia oceanica* meadow. As in Ospidaletti (see §7.2.4), the project anticipates rehabilitating the whole coastal area, now characterized by rip-rap defence work for the national highway and 2 big waste dumps, which profoundly modified the area both above and below sea level before the publication of the European Union's Habitats Directive (Fig. 65, 66 and 67). As a result, the *P. oceanica* meadow suffered marked damage and regressed over the past decades. The damage was all the worse in that the meadow used to reach the coast, where it constituted fringing reef formations (see §2.4) (Fig. 68) (Bianchi and Peirano, 1995).

The requirements of the European Union's Habitats Directive and the targeting of a SCI for the *Posidonia oceanica* meadow in question make an assessment of this habitat necessary, in order to see whether the project is compatible or needs to be modified to protect the meadow.





Fig. 65. Borghetto Santo Spirito (Savona province) Building a coastal defence facility (Settore Ecosistema Costiero archives – Liguria region).

Fig. 66. Laigueglia (Savona province) A facility to defend the coast against erosion (Settore Ecosistema Costiero archives – Liguria region).



Fig. 67. Varazze (Savona province) Enlarging the recreational port (Settore Opere Marittime ed Ecosistema Costiero archives).

The company that carried out the project made a study of the marine biocenoses that tallies fairly well with the data from the Liguria Region services (Fig. 69; Diviacco, 2000). The building would cause the destruction of 4 hectares of meadow, part of which is regressing and about 0.5 hectares in an acceptable state of vitality. According to the criteria adopted by the Liguria Region (see §5.2.3), the authors of the project modified it to minimize the surface area of *Posidonia oceanica* destroyed and planned a replanting operation under strict scientific control (see §14.4) as a compensatory step.

7.3. RECOMMENDATIONS



Fig. 68. 1973 aerial photo of the coast between Noli and Spotorno (Liguria, Italy). In yellow, the coastline in 1954, in red the coastline in 1973. From the aero-photogrammetric archives of the Liguria region. Crafted on Mapinfo Professional®. From Maggioncalda (2002).

7.3.1. Can a meadow remain in good health in a port?

Generally speaking, the Posidonia oceanica meadow cannot be found in ports, either because it has been destroyed, or because it has never existed there. But there are some ports where the meadow has survived: Banyuls, Le Brusc and Porquerolles. These are relatively recent ports (less than 50 years old) and nothing guarantees that the meadow's persistence will continue in the longer term. Moreover, these ports have some unusual features: wide openings onto the sea, strong hydrodynamism, and little visiting by leisure boat/campers⁷⁰. The movement of the water is not, however, a guarantee of the meadow's survival. In the bay of Port-Cros (Var, France), which is very widely open to hydrodynamism, since there is no outside sea wall, and where the port has functioned for several centuries, the P. oceanica meadow has disappeared from the area where there are landing stages for boats to moor (Augier and Boudouresque, 1970a; Belsher et al., 2005).

As regards the few ports where the meadow has survived, we lack the data on the rate of renewal of the water or on average turbidity that would allow us to define the hydrological conditions that are compatible with the meadow's survival.

70 Leisure boat/campers Leisure boaters who live on their boats and thus use the port as a floating camping site.



In the present state of knowledge, the meadow's survival in some ports must not, therefore, hide the general trend: a meadow included in a port is very probably **doomed to disappear** in the short, medium or long term. Thus it is vital to avoid including *Posidonia oceanica* meadows in a port basin. Moreover, building a port sometimes brings about the disappearance of the meadow, replaced by the port, with the meadow sometimes being cut in two on either side of the port (**fragmentation** of the habitat), as can be seen in Pointe-Rouge (Marseille, France) and Saint-Tropez (Var, France).

7.3.2. Minimum distance between artificial rip-rap and the meadow

There are a certain number of examples of artificial rip-rap (ports, breakwaters, areas reclaimed from the sea) which are in direct contact with the *Posidonia oceanica* meadow. This is so for example in Galeria (Corsica), around the Sporting d'été in Monaco, in front of the nautical club of Le Mourillon (Toulon, France) (Fig. 70), in La Madrague de Giens and Le Brusc (Six-Fours, Var, France) (Charbonnel *et al.*, 1996; Verlaque and Bernard, 1997; Bernard *et al.*, 2002; Charbonnel *et al.*, 2002).

However, in most cases there is a "dead matte" area between the rip-rap and the first living *Posidonia oceanica*. This is so, for example, off the port of Saint-Tropez, Aygade port in Hyères (Var, France), and in Pointe-Rouge port (Marseille, Bouches-du-Rhône, France) (Francour *et al.*, 1995; Francour and Marchadour, 1989; Charbonnel *et al.*, 1997b). The absence of data on the meadow's original state, before and immediately after the building of this rip-rap, makes it impossible to determine with certainty the possible responsibility of this rip-rap.

We think it improbable that rip-rap is the **direct** cause of the presence of a strip of "dead matte", several dozen metres to over 100 metres wide, between its base and the living meadow. The existence of meadows in good health directly in contact with rip-rap, as is the case in La Madrague de Giens (Charbonnel *et al.*, 2002), rather argues to the contrary. It is, however, probable that when rip-rap is exposed to intense hydrodynamism this can erode the meadow.

71 Dumping discharge of solid matter into the sea.



Fig. 70. Map of types of seabed and of the Posidonia oceanica meadow in Le Mourillon nautical club (4th port cove in Toulon, Var, France). Rip-rap appears in grey. From Charbonnel et al. (1996).

In fact, the link between rip-rap and the meadow's regression is often **indirect**. Among the more probable indirect causes are: (i) pollution and suspended matter (turbidity) which spread around a port; (ii) discharge of mud from dredging the port basin at the exit to the port, or too near the exit; dumping⁷¹ areas, defined by the competent authorities, are in fact almost never respected by the companies that have obtained the contract, sometimes thanks to an unrealistic minimizing of costs that should have alerted people to the *de facto* non-respecting of dumping constraints; it is, unfortunately, rare that non-respect for dumping areas is denounced and even more rare that it is punished; (iii) the turbidity generated when the facility is built, when fine materials have been dumped in the sea. This was the case for Le Mourillon (Toulon) and Pointe-Rouge (Marseille, France) (Astier *et al.*, 1980; Gravez *et al.*, 1992, 1995, 1997); (iv) the action of site equipment (barges) when building the facility (see §7.3.3); and (v) the modification of the hydrodynamism, in particular coastal currents, brought about by the facility.

Given everything that has been said above, the difficulty of predicting the impact of rip-rap on the meadow, and the precautionary principle, we recommend a minimum distance of **10 metres** between the new rip-rap and the nearest living Posidonia.

7.3.3. Necessary precautions for a building site

A significant part of the impact affecting the *Posidonia oceanica* meadow after coastal development is linked to construction techniques. To minimize this impact, companies that have been the beneficiaries of a tender should be subjected to a certain number of constraints, and a company should not be systematically chosen because it is the **lowest bidder** but because it is the **best bidder** (the company that is most credible as regards respect for the quality and environment protection regulations, even if it is more expensive, precisely for this reason).

When reclaiming land from the sea, dumping fine material (diameter less than 1mm), or blocks mixed with fine material, at sea must be absolutely ruled out. When laying down rip-rap, the blocks of rock should be rinsed beforehand. Despite these precautions, washing the blocks and re-suspending the sediment that is there generates **turbidity**. Protective **geotextile** screens



(Porcher, 1987) must be set up around the building site to minimize the turbidity caused (Fig. 71). The same holds good when rocks have to be cleared away (Fig. 72). Exemplary precautions were taken to this end by the French Navy in the Canier building site (Saint-Mandrier peninsula near Toulon, France) (Fig. 73) (Bonhomme *et al.*, 2001; Bonhomme *et al.*, 2003c; Spina, 2003).

The site equipment is usually fixed to the seabed for reasons of stability, directly and/or with anchors, which has a very negative effect on the seabed – making holes (base of



Fig. 72. . A barge and rock-clearing equipment during the digging of an access channel to a port in the Var (France). The turbidity of the water caused by the work is obvious. Photo by P. Bonhomme.

equipment) or furrows (chains of anchors) in the *Posidonia oceanica* meadow (Fig. 74 and 75). Use of equipment must therefore be avoided as far as possible, and **use of equipment located on dry land** must be encouraged, especially for laying down rip-rap.

Finally, the season when the work is carried out must take *Posidonia oceanica*'s biology into account. **The summer**, a time when the plant is reconstituting its reserves (stored in the rhizomes) for the following year (Alcoverro *et al.*, 2001) must absolutely be avoided.



Fig. 73. A geotextile screen (left) protecting a *Posidonia oceanica* meadow (right) during underwater work on the Canier site (Var, France). Photo by E. Charbonnel.

Fig. 75. Central part (arrow) of a *Posidonia oceanica* clump eroded by the anchor chain of an underwater work barge. Photo by E. Charbonnel.

These constraints should systematically appear in the contract specifications. Moreover, the Contracting Authorities should carry out efficient on-site checks to make sure they are effectively implemented.

7.3.4. Less "harmful" solutions

There are technical solutions that in certain cases mitigate the impact of coastal development. Sea walls and piers must be as "open" as possible, i.e. must close off the man-made lake as little as possible. Piers **built on stilts** should be given preference over rip-rap walls, which have a strong grip on the seabed and hinder the movement of the water.

Furthermore, for any development accompanying steps should be taken: preventing mooring (anchoring and unauthorized mooring; see §4.5 for definitions) around the work, monitoring the evolution of *Posidonia oceanica* meadows (see Chapter 15), eliminating macrowaste, etc.



8. THE POSIDONIA OCEANICA MEADOW AND MOORING

8.1. THE PROBLEM

Posidonia oceanica meadows are threatened by the mechanical action of anchors of leisure boats

(Fig. 76) and bigger ships (warships, cargo boats or cruise ships) when they moor outside ports (Fig. 77). Even though P. oceanica can recolonize patches that have been degraded by anchors (at least, small patches) thanks to growth and the natural ramification of the rhizomes, their rate of growth is very slow, at most some centimetres per year. Beyond a certain density and frequency of mooring, rhizome growth can no longer recolonize the openings and strips made in the "matte", and the meadow gradually deteriorates: shoot density drops, as does cover72 (Boudouresque and Meinesz, 1982; Francour et al., 1997, 1999).



Fig. 76. Mooring between the Lérins Islands (Sainte-Marguerite and Saint-Honorat Islands, Alpes-Maritimes, France). In this photo, taken on 19 August 1996, one can count 415 boats in all. Anonymous photo.

The most beautiful and famous shores of the Mediterranean are particularly attractive to leisure boaters. It is the very great concentration of mooring⁷³ in sheltered, aesthetic sites, that is a problem for some kinds of seabed, like the *Posidonia oceanica* meadow. Now this meadow constitutes a kind of seabed much sought after for mooring – not only is the anchor held strongly by the network of rhizomes, but there is no danger of its catching on rocks and it is thus easy to raise. When a boat is mooring, its anchor slips over the meadow, possibly furrowing it, and then catches hold between the rhizomes (see Fig. 43). During the mooring, the anchor's chain crushes the leaves. The chain moves with the wind and current (sliding over the seabed) and can draw a circle whose radius is the chain's length, crushing and pulling out a large number of leaves.

Finally, when the anchor is raised, the rhizomes to which it was attached are broken off (Boudouresque *et al.*, 1995a; Milazzo *et al.*, 2002). In some cases, when there are erosion scarps¹⁴, whole blocks of "matte" with a number of living shoots, living and dead rhizomes and interstitial sediment are pulled out (see §4.5).

Of course, the damage varies according to the size of the anchor and of the chain (see below), the weather conditions (greater when there is a strong wind than when the sea is calm), and how the anchor is weighed (greater when the boat pulls on its anchor than when it is positioned above it and weighs it vertically). The extent of the damage caused to a *Posidonia oceanica*

⁷² In a meadow, the density is the average number of shoots per square metre; cover is the percentage of surface area of seabed covered by living meadow (of whatever density) and not by "dead matte" or intermatte.

⁷³ Under the term "mooring" we include anchoring (mooring strictly speaking, using an anchor), organised mooring (when boats moor to deadweights mooring that are legally provided within the context of a Temporary Occupation Permit) and unauthorized mooring (when boats moor at illegally placed deadweights mooring (see §4.5).
⁷⁴ Erosion scame a mini-diff inside or on the limit of a meadow.





meadow by anchors also depends on the frequency of anchoring, the size of the ship, the type of anchor and the nature of the "matte" (compact or loose) (Francour *et al.*, 1997, 1999; Milazzo *et al.*, 2004; Ganteaume *et al.*, 2005).

The mechanical damage done to the *Posidonia oceanica* meadow by leisure boat mooring has been the subject of some studies done in France (Porcher and Jeudy de Grissac, 1985; Boudouresque *et al.*, 1995a; Francour *et al.*, 1997, 1999; Ganteaume *et al.*, 2005) and in Italy (Milazzo *et al.*, 2002, 2004) and constitutes a major worry as to its protection (Doumenge, 1992). One should add to this the anchoring or mooring of big ships, particularly cruise liners and warships (Fig. 77; Roy *et al.*, 1999; Ganteaume *et al.*, 2004, 2005).

In many touristic sites, especially in Marine Protected Areas, to avoid the disadvantages of leisure boats' anchoring, fixed moorings constituted by deadweight moorings and surface mooring buoys (organised mooring) have been provided. But indeed the effects of this kind of mooring on the *Posidonia oceanica* meadow may be even worse than those of anchors (Robert, 1983). Concrete

deadweight moorings, though heavy, can move around, eating into the "matte" and eroding it. The chains that link the deadweight moorings to each other and to the surface buoys dig furrows into the "matte" and destroy the *P. oceanica* growing around the deadweight mooring (Fig. 78).

For bigger ships, for example ships of the French Navy in Hyères Gulf (Var), moorings made of a sea surface steel buoy linked to chains (on the seabed) terminating in fixed anchors each weighing several tons have been put down. The impact of such mooring systems on the *Posidonia oceanica* meadow is considerable (Roy *et al.*, 1999).

A particular problem is leisure boats' permanently mooring in sheltered places and laying "personal deadweight mooring" without a TOP (Temporary Occupation Permit delivered, in France, by the *Direction Départementale de l'Equipement*) and thus completely illegally (unauthorized mooring). The impact may be direct, if the deadweight mooring is located over a *Posidonia oceanica* meadow, or indirect, when they have to cross a *P. oceanica* barrier reef (eroding it on the way) to reach their mooring. This last case is illustrated by the many (about a hundred) boats that come into the Brusc lagoon (Six-Fours, Var, France).



Fig. 78. Furrow dug in a Posidonia meadow by the chain of an organised mooring. Anonymous photo.

Finally, we should not forget that a boat at anchor can constitute a source of pollution: discharge of waste water and solid waste. In Elbu Cove (Scandola, Corsica), over a surface area of 1.3 hectares used by leisure boats for anchoring in July and August, the mass of macrowaste

has been assessed at 54 kg (mainly glass), while being almost zero in a neighbouring cove, Petraghja, very little visited by boats (Bianconi *et al.*, 1990).

8.2. CASE STUDIES

The areas most concerned by the problem of mooring, in the RAMOGE area, are, in Italy (Liguria), the Portofino Peninsula, the island of Bergeggi, the island of Gallinara and Capo Mortola and, for France (PACA region), the Villefranche Bay, the Lérins Islands (Alpes-Maritimes), Port-Cros Island and Porquerolles Island (Var) and the Calanques massif (between Marseille and Cassis, Bouches-du-Rhône). Many much visited moorings lie outside the RAMOGE area, for example the Gulf of Ghjirulata (Corsica).

8.2.1. The Italian coast under the RAMOGE Agreement

The seabed of the **Portofino** Marine Protected Area, whose historical and ecological importance is extremely great, is a "must" for divers from all over Italy and various European countries. *Posidonia oceanica* is present there in 2 large, densely covered areas, surrounded by sparser, smaller meadows. The 2 main meadows lie west (between Camogli and Punta Chiappa) and east (between Paraggi and Santa Margherita) of the Portofino Peninsula, and are the 2 sectors most used by leisure boaters for mooring. In the 2 sectors, respectively about 1 000 and 500 metres long and about 100 metres wide, hundreds of boats of every size, from 3 metres to over 20 metres long, moor, especially in summer and at weekends. Since the Marine Protected Area was set up in 1998, there are rules for activities that provide for the defining of anchoring areas and regulated mooring areas. But for the time being, anchoring is permitted in the sectors that contain *P. oceanica* meadows.

A similar situation is found around the **Bergeggi** (not far from Savona) and **Gallinara** (off Albenga) islets. The seabed facing the land (north/north-west) there is still occupied by *Posidonia oceanica* meadows, although over the past few decades these meadows have regressed, particularly because of trawling, anchoring, coastal development and land reclamation. In the summer they are invaded by boats trying to find mooring sites, which can be over the *P. oceanica* meadows. Anchoring, added to other causes of regression, worsens their effects.

Regarding Gallinara Island, Balduzzi *et al.* (1994) have calculated that within a radius of 56 km around the islet there are about 4 000 places in the ports, and that it is one of the leisure boaters' favourite spots. When we consider that the seabed concerned by mooring is no greater than 10 000 m², and that the same boats visit and revisit the site, we can imagine their impact on the seabed, especially on the *Posidonia oceanica* meadow. Late summer diving has enabled the harm done to the benthic habitats to be verified and the many abandoned anchors observed.

8.2.2. The French coast under the RAMOGE Agreement

From the early 1970s, the *Posidonia oceanica* meadows' regression has been remarked in Port-Cros Bay (Var, France) and attributed to (i) the anchoring of leisure boats and (ii) the movement of boats over very shallow meadows, on which they not infrequently run aground, ploughing furrows in the meadow (Fig. 79; Augier and Boudouresque, 1970a; Boudouresque *et al.*, 1975,



Fig. 79. Aerial photo of the bay of Port-Cros (Var, France). The deep *Posidonia oceanica* meadow (H) appears in dark grey and the barrier reef in grey (R). Between the deep meadow and the barrier reef, the light grey areas represent dead matte. The lines (arrows) show the erosion of the superficial meadow by boats that have run aground there. Anonymous photo.

1980a). The passage of boats has therefore been prohibited in the shallow areas of the bay (especially the barrier reef area), using a rope and a line of buoys. Furthermore, mooring was organised by sinking deadweight moorings linked by a parent chain; secondary chains, on which the mooring buoys were fixed, branched off from the parent chain (Robert, 1983). When the mooring buoys were all occupied, anchoring remained possible.

During the summer of 1982, observation of the *Posidonia oceanica* meadow lying in the organised mooring area revealed that, added to the action of the anchors, the deadweight moorings and chains were a new cause of the meadow's degradation, and that this was more serious than the original deterioration (Robert, 1983). The parent chains would move sideways, pulled by the secondary chains, under pressure

from the boats driven by wind and currents. The result was the digging of a channel whose width varied between 1 metre (near the deadweight moorings) and 6 metres (where the secondary chains joined the parent chain); it was estimated that 1 000 m² of meadow (1% of the bay's surface area) had been thus destroyed by chains sunk for mooring (Fig. 80; Robert, 1983).

Given that the deadweight moorings and the chains were sunk precisely to protect the *Posidonia oceanica* meadow, the cure seemed worse than the disease. And directly attaching the secondary chains to the deadweight moorings would not have solved the problem: as well as reducing the elasticity of the mooring, the first part of the chains would have remained in contact with the seabed, destroying the meadow around the deadweight moorings. Among possible solutions, Robert (1983) suggested **(i)** for deep areas (>6 m), fixing the secondary chain to the deadweight mooring and placing an intermediate buoy on the secondary chain so that it would not be in contact with the seabed; and **(ii)** for shallower areas, providing floating pontoons in a daisy shape linked to a single big deadweight mooring submerged in an area of "dead matte".



Then a technical solution to the problem of mooring in *Posidonia oceanica* meadows, Harmony® mooring, was developed by a French company (SMAT Neptune Environnement) and experimented in collaboration with the Port-Cros National Park (Fig. 81). After conclusive trials, the deadweight moorings and chains were removed from Port-Cros Bay and replaced by a set of Harmony-type moorings, that would not harm the meadow.

The Harmony® system has the following characteristics (Smat Neptune Environnement, 2000):

- simple, resistant and reliable anchoring
- negligible environmental impact
- no contact of the mooring line with the seabed
- anchoring point is almost level with the soil and is not an obstacle for fishing gear
- adapts to all kinds of seabed, including Posidonia oceanica meadows
- easy to install and remove

The Harmony system consists of a steel helical spring that screws into the "matte" without harming the surrounding meadow, particularly the rhizomes (Fig. 71). Outside the "matte" only the tip of the helicoid, bearing the fastening ring for mooring, less than 10 cm in diameter (Fig. 81), protrudes. The mooring cable that rises from the fastening ring is kept at a certain distance above the seabed by a small intermediate buoy, itself linked to the mooring buoy on the sea surface (Fig. 82).

The length of the mooring cable is determined so that a traction angle of 45° is obtained, allowing the boat to move on the surface in a radius equal to the depth, whereas in traditional mooring using deep buoys the mooring cables are three times as long as the depth. This method of mooring thus not only permits the *Posidonia oceanica* meadow to be kept from harm but also enables more boats to be accepted for an equal surface area.

The helicoids are designed to bear all the traction forces produced by leisure boats, even large ones. Calculations done by the company which produces Harmony® mooring take winds of 120 km/hour into account. Resistance tests have shown that a mooring of this kind in a *Posidonia oceanica* meadow



Fig. 81. The fastening ring of a Harmonytype mooring, with the first part of the mooring cable, in a *Posidonia oceanica* meadow. From Smat Neptune Environnement (2000).



Fig. 82. A Harmony-type mooring system, in a *Posidonia oceanica* meadow, 10 metres depth. The boats have 10 metres surface radius room to swing. From Smat Neptune Environnement (2000) redrawn.

can bear a force of 21kN, i.e. 2.14t, equal to that borne by a concrete deadweight mooring weighing 4 t and with 1.9 metre sides. To give an example, a 16 metre long 4.5 metre-wide yacht, with a headwind of 120 km/hour, generates a horizontal force of 14 kN, or 1.43 t.

The Harmony® system, and those of the same type developed by other enterprises, do not require a large staff or particularly sophisticated equipment to be installed or removed: 2 underwater divers and hydraulic equipment are sufficient.

In Port-Cros Bay, as well as organised moorings, anchoring is permitted; the density of boats at anchor can reach 7.5/hectare at the height of the touristic season (Ganteaume *et al.*, 2005). Outside Port-Cros Bay, anchoring in the water of the Port-Cros National Park is permitted, except on the northern coast of Port-Cros Island and in a sector of the south-eastern coast of Bagaud Island (Fig. 84). A comparison between this sector, where anchoring has been prohibited since 1993, and a neighbouring sector where anchoring is permitted, though moderate (density of boats: 2.5/hectare at most) does not reveal significant differences in the state of the *Posidonia oceanica* meadow (shoot density, cover, % of plagiotropic shoots) (Ganteaume

et al., 2005). This confirms that the meadow can tolerate the anchoring of little leisure boats when this is very moderate.

8.2.3. Areas outside the RAMOGE Agreement

Milazzo *et al.* (2002, 2004) have studied the impact of leisure boating on the *Posidonia oceanica* meadow in the island of Ustica (north of Sicily, Italy). They did an experimental comparison of the effect of mooring with various kinds of anchor. A preliminary survey in Ustica port showed that the anchors most used by small boats (<5.5 metres) are anchors of the Hall, Danforth and Ombrello (=Folding Grapnel) kind (Fig. 83; also see Fig. 41), weighing about 4 kg. The experiments thus dealt with these 3 models.



The anchors were dropped from a 4.5-metre-long rubber dinghy and raised after staying 10 minutes at the bottom on a meadow between at 8 and 12 metres depth (shoot density: 500-530 shoots/m²). The impact on the seabed was observed by divers who counted the broken shoots after the anchoring phase and that of raising the anchor (Milazzo *et al.*, 2002, 2004).

The average number of shoots broken during the anchoring phase (dropping the anchor) was between 0.47 (Danforth anchor) and 0.93 (Hall anchor). During the mooring phase (stay) the only anchor that caused harm (and this was modest) was the Ombrello (0.07 broken shoots). During the anchor raising phase, the values were between 0.67 (Hall) and 4.87 (Ombrello) (⁷⁵). The presence or absence of a chain did not significantly change the impact. The impact for the 3 phases of mooring was compared for each kind of anchor (Milazzo *et al.*, 2002, 2004):

Hall:	anchoring =raising >stay
Danforth:	raising >anchoring >stay
Ombrello:	raising >anchoring >stay

Similarly, the impact of the 3 kinds of anchor was compared for each phase of the mooring:

Anchoring:	Hall >Danforth =Ombrello
Raising:	Ombrello >Danforth >Hall
Stay:	Ombrello >Danforth =Hall ⁷⁶

In all, Danforth and Ombrello anchors cause greater damage than Hall anchors. Milazzo *et al.* (2002) therefore suggest that, as a possible tool for protecting *Posidonia oceanica*, only boats with low-impact anchors be permitted to moor over a *P. oceanica* meadow.

These reports show that the anchor model can be improved to mitigate the harm done to the meadow. Among the initiatives taken to this end, we can mention the development of an anchor with a pivot system that functions during raise and thus limits the tearing of the meadow (Fig. 84).

8.3. RECOMMENDATIONS

Tourism and leisure boating have a major social and economic role in the Mediterranean. Even if anchoring harms the *Posidonia oceanica* meadow, the impact is certainly not significant for most of the meadows: the density and frequency of leisure boat moorings there is outdone by their regeneration capacity (ramification of rhizomes replacing broken rhizomes; Molenaar, 2000). There is thus no question of forbidding mooring there, which would anyway be unrealistic. However, practices that mitigate the impact should be recommended to leisure boaters: (i) avoid, whenever the choice is possible, anchoring over a *P. oceanica meadow*; (ii) do not raise the anchor by hauling on it but first position the boat vertically above the anchor and then raise it.

⁷⁵ In Elbu Cove (Corsica), Boudouresque et al. (1995) measured a much higher total number of shoots torn out per mooring cycle (anchoring + stay + raise of anchor) 17 on average.

⁷⁶ The authors believe that for the mooring stay the values were negligible and the differences not significant. But we do note that the duration of the stay considered was only 10 minutes, which is tiny compared to the real duration of a stay over a mooring – usually 10 to 100 times longer.



The problem arises at the level of certain areas, often of limited extent, that are particularly appreciated by leisure boaters because of their beauty (landscape) and the shelter they provide. When the density and frequency of anchoring is considerable, during much of the year, management measures become necessary. When the amount of leisure boat mooring is more on average than 2 anchorings/hectare/day (annual average) or more than 10 boats/hectare (at peak period) we recommend that organised mooring be established. However, organised mooring should never be done over deadweight moorings (whose negative impact is much greater than that of anchors) but should call for a non-destructive system⁷⁷.

Unauthorized mooring (installation of deadweight moorings without permits) has as negative an impact as mooring organised over deadweight moorings. It is shocking that this illegal practice should sometimes be tolerated by the authorities concerned.

In Marine Protected Areas, as in all areas where the *Posidonia oceanica* meadow presents good vitality and great heritage value (e.g. in Natura 2000 sites and ZNIEFFs⁷⁸), we recommend regulating mooring. This can be achieved through a fallow period system (alternately banning and permitting over 5-year periods) in some sectors, through a permanent ban in other sectors, and lastly through organised mooring based on a non-destructive system, as is the case in Port-Cros (Var, France; Fig. 84) (Boudouresque *et al.*, 2004). Such regulation of mooring should be done as part of a management plan on a consistent scale, for example that of a bay or a coastal massif.

77 Several kinds of non-destructive mooring already exist in France (e.g. Harmony®, Ancrest®). Others will probably exist in the future.

78 ZNIEFF =natural areas of ecological, floristic and faunistic Interest In sectors where there are *Posidonia oceanica* **barrier reefs**, mooring should be prohibited over the reef itself and in the lagoon lying behind the reef, at the same time as the passage of boats. For this ban to be respected, it is vital that in front of the reef



and the lagoon a line of buoys is placed, if possible joined to each other by a rope, as is the case in Port-Cros Bay (Port-Cros National Park, Var, France).

Finally, regarding the **mooring of big ships**, this should be restricted to areas with no *Posidonia oceanica* (loose sediment or "dead matte") and that are large (at least 100-200 metres in diameter). When a mooring buoy is provided on the surface, linked to a system of chains (3 or 4 branches each terminating in an anchor) all linked centrally by a metal plaque ("forking"), in addition to the above we recommend: (i) reducing to the minimum the impact of the pendant and the "forking"; for this, calculate the minimum length of the pendant that is needed for the buoy's elasticity; install a booster buoy at intermediate depth to prevent the pendant and "forking" from eroding the "matte"; (ii) reducing maintenance work (taking up the chains and anchors) and replacing these structures as exactly as possible where they were (Roy *et al.*, 1999).



9. THE POSIDONIA OCEANICA MEADOW AND THE MARKING OF THE 300 m STRIP



Fig. 85. Outline of principle of the traditional marker device for the 300 metre strip and the alternative device used by the Blue Coast Marine Park (Bouchesdu-Rhône, France).



Fig. 86. Example of the degradation of the *Posidonia oceanica* meadow caused by avoidance of the chain of a deadweight mooring used to mark the 300 metre strip. Photo by F. Bachet.

9.1. THE PROBLEM

Increasingly intensive use of the coast for tourismrelated activities has in recent years brought about a sharp increase in the number of marker buoys delimiting summer bathing areas. These yellow buoys, placed in coves or along beaches, have a marked visual impact on the landscape and also a severe impact on the seabed, particularly on the *Posidonia oceanica* meadows.

In France, it is the local authority ("commune") that is responsible for placing buoys to delimit the bathing areas. Since the Order of 27 March 1991, the law makes the mayor responsible for managing certain nautical activities (bathing, the use of nonregistered and non-motorized craft) within the 300 metre strip. The local authority is responsible for acting and informing the public about this. Installing marker buoys often represents a heavy cost for the local authority, and the systems adopted vary widely from one local authority to another.

To mark the 300 metre strip, the system generally used involves linking a yellow hemispherical buoy

to a deadweight mooring block (Fig. 85 and 86) via a chain whose movement is cushioned on the seabed around the deadweight mooring block. It is the motion of this chain around the deadweight mooring block that causes most of the damage (Fig. 86) (also see §8.2.2).

9.2. CASE STUDY: THE CÔTE BLEUE MARINE PARK

As a marine environmental management organisation, the Côte Bleue Marine Park has for several years taken an interest in the impact marking the 300 metre strip has on the seabed, in particular on the *Posidonia*

oceanica meadow. Inspections carried out at the start and end of the summer season have shown that the motion of the chain around the deadweight mooring causes the degradation and **destruction of 5 to 10 m² of meadow** every season (for each deadweight mooring).

Once the deadweight moorings have been withdrawn, at the end of the season, all that remains is a facies of very damaged "matte" covered with sediment on the most affected zones and very bared *Posidonia oceanica* rhizomes on the limit. The problem is worsened by the fact that the

following year the deadweight moorings **are not put in exactly the same places**, and thus a new hole is dug in the meadow. Once these holes have been opened up, various erosion phenomena can provoke or accentuate the splitting up of the meadow, particularly erosion by currents.

As a result of these first observations, the Marine Park wished to take on responsibility for installing the 300 metre buoys in the Carry-le-Rouet Reserve (a no-take area within the Côte Bleue Marine Park). The degradation noticed was indeed incompatible with the objectives of managing an integrally protected marine area. The Park decided to use an alternative mooring line consisting of a rope, a chain and a **submerged subsurface buoy** that would prevent the seabed from being degraded by dredging and the motion of the chain (Fig. 85 and 87). This system had already been successfully applied to the permanent buoys in the Carry-le-Rouet Reserve. The Park thus tested for 2 years the installing and removal of these buoys in the Reserve.

In early 2003, the Park decided to widen its approach to the scale of the Côte Bleue. Legal research showed that there is no obligation to mark the 300 metre strip when one is in a zone coming under a single set of general regulations as regards navigation. But for some sectors where there is a great deal of nautical and bathing activity, the local authorities find it hard to avoid marking this zone, given that the mayor is responsible for bathing and safety.



Fig. 87. The system of mooring of buoys marking the 300 metre strip used by the Côte Bleue Marine Park (Bouches-du-Rhône, France) the subsurface buoy which prevents erosion of the *Posidonia aceanica* meadow due to the motion of the chain. Photo by F. Bachet – Côte Bleue Marine Park.

In this context, a technical file concerning the use of the alternative mooring system, tested on the Carry-le-Rouet Reserve, was proposed to the local

authorities (*communes*) of the Côte Bleue as a whole. All the *communes* proved very receptive to this problem, and ready to act. In summer 2003, the system advocated by the Marine Park was installed in front of each *commune*.

9.3. RECOMMENDATIONS

When the 300 metre zone marker buoys are positioned over a *Posidonia oceanica* meadow, it is recommended to use at least the alternative system, with subsurface buoy, perfected by the Côte Bleue Marine Park. This system is easy to implement, does not require specialist staff, and its cost is very close (+ 20%) to that of the traditional device (without subsurface buoy).

Other even more efficient systems can be envisaged, but the additional cost is greater. It is, for example, possible to use a sliding buoy on slack adjuster system, as in fish farming, and also fixed deadweight moorings (not removed at the end of each summer season), or to replace the deadweight moorings with screw anchorage (non-destructive Harmony® type anchorage, or similar systems). These systems, which require professional divers to install them, could be implemented in the Marine Protected Areas (MPAs) and in other sites of great heritage value.

In any case, in the medium term it is vital to **stop removing the deadweight moorings** at the end of the summer season; the present precision of the GPS systems, and anyway the incomparable precision of positioning by seamarks, enable deadweight mooring to be found again when the buoy has been removed.

10. THE POSIDONIA OCEANICA MEADOW AND TRAWLING

10.1. THE PROBLEM

Fishing with towed gear is very harmful to the seabed, especially for *Posidonia oceanica* meadows (see §4.6). In France, in theory, benthic and pelagic trawling is prohibited within the ca. 5 600 m (3 nautical miles) coastal strip and at less than 100 metres depth. In Spain and Italy, it is prohibited at less than 50 metres depth. In practice, however, it is widely practiced over shallow depths, which has negative consequences for the meadow (regression) and its ecological and economic functions.

Indeed, trawling causes (also see §4.6): (i) a disbalance in halieutic resources, for this nonselective fishing catches immature juveniles and harms the stocks of many non-target species; (ii) the degradation of habitats, spawning grounds and nurseries, particularly of exploited species. Trawling is thus the main cause of the regression of the *Posidonia oceanica* meadows in Spain, in the Alicante region (Guillén *et al.*, 1994; Bombace, 1995; Ramos-Esplá *et al.*, 2000). Moreover, the degradation of the topography of the seabed and the associated habitats leads to a reduction of spatial heterogeneity, an essential element of biodiversity (Kaiser, 1998).

Furthermore, there is a **user conflict** between trawlers, when they approach the coast and destroy spawning grounds and nurseries, and artisanal fishermen, who depend on the sustainable use of these spawning grounds, these nurseries and thus the resource.

Insofar as the authorities of the countries that border the Mediterranean have not the means, or the will, to make sure the law is respected, which goes against the interests of the artisanal fishermen, but also of the trawlers (although they are not always aware of this), the most realistic solution consists of placing physical obstacles in the way of trawling: **anti-trawl reefs** (Tocci, 1996). These reefs protect the *Posidonia oceanica* meadow (and its ecological and economic functions) and thus permit sustainable management of the halieutic stock through artisanal fishing (Ramos Esplá *et al.*, 2000).

10.2. HISTORY OF ANTI-TRAWL REEFS

Anti-trawl reefs in the Mediterranean were first sunk in the 1970s in the Languedoc-Roussillon region (France), when 3-metre-high concrete stakes were put down in the sediment in Palavas and in Gruissan (Collart and Charbonnel, 1998). This attempt ran into trouble, however.

In 1986, the Côte Bleue Marine Park (near Marseille, France) experimented with the submersion of 83 big slabs of rock (10 to 12 t) arranged in a line and spaced every 40 to 60 metres, particularly intended to **protect the** *Posidonia oceanica* **meadows**. The Côte Bleue meadow is in fact the biggest one in the Bouches-du-Rhône (over 1 000 hectares; Cristiani, 1980; Bonhomme *et al.*, 2003a), and is the last continuous meadow between Provence and the Spanish border. Since then, Côte Bleue Marine Park has diversified its anti-trawl reefs: 5 different types of anti-trawl reef are found there – slabs of rock, sea-rocks, "Fakir," Negri and tripods (Fig. 88 and 89). In all, along the 25 km of Côte Bleue coast, 326 anti-trawl obstacles (i.e. 2 200 m³) have been sunk by



the Marine Park, forming 17.5 km of anti-trawl lines, intended to protect the coastal strip from the trawlers' illegal incursions (Bachet, 1992; Daniel and Bachet, 2003; Charbonnel *et al.*, 2001b; Fig. 88).

In Languedoc-Roussillon (France), many sinking operations have taken place since 1992, with one 8.15 t model of concrete double pipe for 7.1 m³ (Collart and Charbonnel, 1998; Fig. 91): Marseillan (105 pipes in 1992 and 1996), Agde (200 pipes in 1996), Aigues-Mortes (109 pipes in 1999); other projects are under way, anticipated in 2004 and 2005 in Leucate, Valras, Agde and Argelès (Béatrice Pary,



Fig. 88. Map showing the location of anti-trawl protection reefs (purple lines) placed along the coast of the Côte Bleue Marine Park (Bouches-du-Rhône, France). The orange polygons indicate the Reserves of Carry-le-Rouet and Cap-Couronne. From Daniel and Bachet (2003).

personal comm.). Despite the almost complete absence of *Posidonia oceanica* meadows in this region (Boudouresque and Meinesz, 1982), antitrawl reefs enable user conflicts between fishing craft to be managed, by **sharing the space** and the halieutic resource between fixed net fishermen (small-scale fishery) and trawlers. It should be stressed that it is often the artisanal fishermen themselves who spark off the putting down of artificial reefs. The seabed protection role that anti-trawl reefs can have is today giving rise to growing demand from fishermen in the Mediterranean.



Fig. 89. "Fakir" type anti-trawl protection modules based on reconditioned EDF (Electricité de France) electricity poles. 91 modules of this kind were sunk in 1997 inside and outside the Reserve of Cap-Couronne in the Côte Bleue Marine Park. Photo by E. Charbonnel.

In Liguria (Italy), the first artificial anti-trawl reefs were sunk in the Gulf of Tigullio (Genoa province) in December 1980; these reefs were made of wrecks, concrete blocks and tubes, and blocks of stone for a total volume of 16 185 m³ (Relini *et al.*, 1986). After this first experiment, a second reef was sunk in Loano (Savona province) in 1986, under the scientific control of Genoa University and with funding from the European Union (Relini *et al.*, 1995; Relini, 2000). The facility, which also has a repopulating function, lies at between 18 and 23 metres depth, near the lower limit of a regressing *Posidonia oceanica* meadow. It has a central (200 m x 100 m) core of 30 pyramids each made up of 5 2-metre-sided cubes of cement (4 at the first level, + 1 at the second level) with an opening and a cavity (Fig. 92). Around this central core, a vast area (350 hectares), between 5 and 45 metres in depth, constitutes the protection area; 200 blocks of cement with 1.2 metre sides were sunk there (anti-trawl blocks). During the 16 months following their sinking, many anti-trawl blocks were pushed aside by trawls. To enhance protection, 150 new cement blocks, rather bigger (2 metre sides), were sunk.

In 1989, another experiment was carried out in Spotorno (Savona province, Liguria, Italy) under the scientific control of GIS Posidonie and Genoa University with funding from the European Union.









Fig. 92. Artificial reefs (with a repopulating and anti-trawl function) sunk in Loano (Liguria, Italy). Above left map of the central core. Above right cement cubes. Below left cross-section of a pyramid in the central core. Below right bird's eye view of a pyramid. From Relini *et al.* (1995).



As in the above example, an area of repopulating reefs was provided for as well as an area of anti-trawl reefs intended to protect a *Posidonia oceanica* meadow that was regressing because

of trawling. However, the project was never completed, and only a few big cement modules, of Bonna® type, were sunk (Fig. 93).

Between 1997 and 1998, under the scientific control of Genoa University, artificial reefs were sunk in Alassio (Savona province, Liguria, Italy) near the lower limit of a *Posidonia oceanica* meadow. The facility has on its extreme right and left areas 2 "repopulation oases" made up of 3 groups of 8 pyramids each formed of 5 cement cubes with 2 metre sides, at a depth of between 8 and 25 metres. Between the 2 oases, at 20-24 metres depth, many smaller cement cubes (1 metre sides), alternating with tetrapods, were sunk in order to protect the meadow from trawling (Fig. 94).

Anti-trawl reefs have also been sunk in El Campillo and Nueva Tabarca (Alicante province, Spain) and Catalonia (Spain) (Ramos, 1990; Alluè-Puyelo and Olivella-Prats, 1994; Guillén *et al.*, 1994; Ramos-



Fig. 93. Diagram of a Bonna® type module sunk in Spotorno (Liguria, Italy) as part of an unfinished project for coupling repopulating reefs and anti-trawl reefs.



Fig. 94. Diagram of artificial repopulating reefs (pyramids of cubes) and anti-trawl reefs (cubes and tetrapods) sunk in Alassio (Liguria, Italy).

Esplá *et al.*, 1994). These are often (El Campillo in particular) cement cubes with 1-1.5 metre sides, weighing 7-9 t, criss-crossed by bits of railway track (Ramos-Esplá *et al.*, 1994).

10.3. RECOMMENDATIONS

The key to the success of any anti-trawl protection reef is its **design** – both the architecture of its modules (material, shape, height, length, volume) and how they are arranged on the seabed. This design plays a preponderant part in whether the facility is effective and lasting. Protection reefs must constitute an obstacle that is sufficiently dissuasive to stop trawlers entering the area – they would risk damaging their equipment.

(i) The modules of anti-trawl reefs must be sufficiently heavy (at least 8 t) to constitute an effective physical obstacle and not be pulled away by the trawl or harmed by the trawls' side panels. (ii) The modules must offer a sufficient load-bearing surface against the sediment not to sink right into it. Sea-rocks (Fig. 90) are particularly effective on mud bottoms and also are not well detected by trawlers' sounders (Francour *et al.*, 1991). (iii) It is also important that the shape of the anti-trawl module should not risk damaging the nets of artisanal fishermen (small-scale fishery) who must be able to work in the protected areas and benefit from anti-trawl reefs. (iv) The anti-trawl modules must be sunk one

by one to leave spaces between them (50 to 200 metres). (v) The ensembles of anti-trawl reefs must occupy the maximum space possible to really be dissuasive to trawlers.

If the topography and size of the site to be protected allow this, the modules must be arranged in lines running straight out from the coast (most of the trawls are in fact dragged parallel to the coast), thus forming a series of barriers as in the Côte Bleue (Fig. 88). They can also be arranged at regular intervals on the seabed. In Spain, the modules are arranged in a 300-500 metre sided square, or a 400 m x 200 m rectangle, with a module on each side and one in the middle. The area thus protected varies between 1.6 and 5 hectares per module (Ramos-Esplá *et al.*, 2000), according to the distance left between each module (50 to 100 metres).

In addition to their role of physical obstacle on the seabed, certain anti-trawl reef modules are designed with crevices in order to offer a shelter zone for fishes, and thus have a **production** role.

Fig. 95. 2 *Pinna nobilis* noble pen shells (molluscs) in a *Posidonia* oceanica meadow. This threatened species, protected by law (France) and by the Barcelona Convention (all the countries bordering on the Mediterranean), is very vulnerable to trawling, which breaks its shell. Photo by E. Charbonnel.



Artificial anti-trawl reefs are one of the most effective tools for **integrated management** of the coastal resources, after the creation of Marine Protected Areas. This management can concern both uses (sharing the space and the halieutic resource between artisanal fishermen and trawlers) and the ecological aspect (protecting vulnerable or protected species and habitats, restoring degraded environments, diversifying naturally poor substrata).

All in all, artificial protection reefs are a very effective tool against the illegal practice of trawling, for managing user conflicts (sharing space) between artisanal fishermen and trawlers, for the sustainable protection of the resource, for protecting *Posidonia oceanica* meadows and for protecting those species that are dependent on them, such as the noble pen shell *Pinna nobilis* (Fig. 95).

As part of the overall improvement of the coastal zone, the European Union can fund the installing of anti-trawl reefs through the intervention of structural funds such as the FIFG⁷⁹ (Financial Instrument for Fisheries Guidance ; in French IFOP) to the tune of 50% of the investment. The gain in terms of protecting the resource and priority natural habitats is thus recognised, in comparison with the lack of performance of checking systems, respect for rules and regulations, and maritime surveillance.

79 IFOP http://www.info-europe.fr/document.dir/fich.dir/QR001094.htm.

11. THE POSIDONIA OCEANICA MEADOW MEADOW AND FISH FARMS

11.1. THE PROBLEM

The term aquaculture covers all the activities that aim to produce and market aquatic species, whether these are (i) plants or animals, (ii) freshwater, brackish or salt water, (iii) during part or all of the reproduction cycle (Barnabé, 1989). Aquaculture is a very old tradition, already being carried on 4 000 years ago by the Chinese, that involves using natural or artificial aquatic environments to achieve the production of species useful to man.

In 1990, Mediterranean fish farming was producing 130 000 t of fish, molluscs[®] and crustaceans, i.e. 1.6% of world marine aquaculture production (De la Pomélie, 1991). Since then, it has been essentially the marine farming of fish (pisciculture) that has developed strongly, both in the north-western and in the eastern Mediterranean, thanks to the post 1975 mastery of the reproduction of the European sea bass (*Dicentrarchus labrax*) and the gilthead sea bream (*Sparus aurata*) (Dosdat *et al.*, 1994). This production grew by nearly 29% per year in the 1990s (12 500 t) to reach a value of 124 000 t in 1999, i.e. ten times greater than in 1990 (Belias and Dassenakis, 2002). Since the late 1980s, there has been mass production of sea bass and gilthead sea bream in Greece, Spain, Italy, Turkey and France^{\$1}, and today this constitutes the bulk of aquaculture production of fish (92%; Belias and Dassenakis, 2002). In 1998, this fish farming production represented over 654 million dollars, and aquaculture development attracts a growing number of Mediterranean countries (e.g. the appearance of 3 new producer countries in 1999; Belias and Dassenakis, 2002).

But the development of fish farming seems likely in some cases to threaten the quality of the coastal environment (Videau and Merceron, 1992) and thus some of its uses (see Chapter 1; Teinturier, 1993; RAC/PAP, 1996; UNEP, 1999). Because of (i) their specific geographical situation, often sheltered bays where there is little water movement, (ii) the size of the waste (uneaten foodstuff, excretions), (iii) the frequent recourse to "health" substances (antibiotics, trace elements⁸²), aquaculture facilities can actually have a negative impact on the natural environment (Handy and Poxton, 1993; Hevia et al., 1996; Karakassis, 1998; Miner and Kempf, 1999; Boyra et al., 2004; but see Aubert, 1993 and Machias et al., 2004 for a more nuanced, or contrary, opinion), particularly the Posidonia oceanica meadow. Whereas the impact of intensive marine pisciculture (cages) has been the subject of many studies in northern Europe (e.g. the developing of salmon farming; Gowen and Bradbury, 1987; Videau and Merceron, 1992; Munday et al., 1994; Wu, 1995; Merceron and Kempf, 1995) there is fairly little data on its impact in the Mediterranean, where this activity is more recent (e.g. Aubert, 1993; Verneau et al., 1995; Mendez et al., 1997; Delgado et al., 1999; Karakassis et al., 1999; Pergent et al., 1999; Cancemi et al., 2000; Dimech et al., 2000a, 2000b; Mazzola et al., 2000; Karakassis et al., 2000; Ruiz-Fernández, 2000; Ruiz et al., 2001; Karakassis et al., 2002; Cancemi et al., 2003; Machias et al., 2004, 2005).

⁸⁰ In 1990, *Mytilus galloprovincialis* mussels were the main species (in tonnage, about 90%) produced by Mediterranean aquaculture.

81 In the Provence-Alpes-Côte d'Azur region (PACA) (France), the production of sea bass and gilthead sea bream was in 1999 estimated to be about 1 200 t per year (Source Guide de la production aquacole française, 2001 edition), i.e. about 1% of the Mediterranean basin's production for these species (FAO, 2001). In 2003, the production dropped to 990 t (60 t in the Bouches-du-Rhône, 150 t in the Var and 780 t in the Alpes-Maritimes) (Source French Département des Affaires Maritimes).

⁸² The Nutreco® feed presents, for example, a copper content of about 4 mg/kg and the Gouessant® feed a content of about 3 mg/kg (E. Roque and D. Coves, personal comm.).

11.2. CASE STUDIES

Some studies have been done, starting from the 1990s. They concern fish farms that carry on intensive pisciculture, in cages, of the sea bass *Dicentrarchus labrax* (Corsica - Verneau *et al.*, 1995; Pergent *et al.*, 1999; Cancemi *et al.*, 2000, 2003; Sardinia – Pergent *et al.*, 1999), of the gilthead sea bream *Sparus aurata* (Balearic Islands – Delgado *et al.*, 1997, 1999; Malta – Dimech *et al.*, 2000a, 2000b), of the amberjack *Seriola dumerilii* (Spain – Ruiz-Fernández, 2000) or of several of these species (Ruiz-Fernández, 2000; Ruiz *et al.*, 2001). The production of these fish farms varies from some tonnes (Gulf of Sant'Amanza, Corsica, 15 t/year; Figari Bay, Corsica, 18 t/year) to several hundred tonnes (Gulf of Aranci, Sardinia, 200 t/year; Bay of El Hornillo, Spain, 700-800 t/year). The disturbance caused by these farms is measured through abiotic⁸³ (light, sediment, interstitial water) and biotic (density of the *Posidonia oceanica* meadow, leaf biometry, lepidochronology (see §2.2), primary production, leaf epibiota, reserve carbohydrates in the rhizomes) parameters, according to the growing distance from the cages and in geographically close reference sites.

The increased turbidity registered near the cages gives rise to a significant reduction in luminous intensity. This reduction is estimated to be over 30%, on average, under the cages in Figari Bay (Corsica; -10 m; Pergent *et al.*, 1999) and 23% at 40 metres from the cages at El Hornillo (Spain; -8 m; Ruiz *et al.*, 2001). When the cages are placed in shallow areas the lighting at the seabed remains far superior to what it is at the lower limit of the *Posidonia oceanica* meadow. But this factor must be taken into consideration for fish farming facilities located over deeper meadows (Verneau *et al.*, 1995). Furthermore, the shade shed by the cages (independently of turbidity) significantly reduces the density of the *P. oceanica* shoots (Ruiz-Fernández, 2000; Ruiz and



Romero, 2001; see §4.8).

83 Abiotic parameters non-biological (=non biotic), i.e. physicochemical.

Similarly, an increase in the organic matter and silt content can be seen the nearer one gets to the cages (Delgado *et al.*, 1999; Pergent *et al.*, 1999; Dimech *et al.*, 2000a; Karakassis *et al.*, 2000; Ruiz *et al.*, 2001). This enrichment in organic matter can best be seen in the deepest sediment layer (Fig. 96; 10-15 cm layer). In the most serious cases, the sediment presents a black, anoxic⁸⁴ superficial layer, with escape of methane CH_4 and hydrogen sulphide H_2S (Karakassis *et al.*, 2002). Near the cages, the fauna is dominated by the polychaeta *Capitella* (cf *capitata*), a species that indicates extremely high pollution (Bellan *et al.*, 1975, 1999; Karakassis *et al.*, 2000).

Other parameters, such as nutrient salts concentration (Table IX), and chlorophyll a and pheopigments⁸⁵ content, are also greatly influenced by the presence of fish farm cages (Aubert, 1993; Pergent *et al.*, 1999).

However, the main changes are observed at the level of the Posidonia oceanica meadow. Thus the density (number of shoots per m²) shows a significant drop when nearing the cages, with very often the disappearance of the meadow under the cages (Table X). Even at a distance of 300 m, it was seen that the average density, measured in the Figari (Corsica) and St. Paul (Malta) sites, is lower than the density values considered "normal" for that depth (Pergent-Martini et al., 1999). In El Hornillo Bay (Spain), the impact on P. oceanica is also visible up to several hundreds of metres from the cages; 11 hectares of meadow have been destroyed and 10 hectares significantly degraded, representing in all 53% of the meadow's original surface area in the bay; the surface area of destroyed or degraded meadow is 7 times as much as the area occupied by the cages (Ruiz-Fernández, 2000; Ruiz et al., 2001; see Fig. 47). Unlike what was observed for other benthic habitats, for which some authors note an impact that goes no further than 25-30 metres from the cages (Karakassis et al., 2000, 2002; see references in Machias et al., 2004), the impact on the P. oceanica meadow is thus perceptible over a great distance. As for demersal fish, in the oligotrophic water of the Genoa coast, the impact (increase in abundance and in biomass) can be seen over a still greater distance - several kilometres (Machias et al., 2005).

Table IX. Nutrient content in the interstitial water of the sediment (in µM; mean values and 95% confidence interval in brackets) in 3 stations in Figari Bay (Corsica) according to the distance from the fish farm, and in the Moines Islands reference site; the Moines Islands lie 5 km out to sea, about 15 km from the fish farm. From Cancemi *et al.* (2003).

	0m	20m	100m	Moines Islands
Total phosphorus (mg/kg)	2,206 (± 429)	786 (± 229)	568 (± 33)	-
NO3 ⁻ (nitrates)	2.5 (± 0.7)	3.7 (± 0.8)	2.3 (± 0.9)	3.1 (± 0.8)
NH4 ⁺ (ammonium)	19.5 (± 8.7)	12.4 (± 2.3)	8.4 (± 1.6)	1.8 (± 1.1)
PO4 ³⁻ (phosphates)	5.2 (± 0.6)	1.8 (± 0.6)	1.3 (± 0.6)	1.7 (± 0.6)

Table X. Density of the *Posidonia oceanica* meadow (number of shoots per m). Mean values ± confidence interval (95%). *Pergent et al., 1999; **Dimech et al., 2000b.

Sites	Distance from cages				
	Under cage	1m	20m	80m	300m
Figari (-10m) – Corsica*	0	63 ± 11	108 ± 16	250 ± 28	313 ± 41
St Paul (-12m) – Malta **	0	-	-	225 ± 20	310 ± 30
Golfo Aranci (-23m) – Sardinia*	0	110 ± 16	-	175 ± 29	200 ± 32

⁸⁴ The black (anoxic) sediment lying under the cages is termed by Holmer (1991; *in* Karakassis *et al.*, 2002) *"farm sediment"*.
 ⁸⁵ Pheopigments are the products of the degradation of certain chlorophylls.



Also in the Balearic Islands, temporal monitoring of *Posidonia oceanica* and *Cymodocea nodosa* meadows shows a significant regression of these formations after fish farm facilities were set up (Delgado *et al.*, 1997). This regression is expressed in a reduced density of these meadows, and even their disappearance (Fig. 97). Worse still, Delgado *et al.* (1999) show that 3 years after the farms stopped operating, the regressive dynamics of the meadows is continuing.



Although the average number of leaves per *Posidonia* oceanica shoot seems not be influenced by the presence of a fish farming facility⁸⁶, the average length of adult and intermediate leaves does significantly increase in spring the nearer they are to the fish farm (Fig. 98); the same holds good for the leaves' surface area (Leaf Area Index) and the foliar biomass (expressed per m²; Pergent *et al.*, 1999). However, in the summer the average length of these leaves decreases near the cages (Delgado *et al.*, 1999; Dimech *et al.*, 2000b); this phenomenon could express a higher **grazing pressure** at a time of year when foliar growth is reduced, and during which the development of MPOs (Multicellular Photosynthetic Organisms) leaf

epibiota can induce a phenomenon of competition with *P. oceanica*. This hypothesis is corroborated (i) by a rise in coefficient A (definition: Table XII), which expresses the impact of hydrodynamism and herbivores near the cages (Pergent *et al.*, 1999) and (ii) by the values obtained during the yearly cycle (Cancemi *et al.*, 2003). In El Hornillo Bay (Spain), Ruiz-Fernández (2000) and Ruiz *et al.* (2001) attribute most of the direct and indirect regression of the *P. oceanica* meadow to the

⁸⁶ In El Hornillo (Spain), however, the average number of leaves per shoot drops when nearing the cages (Ruiz et al., 2001). increase in herbivore grazing; the original cause of this is probably that the leaves are richer in nitrogen the nearer they are to the cages, and that the herbivores choose their grazing sites according to the richness in nitrogen of the plants they graze (Ruiz-Fernández, 2000); this overgrazing reduces *P. oceanica*'s photosynthetic potential and as a result the storing of reserve carbohydrates in the

rhizomes; now, the plant's yearly growth cycle, whose carbon balance shows a deficit most of the year, depends on these reserves (Alcoverro *et al.*, 2001).

The biomass of the *Posidonia oceanica* **leaf epibiota** increases greatly near the fish farming facilities (Fig. 99; but see Ruiz *et al.*, 2001); it varies on average during the year between 93.5±45.8 mg/shoot (Figari, Corsica, 20m), 51.5±55.3 mg/shoot (Figari, 100 m), and 37.8±35.0 mg/shoot (the Moines Islands reference site; Cancemi *et al.*, 2003). However, the maximal values are not found at the station that is nearest to the fish farm cages (where the nutrient content is, however, the highest) but at a distance of 20-40 m (Pergent *et al.*, 1999; Dimech *et al.*, 2000a). Additions of copper, added in the fish feed (estimated to be between 450 and 500 g/year for the Figari farm; Pergent *et al.*, 1999) could act as an algicide in the immediate proximity of the cages and explain this result.

The number of leaves produced every year, as well as the speed of growth of the rhizomes, do not seem to be affected either by the operations of the fish farms or by the distance from the cages. However, near the fish farm facilities, *Posidonia oceanica* presents a particularly well developed root system that seems to be the result of an adaptation to silting (Pergent *et al.*, 1999).



Fig. 99. Appearance of the *Posidonia oceanica* meadow, 80 m (above) and 300 m (below) from the fish farming cages in Figari (Corsica) (April 1994). Anonymous photo.

Primary production, measured during a yearly cycle by lepidochronology (see §2.2), shows very high values for the reference station (the Moines Islands) compared to those recorded near the fish farm cages in Figari (Corsica; Table X) (Cancemi *et al.*, 2003).

Table XI. Primary production of *Posidonia oceanica* in different stations in Figari Bay (Corsica) according to the distance from the cages of a fish farm. DM = dry mass. The Moines Islands lie 5 km out to sea, about 15 km from the fish farm. From Cancemi *et al.* (2003).

	20m	100m	Moines Islands
Foliar production (gDM/m²)	82.9	123.0	1,022.5
Rhizome production ((DM)g. m ⁻²)	5.0	6.2	48.1

Higher **copper** (Cu) and **zinc** (Zn) contents are recorded in the *Posidonia oceanica* rhizomes the nearer one is to the cages, in Figari (Corsica; Fig. 100). Additionally, the Zn content particularly seems to increase, starting from when the fish farm started operating (Fig. 101). Similar phenomena have been revealed in sediment impacted by the faeces^{\$7} of salmonids (Uotila, 1991; Merceron and Kempf,

1995). It seems that the origin of this enrichment is linked to the supplementing of feed by these 2 trace elements. The content measured is respectively 9.9 μ g/g of Cu and 118 μ g/g of Zn in the feed used[®] (Mendez *et al.*, 1997). But one should

87 Faeces (or feces) solid excrement.

88 The manufacturer of the feed (Super.aquasard®) indicates a content of 5 µg/g of Cu, and fails to mention the presence of zinc. note that, according to present scientific data, a negative effect (with this content) on *P. oceanica* is very improbable (see §4.3).

The benthic **macrofauna** of the *Posidonia oceanica* "matte" shows a higher species diversity in a reference meadow compared to a meadow located near fish farm facilities. Preliminary measurements over 0.01 m³, done in October 2002 in Calvi Bay (Corsica), show that the number of species goes on average from 45.3±4.4 to 31.3±5.3. In Malta, the highest specific richness and abundance are recorded for a distance of 50-170 m from the cages (Dimech *et al.*, 2000b). Additionally, studying the distribution of several species of this macrofauna (echinoderms, decapods and molluscs) reveals zoning according to the distance from the cages (Dimech *et al.*, 2000b). This zoning is like that revealed in Scotland for a salmonid farm located in a fjord (Brown *et al.*, 1987):

- an azoic area (no macroscopic fauna) under the cages
- a highly enriched area
- a transitional area
- a "clean" area (similar to the reference zone).

This distribution is reminiscent of that observed near the sea outlet of a waste water discharge near Marseille (Bellan, 1985).



rhizomes according to the distance (in m) from the fish farm cages in Figari (Corsica). DM = dry mass. From Pergent *et al.* (1999).



The results set out above are of course relative to the sites studied, whose characteristics differ, and anyway these results cannot be transposed to all Mediterranean fish farming sites, nor general conclusions drawn, insofar as they depend on **multiple factors** that are hard to separate, such as hydrodynamism, height of the water column under the cages, density of farming, feed used, rate of feeding and how it is distributed throughout the day (Karakassis *et al.*, 2000). Furthermore, many of these parameters (e.g. density of farming, rate of feeding, feed used) are not clearly stated in studies on the impact of fish farms on benthic settlements, making it even more difficult to try to generalize. Lastly, it is rarely possible to completely separate, in the fish farming sites studied by the authors, the impact of fish farms from the other possible impacts (discharge of waste water from the coast, anchoring, etc.).

Regarding the impact of fish farms on *Posidonia oceanica* meadows, it is important to stress the fact that this impact is **irreversible** on the human scale (when the meadow has been destroyed).

Works that emphasize the relatively rapid reversibility of the impact (after the operation of a fish farm is stopped) do not concern the *P. oceanica* meadows (Mazzola *et al.*, 2000). On the contrary, in the case of the *P. oceanica* meadow, the impact can even go on becoming worse after the operation stops (Delgado *et al.*, 1999).

11.3. SUMMARY AND RECOMMENDATIONS

Studies on the impact fish farms have on *Posidonia oceanica* meadows are still recent and sparse. However, the results are fairly homogeneous and show **significant degradation** of these plant formations in all the sectors studied. Generally speaking, when the fish farm cages have been set above a *P. oceanica* meadow, the meadow has been greatly degraded or has disappeared, according to the age of the farm. Although many descriptors already seem to be pertinent, others still deserve to be refined (Table XII).

Compartment	Descriptor	Measurements taken	Impact
Water column	Turbidity	Light	ΟÌ
	Nutrients	NO ₃ (nitrates)	Ν
		NH4 (ammonium)	Ν
		PO ₄ (phosphates)	Ν
Sediment	Organic matter	Organic matter content	O 🖌
	Nutrients	NO ₃ (nitrates)	Ν
	(interstitial water)	NH ₄ (ammonium)	O 🖌
		PO ₄ (phosphates)	O 🖌
		Total phosphorus	O 🖌
	"Trace metals"	Zinc and copper	Ν
	Benthic "micro-algae"	Chlorophyll a	O 🖌
		Pheopigments	O 👠
Posidonia oceanica	Phenology	Density of meadow	ΟÌ
		Leaf length	O 👠 (¹)
		Epiphytic cover	O 🖌
		Coefficient A [∞]	O 🖌
		Leaf Area Index ⁹⁰	ΟÌ
		Foliar biomass (per m²)	ΟÌ
		Primary production (per m²)	ΟÌ
	Lepidochronology	Rhizome growth	Ν
		Number of leaves produced	Ν
		"Trace metals" (Zn, Cu) rhizomes	O 👠
	Associated macrofauna	Biodiversity	ΟÌ

Table XII. Main descriptors enabling characterization of the impact of fish farms on the coastal environment. N no significant difference; ov significant drop; ou significant increase. () high seasonal variability.

⁸⁹ Coefficient A percentage of broken (by hydrodynamism) or grazed (mainly fishes and sea urchins) leaves.

⁹⁰ Leaf Area Index=foliar surface area in m of leaves per m of seabed surface area.

In view of their effect on the environment, especially on *Posidonia oceanica* meadows (in many countries a protected species; see §5.1.2), setting up of new fish farms must take into account:

- the characteristics of the site where the farm is to be installed (physicochemical factors, e.g. currents, biological factors and user conflicts).
- the operating practices envisaged (species farmed, type of feed, mode of distribution, management of the daily ration⁹¹, waste control, health products, etc.).
- the production (tonnage) envisaged in relation to the site's characteristics (carrying capacity).
- the regulatory constraints (see Chapter 5), in particular the presence of the *P. oceanica* meadow.

Only by implementing an **overall policy** can these requirements be satisfied. From this point of view, the approach initiated by the *collectivité territoriale* of **Corsica** (in particular the *Agence de Développement Economique de la Corse*), the *Direction Régionale de l'Environnement de Corse,* the *Direction Régionale des Affaires Maritimes* de Corse, and people in the fish farming business, working in collaboration with scientists, is exemplary. From this perspective, a Methodological Guide for the development of files to request permission for "Installations Classées pour la Protection de l'Environnement" (ICPE)⁹² and an atlas for ecological awareness of the coastal environment regarding fish farming, were published in 2003. Among the criteria selected were: (i) the presence of strong currents (preferentially directed seawards), (ii) the absence of "sensitive" habitats (in particular *Posidonia oceanica* meadows), (iii) remoteness from other potential sources of disturbance (waste water discharge, coastal rivers) and (iv) significant depth underneath the fish farm facilities.

The **Liguria** Region (Italy), for example, has a document (adopted on 28 March 2001) that, as part of an environmental impact assessment (*Valutazione di Impatto Ambientale*, VIA), lays down a set of technical norms for fish farm projects. The criteria for positioning fish farm facilities in exposed sites (*Criteri di Posizionamento di Impianti di Maricoltura in Siti exposti*), accompanied by a fish farming map, constitute a useful tool for authors of projects in that they provide useful indications for the identification of favourable sites and for the impact study. The things that must be taken into account when positioning a fish farm are:

- **Sites of Community Importance** (SCI Habitats Directive). A safety distance is suggested according to the characteristics of the environment (currents, type of seabed, etc.) and those of the fish farm (number of cages, quantity of fish, etc.)
- Marine Protected Areas (existing or foreseen). A safety distance must be respected according to the characteristics of the environment and of the fish farm
- **Terrestrial Protected Areas**. The fish farms must not have a negative visual impact from the land (distance, visual angles, size, etc.) for landscape reasons
- Posidonia oceanica and Cymodocea nodosa. A safety distance must be respected from the meadows which these species form, according to the characteristics of the environment and of the fish farm
- **Bathymetry**. A depth of at least 30 m is required, which usually distances the fish farm from the most sensitive habitats and ensures better dilution of effluent from the farm
- Distance from the coast: At least 1 000 metres
- **Mouths of rivers.** Attention to these outflows is required for several reasons: entry of fresh water, interaction with currents, entry of pollutants, etc.

92 Guide méthodologique pour l'élaboration des dossiers de demande d'autorisation d'Installation Classée pour la Protection de l'Environnement (ICPE) en matière de pisciculture marine pour la Région Corse, 2003. Rapport Ifremer DEL/PAC/03-04 pour la collectivité territoriale Corse.

⁹¹ Rigorous management of the daily ration, which allows the amount of uneaten feed introduced into the surrounding environment to be minimized, is anyway interesting for fish farmers, for whom uneaten feed constitutes an unproductive load.


- waste water discharge. A sufficient distance must be kept to avoid contamination of fish in the fish farm
- Zone under regulations of the port authorities. (Anchoring merchant ships, etc.)
- Undersea lines. A safety distance must be kept from underwater water pipelines, telephone cables and electricity cables
- Archaeological sites (for example, wrecks). A safe distance must be kept.

The Liguria Region's impact assessment procedure also provides for setting up a **monitoring** programme to evaluate over time the environmental situation of the area in which the fish farm is established.

In France, the law regulates the permission to use the MPD (Maritime Public Domain) for fish farming via the "permission to exploit marine crops"93 procedure, prepared by the Affaires Maritimes as a public and administrative enquiry which requires that producers provide a file presenting the projected installation and how it integrates with the other uses of the area. If need be, permission is granted for a limited period, usually 5 or 10 years. Since 1993, marine fish farming facilities producing over 5 t/year have, in France, been subject to restrictive general regulations concerning all the kinds of installation that could present dangers or disadvantages, particularly for environment protection³⁴. Facilities with less than 5 t/year are the subject of a simple declaration, with deposit of a complete file; for those with 20 t/year or more (new installation or extension) a request for permission is required. This is prepared by the Veterinary Services and is the subject of an administrative and public inquiry among the potentially concerned stakeholders, with an investigating commissioner, on the basis of a file presented by the producer. This very complete file must contain a detailed impact assessment, whose contents are indicated by legislative decree⁹⁵. This study, usually carried out by a research department and paid for by the producer, has several parts, including an analysis of the original state of the site, an analysis of the direct and indirect effects on the environment, landscapes, surroundings etc., the reasons why the site was chosen, and the steps envisaged for eliminating, minimizing or compensating for the disadvantages of the installation. In the French Mediterranean, the most delicate item is usually the potential impact on the Posidonia oceanica meadow, a protected species, as the remarks made during public enquiries and litigation show. Permission, in the shape of a bylaw, is granted for a limited period and is accompanied by prescriptions which usually contain a demand for regular monitoring of the potential effects of the installation on the environment and, if need be, on the P. oceanica meadows located nearby, with a report made to the administration⁹⁸.

In the absence of a predictive⁹⁷ model that enables the impact on the *Posidonia oceanica* meadow of a fish farm project to be precisely anticipated, according to where it is (depth, distance from the coast, movement of water, etc.) and its characteristics (species produced, anticipated tonnage, anticipated load in the cages in kilos of fish/cubic metre, farming techniques, kind of feed used, etc.), and given the **irreversible** nature of the harm possibly done to the *P. oceanica* meadow, our recommendations are clearly based on the precautionary principle⁹⁸.

94 Law no. 76-663 of 19 July 1976 on Installations Classées pour la Protection de l'Environnement.

⁹³ This procedure is governed by Decree no. 83-228 of 22 March 1983, modified on 14 September 1987.

⁹⁵ Decree of 25 February 1993.

⁹⁶ French fish farmers are of the opinion that this set of regulations constitutes a very major restraint on the development of marine fish farming in the French Mediterranean, which has stagnated since the 1990s.

⁹⁷ European research programmes are under way to attempt to anticipate the consequences for the environment of discharge from fish farms MERAMED, Development of monitoring guidelines and modelling tools for environmental effects from Mediterranean aquaculture (www.meramed.com) and MEDVEG, Effects of nutriment release from Mediterranean fish farms on benthic vegetation in coastal ecosystems.

⁹⁸ An impact assessment, according to the model on which it is based, can propose optimistic predictions, i.e. the probable absence of impact. But if it proves to have an

Regarding the sectors where *Posidonia oceanica* meadows are present, the following recommendations are thus suggested:

- (1) No fish farm structure must be directly established over a *P. oceanica* meadow.
- (2) If there is a meadow nearby, a minimum distance of 100 m from the cages must be respected. This distance must be raised to 200 m near the lower limit of a meadow (more sensitive to turbidity than the superficial meadows) and adjusted according to the currents and the size of the farm.
- (3) Generally speaking, an installation on a 45-50 m seabed must be given priority⁹⁹ whenever possible.
- (4) An impact assessment should accompany any request for setting up a fish farm¹⁰⁰.
- (5) Permission to set up a fish farm should be reviewed every 4 years for possible extension, according to the demonstration that the *P. oceanica* meadows located nearby have not regressed (spatial extent and vitality; parameters in Table XII). This constraint, that implies setting up **monitoring of meadows** (see Chapter 16) should lead fish farmers to distance themselves as much as possible from the meadows.

Table XIII. A grid that shows the eligibility of fish farming sites according to the distance from the nearest *Posidonia oceanica* meadow, the depth and movement of water (openness of site). In green the combinations of ineligible factors. The eligible values shown correspond to the maximum annual production (in tonnes of fish produced per year). Open sites or not open sites located outside a bay or within a bay. For seabeds deeper than 40 m, the extrapolations suggested should be validated by case studies. Furthermore, no minimum distance has been suggested regarding farms producing more than 1 000 t, in the absence of concrete cases, and, thus, scientific data on the Mediterranean and *P. oceanica* meadows.

Depth	Openness	Distance of the nearest Posidonia oceanica meadow								
		< 100 m	100-200 m	200-300 m	300-400 m	> 400 m				
< 5 m	Open				< 100 t	< 500 t				
	Not open					< 100 t				
5-10 m	Open			< 100 t	< 500t	< 1 000 t				
	Not open				< 100 t	< 500 t				
10-20 m	Open		< 100 t	< 500 t	< 1 000 t	< 2 000 t				
	Not open			< 100 t	< 500 t	< 1 000 t				
20-40 m	Open			< 100 t	< 500 t	< 1 000 t				
	Not open				< 100 t	< 500 t				
> 40 m	Open		< 500 t	< 1 000 t	< 2 000 t	< 5 000 t				
	Not open		< 100 t	< 500 t	< 1 000 t	< 2 000 t				

In order to help applicants for a fish farm project optimize their choice of site, as a rough guide we are suggesting an eligibility grid for fish farming sites (Table XIII). This grid is empirical, in the absence of sufficiently precise scientific data (see §11.2). It cannot therefore replace a precise study of the site for the envisaged fish farm (currents, water movement, modelling of the diffusion of the mineral or organic substances produced) and cannot take into account the characteristics that are peculiar to a project (types of feed used, economics of their distribution, presence or not of antibiotics, mass of fish per cubic metre, etc.).

⁹⁹ Posidonia oceanica meadow is indeed absent at this depth (see §2.3).
 ¹⁰⁰ The impact assessment is mandatory in some countries or regions (see above).



The main element of the decision-making process that we are suggesting lies in the fact that permission to occupy the Maritime Public Domain for fish farming purposes should be restricted in time (as is anyway the case in some countries), and especially that their possible prolongation should depend on a demonstration (by the applicant or not) of an **absence of impact** on the *Posidonia oceanica* meadows. This constraint should lead applicants, in accordance with the precautionary principle, to interpret in the broadest sense the recommendations made in Table XIII.

12. THE POSIDONIA OCEANICA MEADOW AND DISCHARGE OF EFFLUENTS

12.1. THE PROBLEM

Among the many causes of the *Posidonia oceanica* meadows' regression, the discharge of effluents, whether urban, industrial or from leisure boats, bears a heavy responsibility (see Chapter 4). Domestic effluent represents 1% of the annual renewal of the water of the Mediterranean (UNEP, 1996). In the Mediterranean, 33.3% of these effluents has not been treated at all, 13.5% has been pretreated, 12.1% has undergone primary treatment, and 41.1% secondary treatment (UNEP, 1996).

Generally speaking, the discharge of effluents mainly acts at 3 levels on coastal marine habitas: (i) reducing water transparency, (ii) increasing the nutrient concentration, (iii) adding chemical contaminants. It can also bring about local drops in salinity that can harm *Posidonia oceanica* in that the species is stenohaline¹⁰¹ (Ben Alaya, 1972). For *P. oceanica*, a photophilous species¹⁰² that is sensitive to pollution, this discharge is thus a major factor of disturbance, on top of the other factors of regression (see Chapter 4).

Urban discharge presents a high nutrient and organic matter content. It directly (through turbidity) or indirectly (by encouraging plankton to develop) reduces water **transparency**. This has a negative impact on the *Posidonia oceanica* meadows, especially at depth: reduced shoot density, breaking up of the meadow and rise of the lower limit (Fig. 38).

The input of nutrients encourages the development of the *Posidonia oceanica* **leaf epibiota**¹⁰³ that intercept the light and thus harm their host's photosynthesis. Moreover, both directly (via the increase in the leaves' nutritive value) and indirectly (via the leaf epibiota) nutrients encourage grazing of *P. oceanica* by herbivores (see Chapter 4).

Urban and industrial discharge also acts on *Posidonia oceanica* meadows through the presence of **xenobiotics** (detergents, hydrocarbons, pesticides etc.): direct effects on *P. oceanica*, indirect effects on the flora and fauna of the ecosystem. Pollutants act on different levels according to their chemical characteristics: roots, rhizomes and/or leaves (Pérès and Picard, 1975).

The presence of pollutants causes a fairly marked change in the physiological activity of *Posidonia oceanica*: histological damage, impact on photosynthetic pigments, reduction in the growth rate of the leaves (Augier *et al.*, 1984b). Pollutants also have an impact on the other species of the ecosystem, giving rise to a drop in specific diversity, more marked for fauna (proportionally favoured) than for flora (Eugène, 1979).

- 101 A stenohaline species is one whose tolerance of variations in salinity is little.
- 102 A photophilous species (etymologically "light-loving") is one that lives in well-lit habitats.
- 103 Leaf epibiota (= epiphytes) are organisms that attach themselves to a plant (in this case, P. oceanica leaves) which only constitutes a substratum for them. Thus they are not parasites.

12.2. CASE STUDIES

12.2.1. Meadows in the Genoa Region

Balduzzi *et al.* (1984) studied the situation of some *Posidonia oceanica* meadows in the Ligurian Sea (Genoa

Region, Italy) subject to several kinds of pollution and compared it with Issel's description (1912, 1918b) in the early 20th century. The massive pollution from the central parts of Genoa and its port caused the disappearance of *P. oceanica* from the most directly affected areas, whereas, further east, along the Ligurian coast, the mainly domestic pollution and earthy discharge provoked a regression of the meadow's upper limit without having any visible impact on the meadow's interior, except in certain particular cases.

Along the Genoa coast, for example, the *Posidonia oceanica* meadow is in a state of irreversible degradation in Foce, i.e. at the mouth of the Bisagno torrent and near the entrance to the port, whereas moving eastwards, thus away from the town centre, in Sturla, some isolated clumps can be found, only at 10 metres depth. Moving eastwards, the state of the meadow seems to resemble that formerly described by Issel (1912, 1918b), although at that time the upper limit was much nearer the shore. In some cases, the concentration of discharge in the big sewage outfalls opening out in front of the meadow has caused degradation there: a sizeable reduction in shoot density, a drop in the specific diversity of the epifauna, etc. In Nervi, i.e. 10 km from the town centre, the state of the meadow is much better, and the withdrawal of the upper limit seems more due to local human pressure than to pollution from the town of Genoa.

12.2.2. The Marseille sewage outfall

The waste water discharge of the town of Marseille (France) has from 1886 opened out east of the town, some kilometres from Cap-Croisette, in the Cortiou *calanque* (Fig. 102).

At the time (19th century) this was a considerable advance, since it was one of the first complete waste water collection and discharge networks in France. In 1959, the water of the Huveaune, a very polluted little coastal river that ran into the Prado Bay (Marseille), was partially diverted to the discharge (in summer) and then, completely, from 1977 on (Belsher, 1977; Bellan, 1994; Bellan et al., 1999; Arfi et al., 2000a). Usually the current pulls this waste water westward, i.e. towards Cap Croisette (Fig. 102). Since a physicochemical treatment



plant started operating (October 1987), over 80% of suspended matter, 50% of organic matter, 15% of nitrogen, 40% of phosphates, 55% of hydrocarbons and 60-80% of heavy metals have been withdrawn from the effluent (Bellan, 1994; Bellan *et al.*, 1999). However, in all, the depollution rate only reaches an average 47%.

However, the imposed norms for discharge (e.g. reduction at source of pollutant additions, discharge of suspended matter limited to 50 mg/l with a maximum flow of 4.1 m³/s, Bertrandy, 1990) have brought about a clear drop in turbidity and more generally in pollution in this sector. The regression of the *Posidonia oceanica* meadow has been spectacular, mainly between Cortiou

and Cap Croisette (Fig. 102 and 103). This regression, as is obvious from the assessments of Blanc and Jeudy de Grissac (1978) resumed by Pergent-Martini and Pergent (1990) (Fig. 103), does seem to have been exaggerated; for example, it does not appear that the meadow ever existed near Cortiou, and also no "dead matte" is observed there (Pergent-Martini, 1994). An attempt to reconstitute the historical data allows us to estimate at **471 hectares** the surface area occupied by the meadow in the early 20th century, and at **263 hectares** in 1994, i.e. a regression of nearly 46% (Pergent-Martini and Pergent, 1996; Arfi *et al.*, 2000b).

Since the treatment plant started operating, this regression seems to have stabilized. Signs of recuperation, naturally very slow, have even been seen at the level of a characteristic structure (aureole produced by a bomb; Fig. 104) as well as recolonization by the meadow as regards its lower limit (Pergent *et al.*, 1988; Pergent-Martini and Pergent, 1990; Pergent-Martini *et al.*, 2002).

The case of the Cortiou sewage outfall is particularly interesting in that the pipe was laid down in a sector where the other causes of the *Posidonia oceanica* meadow's regression do not, or did not, at the time when it regressed, have a major role. Indeed, (i) there are no ports in the sector; (ii) trawling cannot (in theory) be blamed when the seabed is too superficial¹⁰⁴; (iii) the development of leisure boating, and thus anchoring, postdates most of the regression and (iv) there are no fish farms in the sector. Thus, unlike in other Mediterranean regions where many causes of regression act in synergy and are thus hard to separate (see §4.13), the regression of the *P. oceanica* meadow in the Cortiou sector can clearly be related to the discharge of waste water by an urban sewage outfall.

Fig. 103. Changes in the position of the lower limit of the Posidonia oceanica meadow (dotted line) between Cortiou calanque (where the waste water of the town of Marseille is released) and Callelongue calangue (east of Cap Croisette, Marseille) since 1945 (according to Blanc and Jeudy de Grissac, 1978). The present extent of the meadow in the sector is shown by the letters "v". Continuous lines show the isobaths. The meadow is present further south, between Calseraigne and Riou Islands, but has not been shown on this map. Since 1987, waste water from the town of Marseille has been treated by a physicochemical treatment plant. According to Pergent-Martini (1994), this map exaggerates the regression of P. oceanica; the meadow was certainly never present near Cortiou.



104 In fact, this point should be considered with caution illegal trawling has indeed been regularly seen in this sector over a seabed of only 8 metres.

12.3. RECOMMENDATIONS

No waste water discharge should open out into a *Posidonia oceanica* meadow. This is valid whatever the level of water treatment; indeed, this is fresh water that would usually rise to the surface, but the base of whose dilution cone can have a certain grip on the seabed, according to currents and during storms; now, *P. oceanica* is very sensitive to the low salinity (see §2.3). Additionally, even if phenomena of natural recolonization can be seen, they should not blind us to the fact that the species' biology means that the reconquest of a single hectare can take almost a century (Pergent-Martini, 1994).

Whether the pipes are old or new, there should be **monitoring** of the nearest *Posidonia oceanica* meadows, using markers and permanent quadrats (see §16.2), to check that the level of treatment is sufficient.



In the case of **old sewage outlets**, as is the case of that of Hyères-Carqueiranne¹⁰⁵, in Giens Gulf (Var, France) (Boudouresque *et al.*, 1988), a "dead matte" area already surrounds the point of release. Its evolution should be monitored using markers and permanent quadrats (see §16.2). If the situation has stabilized, especially if the meadow has begun to recuperate, as is the case in Giens Gulf (Charbonnel *et al.*, 1997a), after a treatment plant has started operating, or there is an improvement in its rate of treatment of waste water, it is usually not necessary to undertake expensive work to move the pipe or extend it beyond the limits of the meadow.

In the case of new sewage outlets, a **minimum distance** should be planned between the point of release and the nearest meadows (Table XIV). This grid is obviously very simplistic; in fact, the most important thing for a *Posidonia oceanica* meadow is firstly a big reduction in SM (suspended matter) and secondly a reduction in the discharge of nutrients and detergents, aims that should be attained without difficulty with the application of the European Directive on Urban Waste Water Treatment (EUR of 30 May 1991). The effectiveness of the reduction also depends on the kind of treatment used (Table XV), so that it would be a good idea to increase the distances by 50% in the case of treatment of the physicochemical type alone.

Regarding the waste water **pipe**, when a new pipe is laid, crossing a *Posidonia oceanica* meadow should be avoided, or the length of meadow crossed minimized (see Chapter 14). When the pipe crosses a meadow, it must not be buried (see Chapter 14). Furthermore, for various reasons (ageing of materials, collision with fishing gear, etc.), it is not unusual that these pipes present leaks. There must therefore be regular inspections (every year).

¹⁰⁵ District Association of Hyères-Carqueiranne for cleaning up Giens Gulf.

Table XIV. Minimum distance recommended between the point of release of a sewage outlet and the nearest *Posidonia oceanica* meadow, according to the size of the discharge and the rate of treatment.

	Wastewater load (population equivalent)								
Rate	<1 000	1 000 - 10 000	10 000 -	100 000 -	>1 million				
of treatment			100 000	1 million					
<10%	100 m	200 m	300 m	500 m	800 m				
10-50%	50 m	100 m	200 m	300 m	400 m				
>50%	20 m	50 m	100 m	150 m	200 m				

Table XV. Effectiveness of reduction of various elements in domestic effluent, according to type of treatment installed after mechanical pretreatment. * low effectiveness; *** good effectiveness; SM = Suspended Matter

	SM	Organic C	Ν	Р	Metals	Detergents	Bacteria
Physicochemical	* * *	* *	* *	*	* *	*	*
Biological	* *	* *	* *	* *	* *	* *	* *
Lagooning	* *	* * *	* *	* * *	* * *	* * *	* * *

13. THE POSIDONIA OCEANICA MEADOW AND SOLID WASTE

13.1. THE PROBLEM

Once built, ports have a tendency to silt up, at variable speeds. Regular dredging is necessary, and the problem then is that of the release of **dredged mud** and its impact. This mud is rarely stored on land; in fact, to cut costs, it is often released in the sea in dumping sites intended for this purpose, as long as its pollutant content is moderate (Mauvais, 1990). In Italy, between 1988 and 1997, 4.8 million cubic metres per year on average were dumped at sea, 0.7 million cubic metres per year of this in Liguria (Barbera, 2000).

The vertical growth of orthotropic *Posidonia oceanica* rhizomes does not allow the meadow to resist sedimentary entry of more than 5-7 cm/year (see §4.1). Dumping fairly dissolved materials from the dredging of ports or canals on the *P. oceanica* meadow thus has an extremely negative impact (see §4.10).

This impact on the *Posidonia oceanica* meadow is **direct** – it is **buried** at the site of waste release. The death of the meadow is quick, even if, over later months or years, the sediment can be re-suspended by hydrodynamism. It is also **indirect**: re-suspending the sediment, which settles further away, gives rise to the silting of the surrounding areas of meadow. Moreover, re-suspending fine particles increases the water's **turbidity**¹⁰⁶ (see §4.3); now, *P. oceanica*, a photosynthetic organism, needs light. The negative impact of dumping on the meadow has been demonstrated particularly in Liguria (Italy; Peirano and Bianchi, 1995), in Corsica in the Gulf of Porti-Vechju (Pasqualini *et al.*, 1999), and in Portmán (Murcia, Spain), where mining waste rich in mercury, lead, cadmium, zinc and manganese (2.5Mt/year) is released out at sea, including over *P. oceanica* meadows (Ros, 2003).

When the discharge is made up of **rock slabs** from coastal work, especially the removal of rocks, the direct impact is of course due to the (irreversible) covering of the *Posidonia oceanica* meadow. An indirect impact is also possible, because of hydrodynamism, through erosion around the blocks if they are big and movement of them if they are small. This impact is partly comparable to that of the deadweight moorings sunk in organised or unauthorized mooring (see §8.2.2).

Lastly, we should mention **macrowaste** discharge of human origin (bottles, batteries, tyres, engines, etc.) that as well as causing possible pollution and the aesthetic deterioration of underwater landscapes, whatever the ecosystem concerned, has the same effect on the *Posidonia oceanica* meadow as slabs of rock (see below) (e.g. Relini, 1972; Clark, 1986; Bianconi *et al.*, 1990; Guéna and Thomas, 1997a, 1997b; Thomas, 1997; Meinesz *et al.*, 2001b).

13.2. CASE STUDIES

In spring 1989, Saint-Jean port (La Ciotat, Bouches-du-Rhône, France) was dredged. 1 013 m³ of dredged mud was dumped over the *Posidonia oceanica* meadow (between 27 and 35 metres depth) in the bay of La Ciotat. The site indicated by the authorities lays much

106 The negative effect of turbidity increase not only concerns the discharge of mud from dredging over the *Posidonia oceanica* meadow but also discharge that is outside the meadow, but too close to it. further out to sea. This mud formed a number of heaps (each certainly equal to the discharge from a barge) from 80 to 100 cm high and with a surface area of about 30 m² (Fig. 105). The mud piles slowly spread out over the seabed; 6 months later, their height had been reduced by half (40 cm) and their diameter had doubled (Fig. 105). Under these heaps, the *P. oceanica* meadow died (Rivoire and Ceruti, 1989). Diving explorations in the bay of La Ciotat enabled many circular patches of dead meadow to be discovered, which could correspond to older dumping of dredged mud (Rivoire and Ceruti, 1989). It should be pointed out that it was not the authorities who discovered that the company responsible for the work was not respecting the contract specifications, but the residents of La Ciotat and local NGOs.



Fig. 105. Dumping of dredged mud (in grey) from Saint-Jean port (La Ciotat, France) over the *Posidonia oceanica* meadow. Above 2 days after the dumping. Below 6 months after the dumping. From Rivoire and Ceruti (1990), redrawn (C.F. Boudouresque).

In Cassis (Bouches-du-Rhône, France), as part of the improvement of Saint-Pierre quay, the sinking of **materials from rock removal** (about 600 m³) was permitted on the outer embankment of the port's southern protective sea wall, in the late 1990s. Furthermore, only the rocky materials, with the exception of silt and other products from the demolition, could be dumped out at sea. But, contrary to what had been provided for, the company responsible for the work dumped the discharge, or part of the discharge, directly over the *Posidonia oceanica* meadow located between 50 and 100 metres from the port's sea wall. Blocks of rock up to 1 metre in diameter, gravel and fine materials caused major degradation of this meadow (Fig. 106). Later on, the company responsible for the work was asked to remove blocks of sufficient size that the withdrawal operation would not cause additional degradation to the *P. oceanica* meadow (Bonhomme and Palluy, 1998). As in La Ciotat, it was not the authorities who discovered that the company responsible for the work was not respecting the contract specifications, but a local NGO.

In 1995, a public works company, acting in the Madrague district (Saint-Cyr-sur-Mer, Var, France), under the control of the town council and the Regional Equipment Department (the DDE), chose to dump rubble from a sea building site over a *Posidonia oceanica* meadow, and not in a dumping site, as was provided for. Given the protection that *P. oceanica* enjoys in France (see §5.1.2), the case was referred to the court and investigations were ruled for "*degradation of non-cultivated*"



and protected plant species". In 1989, in Cavallo (southern Corsica), the sponsor and the 3 entrepreneurs who had started to build a private port were sentenced by the Tribunal Correctionnel (court trying fairly serious criminal cases) of Ajaccio (Corsica); the ruling concerned 5 offences, including "execution of building work without a building permit in a 100 m coastal strip", and "mutilation of protected plants"; indeed, the slabs of rock had been placed over a *P. oceanica* meadow (Pergent, 1991a).

We do not go into here the cases, fortunately the most frequent, of discharge of dredged mud that has indeed been done in sites permitted by the administration that has competence in the matter, defined on scientific (ecological, toxicological) and economic grounds very far away from *Posidonia oceanica* meadows (e.g. Cocito *et al.*, 1994; Salen-Picard *et al.*, 1997; Ausili and Gabellini, 2000; Matteucci, 2000; Pellegrini, 2000; Virno-Lamberti, 2000) or the possible impact on other ecosystems than the meadow.

There are also some cases of steps that have been taken to protect the surrounding *Posidonia oceanica*



Fig. 106. Dumping of slabs of rock from rock removal onto the *Posidonia oceanica* meadow off the port of Cassis (Bouches-du-Rhône, France). A few clumps of *P. oceanica* (arrows) have escaped being buried. Photo by J. Laborel.

meadows from any impact, direct or indirect, during dredging and rock removal operations. The port development of the Pointe du Canier (Saint-Mandrier, Var, France) is exemplary from this angle. From the start it was given scientific assistance. The meadows near the development area were marked at the surface during the whole period of the work; this marking defined the area of the building works which the various site equipment in operation should never cross. Protective geotextile screens (see §7.3.3) were placed in front of the main areas of meadow in order to restrict the diffusion of fine particles. Lastly, regular inspections of the seabed to assess the state of the meadow's vitality were made during the entire period of the building work. When situations that were critical for the meadow were observed, the frequency and intensity of the dredging operations were reduced, as is (moreover) stipulated by the legislation in force (see §13.3). Monitoring the meadow confirmed that the meadow had in all only been very locally affected by the building works (Bonhomme *et al.*, 2001, 2003b, 2003c, 2004).

13.3. THE LEGISLATIVE FRAMEWORK

Internationally, discharge at sea comes under the London Convention¹⁰⁷, signed by most of the Mediterranean countries, including those within the RAMOGE Agreement. Discharge of any substance or material not appearing on the reverse list is forbidden¹⁰⁸ (Barbera, 2000).

In Italy, the discharge of dredged mud at sea is controlled by the Ministerial Decree of 24 January 1996, which complies with 107 The London Convention (Convention on the Prevention of Marine Pollution by Dumping of Wastes and Other Matter) was signed in 1972. Its additional Protocol dates back to 1996.

108 This reverse list was introduced by the 1996 Protocol. It replaces the list of permitted substances and materials, and is thus much more restrictive (Pellegrini, 2000). the provisions of the London Convention (Barbera, 2000). In Leghorn (Italy), a series of discharge sites was defined more than 30 km out to sea, very far away from the nearest *Posidonia oceanica* meadows (Secche della Meloria), with the site changing from one discharge operation to the next (Barbera, 2000).

In France, dredging and discharge are subject to declaration, in accordance with Law no. 92-3 of 3 January 1992 on water and the Interdepartmental Order of 23 February 2001(¹⁰⁹). Permission is delivered by the *département* (the chief administrative division in France) prefect, after preparation of the file by the Maritime Department (Coastal Water Quality Unit, Facilities Board of the *département*).

13.4. RECOMMENDATIONS

No discharge of products of dredging or slabs of rock should be permitted over the *Posidonia oceanica* meadow, as, besides, over any ecosystem with great ecological or economic value (GESAMP, 1975, 1982).

In the European Union countries (and thus in the RAMOGE area), dumping permits usually specify sites for release that are far away from the coast, and are thus not located over *Posidonia oceanica* meadows. However, it has often been noticed that public works companies responsible for this discharge, in the absence of active monitoring by the authorities, **shorten**, sometimes considerably, the distance of the discharge. Slabs of rock or products from dredging have thus been dumped directly over the *P. oceanica* meadow.

The **choice of the company responsible for the work** is thus very important. The *collectivités territoriales* (regional authorities) which, within the context of a call for tender, very logically choose the *lowest bidder* (the cheapest) must be aware of the fact that an exaggeratedly low cost can imply *ipso facto* that the **contract specifications will not be respected**, and therefore the discharge will be much nearer the dredging site than provided for (possibly over the *Posidonia oceanica* meadow).

Moreover, it is very shocking that it is often private individuals and **NGOs** who **alert** the authorities to the dumping of products from dredging and slabs of rock over a *Posidonia oceanica* meadow, thus non-respect of the contract specifications. This absence of vigilance could in fact be interpreted as condonation.

¹⁰⁹ Also see the Journal Officiel (JO, government publication) of 30 March 1993.

14. POSIDONIA OCEANICA MEADOWS AND THE LAYING OF CABLES AND PIPES ON THE SEABED

14.1. THE PROBLEM

It is often necessary, to provide an island with electricity or water, that a cable or "sea-line" pipe be installed on the seabed. The questions asked are: (i) Do these installations have an impact on the natural environment, first and foremost on the *Posidonia oceanica* meadow? (ii) Are some of the techniques for laying these less harmful than others? (iii) What overall strategy must be implemented to mitigate the impact?

Today we possess many impact assessments that were done before cables or pipes were laid and which, according to the local features (topography, biocenoses crossed, potential risks linked to uses – anchorage, trawling), make recommendations as to the technique to be implemented and the route to be taken (Meinesz and Bellone, 1989; Avon *et al.*, 1992; Francour *et al.*, 1992; Pergent-Martini *et al.*, 1992a; Pasqualini and Pergent, 1993; Charbonnel and Francour, 1994; Charbonnel *et al.*, 1994b, 1995c, 1995e; Bellone and Meinesz, 1995; Ruitton and Chiaverini, 1997; Charbonnel *et al.*, 1998, 1999; Bonhomme *et al.*, 1999; Charbonnel *et al.*, 2000a; Pergent *et al.*, 2002b; Bernard *et al.*, 2003; Pergent *et al.*, 2003). This experience allows a decision-making process to be formalized. Unfortunately, little data exists on the actual evolution of the settlements over the years which follow the work of laying down, that would enable the choices that were made to be validated (or not) (Pasqualini and Pergent, 1993; Molenaar, 1994; Charbonnel *et al.*, 2000a; Pergent *et al.*, 2002b).

Furthermore, in almost every case, the Contracting Authority has decided beforehand on the sites of departure and arrival on land ("landing" or "grounding") of the pipe or cable, according to 3 imperatives: (i) the shortest possible sea route (supposedly a straight line); (ii) the cost of land work and of burying to reach these sites of departure and arrival; (iii) safety of the placing in relation to the risks of being moved and degradation linked to uses (anchorage, fishing with towed gear). This practice is incompatible with a correct strategy for minimizing the environmental impact.

14.2. CASE STUDIES

14.2.1. Drinking water pipe between Hyères and the island of Porquerolles

To improve the provision of drinking water to the island of Porquerolles (Var, France), the town of Hyères planned to lay a 15-16 cm diameter pipeline with a flow of about 100 cubic metres a day. According to technical imperatives, the route originally proposed was from Tour Fondue to Bon-Renaud cape (Fig. 107). A major constraint was that the route had to enter an area that was forbidden to mooring and trawling because of the lines of cables already laid down (Francour *et al.*, 1992; Bernard *et al.*, 2003).

Benthic habitats, especially the *Posidonia oceanica* meadow, have been mapped with precision (7 km²) using aerial photography, field verifications and transects observed by divers (Francour *et al.*, 1992). Such mapping has shown that the map then available for the sector (Blanc, 1975) was very inexact. Moreover, a quality bathymetric profile of possible routes was established.

It is important to know whether there are unevennesses such as erosion scarps or rocky cliffs; indeed, the pipe cannot adjust exactly to a very uneven seabed, and as a result is exposed and precariously balanced (Fig. 108) and thus vulnerable to hydrodynamism, trawling and anchors.



Fig. 107. Route originally traced for the pipeline between Tour Fondue (Giens Peninsula, Hyères, Var, France) and Bon-Renaud Cape (Porquerolles Island).





On the basis of this new map, it appeared that 100% of the **original route** (the route originally traced) lay on the meadow. The **optimal route** for the pipe, that kept the crossing of the *Posidonia oceanica* meadow to a minimum (about 65% of its route) would have been the following: Tour Fondue–islet of Petit-Langoustier–Sainte-Anne cape (Fig. 109). However, this route presented many disadvantages (from economic and technical criteria) : (i) it became too deep (almost 40 metres over one-third of the route); (ii) the land route, on Porquerolles Island, was much longer; (iii) one part of the sea-line was outside the area where mooring and trawling are prohibited. A first compromise line (**compromise route 1**) was suggested: 80% of it still ran over the meadow, and 55% over dense meadow, and it ended in the Rousset cape (Fig. 109; Francour *et al.*, 1992). After much more precise mapping (aerial photography, side-scan sonar¹¹⁰ and field verifications), a new compromise route (**compromise route 2**) was suggested (Fig. 109; Bernard *et al.*, 2003).



Fig. 109. Map of the seabed between the Giens Peninsula (top) and the island of Porquerolles (bottom) (Var, France). Original route (suggested by the Hyères town council), optimal route (minimizing the impact on the *Posidonia oceanica* meadow) and compromise routes 1 and 2. From Bernard et al. (2003).

The route selected by the Contracting Authority (compromise route 2) crosses, near Tour Fondue, a meadow that is highly uneven and has erosion scarps (shifting intermattes in particular; see §2.5). To avoid parts of the pipe being exposed and precariously balanced, it seemed necessary to **bury it in a trench**¹¹¹, despite the negative consequences for the meadow: (i) direct, by destruction of the meadow; (ii) indirect, through the trench being widened by hydrodynamism and erosion; (iii) indirect re-suspending sediment and increasing turbidity (Francour *et al.*, 1992). It should be noticed that in cases where the pipe is not buried, when the meadow presents great unevenness

and the depth is less than 20 metres, hydrodynamism alone can dig out a trench around the pipe (Bernard *et al.*, 2003). The surface area of the meadow directly destroyed by laying the pipeline was estimated to be between 0.6 and 2.1 hectares (Table XVI; Francour *et al.*, 1992; Bernard *et al.*, 2003).

110 Side scan sonar data was acquired in 2000 as part of the "Posicart" campaign funded by the PACA Regional Council, the Rhône-Mediterranean-Corsica Water Board, and DIREN PACA.

111 Burial digging a trench at the bottom of which the pipe or cable is laid, then covering it up again with sediment.

 Table XVI. Estimate of the surface areas of the *Posidonia oceanica* meadow destroyed by the laying of a pipeline between Hyères (Tour Fondue) and the island of Porquerolles (Var, France). Compromise route 1. From Francour *et al.*, 1992.

 Type of meadow
 Trenching and
 Length
 Direct
 Indirect
 Total

Type of meadow		Trenching and rock removal	Length (m)	Direct impacts (m²)	Indirect impacts (m²)	Total impacts (m²)	
Sand		absent	No	400	0	0	0
<i>Posidonia</i> oceanica meadow	North Strong relief, high cover Centre Low cover		Yes	500 820	500	2 500-7 500	3 000-8 000
			No		160	1 600	1 760
	South	High cover	No/Yes	500	100-500	500-7 500	600-8 000
	South on rock	High Cover	No/Yes	180	40-200	180-2 700	220-2 900
			TOTAL	2 400	800-1 360	4 780-19 300	5 580-20 660



Fig. 110. Diagram of the direct and indirect impacts caused by the digging of a trench in the *Posidonia oceanica* meadow to hold a pipe carrying fresh water from Cannes to Sainte-Marguerite Island (Lérins Islands, Alpes-Maritimes, France). From Molenaar (1994).

14.2.2. Water pipes between Cannes and Sainte-Marguerite Island

2 water pipes were laid in 1992 between Cannes and Sainte-Marguerite Island (Alpes-Maritimes, France). Digging the trench for them, over a length of 1 500 metres (1 200 metres of this in the meadow), caused directly or indirectly the destruction of 2.13 hectares of meadow: the trench itself (0.57 hectares), the widening of the trench by hydrodynamism (0.18 hectares) and the

112 EDF Electricité de France.

burial of the meadow under sediment from the trench (1.38 hectares) (Fig. 110; Molenaar, 1994). The indirect destruction of the meadow thus almost tripled the surface area that had been directly destroyed.

It should be added that subsequently, after 1994, the regression continued, worsened by local hydrodynamic conditions (shallow areas and presence of currents from the east speeded up in the channel by the Venturi effect).

In 1992, in the same sector, between the Lérins Islands (Sainte-Marguerite and Saint-Honorat), the digging of a trench using a mechanical digger on a barge to protect an EDF¹¹² electric cable had a very great effect on the *Posidonia oceanica* meadow (Fig. 111; Ruitton and Chiaverini, 1997). Indeed, over a 760 metre underwater route, 2.65 hectares of meadow were destroyed, and a 30-65 metre wide channel of "dead matte" appeared around the trench.

14.2.3. Telephone cable between the continent and the island of Port-Cros

To improve the telephone link between the continent and the Hyères Islands (Var, France), France Télécom decided

to lay down a new optic fibre underwater cable to replace the existing network. Nearly 50 km of cable was to link Tour Fondue (Giens peninsula, the continent) with the islands of Porquerolles, Port-Cros and Le Levant, and thence to Bormes-Le Lavandou on the continent (Charbonnel *et al.*, 1995f). Much of the route was over the Hyères Gulf *Posidonia oceanica* meadow, the biggest on the French continental coast (Boudouresque and Meinesz, 1982; Astier and Tailliez, 1984).

The impact assessment, with its mapping of the benthic habitats in the areas where the cable comes up to the land, showed that housing the cable in a trench (near these areas) is possible in Tour Fondue (over 280 m distance), in Port-Cros (over 210 m) and in Bormes (over 800 m) without causing any particular ecological impact: the seabed is in fact occupied by "dead matte" or sand. The technique of trenching is generally preferred by the Contracting Authority, since it guarantees maximum safety from the



Fig. 111. Trench opened in the *Posidonia oceanica* meadow to house an electric cable (arrow). Lérins Islands (Alpes-Maritimes, France). Aerial® photo.



Fig. 112. Suggested route (in red and mauve) for the France Télécom cable between the continent and the Hyères Islands (Var, France) coming up to land at Port-Cros. From E. Charbonnel et al., 1995f.

risks of use-linked deterioration and displacement. In Le Levant, it also seems possible to remove the rocks over a length of 30 metres without harming the *Posidonia oceanica* meadow. Lastly, the depth increases quickly and it is no longer necessary to protect the cable by housing it in a trench; it can just be laid on the seabed (Charbonnel *et al.*, 1995f).

In Port-Cros, in the water of the National Park, where the cable comes up to the land, the impact study led to the suggestion that the original route should be moved 70 metres southward, allowing the live meadow to be avoided and the cable to be housed in a trench, since the seabed was made up of "dead matte". Moreover, the presence of a historic wreck (sunk in 1710; Guérout, 1981) also made it necessary to shift the originally traced route (Fig. 112; Charbonnel *et al.*, 1995f).

14.2.4. Electric cable between Corsica and Sardinia

There is an electricity link between Italy and Sardinia, via Corsica. The Corsica-Sardinia cable ("SACOI cable") was laid in 1967 by EDF. It starts from Cala di Sciumara (Bouches de Bonifacio,

Fig. 113. The electric cable between Corsica and Sardinia, simply laid over the meadow near Cala di Sciumara (Corsica). 35 years after it was laid (in 1967) there is no negative impact on the meadow. From Pergent *et al.* (2002b).



southern Corsica) and crosses various settlements and types of bed, including a *Posidonia oceanica* meadow (down to 33 m depth) (Fig. 116). The interest of the cable lies in its being one of the rare cases for which we have a study of its impact, 35 years after it was laid (Pergent *et al.*, 2002b, 2003).

When the cable crosses areas of sand it seems to bury itself naturally under its own weight, gaining a good protection. When the cable was laid over a *Posidonia oceanica* meadow, without either being buried in a trench or material added, the natural growth of the rhizomes has partially covered it again, especially at depth (Fig. 113). When the cable was buried under imported material (little blocks of

cement) the meadow has been destroyed over a width of 1 metre; given the slowness of *P. oceanica*'s growth, the recolonization has only been partial: 42% of the surface area recovered (Fig. 114 and 115). In any case, it does not seem that the cable has moved significantly under the effect of hydrodynamism; there are thus no secondary effects (Pergent *et al.*, 2002b).



Fig. 114. The electric cable between Corsica and Sardinia, near Cala di Sciumara (Corsica) was laid in 1967. It was fixed over the superficial meadows (10-15 metres) by covering it with little slabs of rock. 35 years later, the impact of this covering has not evolved. The line crossing it is a measuring tape unrolled over the bed during observations. From Pergent et al. (2002b).



Fig. 115. The electric cable between Corsica and Sardinia, near Cala di Sciumara (Corsica) was laid in 1967. Its hold on the meadows has been reduced by a start of recolonization of the slabs of rock by *Posidonia oceanica*. From Pergent *et al.* (2002b).



Fig. 116. Mapping the main habitats and seabed types present in Cala di Sciumara (Bouches de Bonifacio, southern Corsica). The route of the SACOI electric cable, laid in 1967 by EDF, and the optimized route that could have been selected if mapping had been done then, are shown. From Pergent *et al.* (2002b).



Fig. 117. An unused length of cable (arrow) left when the EDF cable was laid between Corsica and Sardinia lying in a *Posidonia oceanica* meadow, Cala di Sciumara. The line crossing the photo is a measuring tape unrolled over the seabed during observations. From Pergent *et al.* (2003).

The map of habitats and seabed types made by Pergent *et al.* (2002b) shows that the route taken by the cable could easily have been bettered, mostly avoiding the *Posidonia oceanica* meadow (Fig. 116). When the cable was being laid, unused lengths of the cable were left lying about, which constitutes visual pollution, especially regrettable in that the sector is today in the Bouches de Bonifacio Nature Reserve (Fig. 117; Pergent *et al.*, 2003).

14.3. RECOMMENDATIONS

(1) The Contracting Authority must propose a minimum 3 sites of departure and/or arrival at land. It can put these variants into its preferred order, explaining the reasons:

additional cost due to the length of the underwater route, additional cost linked to the work on land or at sea (removal of rocks), etc.

(2) It is necessary to have a precise map (between the 1:1 000 and the 1:5 000) of the nature of the seabed (rock, sand, mud, etc.) and its settlements, particularly of the extent of the *Posidonia oceanica* meadow and its cover and types of meadow present, and also of other habitats of heritage value (coralligenous bioconstructions, *Cystoseira* forests, *Cymodocea nodosa* prairies, etc.) (Table XVII; Pergent *et al.*, 2002b). The rate of baring of the *P. oceanica* rhizomes must also be assessed (measuring procedure: Boudouresque *et al.*, 1980a; Charbonnel *et al.*, 2000b; see Chapter 16). In most cases, the preexisting maps are unsuited to the problem (Charbonnel and Francour, 1994) and a precise map must thus be drawn before any choice of route is made (Bonhomme *et al.*, 2003a; Denis *et al.*, 2003).

Ecological sensitivity	Settlement and seabed type
Very high 6	Posidonia oceanica meadow on "matte" (striped, atolls) and on rock (staircase)
Very high 5	Posidonia oceanica meadow on "matte" (hill, sugar-loaf)
Very high 4	Posidonia oceanica meadow on rock (other), bioconstructions
High 3	Posidonia oceanica meadow on "matte" (plain meadow)
High 2	Cystoseira forest (on rock)
Low 1	Cymodocea nodosa meadow
Low 0	Other habitats and seabed types (sand, other "algae" on rock, etc.)

Table XVII. Ecological sensitivity of various habitats and seabed types. From Pergent et al. (2002b), modified.

(3) Along the various routes envisaged or proposed (scenarios) it is necessary to establish very precise **bathymetric profiles** (vertical precision: 10 cm; horizontal: 1 metre), at least in sectors where there is great unevenness (Francour *et al.*, 1992; Charbonnel *et al.*, 1998; Bonhomme *et al.*, 1999). The presence of this unevenness (erosion scarp, erosive intermatte, etc.) may in fact make burial in a trench necessary. Also, the **hydrodynamism** must be assessed from indicators that can be observed on the seabed: erosive structures, baring of the rhizomes, undulating meadow, ripple marks etc. (Charbonnel and Francour, 1994; Bonhomme *et al.*, 1999).

(4) Digging a trench in the *Posidonia oceanica* meadow to house the pipe or cable (**burial in a trench**) is a technical choice to be avoided as far as possible (Table XVIII; Charbonnel *et al.*, 1995c, 1995f; Bernard *et al.*, 2003). In fact, the trench rarely keeps the sediment that covers the

pipe (=filling in sediment) and hydrodynamism tend to widen the trench. The baring of the rhizomes, expressing strong hydrodynamim and/or a sedimentary deficit, is an indicator of this risk (Charbonnel and Francour, 1994). The fact that the filling sediment has gone has been verified between Cannes and Sainte Marguerite Island (Alpes-Maritimes, France; Molenaar, 1994) and in the Giens Gulf (Var, France); in the last case, the original (1 metre wide) trench has been widened to 35 metres (Gravez et al., 1988). On the other hand, burial is well suited to crossing soft bottoms. Just laying the pipe on the seabed is possible when there is not much hydrodynamim, especially from 10 metres depth (Avon et al., 1992). Apart from the low destruction of the meadow by direct burial, the indirect impacts on the meadow are very limited (Fig. 113; Table XVIII; Avon et al., 1992; Pergent-Martini et al., 1992a; Charbonnel et al., 1995f, 2000a). In the case of meadows in good health, the impact can even be zero: in fact, the meadow tends to cover the cable and



Fig. 118. A cable laid in 1948 between Tour Fondue and Porquerolles (Var, France) is today cover in "matte". The photo was taken on a segment where the *Posidonia oceanica* meadow is bared, revealing the cable. From Charbonnel *et al.*, 1995e). Photo by E. Charbonnel.

incorporate it in the matte, as has been seen for the cables that were laid down in former days between Tour Fondue and Porquerolles: a cable laid in 1948 is now covered by 35 cm of "matte" (Fig. 118; Charbonnel *et al.*, 1995c, 1995f). An intermediate solution consists of covering the cable under slabs of rock or cement to secure it (at shallow depth); this is particularly suited to rocky beds (Pergent *et al.*, 2003); in Cala di Sciumara (southern Corsica), where this technique was used over the meadow, the immediate impact was significant, but appears to be relatively modest after 35 years (Fig. 114 and 115; Pergent *et al.*, 2002b).

(5) When the pipe or cable is simply laid down on the seabed, it may be necessary to secure it, especially where there is strong hydrodynamism; the movement of the structure under the effects of the hydrodynamism can harm the bed, as well as the risk of the structure's deteriorating (Charbonnel et al., 1995c; Bonhomme et al., 1999; Pergent et al., 2002b, 2003). In a meadow on "matte", concrete deadweight moorings are effective, but have a big grip on the bed (1-1.5 metres) and may give rise to secondary effects on the meadow when the hydrodynamism is strong; this solution was adopted, for example, between the Ratonneau Islands and Château d'If Island (Marseille, France) (Fig. 119; Charbonnel et al., 1998, 2000a). The best thing would be to lay cast iron half-clamps or staples around the pipe or cable (Table XVIII; Charbonnel and Francour, 1994; Bernard et al., 2003), or non-destructive clamps of the Harmony® or similar kind about every 10 metres (Fig. 120; Charbonnel et al., 1995c; Francour and Soltan, 2000; Pergent et al., 2002b, 2003). When the meadow is on rock, the best securing system is pegging, using a securing gun for underwater work of the Spit® Rock HD 200 kind, or one like it (Augier, 1969; Pergent et al., 2003). Usually there is no danger of a pipe or cable being damaged by trawls since trawling (towed gear) is prohibited within 3 miles of the coast (France, Italy, Tunisia), and above the 50 m (Spain, Italy, Algeria, Gulf of Tunis), or 20 m (rest of Tunisia) isobath (Boudouresque, 1996). This law is, however, not respected in any country. It is thus necessary to study local fishing practices and envisage, with the maritime authorities (in France, the Maritime Affairs Department), an order



Fig. 119. Clamping a drinking water pipe crossing a *Posidonia oceanica* meadow by using deadweight moorings. Underwater pipe between the Ratonneau and If Islands (Marseille Gulf, Bouches-du-Rhône, France). From Charbonnel *et al.* (1998). Photo by E. Charbonnel.

Fig. 120. A Harmony®-type screw to hold a cable or pipe in a *P. oceanica* meadow on matte. From Francour and Soltan (2000) and Pergent *et al.* (2003).

Table XVIII. Direct and indirect impacts of laying a pipe over a *Posidonia oceanica* meadow, according to technique used. (-) negligible, (+) low, (++) great, (+++) very great. From Charbonnel *et al.* (1995b).

Technique	Direct Indirect impact impact		Potential recolonization by meadow	Cost of implementation	Protecting the pipe	
Ballast	+++	+++	12	+++	++	
Staplers	-		+++	+	+	
Half-shells	+	0(20	±	+	+	
Weights at intervals	-		++		+	
Small trench opened by jet of water under pressure (jetting)	++	++	÷	++	+++	
Big trench + filling	+++	+++	+	+++	+++	
Anchorage on the seabed + resistant piping	+	1. 	+++	++	+++	

forbidding mooring in the sector plus strict respect for the ban on trawling. If this is not enough, one solution is to lay down anti-trawl reefs (see Chapter 10). For example, in Cap Couronne (Bouches-du-Rhône, France) 13 t modules have been laid down to protect the telecommunications cables between France and Africa (Charbonnel *et al.*, 2001b; Frédéric Bachet, personal comm.; see Chapter 10).

(6) The presence of heritage elements other than *Posidonia oceanica* and the settlements mentioned in (2) (protected species such as the noble pen shell *Pinna nobilis*, wrecks of archaeological interest, etc.) must also be taken into account (Charbonnel *et al.*, 1995c, 1995f, 1999).

Table XIX. Grid for assessing the various scenarios for laying an underwater pipe or cable, from 0 (the worst scenario) to 100 (the best). An example of the calculation is given below.

Criteria		Method of calculation	Maximum number of points
	C1. Length of the <i>Posidonia oceanica</i> meadow and other intercepted habitats	Percentage (P) of habitat intercepted (compared to the shortest route) and contribution (C) per habitat: <i>Posidonia oceanica</i> meadow (p): $C_p = P_p \times 0$ Bioconstructions (b): $C_b = P_b \times 0.1$ <i>Cystoseira</i> forest on rock (f): $C_f = P_f \times 0.2$ <i>Cymodocea nodosa</i> meadow (c): $C_c = P_c \times 0.3$ Other habitats (a): $C_a = P_a \times 0.4$	40
Ecological criteria	C2. Types of <i>P. oceanica</i> meadow intercepted ^a	From 8 (no meadow or only plain meadow), to 6 (undulating meadow present), to 4 (hill meadow or sugar-loaf meadow present), to 2 (staircase meadow present) to 0 (striped meadow, atolls, barrier reef)	8
	C3. Other heritage elements	Protected species, wrecks of archaeological value: from 4 (absent) to 0 (many, abundant)	4
	C4. Trench <i>versus</i> laying flat	Percentage (P) of habitat intercepted (compared to the shortest route) and contribution (C) per type of seabed (T) and method of laying (M). $C = P \times T \times M$. Type of seabed: <i>P. oceanica</i> and bioconstruction: = 0.1, <i>Cystoseira</i> on rock = 0.2, <i>Cymodocea nodosa</i> and other habitats = 0.3. Method of laying: trench = 0.1, covered by slabs = 0.2, just laying it on the seabed = 0.3.	9
	C5. Length of sea route	From 25 (shortest route) to 0 (route 100% longer or more)	25
Technical and economic	C6. Additional cost, land route	From 9 (cheapest route) to 0 (most expensive route)	9
criteria	C7. Hydrodynamism in landing areas	From 5 (very weak) to 0 (very strong)	5

^a For the definition of these types of meadow, see §2.5.

(7) We suggest above (Table XIX), as a rough guide, an assessment grid of the scenarios for laying a pipe or cable that will enable the best choice of scenario to be made, according to ecological, technical and economic criteria. This grid uses the elements of an environmental awareness scale proposed by Pergent *et al.* (2002b).

(8) When **laying** a cable or pipe, it is vital that the cable-carrying ship does not anchor in the *Posidonia oceanica* meadow: it must anchor outside the meadow's lower limit. Work done near the coast should be done using a small service boat (Pergent *et al.*, 2003). If burial work is done inside the meadow, geotextile screens (Porcher, 1987) (see §7.3.3) should be placed all around so that fine particles do not settle on the meadow (Charbonnel *et al.*, 1995f; Pergent *et al.*, 2003). Lastly, it is vital that the cable-carrying ship does not leave unused segments of cable or pipe lying about, likely to constitute visual pollution (Fig. 117; Pergent *et al.*, 2003).

Example of how to apply an assessment grid for a route scenario

C1. The length of the route to be assessed is 1 300 metres; the shortest route is 1,000 m. The route to be assessed crosses 500 m of *Posidonia oceanica* meadow, i.e. 50% (of the 1 000 m shortest route) ($C_p = 50 \times 0 = 0$), 10 m of coralligenous bioconstruction ($C_b = 1 \times 0.1 = 0.1$), 100 m of *Cymodocea nodosa* meadow ($C_c = 10 \times 0.3 = 3$), and 690 m of other settlements, of which only 390 m will be taken into account, to refer to the shortest route ($C_a = 39 \times 0.4 = 15.6$). C1 is thus gets 0 + 0.1 + 3 + 15.6 = 18.7/40.

C2. Over part of the route, the cable crosses a hill meadow, and over another part a staircase meadow. The length of the crossing is not taken into account in the figure: types of meadow of great heritage value must be avoided. If many types of meadow of heritage value are crossed, that which is of the highest value (here, a staircase meadow) is taken into account. C2 thus gets a mark of 2/8.

C3. Only one species of heritage value (as well as *Posidonia oceanica* and *Cymodocea nodosa*, already taken into account in C1) has been found: the noble pen shell *Pinna nobilis*. Its density (0.5 individuals/hectare) is modest. So C3 gets the mark 3/4. Generally speaking, heritage species are those that appear in the Annexes to the Habitats Directive and the Berne and Barcelona Conventions (Boudouresque *et al.*, 1996).

C4. In crossing various types of seabed, a trench will be necessary over 100 m of *Posidonia oceanica* meadow ($C = 10 \times 0.1 \times 0.1 = 0.1$) and 100 m of *C. nodosa* meadow ($C = 10 \times 0.3 \times 0.1 = 0.3$), the cable will be covered by slabs over 100 m of meadow ($C = 10 \times 0.1 \times 0.2 = 0.2$) and will just be laid on the bed over the rest of its route in the meadow ($C = 30 \times 0.1 \times 0.3 = 0.9$), over bioconstructions ($C = 1 \times 0.1 \times 0.3 = 0.03$), and other settlements ($C = 39 \times 0.3 \times 0.3 = 3.51$). C4 thus gets a mark of 4.94/9.

C5. The route is 300 m longer than the shortest route (1 000 m), i.e. a 30% increase in length. C5 thus gets $25 - 25/100 \times 30 = 17.5$.

C6. The additional cost of the land route, compared to the less expensive route, is 40%. C6 thus gets a mark of $9 - 9/100 \times 40 = 5.4/9$.

C7. There is little hydrodynamism in the departure site; it is strong in the arrival site. The mark is thus 2.5/5.

The total mark (C1 + C2 + C3 + C4 + C5 + C6 + C7) is 53.9/100.

(9) In any case, **impact monitoring** of the pipe or cable should be provided for (Charbonnel *et al.*, 1998, 2000b; Pergent *et al.*, 2003), after it has been laid, and after 2 years, 5 years and 10 years, to validate (or not) the selected choice of scenario and to allow management of this kind of development to be improved, as is suggested in the present work. This monitoring can be done by setting up markers (at least ten or so) just after the pipe or cable has been laid (Fig. 121) and monitoring these (initial reference state, and later monitoring) using the techniques of the Posidonia Monitoring Network (RSP; Boudouresque *et al.*, 2000; Charbonnel *et al.*, 2000b).

14.4. CONCLUSIONS

All in all, laying a pipe or cable over the *Posidonia oceanica* meadow constitutes a development that today we know how to best manage. As long as the decision-making strategy (see §14.3) is respected, the impact on the meadow can be extremely modest, especially when the meadow's health (shoot density, cover) is good. If it is accompanied by a real ban on trawling and anchorage (intended to protect the pipe or cable), this development can even be **positive** in an overall way: restoring the seabed and its habitats. But on the other hand, when the *P. oceanica* meadow is already degraded, laying cables or pipes can be an aggravating factor that is likely to speed up its local regression.





15. CAN DEAD MEADOWS BE RESTORED?

15.1. THE PROBLEM

Posidonia oceanica meadows play a central part in **how the Mediterranean coastal environments function** (see Chapter 3): strong primary production and exporting part of this production to the many coastal ecosystems, controlling sedimentary flows, mitigating the hydrodynamism and protecting the beaches from erosion, recruiting species of fish and shrimps of commercial interest, protecting threatened species like the noble pen shell *Pinna nobilis* (Fig. 95) (Boudouresque and Jeudy de Grissac, 1983; Gambi *et al.*, 1989; Boudouresque *et al.*, 1994b; Pergent *et al.*, 1994; Jiménez *et al.*, 1996; Pergent *et al.*, 1997; Orth, 2000). The meadows also constitute a habitat with a particularly high specific diversity (biodiversity) (Boudouresque, 1996). Their protection is therefore a must, not only for reasons of ecological balance and heritage protection, but also for economic reasons (see Chapter 5) (Boudouresque and Meinesz, 1982).

Posidonia oceanica meadows are very vulnerable to anthropogenic pressures (see Chapter 4). Their **regression** has been considerable, especially near the great urban, industrial and port centres (Bourcier *et al.*, 1979; Boudouresque and Meinesz, 1982; Meinesz and Lefèvre, 1984; Pérès, 1984; Ramos-Esplá, 1984; Blanc and Jeudy de Grissac, 1989; Shepherd *et al.*, 1989; Meinesz *et al.*, 1991b; Boudouresque *et al.*, 1995a; Peirano and Bianchi, 1995; Pasqualini *et al.*, 1999; Orth, 2000; Pergent-Martini and Pasqualini, 2000).

Natural recolonization of marine Magnoliophyte ecosystems, **when the causes of their regression have ceased to operate**, is slow to extremely slow. In Australia, the horizontal progression of the rhizomes of *Posidonia australis* and *P. sinuosa* is 8-26 and 8-15 cm a year respectively (West *et al.*, 1989; Cambridge *et al.*, 2000). The average horizontal progression of a *Posidonia oceanica* meadow's front is no more than 3-4 cm a year (Meinesz and Lefèvre, 1984). Near Marseille, an area of 1.13 hectares destroyed by a bomb in 1942 was still not completely recolonized by 1999, i.e. 57 years later; 0.39 hectares of sand remain without any *P. oceanica* (see Fig. 104; Pergent-Martini, 1994; Pergent-Martini and Pasqualini, 2000). Furthermore, the ceasing of a pressure does not imply that recolonization will start immediately. In Menorca (Balearic Islands), 3 years after a fish farm stopped its operations, *P. oceanica* continues to regress. This persistence could be linked to the storing of organic matter in the sediment of the "matte" (Delgado *et al.*, 1999). In the Provence-Alpes-Côte d'Azur region (French Mediterranean), the Posidonia Monitoring Network (RSP) has reported an increase in the number of progressive meadow limits since practically all the waste water has started going through a treatment plants, but many meadows continue to regress (Boudouresque *et al.*, 2000; Charbonnel *et al.*, 2003).

It is the size of the regression of marine Magnoliophyte meadows, joined to the slowness of the natural recolonization, that has led to the idea that it could be necessary to start **restoring** the meadows, via transplanting or seeding (Meinesz *et al.*, 1990b, 1991a; Cinelli, 1991; Molenaar and Meinesz, 1992c, 1992d; Calumpong and Fonseca, 2001). The first attempts at transplanting marine Magnoliophytes dates back to 1947. These were done on the eastern coast of the USA and concerned *Zostera marina* (Addy, 1947a, 1947b). Later, on the eastern and south-eastern coasts of the USA there was an attempt to transplant in shallow (less than 6 metres) areas a whole set of species, mainly *Thalassia testudinum*, *Halodule wrightii*, *Syringodium filiforme* and *Zostera marina* (Thorhaug, 1979; Fonseca *et al.*, 1982b, 1982c; Meinesz *et al.*, 1990b;

Sheridan *et al.*, 1998). In Florida (USA), *Thalassia testudinum* has been successfully planted from seeds in a site where the species had been destroyed by heat pollution; 4 years after sowing, the reconstituted meadow was seen flowering and fruiting (Thorhaug, 1979). In Japan, many attempts have been made to reconstitute *Zostera marina* meadows from seeds germinated in an aquarium (Kawasaki *et al.*, 1988) or cuttings.

We call it **re-introduction** when a species is reintroduced into a region where it has existed in the past and where human action has caused it to disappear. We call it **re-stocking** when we release or replant individuals of a threatened species into a region from which it has not disappeared but where its numbers are thought to be too low. The restoring of seagrass meadows we mention here in fact always involves re-stocking (via transplanting or seeding).



Fig. 122. The different stages of transplanting Nanozostera noltii using clumps (plant, roots and sediment) placed in two PVC tubes that fit into each other, and are later removed. From Jeudy de Grissac (1984b).

15.2. RESTORING TECHNIQUES

In the Mediterranean, attempts to restore *Cymodocea nodosa* and *Nanozostera noltii* meadows by transplanting **clumps** (plant, roots and sediment) have been made in the Bouches-du-Rhône, Var and Alpes-Maritimes (France) (Fig. 122; Meinesz, 1976, 1978; Meinesz and Verlaque, 1979; Jeudy de Grissac, 1984b). In the Venice Iagoon (Italy), experiments with transplanting *Zostera marina*, *Nanozostera noltii* and *Cymodocea nodosa* have given interesting first results (Curiel *et al.*, 1994; Rismondo *et al.*, 1995; Faccioli, 1996). It is, however, *Posidonia oceanica* that has given rise to the greatest amount of work (Meinesz *et al.*, 1990b).

There are many techniques¹¹³ perfected for transplanting marine Magnoliophytes (e.g. Phillips, 1980b; Lewis, 1987; Meinesz et al., 1990b; Cinelli, 1991; Piazzi and Cinelli, 1995; Calumpong and Fonseca, 2001) and involve (i) laying cement slabs with holes in them, in which the cuttings are placed (Maggi, 1973); (ii) laying cement frames at the centre of which are placed a large number of cuttings held by wire mesh (Fig. 123; Cooper, 1976; Giaccone and Calvo, 1980; Cooper, 1982; Chessa and Fresi, 1994); (iii) plastic or metal mesh laid flat on the bed, onto which cuttings are fixed (Larkum, 1976; Molenaar and Meinesz, 1992a, 1992b; Molenaar et al., 1993; Piazzi and Cinelli, 1995; Piazzi et al., 1998, 2000); (iv) systems for fixing the cuttings directly onto the bed by pegs ("support stakes") or hooks (Fig. 124; Fonseca et al., 1982c; Molenaar, 1992; Charbonnel et al., 1995f; Rismondo et al., 1995; Davis and Short, 1997); (v) digging holes in which blocks of matte (clumps) are placed (Fig. 122; Addy, 1947a; Phillips, 1980a; Noten, 1983; Jeudy de Grissac, 1984b; Dennison and Alberte, 1986; Chessa and Fresi, 1994; Rismondo et al., 1995; Faccioli, 1996); in Australia, an amphibious machine weighing 3 t (ECOSUB1) was perfected for removing rootballs and then planting them (Paling et al., 2001a, 2001b; Calumpong and Fonseca, 2001; Paling et al., 2003); (vi) biodegradable nets (Fonseca et al., 1979; Kenworthy et al., 1980); and finally (vii) planting out young individuals (plantlets) that have germinated in the laboratory (Addy, 1947a; Cooper, 1976; Thorhaug, 1979; Lewis and Phillips, 1980; Kawasaki et al., 1988; Piazzi and Cinelli, 1995; Balestri et al., 1998; Piazzi et al., 2000).





Techniques that do not involve laying cement structures are preferable in that if the attempt fails there is no impact on the environment (Jeudy de Grissac, 1984b).

The **cuttings** are either wrecked rhizomes (Cooper, 1976, 1982; Sougy, 1996) or rhizomes removed from living meadows. Cuttings from wrecked rhizomes have the advantage of being available in tens of thousands, naturally produced by hydrodynamism, while their chances of being naturally replanted are very remote (Meinesz and Lefèvre, 1984). As to the removal of cuttings in living meadows, the advantage is that we know exactly where they come from (depth), we can decide on the number of shoots per cutting and the type of rhizome (plagiotropic or orthotropic), and

113 Some of these techniques have been patented. Thus use of them does not come under the Public Domain. thus optimize the restoring conditions (see below). In countries where *Posidonia oceanica* is a protected species, the



removal of cuttings, wrecked or not, is prohibited and is therefore subject to specific permission being granted.

The best **season** for transplanting *Posidonia oceanica*, for the cuttings to survive and develop, is spring for plagiotropic cuttings (creeping rhizomes; Fig. 124), with an average survival rate of 92% (after 3 years), and autumn for originally orthotropic cuttings (erect rhizomes), with a survival rate of 45% (Molenaar, 1992; Meinesz *et al.*, 1992). **Plagiotropic** cuttings give better results (74-76% survival on average) than orthotropic cuttings (30-60% survival), and their growth is quicker (Meinesz *et al.*, 1992; Molenaar *et al.*, 1993; Piazzi and Cinelli, 1995; Piazzi *et al.*, 1998, 2000). For orthotropic cuttings, the optimal **length** of the rhizome is 10-15 cm (Meinesz *et al.*, 1992). Cuttings from **deep meadows** give better results than those from superficial meadows (Molenaar and Meinesz, 1992a; Chessa and Fresi, 1994; Génot *et al.*, 1994). Furthermore, it is a good idea to put the cuttings **near to each other** (5-10 cm apart) (Molenaar and Meinesz, 1993, 1995). The survival rate depends on the **substratum**: in the case of plantlets grown from seeds, after 3 years it is 68% on "dead matte" as against 0% on a pebble seabed (Balestri *et al.*, 1998). However, for *Thalassia testudinum*, replanting from seeds has given poor results (Thorhaug, 1974).

With certain species (e.g. *Zostera marina*) cuttings taken from **distant sites** have not given good results. The reason could be that they present small genetic differences that make them less suited to the replanting site than the indigenous stock (Hartof, 2000). But for *Posidonia oceanica* transplanting cuttings from very distant sites has given very good results (Meinesz *et al.*, 1993).

If one bears in mind, when transplanting, all the above-mentioned elements, the survival rate of *Posidonia oceanica* cuttings can be very good: for example, 84% after 4 years in the Prado Bay in Marseille (Niéri *et al.*, 1991; Charbonnel *et al.*, 1994a, 1995e). But recolonization is always **slow**: on the same Prado site, the total number of shoots (about 1 240) did not differ significantly between 1991 and 1993; their growth on the surviving cuttings merely compensated for the drop in the number of cuttings (Table XX). It is only from the third or fourth year that growth compared to the number of shoots originally replanted becomes significant (Table XX and Fig. 125; Charbonnel *et al.*, 1995e).

Posidonia oceanica transplants, usually experimental, involving over 150 000 cuttings, have especially been done in Marseille, Toulon, Hyères, Port-Cros, Cannes, Golfe-Juan, Nice, Villefranchesur-mer, Galeria and the Lavezzi Islands (France) (Maggi, 1973; Cooper, 1976; Loques *et al.*, 1989; Molenaar *et al.*, 1989; Molenaar and Meinesz, 1991; Niéri *et al.*, 1991; Molenaar and Meinesz, 1992b, 1992c, 1992d, 1993; Meinesz *et al.*, 1993; Molenaar *et al.*, 1993; Génot *et al.*, 1994; Molenaar and Meinesz, 1995), in Monaco (Sougy, 1996), in Tuscany (Piazzi *et al.*, 1998), north of Civitavecchia (Piazzi and Cinelli, 1995), in Naples (Cinelli, 1980; Chessa and Fresi, 1994), in Sardinia (Chessa and Fresi, 1994), and in Sicily (Italy) (Giaccone and Calvo, 1980). These transplants are however of very limited size if compared to those of *Thalassia testudinum*, *Halodule wrightii*, and *Syringodium filiforme* done in the south-east of the USA. Lewis (1987) mentions 13 major operations there, the largest of which concerns a surface area of 49 hectares. In Japan, many large-scale *Zostera marina* transplanting campaigns have also taken place (Kawasaki *et al.*, 1988).

Campbell (2000) considers that transplanting is successful if the survival rate of the transplants is at least 50% and if the rate of advance of the rhizomes is at least 50%. Out of all the operations done, the success rate was less than 50% in the USA and less than 22% in Australia (Fonseca *et al.*, 1996; Campbell, 2000). In the Mediterranean, it is hard to assess the success rate with any precision.

Table XX. Experimental transplanting of *Posidonia oceanica* in the Prado Bay (France), using the method patented by the University of Nice-Sophia Antipolis. Changes in the number of cuttings and the total number of shoots (one cutting contains several of these) between 1991, date when the transplant was done, and 1995. dm=missing data. From Charbonnel *et al.* (1994b, 1995e) modified, in Boudouresque (2001).

Station	Original number of cuttings	Survival rate of cuttings	Survival rate of cuttings	Average number of shoots per cutting	Average number of shoots per cutting	Average number of shoots per cutting	Total number of shoots	Total number of shoots	Total number of shoots
	(1991)	(1993)	(1995)	(1991)	(1993)	(1995)	(1991)	(1993)	(1995)
1	132	84%	dm	2.9	2.7	dm	383	299	dm
2	100	85%	dm	2.9	3.4	dm	290	289	dm
3	139	89%	84%	2.8	3.1	6.4	389	384	747
4	100	87%	84%	2.8	3.1	5.2	280	270	437

In fact, most of the operations mentioned in the literature on the subject have been experimental tests, and in some cases the transplanted cuttings have been pulled up at the end of the experiment to analyse the features of their growth. Furthermore, most of the transplantings have not yet been assessed, several years after they were done. We empirically assess it as about 30-40% (Boudouresque, 2001).

15.3. RESTORING THE MEADOWS

We use the word **mitigation** (or compensatory measures, or even accompanying measures) to describe the measures intended to mitigate the effects of human impact on the environment, to compensate for such effects, or to get back to a former situation. Creating Marine Protected Areas (MPAs), putting down artificial reefs, optimizing artificial ripraps and jetties, enhancing the population of a species (transplanting marine Magnoliophytes, for example) can also be mitigating steps (Boudouresque, 2001). The notion of mitigation must, however, be used with the greatest caution: the risk does exist of mitigation being used as an excuse to enable destructive development to continue, deceiving the public and leaving the elected representatives with a clear conscience. It must in fact be clearly understood that no real compensation can be made for a development; the destruction of a *Posidonia oceanica* meadow when it has been covered under a facility is irreversible, for it is the biotope that has been definitively destroyed. Mitigation must therefore be seen only as an attempt at restoring approximately what was destroyed in the past, not as a justification for future destruction based on a hypothetical compensation (Boudouresque, 2001). Furthermore, any compensation measures announced at the time when a development decision is made do not legally commit the developing company, which usually has not the (legal and financial) authority to put them into effect. Oliver (1993) cites the very instructive case of the development of the Languedoc-Roussillon coast; in 1978, the Interdepartmental Team for developing the Languedoc-Roussillon coast (France) had accepted the principle of creating some fifteen or so "Biological Protection Zones" or "Nature Reserves" to compensate for the foreseen development; the CNPN (National Nature Protection Council, Ministry of the Environment, France) had approved these mitigation measures. But what really happened was that the creation of Nature Reserves necessarily had to comply with the normal procedure, in which the collectivités territoriales (regional authorities) have a decisive role; further, the Interdepartmental Team disappeared once the development had been completed; 20 years later, only 3 sites (out of the fifteen originally foreseen) are protected.

Improving marine Magnoliophytes transplanting techniques, and then actually implementing



them, seems urgent. The natural regeneration of the meadows is in fact very slow and it can prove necessary, in sectors where the regression has been considerable, to speed up natural regeneration by transplanting activity. But it should first be ascertained that the causes of the regression have ceased to operate (Fig. 126). The constraints specific to the marine environment do make such transplantation fairly expensive: for example, 250 to 500 man-hours, i.e. \$30 000-45 000 (¹¹⁴), are necessary to replant one hectare of *Zostera marina* (Thorhaug and Austin, 1976; Fonseca *et al.*, 1979, 1982a, 1982b, 1998; Chessa and Fresi, 1994). It would not therefore be very consistent to try to regenerate 1 or 2 hectares of meadow (at the end of 50-100 years, perhaps more, in the case of *Posidonia oceanica*) in a sector where many hectares of meadow continue to disappear

every year because of human activity. In all, transplanting must be integrated within an overall meadow management strategy on the scale of a bay or a region (Campbell, 2000; Hartog, 2000; Orth, 2000). This strategy must take into account the following elements (Fig. 126; Boudouresque et al., 1994a, 2000): (i) the total surface area of the existing meadows; (ii) the area lost every year due to regression and the causes of this regression; (iii) the area reclaimed every year through natural regeneration (if this exists); (iv) the area that one



can hope to reclaim through transplanting, with a 10, 20 or 50 year schedule; (v) the cost of transplanting, and a comparison with the effects of an identical alternative investment in mastering the causes of the regression (water treatment, laying anti-trawl reefs, providing organised moorings for leisure boats, creating Marine Protected Areas, etc.). Also, one should make sure that the marine Magnoliophyte population used in the transplanting is as close as possible to the population that has disappeared¹¹⁵, geographically, ecologically and genetically (Lambinon, 1994; Hartog, 2000); (vi) lastly, in any case, it is necessary that a trial be done on a small control plot, and monitored over at least 3 years. Only a favourable result can justify a large-scale operation (Boudouresque, 2001).

Unfortunately, there is a serious risk that the technical possibility of transplanting is misused to **serve as an excuse for further destruction** (Fonseca *et al.*, 1979, 1987). Transplanting *Posidonia oceanica* in the Mediterranean does indeed offer many examples of "planting for planting's sake" with no overall strategy, at the whim of local representatives' appeals (Boudouresque *et al.*, 1994a, 2000). (i) Planting has been done where *P. oceanica* does not exist naturally and seems never to have existed: what justification can there be for trying to replace an infralittoral sandy bed (certainly not a biological desert, but the public does not always know this) with a few clumps of

P. oceanica? (ii) Planting has been done where the meadow is rapidly regressing.
(iii) In Cannes (Alpes-Maritimes) some transplanting has been done in a *Cymodocea nodosa* meadow, another marine Magnoliophyte which, like *P*.

114 Dollars the late 1970s value, not corrected to reflect the dollar's present value.

¹¹⁵ This point does not perhaps concern the special case of Posidonia oceanica (Meinesz et al., 1993). oceanica, is protected by law in France (Order of 19 July 1988); destroying one protected species to replace it with another protected species is a rather inconsistent strategy. **(iv)** More seriously, transplanting *P. oceanica* has been suggested as a compensatory measure in the context of projects to build or enlarge marinas/ports. This is the case, for example, of the project to enlarge the port of Sanary-sur-Mer: to compensate for the (certain) destruction of a large area of meadow, it was planned to plant several thousand cuttings in an area where nothing suggested that meadows had ever existed in the past, or that *P. oceanica* would be able to survive there. Be that as it may, in France, because of the legal protection *P. oceanica* enjoys, transplanting operations that involve collecting and transporting cuttings are not permitted by the Ministry of the Environment; the only dispensations that have been granted concern scientific research (Boudouresque *et al.*, 1994a, 2001).



Fig. 126. Decision-making strategy for transplanting *Posidonia oceanica* and other marine Magnoliophytes. The question-answer sequence first looks at the local level (the site of the anticipated transplanting) and then the regional level (a homogeneous area, such as a bay). "No" answers should lead to the project's being abandoned. From Boudouresque (2001), modified.



15.4. A CODE OF GOOD CONDUCT

To avoid techniques of transplanting *Posidonia oceanica* being used as an excuse for going ahead with the destruction of existing meadows, a code of good conduct has been proposed, at the request of the French Ministry of the Environment (Boudouresque *et al.*, 1994a, 2001). Its principles appear below:

(1) The exact site and the biotope where the transplanting will be done must have been occupied by *Posidonia oceanica* before¹¹⁶.

(2) The causes of the disappearance of *Posidonia oceanica* (pollution, trawling, anchorage, etc.) from the site where the transplanting will be done must have ceased to operate. Thus, before any transplanting is done, one must demonstrate that the meadows or isolated clumps of *P. oceanica* that are nearest to the transplanting site have started a process of natural recolonization.

(3) Transplanting must not be done near very extensive meadows. It is useless to add several dozen or hundred square metres (0.001 to 0.01 hectare) to a meadow consisting of several hundred or thousand hectares¹¹⁷.

(4) Transplanting cannot be done to compensate for the destruction of a meadow. To avoid this abuse, no transplanting must be done within a distance of 10 km from a deliberate destruction (as part of coastal development) for a 10-year period.

(5) Transplanting on the exact site of a temporary destruction should however be possible, at least in the countries where the legal protection of *Posidonia oceanica* is not opposed to this. This is the case when an open trench for an archaeological dig is closed, or a pipe (or cable) crossing a meadow is buried.

(6) With the exception of the special case above (point 5), any transplanting of *Posidonia oceanica* must come after an experimental transplanting of several hundred cuttings; scientific monitoring for at least 3 years must show that the experiment has been a success before a larger operation can be envisaged.

(7) The removal of cuttings for transplanting must not endanger existing meadows. It must therefore be spread over a large area of meadow (less than 2 cuttings/m²). The use of cuttings detached naturally, although giving less good results, or plantlets from seeds, can also be envisaged.

(8) Lastly, transplanting must be done within an overall strategy of *Posidonia oceanica* meadow management of the concerned region (see §15.3).

¹¹⁶ But the special case of a coast profoundly changed by heavy development that has modified the pre-existing habitats (changes in sediment transfer, river diversion, etc.) should be considered. In this case, the new habitat can be tested, for it is likely to be naturally colonized by *Posidonia oceanica* over several centuries without human help. This is the case of the beaches of Nice (Alpes-Maritimes, France), where environmental conditions were shattered when 250 hectares of platform were laid down to build an airport, with the lasting diversion of the main coastal river in the Alpes-Maritimes, the Var, and the rapid erosion of the beaches. Similarly, other marine Magnoliophytes have been successfully transplanted in sites that have been profoundly changed by coastal development. This is the case of the artificial beach of Beaulieu (Alpes-Maritimes), and the area where the Martigues-Ponteau thermic effluent is released (Bouches-du-Rhône, France) (Meinesz, 1976, 1978; Meinesz and Verlaque, 1979).

¹¹⁷ But one can consider the special case of sites of great heritage value (Marine Protected Areas) or educational value (underwater trails), where transplanting would allow the natural recolonization to be speeded up.

Similar procedural guides for decision-making, suited to local problems and species, have been crafted in the US (Fonseca *et al.*, 1996, 1998) and in Australia (Campbell, 2000; Calumpong and Fonseca, 2001).

15.5. CONCLUSIONS

Despite a considerable research effort on a global scale, the tangible results of transplanting marine Magnoliophytes remain uneven. The result has been that the restoring of marine Magnoliophyte meadows cannot yet be compared to reafforestation, as practiced in the continental domain. The least disappointing results come from the US and, especially, Japan, where transplanting has been done on a wider scale than in the Mediterranean. This is explained by the much faster growth of the concerned species, *Zostera marina, Thalassia testudinum* and, especially, *Halodule wrightii* and *Syringodium filiforme* compared to *Posidonia oceanica*. True meadows, that is, areas of several hectares occupied almost continuously by marine Magnoliophytes that have grown from transplanted plants, have been reconstituted.

As for *Posidonia oceanica*, one of the slowest-growing marine Magnoliophytes in the world, the operational applications of the results obtained, from the experimental transplanting that has been done, are not very encouraging. More than 25 years after the first transplanting was done, there is still no true meadow reconstituted from transplants.

The cost remains very high¹¹⁸, even if one could imagine that this could fall for large-scale operations. Moreover, the cost/result ratio of transplanting does not seem to be competitive compared to other operations of protecting the quality of the environment or restoring coastal ecosystems.

All in all, transplanting *Posidonia oceanica* does not constitute a pertinent tool for managing Mediterranean coastal environments, at least in the present state of knowledge and considering the state of the environment of most of the Mediterranean coast. This conclusion does not rule out the possibility that in some particular cases transplanting can be envisaged; it should in these cases be very solidly governed by the **code of good conduct** mentioned above (see §15.4) and by an overall reflection on the integrated management of coastal environments on a wider regional scale.

118 The fact that transplanting must be done by professional divers explains, in particular, the high cost.

16. METHODS OF MONITORING POSIDONIA OCEANICA MEADOWS

16.1. INTRODUCTION

The aim of monitoring *Posidonia oceanica* meadows is threefold: (i) Monitoring an ecosystem that is of great heritage value but is vulnerable (see Chap. 3 and 4) in order to notice any new degradation rapidly. (ii) Using this ecosystem as a biological indicator of the overall quality of coastal environments (see Chap. 17). (iii) Measuring the effectiveness of a local or regional coastal environment policy, the set up of waste water treatment plants, the improvement of water quality, the reduction of domestic and industrial pollutant levels in the rivers flowing towards the Mediterranean, the set up of Marine Protected Areas (MPAs) and the freezing of coastal development at the expense of the sea (ports, reclaimed land, etc.) (Boudouresque *et al.*, 1990b, 1994a; Charbonnel *et al.*, 1993; Pergent *et al.*, 1995; Boudouresque *et al.*, 2000; Sandulli, 2004).

16.2. MONITORING TOOLS

Tools for monitoring *Posidonia oceanica* meadows concern 3 spatial scales: (i) ecosystem scale: mapping, use of aerial photographs other than for mapping, measuring cover, permanent transects; (ii) local scale: placing markers at the upper and/or lower limits of the meadow, permanent quadrats; (iii) microscale – shoots or groups of shoots: shoot density, plagiotropic to orthotropic rhizome ratio, degree to which rhizomes are bared, sediment granulometry, lepidochronology, leaf biometry (leaf length, number of leaves per shoot, etc.), cover and biomass of leaf epibiota. These tools have been gradually diversified and perfected since the 1980s, according to the experience drawn from their implementation and the progress made in scientific and technological research (Charbonnel *et al.*, 1993; Niéri *et al.*, 1993a; Boudouresque *et al.*, 2000; Pergent-Martini *et al.*, 2005).

Methods of mapping marine Magnoliophyte meadows and managing the cartographical data in Geographical Information Systems (GISs) were addressed by many authors (e.g. Augier and Boudouresque, 1970a, 1970b; Meinesz *et al.*, 1981b; Calloz and Collet, 1997; Lehmann and Lachavanne, 1997; Ward *et al.*, 1997; Dahdouh-Guebas *et al.*, 1999; McRea *et al.*, 1999; Stanbury and Starr, 1999; Bernard *et al.*, 2001; Denis *et al.*, 2001; McKenzie *et al.*, 2001; Pasqualini *et al.*, 2003; Bonhomme *et al.*, 2003a; Denis *et al.*, 2003; Bianchi *et al.*, 2004; Leriche *et al.*, 2004) and do not form the subject of the present work.

16.2.1. Cement markers

A dozen or so permanent cement markers are laid down at the lower limit of the meadow (one marker about every 5 metres) or its upper limit (one marker every 5-15 metres) (Harmelin, 1976, 1977; Meinesz, 1977; Buia *et al.*, 2004). A buoy is placed above each marker to make it easier to locate it when diving (Fig. 127). A "photo-stake" of 0.5 m height is placed at 1.5 m from each marker, facing the limit of the meadow.

The markers are placed in 4 stages (Fig. 127). (i) the limit is defined using a rope. When the limit is not clear cut, with isolated clumps of *Posidonia oceanica*, the choice of where to place the marker can be difficult (Fig. 128). It is better to leave aside spots that are completely isolated





from a sharp limit. If there is no clear cut limit, but just spots of meadow fairly distant from each other, the marker is placed from spot to spot, leaving out the most offshore spots (Fig. 128). However, the importance of these placing problems should not be exaggerated: the aim is to highlight the stability or changes in the overall meadow's limit, not of specific spots. Moreover, often a meadow progresses with respect to some markers and regresses at others; this is why about a dozen markers are used. (ii) positioning and fixing of markers (Fig. 127); Fixing them should enable them to resist hydrodynamism and, especially, trawling. However, sites that are too often trawled should be avoided. (iii) Placing "photostakes" in front of each marker. A number is fixed on each marker, and a plague with the marker's number is also slid underneath it¹¹⁹. (iv) Finally, the first photographic monitoring is done (starting point) (Fig. 127).

During the first photographic monitoring and the following photomonitorings, 3 photos (one off-centre to the left, one full-front, and one off-centre to the right)

are taken of each marker and the surrounding meadow, in addition a measuring scale set in front of the marker (Fig. 129); photos are taken from the tip of the "photo-stake" using a camera with a 15 or 35 mm lens (Charbonnel *et al.*, 2000b).

The monitoring of markings is aimed at highlighting the stability, progression or regression of the meadow at the level of each marker (Fig. 130). Changes are measured (to the nearest centimetre). When groups of shoots (clumps) stand out well immediately around the markers, the shoots are counted in these clumps.



Fig. 128. Choice of where to place the marking, when the limit is sharp (on the left) or not clear cut (on the right). In green the meadow, in yellow a soft bottom or a "dead matte". The black squares correspond to markers (their size is exaggerated). After C. F. Boudouresque (unpublished).

¹¹⁹ This plaque allows to identify more easily the marguer below which it is slid in case its number was torn away by the hydrodynamism or by recreational divers.


The regression of a meadow's limit can be relatively quick, so that the marker can be found several metres from the new limit. The photos become difficult to interpret, and the precision of the *in*

situ measuring of the distance between the meadow and the marker lessens. In this case, a new marker must be placed where the limit now is (Fig. 131).

The traditional technique of horizontal photographic shots can be supplemented by a mosaic of vertical shots of the lower limit of the meadow between the markers (Fig. 132). This technique was developed in Corsica for the Posidonia Monitoring Network.

16.2.2. Acoustic positioning

Acoustic positioning is an interesting alternative to placing markers (see § 16.2.1.) (Descamp et al., 2005). An acoustic marker is placed in the neighbourhood of the Posidonia oceanica meadow's limit that is to be monitored (less than 100 metres off) (Fig. 142; Aquamètre D100®120). Its position must be very exactly noted so that later return visits are possible, for example by a cement deadweight that remains in place when the acoustic marker is removed. The acoustic marker consists of 4 hydrophones that detect signals from a **pointer**. The pointer is used by divers, who direct it towards points at the limit of the P. oceanica meadow whose position they wish to record. The precision of the positioning is between 2 and 20 cm. This precision is academic, for it does not take into account errors of positioning on a later return visit due (particularly) to the acoustic marker's new positioning on the deadweight (Descamp et al., 2005). Anyway, this method enables the limit of a meadow to be traced with very many points, not just from a dozen markers (see § 16.2.1.). For the time being, despite its great interest, this method is experimental (but see § 16.3.3.).



Fig. 129. Marker No. 9, at the lower limit of *Posidonia oceanica* meadow, in Golfe-Juan (Alpes-Maritimes, France), 31m deep. 3 photos (offset to the left, centered, offset to the right) are taken during monitoring dives. Photos A. Meinesz.



Fig. 130. Marker No. 4, at the lower limit of the *Posidonia* oceanica meadow, in Niolon (Bouches-du-Rhône, France), 26 m deep, in 1987 (on the left) and in 1996 (on the right). Photos CQEL 13.

Marker No. 6, at the upper limit of the *Posidonia* oceanica meadow, in Cassis (Bouches-du-Rhône, France), 10 m deep, in 1994 (on the left) and in 1999 (on the right). Photos E. Charbonnel. In the 2 cases, the meadow is in progression.



Fig. 131. Marker No. 6, at the lower limit of the *Posidonia oceanica* meadow, in Saint-Tropez (Var, France), 36 m deep. In 2000 (left), the regression of the meadow is already clear, but shoots of *Posidonia* are still present around the marker.



In 2002 (right), the regression continued, isolated shoots around the marker (in the foreground) disappeared. A new marker has been put in place (in the background). Photos E. Charbonnel (GIS Posidonie).



Fig. 132. Mosaic of vertical shots, between two markers, in the lower limit of the medow of Lavezzi (-31 m). Photos G. Pergent.

16.2.3. Measuring meadow cover and shoot density

Cover is the average percentage of substratum covered (in vertical projection) by the *Posidonia oceanica* meadow (whatever the shoot density within the meadow or the clumps of *P. oceanica*) compared to the total surface area of the sector considered (sand, mud, "algal" settlements on the hard substratum, "dead matte" and living meadow). In fairly shallow, healthy meadows, cover can be high (80-100%). But at the lower limit of the healthy

meadow and in meadows subject to great human pressures, cover is usually low (between 5 and 40%) (Pergent *et al.*, 1995; Charbonnel *et al.*, 2000b); however, there may be exceptions (see below).

Cover is measured using a 30 cm x 30 cm see-through plastic slide divided into nine 100 cm² squares. The diver swims 3 metres above the seabed, holding the plaque at arm's length, and counts the number of squares that are (more or less completely) occupied by *Posidonia oceanica* (Fig. 133). He takes 30 measurements at fairly regular intervals (for example, every so many flipper strokes). Reproducibility is good¹²¹.

The Conservation Index proportion of "dead matte" compared to live *Posidonia oceanica* meadow in a given sector was suggested to act as an indicator of anthropogenic disturbance. CI = L/(L+D), where L is the surface area of the live meadow and D that of the "dead matte"

4 classes of CI are considered (Moreno et al., 2001):

(1) Cl < (x - 1/2 s)

(2) CI between (x – 1/2 s) and x

(3) CI between x and (x + 1/2 s)

(4) Cl > (x + 1/2 s),

where x is the average of CI in the sector under consideration and s is the standard deviation.



This index is however to be used with the greatest prudence; indeed, the presence of "dead matte" in a *Posidonia oceanica* meadow may be **of natural origin** (see § 2.5.) (Leriche, 2004). In some types of meadow, for example in the striped meadow, with very great heritage value, the surface area of "dead matte" may be much greater than that of the live meadow. It is in reality the evolution of the CI over time in a given sector that expresses anthropogenic disturbances, and not its absolute or relative value.

Shoot density is the number of live *Posidonia oceanica* shoots per unit of surface area (Giraud, 1977a, 1977b; Romero, 1986; Duarte and Kirkman, 2001; Buia *et al.*, 2004). Only those areas that are effectively covered by the meadow (thus excluding the intermatte) are taken into consideration when measuring density (Giraud, 1977a, 1977b). The measuring is done by counting, when diving, within 20 cm x 20 cm quadrats randomly set with at least 30 replicates per site (Pergent-Martini and Pergent, 1996; Charbonnel *et al.*, 2000b). Larger quadrats (30 cm x 30 cm or 40 cm x 40 cm, for example) are sometimes used, but they increase the potential measuring error; for the same sampling effort (time spent diving), it is better to use small quadrats, which allow the number of replicates to be increased¹²². It is important to note that depth explains 54% of the **variability** of shoot density: it decreases naturally with depth (Table XXI; Pergent *et al.*, 1995). Furthermore, the variability of shoot density is considerable, at short or mid distance, within the meadow, thus interpreting this parameter is very tricky and requires the greatest prudence (Panayotidis *et al.*, 1981; Balestri *et al.*, 2003; Leriche, 2004). To avoid simplistic mistakes, we suggest that it should not be routinely used by administrations responsible for the coastal environment.



122 It is important that the number of replicas is as high as possible, for the comparison of the average density as a function of time and the use of statistical tests (but see Romero, 1986). Table XXI. Abnormal, close to normal and normal average density of *Posidonia oceanica* shoots of leaves, according to the depth, in Western Mediterranean sea. From Pergent et al. (1995), modified.

Depth (m)	Abnormal density	Close to normal density	Normal density	
1	< 822	822-934	> 934	
5	< 413	413-525	> 525	
10	< 237	237-573	> 573	
15	< 134	134-246	> 246	
20	< 61	61-173	> 173	
25	< 4	4-116	> 116	
30		< 70	> 70	
35	5 8	< 31	> 31	
40	N78		> 1	

16.2.4. Permanent transects

Permanent transects are set perpendicularly to the limits of the *Posidonia oceanica* meadow, or perpendicularly to the isobaths. They are a few tenths to several hundred metres long. On the seabed, they consist of permanent markers of the same type as those used for marking the meadow limits (see § 16.2.1.), placed every 50-100 metres. At the start and end of the transect there is usually a group of 3 markers, so that the loss (e.g. by being buried under the sand) or removal of one or two of them does not compromise the later localizing of the permanent



Fig. 134. Example of habitats and bottom types intercepted along a pemanent transect, in Prado Bay (Marseille, France), between metres 100 and 200 of the transect. The values indicated along the transect correspond to the distance (in m) since its origin. From Gravez *et al.* (1992).

transect.

When monitoring the transect while diving, a measuring tape is held between the markers. The habitats and types of seabed intercepted by the measuring tape over a width of 2 m (1 m each side of the tape) are noted: sand, mud, "algal" settlements on hard substratum, "dead matte" and live meadow (Fig. 134). Limits between habitats and types of seabed are noted to within 20 cm, but actual accuracy does not exceed 50 cm, due to imprecision in the tape positioning (Gravez *et al.*, 1992).

Permanent transects are placed in Port-Cros (Var, France) (Boudouresque *et al.*, 1980a; Nédélec *et al.*, 1981), in the Giens Gulf (Var) (Charbonnel *et al.*, 1995d, 1997a; Bernard *et al.*, 2000) and in the Prado Bay (Marseille, Bouches-du-Rhône, France) (Gravez *et al.*, 1992, 1995, 1997, 1999).



16.2.5. Permanent Quadrats

Permanent quadrats are made up of 8 markers (placed at the angles and in the middle of the sides). They are usually measure 6 m x 6 m (Fig. 135; Gravez *et al.*, 1992). Bigger permanent quadrats

(10 m x 10 m) have also been placed (Boudouresque *et al.*, 1981, 1986b); but these take too long to map. During cartographical monitoring when diving, the permanent quadrat is divided by ropes into 1 m² squares; the ropes are removed after mapping. In each of the squares the habitats and types of seabed (the same as for permanent transects) are mapped to within 20 cm; naturally, the real actual accuracy, bearing in mind any possible error on the positioning of the ropes stretched out at each cartographical monitoring, is no more than 40 cm (Fig. 136; Boudouresque *et al.*, 1986b; Gravez *et al.*, 1992; Bernard *et al.*, 2000).



Fig. 135. Schematic drawing of a permanent quadrat of 6 m side, in degraded *Posidonia oceanica* meadow. The grid in small squares of 1 m, with ropes, intended for the mapping, is in place. These ropes will be removed after the mapping. After S. Ruitton (unpublished).





16.2.6. Aerial photographs

Shallow sites (< 10-15 m) can be monitored by means of aerial photographs. The aim is not to map the meadow itself (even if a map is made) but to closely monitor the possible changes in its limits, or in some easily recognizable structures (e.g. intermattes). The photos are taken according to a standardized protocol: altitude, lens, angle of shot, hour of day, definition and contrast of the film, etc. (Lefèvre *et al.*, 1984). The photos (monochrome, scale 1/4 500) are digitized and computer processed to get a 1/1 000 scale orthophoto map. The processing technique involves geometrical corrections on the basis of accurately located landmarks (Leriche-Guichard, 2001; Leriche *et al.*, 2004).

On these orthophoto maps, the borders between benthic structures are traced and an initial interpretation made. A light area usually corresponds to a spot of sand, whereas a dark area can reveal either the presence of *Posidonia oceanica* meadow or accumulations of dead *P. oceanica* leaves on the sea floor. The intermediate grey area usually corresponds to "dead matte" or settlements of photophilous "algae" on the rock. Afterwards, these borders (or a certain number of them selected for subsequent monitoring) are validated *in situ* by divers ("ground truth") (Fig. 137; Lefèvre *et al.*, 1984; Niéri *et al.*, 1993b; Pergent-Martini *et al.*, 1995a; Charbonnel *et al.*, 2000b). This validation is extremely important as many errors can result from photointerpretation not supported by this check; there are many examples of maps made without a "ground truth", or where the "ground truth" has been hastily done, and which contain basic errors – sometimes over vast stretches.

16.2.7. Microscale tools

A first microscale monitoring tool, **shoot density**, has for logical reasons been dealt with in the section on meadow cover and shoot density (see § 16.2.3.).



Fig. 137. Orthophoto map (on the left) and map of the habitats and bottom types (on the right) of Plage d'Argent (Porquerolles, Var, France). To the left of the photograph (white arrow), a vast dark zone corresponds to dead leaves of *Posidonia oceanica* which recover a sand bottom. To the right of the photograph, a dark spot corresponding to a "dead matte" (black arrow) appears with the same nuance of grey as a spot of a nearby alive *P. oceanica*. Data of the Posidonia Monitoring Network (RSP). From Charbonnel *et al.* (2003).

At the limit of a meadow, or of a Posidonia oceanica spot, the presence of **plagiotropic** shoots (horizontally growing rhizomes) is a sign of good health, since it expresses the meadow's tendency colonize (or recolonize) to neighbouring areas (Table XXII and Fig. 138; Charbonnel et al., 2000b). But within a meadow the importance of plagiotropic rhizomes can also express the meadow's (positive) reaction to stress, for example mooring pressure (Francour et al., 1999).

Rhizome **baring** is the result of a sedimentary deficit in the meadow (see § 4.2.): the amount of sediment trapped by the canopy (all the leaves together)



Table XXII. Measure of Posidonia oceanica vitality to the limit of the meadow pourcentage of plagiotropic shoots compared to the whole of the shoots, plagiotropic (creeping) ant orthotropic (erect). From Charbonnel et al. (2000b).

Pourcentage of plagiotropic shoots	Interpretation
< 30%	Stable meadow
30 à 70%	Slight trend to progress
> 70%	Net trend to progress

and the biogenic sediment produced in situ (remains of calcified organisms that had lived in the meadow: mollusc shells, sea urchin tests and spines, calcareous "alga", etc.) is less than the amount of sediment that has left the meadow, for example during storms (see Fig. 41). Conventionally, the degree of regression is measured as follows (Boudouresque et al., 1980a): (i) (creeping) plagiotropic rhizomes: the distance between the level of the sediment (soil) and the lower part of the rhizomes; (ii) (erect) orthotropic rhizomes: the distance between the sediment and the base of the outermost leaf, minus 2 cm (Fig. 139).

Granulometry (size of sediment grains, and distribution by size class) is indicative of hydrodynamism. Sediment traps (Gardner, 1980) provide information about the sedimentary balance and about silting. The sedimentary balance may account for the decrease of certain *Posidonia oceanica* meadows. When the yearly balance is more than 6-7 cm, the growth of orthotropic rhizomes' vegetative tips is not enough to compensate for sedimentation, and the vegetative tips die (see Fig. 40; Boudouresque *et al.*, 1984).

Big ripple marks and, especially **sand inundation**¹²³, can also bury the vegetative tips of shoots for a sufficiently long period (weeks or months) to kill them.



Fig. 138. Plagiotropic shoots (rhizomes growing horizontally) in the lower limit of a *Posidonia oceanica* meadow. Photo C.F. Boudouresque.



Fig. 139. Conventions for the measurement of the baring of plagiotropic (on the left) and orthotropic (on the right) rhizomes. See the text. From Boudouresque *et al.* (1980a).

In the context of the Posidonia Monitoring Network (PMN; RSP in French, see § 16.3.1.), banks of driven sand that temporarily hid the markers were observed. Such movements of sediment must be taken into account to avoid blaming on human pressure local regressions of *Posidonia oceanica* whose origin is in fact natural (Fig. 141).

When *Posidonia oceanica* leaves die, only the limb falls off. The base of the leaf (sheath) remains attached to the rhizome. The thickness of the sheaths and their anatomical structure present cyclical

¹²³ Sand inundation are natural movements (but irregular) of the sediment, which can bury settlements (meadows, hard substrate), then withdraw, under the effect of the hydrodynamism. The thickness of sediment can reach several dozen centimetres and even exceed the metre.

Fig. 140.

Effect of a sand inundations on a limit (a) of *Posidonia* oceanica meadow.

b the sediment submerges the margin of the meadow during an hydrodynamic event.

c the shoots of leaves die.

d The sediment withdraws following a new hydrodynamic event; a zone of "dead matte" appears (C.F. Boudouresque, unpublished).



variations according to their order of insertion along the rhizome, and thus to the period of the year when the leaves were born, lived and died. These cyclical variations have chronological significance: each cycle corresponds to a 1 year period (Fig. 3). This is like the variations in the thickness of growth rings in tree trunks (dendrochronology). By analogy, the analysis of the thickness cycles of *P. oceanica* sheaths has been called "lepidochronology" (Crouzet, 1981; Boudouresque *et al.*, 1983; Crouzet *et al.*, 1983; Pergent *et al.*, 1989b; Pergent, 1990a). As well as the chronological signal (years) *P. oceanica* sheaths enable a whole set of other parameters to be memorized: speed of growth of the rhizomes, number of leaves produced each year, primary production, and how these change from one year to the next, according to sedimentation rate, water quality and climate parameters. Lepidochronology is thus a powerful tool for backdating events that happened before the date of monitoring (Calmet *et al.*, 1986, 1988, 1991; Pergent *et al.*, 1992; Pergent-Martini and Pergent, 1994; Roméo *et al.*, 1995; Pergent-Martini, 1998).

Posidonia oceanica leaves constitute a substratum for **leaf epibiota**¹²⁴: Phaeophyceae (mainly Ectocarpales), Rhodobionta (mainly Acrochaetiales and Corallinales), Bryozoa, Hydroids, etc. Their biomass presents a seasonal cycle with a maximum from March to September (Thélin and Bedhomme, 1983). This biomass is particularly high in sites where there is a high input of nutrients and/or organic matter (Jupp, 1977; Pergent-Martini *et al.*, 1995b, 1999). The biomass of the epibiota is thus a water quality indicator and allows the impact of the waste water discharges, fish farms and recreational ports, to be measured. However, because of the seasonal and bathymetric variations of the epibiota biomass, comparisons must be drawn from the same season and the same depth (Romero, 1986; Pergent-Martini *et al.*, 1999). In addition, grazing is likely to reduce the epibiota biomass; indeed, epibiota are particularly palatable to herbivores (Kitting *et al.*, 1984; Ott and Maurer, 1977; Traer, 1979; Tomàs-Nash, 2004); consequently, interpretation of data on epibiota biomass is an extremely complex matter.

The **biometry of the leaves** includes several descriptors. (i) Number of leaves per shoot. (ii) Length of adult leaves (leaves whose growth is over). (iii) Foliar surface area per shoot (in cm²). (iv) Leaf area index¹²⁵ (LAI, in m²/m²). (v) Coefficient A (percentage of leaves which have lost their apex).

124 Leaf epibiota organisms that grow on a plant that they use as a substrate.



Parameters (i) to (iv) provide information on *Posidonia oceanica*'s vegetative development, whereas coefficient A gives information either on herbivore pressure (characteristic grazing marks) or on hydrodynamism (breakage of leaves) (Drew and Jupp, 1976; Giraud, 1977a, 1979; Giraud *et al.*, 1979; Panayotidis and Giraud, 1981; Boudouresque and Meinesz, 1982; Pergent-Martini *et al.*, 1999). As for leaf epibiota biomass, natural seasonal variations mean that comparisons should only be drawn for the same season.

16.3. THE MAIN MONITORING SYSTEMS

The tools described above (§16.2) interact differently to constitute monitoring systems on a local or regional scale.

16.3.1. RSP (the Posidonia Monitoring Network)

The *Réseau de Surveillance Posidonies* (RSP; the Posidonia Monitoring Network) is the main system for monitoring *Posidonia oceanica* meadows in the Provence-Alpes-Côte d'Azur region (PACA, France). It was set up in 1984 by the PACA Regional Council, the *services maritimes* of the concerned *départements* (chief administrative divisions in France), the Alpes-Maritime *Conseil Général* (the council who administer the *département*) a scientific NGO, GIS Posidonie (Boudouresque *et al.*, 1990b; Charbonnel *et al.*, 1993; Niéri *et al.*, 1993; Charbonnel *et al.*, 1994a, 1995a, 1995b; Boudouresque *et al.*, 2000; Charbonnel *et al.*, 2000b, 2000c, 2003; Cadiou *et al.*, 2004). It continued until 2004 (anonymous, 2005a, 2005b, 2005c).

In 1984, 24 monitoring sites were selected along the 650 km of PACA region coast. 9 new sites were added in 1994, bringing the number of sites up to 33 (Fig. 141). These sites were located (i) in sites where human pressures are important (and where one can expect regression of *Posidonia oceanica* meadows), (ii) in reference areas where *P. oceanica* is not a priori exposed to human pressures and where one might think that the meadows are stable or progressing, and (iii) in sites with intermediate characteristics (Boudouresque *et al.*, 2000; Charbonnel *et al.*, 2000b, 2003). The RSP sites were located at the 2 extremes of *P. oceanica*'s bathymetric area: the upper limit (15 sites) and the lower limit (18 sites). Indeed, it is at these two levels that the meadow is most sensitive to environmental change. The tools used are mainly: aerial photographs (validated by "ground truth") at the upper limit, monitoring markers at both the lower limit and the upper limit of the meadow, measuring meadow cover and shoot density, measuring rhizome baring, assessing the proportion of plagiotropic rhizomes, measuring lepidochronology and leaf biometry characteristics of shoots.

Given the slow growth of *Posidonia oceanica*, each site is monitored every 3 years. The RSP's chronology was thus the following: the period 1984-87 (selecting the sites, setting up markers and description of the initial state against which future measurements are compared), the period 1988-90 (first return to the sites), the period 1991-1993 (second return), the period 1994-96 (third return), the period 1997-1999 (fourth return) and the period 2000-2002 (fifth return) (Boudouresque *et al.*, 1990b; Charbonnel *et al.*, 1993; Niéri *et al.*, 1993a; Charbonnel *et al.*, 1995a, 1995b; Boudouresque *et al.*, 2000; Charbonnel *et al.*, 2000c, 2001a; Ruitton *et al.*, 2001b; Charbonnel *et al.*, 2004).

During the 15 years of monitoring, along the coast of Provence and the French Côte d'Azur, the *Posidonia oceanica* meadows presented 2 opposite trends (Table XXIII). At the upper limit, the number of **regressing** sites dropped. But conversely, at the lower limit the regression continued or even worsened. All in all, between 1988-90 and 1997-99, the percentage of regressing or stable limits dropped (from 79% to 59%) whereas that of the progressing limits increased (Table XXIII; from 21% to 42%). For each site and each period, a dynamic score was calculated as follows: 0=very marked regression; 1=marked regression; 2=slight regression; 3=stability; 4=slight progression; 5=relatively great progression¹²⁶. The "no change" between 1988-90 and 1997-99 hypothesis was rejected for the upper limits ¹²⁷ (Table XXIV). More detailed analysis of the data highlighted marked differences between sectors, for example between the east (Côte d'Azur) and the west (Provence) of the region. Furthermore, for a given site change often took place irregularly over time, with alternating phases of regression and progression (Boudouresque *et al.*, 2000).

16.3.2. The Prado Bay monitoring system

Prado Bay (Marseille, France) was in the past occupied by a vast *Posidonia oceanica* meadow (Marion, 1883; Picard, 1965b). Then this meadow was subject to heavy human pressures: artificial beaches reclaimed from the sea, 2 marinas/ports, turbidity caused by the construction of one of these ports, industrial pollution brought by the coastal river (the Huveaune), domestic and industrial



Fig. 141. The 33 RSP (Posidonia Monitoring Network) sites in Provence and the Côte d'Azur (France). From Charbonnel et al. (2003).

- 126 The regression can be considerable (the limits of the meadow can retreat several metres). On the other hand, the progression can only be slow a few dozen centimetres is seen as a relatively sizeable advance.
- 127 McNemar test for the significance of changes and binomial test; Siegel, 1956.

Table XXIII. Changing limits of Posidonia oceanica meadows (regression, stability or progression) as a percentage of the 24 sites set up in 1984-1986 by the RSP (Posidonia Monitoring Network) in the Provence-Alpes-Côte d'Azur region (France) over 5 monitoring periods. From Boudouresque et al. (2000), updated data.

		1988-1990	1991-1993	1994-1996	1997-1999	2000-2002
Regression	Upper limit	42	25	17	17	8
	Lower limit	50	45	67	67	67
	All sites	46	35	42	42	37
Stability	Upper limit	50	42	58	33	50
	Lower limit	17	36	8	0	0
	All sites	33	39	33	17	25
Progression	Upper limit	8	33	25	50	42
	Lower limit	33	18	25	33	33
	All sites	21	26	25	42	38

Table XXIV. Changing average dynamic score (see text) of the sites monitored by the RSP, at the upper and lower limits of the *Posidonia oceanica* meadows (Provence-Alpes-Côte d'Azur region, France). Data *in* Charbonnel *et al.* (2001a). NS=Not significant. Binomial test between 1988-90 and 1997-99. The period 2000-2002 is incomplete.

	Number of sites	1988- 1990	1991- 1993	1994- 1996	1997- 1999	2000- 2002	Binomial test
Upper limite	12	2.5	3.1	3.1	3.6	3.5	p = 0.002
Lower limite	12	2.4	2.4	1.9	2.2	2.3	NS

pollution from an untreated sewage outfall (until 1987) located 10 km upstream of the dominant current (Harmelin and True, 1964; Niéri *et al.*, 1986; Pergent and Pergent, 1988). Between 1970 and 1980, the water of the coastal river was diverted towards the Cortiou seawage outfall (see §12.2.2). Lastly, in 1987, a (primary) water treatment plant started operating. The result was a significant drop in the pollution level (Bellan *et al.*, 1999; Soltan *et al.*, 2001).

The Prado Bay monitoring system was set up in 1986 at the request of the town of Marseille¹²⁸. The monitoring tools used were the following: 2 permanent transects, 4 permanent quadrats, marking the meadow's lower limit, measuring meadow cover, quantitative mapping of part of the meadow (based on cover; kriging method) and laying sediment traps (Niéri *et al.*, 1986, 1993b; Gravez *et al.*, 1992, 1995). From 1988 on, i.e. 1 year after the town of Marseille's sewage treatment plant started operating, *Posidonia oceanica* not only stopped regressing but actually progressed significantly in certain sectors (Gravez *et al.*, 1995, 1997, 1999).

16.3.3. The Monaco monitoring system

The Larvotto Underwater Reserve (Principality of Monaco) occupies 50 hectares. It was set up in 1976. The *Posidonia oceanica* meadow constitutes the most important habitat in this Reserve. This meadow is the sole representative of this habitat in Monaco. Preserving it is thus a priority, both because of its heritage importance and because of its exemplary value in the Principality of Monaco's environmental policy.

Monitoring this *Posidonia oceanica* meadow started in 1977 with the placing of cement markers on a portion some 100 metres long of its lower limit (Alexandre Meinesz, unpublished data), which makes it one of the longest long-term set-ups concerning *P. oceanica* in the Mediterranean. To study

¹²⁸ The town of Marseille decided in 1999 to abandon the site. It is not for us to judge the political and financial reasons for this decision. From the scientific point of view, abandoning a "long-term monitoring site" is always regrettable in that its interest grows exponentially with time.

Fig. 142. Positioning the limit of a *Posidonia oceanica* meadow using the Aquamètre® acoustic positioning system. Photos by J. de Vaugelas and A. Meinesz.



Locating the acoustic marker.



Divers handling the pointer.



The acoustic marker.

the long-term evolution of this meadow, it was mapped in 2001 (Jean de Vaugelas, unpublished data). The position of the lower limit was established using an Aquamètre D100® (Fig. 142, and see §16.2.2). This position was compared to that established by Alexandre Meinesz in 1977. More than 25 years later, it appears that the lower limit has remained stable.

In 2004, using the Aquamètre D100®, the *Posidonia oceanica* meadow's lower edge was positioned over a distance of about 1.5 km. (Fig. 142). This positioning will enable an extremely exact monitoring to be done later.

16.3.4. The Liguria Region (Italy) monitoring system

The Ligurian monitoring network of *Posidonia oceanica* meadows was set up in 2002 by the Liguria Region, in cooperation with the Italian Ministry of the Environment. It is based upon two procedures.

In 3 sites (Imperia, Cogoleto, Punta Mesco) the **marking** technique is being used: this consists in monitoring the meadows' lower limit using the cement markers, according to the specifications suggested by the Ministry of the Environment (drawing their inspiration from the French model); this procedure involves placing 10 markers at about 5-metre intervals to check every year how this limit is changing, documented by measurements and photographs; for each site, over a strip 10 metres behind the markers, the following data is also noted:

- type of limit (sharp, progressive, erosive, regressive)
- estimate of meadow cover
- estimate of shoot density
- type of rhizome growth (% of plagiotropic rhizomes)
- estimate of the percentage of bared meadow
- shoot collection for phenological and lepidochronological analyses.

In 6 other sites (Sanremo, Arma di Taggia, Borghetto S. Spirito, Genoa Quarto, Camogli, Riva Trigoso) the "**transect"** procedure

is used: this biennially checks the state of conservation of the meadows along a transect that runs straight out from the coast, where 3 stations were identified, the first near the upper limit, the second at intermediate level and the third near the lower limit of the meadow; in correspondence to these 3 stations, the following data is noted:

- type of limit (sharp, progressive, erosive, regressive)
- estimate of meadow cover
- estimate of shoot density
- type of rhizome growth (% of plagiotropic rhizomes)
- estimate of the percentage of bared meadow
- shoot collection for phenological and lepidochronological analyses.



16.3.5. Other monitoring systems

Many *Posidonia oceanica* meadow monitoring systems have been set up along the Provence and Côte d'Azur coast (France) in response to local needs: the Riou Archipelago, near Marseille (Pergent-Martini, 1994; Pergent-Martini *et al.*, 1995a; Pergent-Martini and Pergent, 1996; Pergent*al.*, 1995d, 1997a; Bernard *et al.*, 2000); the Maures coast, between Cavalaire and Saint-Tropez (Var; Bonhomme *et al.*, 2000); and Cap-Martin, between Monaco and Nice (Alpes-Maritimes), in a site colonized by the introduced Chlorobionta *Caulerpa taxifolia* (Ruitton *et al.*, 2001a).

In Spain, *Posidonia oceanica* meadow monitoring systems have been set up in Catalonia¹²⁹ (33 sites), in the Comunitat Valenciana¹³⁰ (15 sites), in the Balearic Islands¹³¹ (13 sites) and in the Murcia region (Alvárez and Marbà, 2001). A monitoring system very similar to that of the RSP in the PACA region (France) was set up in Algeria in the Algiers region (Semroud *et al.*, 1998; Boumaza and Semroud, 2000).

16.4. CONCLUSIONS

The great heritage, ecological and economic value of the *Posidonia oceanica* meadows, as well as the need to assess how effective the measures for protecting and managing coastal areas that have been implemented are, make the monitoring of meadows necessary. Today a wide range of monitoring tools is available.

These tools can be combined in various ways to constitute monitoring systems according to local objectives and situations, and are today widely used, particularly in the Provence-Alpes-Côte d'Azur region (France). There, the largest of these systems is the RSP (Posidonia Monitoring Network), which covers the entire coast of the region. Similar systems have been, or are now on the point of being, set up in other Mediterranean regions (Corsica, Spain, Algeria) (Semroud *et al.*, 1998; Boumaza and Semroud, 2000; Alvárez and Marbà, 2001).

Above and beyond the original aim of these monitoring systems, *Posidonia oceanica* is used as a biological indicator of the overall quality of the coastal environment (Pergent-Martini *et al.*, 1993; Pergent *et al.*, 1995; Pergent-Martini *et al.*, 1999; Boudouresque *et al.*, 2000) (see Chapter 17).

All in all, the *Posidonia oceanica*-based monitoring systems can provide policy-makers, local authorities and all coastal area managers with **effective**, relatively **cheap**, user-friendly tools to measure both the health of the meadows and at the same time of the coastal environment.

¹²⁹ Contact www.cram.es.

¹³⁰ Contact ecologic@dip-alicante.es and www.dip-alicante.es/IEL.

¹³¹ Contact ealvarez@dgpesca.caib.es.

17. POSIDONIA OCEANICA MEADOW AND THE WATER FRAMEWORK DIRECTIVE

17.1. SOME KEY ELEMENTS OF THE WATER FRAMEWORK DIRECTIVE

Adopted on 23 October 2000, and published in the European Communities' Official Journal of 22 December 2000, the Water Framework Directive (WFD) was transcribed into French law by the Law of 21 April 2004. A major text, which will structure water policy in each European Member State, this Directive commits the countries of the European Union to improving the state of water quality and aquatic environments. It is ambitious: aquatic environments (rivers, man-made lakes, lakes, groundwater, coastal water and transitional water) must reach good ecological status by 2015, except when technical or financial reasons explain why this aim cannot be achieved. To carry out this work successfully, the WFD advocates working on the scale of the hydrographic basins called "hydrographic districts", in this case the Rhône district and the Mediterranean as regards the Provence-Alpes-Côte d'Azur region and Corsica (France). It gives as the main deadlines the drafting of an assessment (late 2004) and of a management plan will state the objectives to be achieved by 2015. In France, the management plan will consist of a modified SDAGE (*Schéma Directeur d'Aménagement et de Gestion des Eaux* = Regional Planning Programme for Water Development and Management) plus a programme of measures to be defined by 2009.

The WFD confirms and enhances principles of water management that have been verified and tested: management by drainage basin (catchment area), balanced management of the water resource, and participation by stakeholders. But it goes further, introducing 3 major innovations: setting out environmental targets, taking socio-economic factors into account, and public participation.



17.1.1. A major innovation of the WFD: targets for all aquatic environments

It is no longer a question of "doing better" but of acting to achieve good ecological status by 2015, or else explaining why this objective of "good ecological status" cannot be achieved. From this simple objective a certain number of logical consequences derive, such as: the need to take coastal planning and economic data into account when setting out relevant objectives, affirming the principle of non-deterioration of water resources, defining specific strategies that tackle, for example, the fight against chemical pollution or the protection of

groundwater.

For surface water, "good ecological status" consists of (Fig. 143):

- "good chemical status" of the water, this being assessed using other active Directive (bathing, shellfish, drinking water directives etc.),
- "good (or high) ecological status", assessed on the basis of biological quality elements.

Today, systems to assess water quality and to formulate objectives to be achieved vary greatly from one country to the next within the European Union. The Directive thus requires the development of a common framework for the assessment of water quality, in order to have an objective view of the assessments and strategies in the various member states. This European reference framework has been developed over the years 2003 to 2006.

Exceptions will however be possible, to objectives that are less ambitious than that of "good ecological status by 2015", whether in terms of the final date (postponing the objectives to 2021, 2027) or in terms of level of objectives. Such exceptions must be justified by:

- financial reasons (estimation of disproportionate cost)
- natural conditions (e.g., time taken for pollutants to migrate)
- specific technical reasons (e.g. very big discharge into a little river with low dilution ability), to take into account existing uses that cannot be challenged and whose impact is such that the good ecological status of the environmental objective is technically unattainable. This takes up the idea of "highly modified water bodies" for which the objective will be adapted (the idea of "potentially good") owing to heavy physical development, linked, for example, to navigation, to certain hydroelectric facilities, or to the crossing of certain urban areas, to bear in mind the origin of the environment itself, with the idea of "artificial water bodies" for which the objective will also be adapted by defining the "potentially good".

17.1.2. The WFD's analysis unit: the waterbody

The unit for analyzing whether or not the WFD's objectives have been achieved is the waterbody. A waterbody is a stretch of a water course, or a lake, a pond, a section of coastal water, all or part of one or several aquifers of sufficient size bearing in mind the homogeneous biological and physicochemical features.

From both the qualitative and the quantitative point of view, waterbodies can thus be the subject of a **well-determined management objective**. Thus, when one is on a mountain torrent, a water course on the plain, a Mediterranean river, a lake, or the coast, the condition of the environment will not be characterized by the same (particularly biological) indicators. The waterbodies consequently correspond to a type of environment on a scale for which a homogeneous objective may be set and monitored according to a given indicator: "objective good ecological status 2015," "potentially good 2015" or "good ecological status 2021".

17.1.3. Good ecological status and coastal waterbodies

To assess the good status of waterbodies, the WFD requires that **monitoring programmes** are implemented for the ensemble of aquatic environments concerned. This monitoring must be based on "biological quality elements" (BQE). For coastal water, the BQE recommended in the Mediterranean by the WFD are: phytoplankton, "macroalgae", magnoliophyta (of which the *Posidonia oceanica* meadow), and benthic invertebrates. For each of these elements, work is in progress to assess their relevance in the WFD's future monitoring programme.

17.2. THE POSIDONIA MEADOW AS A "BIOLOGICAL QUALITY ELEMENT"

Given the importance of the Posidonia meadows in the Mediterranean (see Chapter 3) and the many studies that have been done on them, it is quite natural that they have been selected as one of the biological quality elements, able to give information on the ecological status of coastal waterbodies, and more particularly on the soft bottoms. However, using the meadow as a biological quality element requires perfecting a **Comprehensive Posidonia Index** able to characterize the general condition of the ecosystem. Although it is not yet possible to present this index, **applicable by all** the EU countries of the Mediterranean, since the approach is currently ongoing, several elements may already be acknowledged:

(1) The index will necessarily be based on several parameters. The available parameters are numerous, and many are already being used to assess meadow health (Fig. 144; Buia *et al.*, 2004; Romero *et al.*, 2005; Pergent-Martini *et al.*, 2005). Thus, the parameters most widely used at present are:

- shoot density, which expresses the number of shoots per unit of surface area (see §16.2.3)
- depth of the lower limit, which gives information on the general transparency of the water and its evolution over time



Fig. 144. Parameters used to assess the health of a *Posidonia oceanica* meadow and percentage of use (answers from 25 research institutes), criteria measured and methods of acquisition 1) Lefèvre *et al.*, 1984 ; 2) Pasqualini *et al.*, 1997a ; 3) Mc Kenzie *et al.*, 2001 ; 4) Pasqualini *et al.*, 2005 ; 5) Balduzzi *et al.*, 1981 ; 6) Cinelli *et al.*, 1984 ; 7) Morri, 1991 ; 8) Buia *et al.*, 2003 ; 9) Giraud, 1977b ; 10) Giraud, 1979 ; 11) Drew and Jupp, 1976 ; 12) Romero *et al.*, 2005 ; 13) Blanc, 1956 ; 14) Clarefond and Jeudy De Grissac, 1979 ; 15) Willsie, 1987 ; 16) Pergent *et al.*, 1995 ; 17) Pergent, 1990a ; 18) Duarte, 1991 ; 19) Cebrian *et al.*, 1994 ; 20) Mateo *et al.*, 1997 ; 21) Pergent *et al.*, 1989 ; 22) Panayotidis *et al.*, 1981 ; 23) Romero, 1986 ; 24) Duarte and Kirkman, 2003 ; 25) Pergent-Martini *et al.*, 1999 ; 26) Ramos-Martos and Ramos-Espla, 1989 ; 27) Pasqualini *et al.*, 2000 ; 28) Blanc-Vernet, 1984 ; 29) Russo and Vinci, 1991 ; 30) Harmelin-Vivien and Francour, 1992 ; 31) Hamoutene *et al.*, 1995 ; 32) Ferrat *et al.*, 2002 ; 33) Mateo and Sabaté, 1993 ; 34) Gobert *et al.*, 1995.



- the upper limit, which expresses more specifically the impact of human activity on the coast (coastal development; also see Chapters 4 and 7)
- epiphytic cover, which gives information on nutrients richness
- foliar biometry and/or the structure of the "matte", which gives overall information on the environment.

(2) Parameters whose **response times are relatively low** (Table XXV) should be integrated, enabling their use to be envisaged for the WFD's surveillance monitoring programmes. Correctly interpreted, they enable a meadow's response to be identified after a change of environmental conditions. Generally speaking, descriptors corresponding to individuals react more quickly than descriptors that correspond to populations.

(3) The parameters selected may **differ from one member state to the next** insofar as the Comprehensive Posidonia Index provides, in a given situation, a comparable assessment of the ecological status of the environment. It is, nevertheless, desirable for a small number of common parameters to be identified. Three parameters were chosen to this effect at the Ispra meeting (WFD, MED-GIG, February 2005):

- shoot density
- percentage of plagiotropic rhizomes, i.e. percentage of rhizomes which grow horizontally. At the limit of the meadow, this parameter gives information on the ability to colonize new substrata; on the other hand, in a meadow, it expresses the presence of a degradation within it and attempts at recolonization
- shoot foliar surface, which integrates all the phenological variables (number and size of leaves).

(4) The parameters selected may differ according to the **type of monitoring** required by the Water Framework Directive: surveillance monitoring (medium field) or operational monitoring (close field). Thus, in the context of an operational monitoring, parameters will be selected to reflect the nature of the disturbance (e.g. nutrient enrichment /eutrophication, moorings; Table XXV), whereas in a surveillance monitoring, parameters giving information on the condition of the population should be included (Table XXV), while permitting comparisons between sites (at regional and national level) and overcoming bathymetric constraints (e.g. a parameter independent of depth or assessed at a previously set homogeneous depth, of for example 15 metres).

Experience acquired over many years as to the evolution over time of *Posidonia oceanica* meadows (see Chapter 16) shows that regression is always quicker than recolonization (Boudouresque, 2001). In this respect, the *P. oceanica* monitoring networks set up in several Mediterranean countries (Boudouresque *et al.*, 2000; Buia *et al.*, 2004; Romero *et al.*, 2005; Pergent *et al.*, in press) constitute an element of appraisal that can validate the most pertinent descriptors.

The WFD monitoring programmes must become operational in the year 2006. So must the Comprehensive Posidonia Index. As regards France, it will come under the general strategy of the acquisition of data on coastal water (Fig. 145). The years 2005 and 2006 will from now on be devoted to putting the final touches to this index, and also to the work of intercalibration with the countries bordering the Mediterranean, which will in the end enable comparisons to be made on the health of *Posidonia oceanica* meadows throughout the Mediterranean basin (the text, written in 2005, has not been updated).

Table XXV. Main parameters of the *Posidonia oceanica* meadow, main anthropogenic pressures able to modify them and average response time to an improvement or deterioration in environmental conditions.

Parameter	Information level	Main impact	Response time		
			Deterioration	Improvement	
Upper limit	Population	Coastal			
Bathymetric position		developments	Yearly	X-ten-year	
Lower limit	Population	Turbidity			
Bathymetric position			X-monthly	X-yearly	
Type of limit			Yearly	X-ten-year	
Density	Population	Turbidity Eutrophication	Yearly	Yearly	
Cover	Population	ppulation Trawling & dredging Mooring		X-yearly	
"Matte" structure % of plagiotropic rhizomes Compactness	Population	Coastal developments Dredging discharge	X-yearly X-yearly	X-yearly X-ten-year	
Burial/baring Endofauna biodiversity	Mooring		Yearly X-monthly	X-yearly X-yearly	
Associated species	Population	Overfishing	ested areas areas atta		
Ichthyological populations		Invasive species	X-monthly	X-yearly	
Epiphytic cover	Individual	Eutrophication	Monthly	X-monthly	
Leaf biometry Leaf surface area % of necrosis Coefficient A	Individual	Eutrophication Overfishing	Yearly X-monthly	Yearly Yearly Yearly	
Dating measurements Number of leaves produced/year Rate of rhizome growth/year	Individual	Eutrophication Coastal developments Dredging discharge	Yearly Yearly	Yearly Yearly	
Contaminants	Individual	Urban sewage outfall Industrial waste	Monthly	X-monthly	
Chemical and biochemical composition Carbohydrates and CNP content Phenols and stress enzymes	Individual	Eutrophication Urban sewage outfalls Industrial waste Invasive species	X-monthly Monthly	X-monthly Monthly	



of the environment. There is thus symbiosis between these three aspects.

It is regrettable that the idea of sustainable development is very often **misinterpreted** by ecologists, sociologists and policy makers. (i) Ecologists, for whom nature takes precedence over man; (ii) sociologists, who do not accept that animals can be protected while human beings are threatened, and (iii) certain politicians who only tolerate protection of nature if this does not interfere with their naïve, archaic and very (very) short-term idea of economic development.

In order to make it easier for readers to refer back to the specialist chapters that make up this work, the **paragraphs** in this chapter **bear the same number** as the chapters where the information is laid out in detailed form: the information corresponding to paragraph 18.1 is found in chapter 1, the information in paragraph 18.2 is in chapiter 2, and so on and so forth (even if the titles of the chapters differ slightly).

Each time it was not necessary to change the text to make the meaning clearer, we have used the same words, and sometimes the same phrases, as in the corresponding chapter. **Repetitions** are thus deliberate. When we were unable to avoid using a technical term, we gave its definition again, even if this had already appeared in the specialist chapters

Beyond the regions in the RAMOGE area (Provence-Alpes-Côte d'Azur Region, Liguria, Monaco), our recommendations are addressed to the whole of the **Mediterranean**. Readers from a country or region in the RAMOGE area should therefore not be surprised at recommendations that can already be part of the national law of their country or region.

These recommendations are the fruit of the collective experience of the authors and of the existing **literature** (more than a thousand references) on Posidonia. Although it was necessary to take them into account, with the aim of protecting and conserving the Posidonia meadows, they do not commit the RAMOGE Agreement. Furthermore, they are not a substitute for the regulatory texts which may exist in Mediterranean countries or regions.

18. THE POSIDONIA OCEANICA MEADOW. A SUMMARY

This chapter addresses **non-specialist** managers and decision-makers as well as policy makers. It offers them a quick read free from scientific references or explanations. The recommendations appearing in this chapter are "**robust**," i.e. it is extremely unlikely that current research will lead to them being modified. The manager or representative who wishes to make a decision, one that is obviously in compliance with the regulatory texts in force but is also the best possible one for the underwater heritage and for the economy, and thus falls within a **sustainable development** approach, will be taking no risks in following the recommendations that we propose.

Sustainable development – don't let its meaning be abused!

Sustainable development involves 3 inseparable elements – **protection** of the environment, **economic development** and **social justice**. There can be no sustainable economic development without protection of the environment, no protection of the environment without economic development and social justice, and no social justice without economic development and protection

18.1. INTRODUCTION: WHY BE INTERESTED IN POSIDONIA ?

The seagrass *Posidonia oceanica* and the meadows it constitutes have over the past decades become a major focus for protecting and **managing** the marine environment throughout the Mediterranean.

Indeed, the Posidonia meadows are an element that is fundamental for the **quality of the** coastal **environments** on which artisanal fishing and tourism development depend. Because of its socio- economic importance and the contribution it makes to maintaining the balance of trade, tourism is a key element that no Mediterranean country can do without (about 10% of GDP (Gross Domestic Product) of the Mediterranean states). artisanal fishing, whose economic importance is more modest, has a major social and cultural dimension, with positive repercussions on tourism.

The protection and conservation of the Posidonia meadows is thus justified not only by their very great heritage value but also for social and **economic** reasons. It thus illustrates the idea of sustainable development (see insert).

18.2. POSIDONIA AND THE MEADOWS

475 million years ago, life (until then restricted to the marine environment) started to conquer the continents. Evolution then speeded up, giving first of all mosses and then ferns and lastly the plants with flowers and roots that we know. A little over **100 million years** ago, terrestrial flowering plants that were like today's rushes returned to the marine environment, while retaining the "technology" (and thus superiority) they had acquired on the continents: flowers, roots etc. These were Posidonia's ancestors.

A few dozen million years later, terrestrial **mammals** took the same route; they became dolphins and whales. They also retained the "technology" they had acquired on the continents: lungs, warm blood, etc. There is as much difference between a Caulerpa and a Posidonia as between a fish and a dolphin.

Posidonia is present in almost the whole Mediterranean, from east to west and from north to south, and is **only** present in the Mediterranean. Posidonia can thus be seen as emblematic of this sea, in the same way the olive tree is of its shores (Fig. 146 and 147).

Like the oak and the olive tree, Posidonia can live a very long time: more (and even a lot more) than a thousand years. As with the oak and the olive, growth is very slow. Its **superiority** to other species is therefore not manifest in the short term (the time-scale of a human life) but on a scale of centuries. But Posidonia's special biological characteristics, which guaranteed its success for millions of years, explain its **vulnerability**, its fragility, when faced by the very rapid change and disturbances that are a feature of the present decades.

Posidonia grows between the surface and 20-40 metres depth. As a photosynthetic¹³² plant, the

132 Photosynthetic it manufactures organic matter out of carbon dioxide and mineral salts, using the energy of light. maximum depth at which it can develop depends on the transparency of the water. It dislikes lack of salt (and therefore does not live



near river mouths). The density of its leaves allows it to trap sediment. It resists being buried

under sediment by the vertical growth of its rhizomes¹³³. Thus a matte is built up, consisting of rhizomes, roots and the sediment that fills the interstices (Fig. 148).

Over the centuries, the "matte" thickens and the meadow then grows closer to the surface of the sea. At the back of sheltered bays, it may reach the surface. Thus are formed what are called barrier reefs. The most typical Mediterranean **barrier reefs** are those found in Port-Cros and the Brusc (Var, France). Almost 10 000 years were needed to build them up. Their destruction would therefore be irreversible on a human scale.

The Posidonia meadow is not present as a uniform facies in a given region, and of course on the Mediterranean scale. As well as barrier reefs there are plain meadows, hill meadows, striped meadows, staircase meadows, etc. As with the barrier reefs, it took thousands of years to build them up, and their destruction is **irreversible**.

The Posionia meadow presents a **biomass**¹³⁴ that is exceptionally high for the marine environment. Its primary production¹³⁵ is one of the highest in the world (terrestrial and marine environments taken together). A large part of this primary production is **exported** in the form of dead leaves to other types of seabed (for example, beds at several hundred metres depth), where it constitutes a vital alimentary resource.



Fig. 146. Olive tree. Photo by S. Ruitton.



Fig. 147. Posidonia. Photo by A. Meinesz.

Furthermore, the meadow is host to 25% of species present in the Mediterranean (**biodiversity**). The importance of the Posidonia meadows is thus far greater than the surface areas (modest, on a Mediterranean scale) they occupy.

18.3. THE ROLE OF THE POSIDONIA MEADOWS

In the Mediterranean, the Posidonia meadow has a role that is often compared to that of the **forest** in a land environment, but that actually goes far beyond this.

(1) The meadow acts as a refuge for a quarter of the (flora and fauna) species that live in the Mediterranean, which is impressive when we consider that it covers less than 1% of Mediterranean seabed. Like coral reefs and the Amazonian forest, it therefore constitutes a **biodiversity hot spot**.

(2) The meadow produces enormous quantities of vegetal matter. This vegetal matter feeds the

rich fauna it shelters (Fig. 149). Furthermore, a great deal (about **40%**) of this matter is **exported** as dead leaves to other types of seabed. This entry of organic matter is a boon to

133 Rhizome=underground stem.

¹³⁴ Biomass the mass of living matter per square metre.

¹³⁵ Primary production production of living matter by photosynthetic plants from carbon dioxide and mineral salts, using the energy of light.



Fig. 148. A Posidonia meadow. Rhizomes (underground stems), shoots and "matte" (made up of rhizomes, roots and sediment that fills in the interstices) can be seen. From Boudouresque and Meinesz (1982).

the organisms that live at depths of more than 50-100 metres. Indeed, at such depths there is little (or no) light, and thus little (or no) photosynthesis. So the organisms that live there depend on organic matter that comes from elsewhere, particularly dead Posidonia leaves. The **fish** that are caught on the seabed out at sea are thus in a certain way the product of Posidonia. **By protecting Posidonia**, **we are defending the fishermen**.

(3) The meadow constitutes a **spawning ground** (egg-laying) area or a **nursery** (place where juveniles grow up) for many species of fish and crustaceans of economic interest. This role is easily understood when we notice that the thousands of long Posidonia leaves (Fig. 150) per square metre on the seabed constitute an inviolable refuge from predators. All

fishermen know that without spawning areas and nurseries there can be no adult fishes.

(4) By means of photosynthesis, the Posidonia meadow produces **oxygen** and is thus an important factor in oxygenating the water. At 10 metres depth, 1 m² of meadow releases up to 14 litres of oxygen per day.

(5) The Posidonia meadow traps and stabilizes **sediment**, as Marram grass (European Beachgrass) does on the sand dunes of the Atlantic coast. It thus stops it moving around during storms. By preventing sediment from re-suspended, it helps to keep the water **transparent**.



Fig. 149. The Posidonia meadow part the leaves a little and you will discover an oasis of living things. Photo by G. Pergent.

(6) The Posidonia meadow with its maze of leaves reduces **hydrodynamism** (swell, current) not only under the blanket of leaves, which is easily understood, but also in the water column. This is how it cushions the force of the waves on the coast and protects the **beaches** from erosion. By protecting the meadow, we protect the beaches.

(7) Before reaching deep seabeds, the dead Posidonia leaves often pile up on beaches in **banquettes** (see Fig. 27 and 33). These banquettes also protect the beaches from erosion during the autumn and winter storm.

(8) By its presence, state of health, or absence, the Posidonia meadow indicates the average quality of the water in which it is bathed all year round. It is thus an **efficient**

tool for monitoring the overall quality of the coastal water and the coastal environments.



All in all, the **economic value** of the meadows is considerable. Readers will perhaps be amazed to learn that, according to internationally recognized economists, this is 3 times greater than that of the coral reefs, 10 times greater than that of the tropical forests and 100 times greater than that of a terrestrial meadow.

18.4. REGRESSION OF THE POSIDONIA MEADOWS

During the 20th century, especially since the 1950s, the Posidonia meadow has regressed considerably, especially around the great industrial-port centres: Barcelona, Marseille, Toulon, Nice, Genoa, Naples, etc. This regression may be up to 90% in the case of the Prado Bay in Marseille. **Human activity** is clearly the cause of this regression.

The regression concerns the lower limit, which very logically rises because of the drop in water transparency, and also the upper limit and intermediate depths. First of all this is expressed in a reduced shoot density and the formation (or extension) of intermattes¹³⁸. The meadows' regression is due to many factors, especially when they act together:



Fig. 150. The damselfish Chromis chromis above a Posidonia oceanica meadow. Port-Cros National Park. Photo by S. Ruitton.

(1) Being covered under coastal development (land reclaimed from the sea=reclamation).

(2) Modification of the **sedimentary flow**. Developments on river catchment areas reduce the entry of sediment into the marine environment. Coastal facilities (ports, rocky groynes) hamper the lateral transfer of sediment along the beaches. The result is that the meadow can either be bared because of a sedimentary deficit (the rhizomes lie above the sediment) or buried under a layer of sediment that the shoots are unable to pierce. In the first case, the meadow is very vulnerable (anchorage, trawling, storms) and in the second it is destroyed.

(3) A reduction in water **transparency**. This may be due to sedimentary particles brought by rivers, to waste water, and to the sediment being re-suspended by hydrodynamism. It can also be due to the development of plankton in water rich in nutrients (nitrogen, phosphorus).

(4) Anchoring. Obviously in areas where there is very frequent anchoring, an impact is perceptible.

(5) **Trawling**. More or less forbidden according to depth, distance from the coast, type of towed gear and country, but everywhere tolerated by the authorities. Trawling is one of the main causes of the meadows' regression, especially at important depth.

(6) **Pollution**. Whether this is due to urban discharge, leisure boats or fish farms, pollution has a direct or indirect negative impact on Posidonia.

136 Intermatte a patch of sand or of "dead matte" (Posidonia rhizomes that have lost their leaves) inside the meadow. (7) Competition with **invasive species** is a problem whose importance is currently growing. It will take some decades to know the outcome of the competition between Posidonia and various Caulerpa species and other introduced Macrophytes¹³⁷. The precautionary principle must thus be applied here.

(8) Overgrazing by sea urchins or herbivorous fish. This may be due to the fact that fishing has eliminated the fish that used to be predators of sea urchins. It may also be due to pollution, which encourages sea urchins. Moreover, nitrogen pollution increases the nitrogen content of the Posidonia leaves and makes them more attractive to herbivores.

In most cases, the regression of the Posidonia meadow has no single cause but is the result of the accumulation (**synergy**) of several causes. And it is important to stress the fact that unlike other disturbances, which are reversible on a human scale¹³⁸, the disappearance of a Posidonia meadow is **irreversible**: natural recolonization takes centuries.

18.5. THE REGULATORY TEXTS THAT APPLY TO THE MEADOWS

The importance of the Posidonia meadows from both the **ecological** (protection of biodiversity) and the **economic** (fishing, tourism, etc.) points of view has led the national, European and/or international authorities to protect them. There are direct and indirect protection measures.

Posidonia (and the meadows it forms) is **directly** protected by international conventions ratified by most of the Mediterranean countries, in particular the countries of the RAMOGE area: the **Bern** Convention (on the conservation of European wildlife and natural habitats), the **Barcelona** Convention (on the protection of the marine environment and the coastal region of the Mediterranean) and the European Union's (1992) **Habitats Directive**. It is also protected at national or regional level in France, Liguria (Italy), Catalonia, the Comunitat Valenciana (Spain) and Slovenia. It is important to note that in France, protecting Posidonia involves banning the destruction, removal and transporting of Posidonia, or of parts of Posidonia, both living and dead. That means that the widely practiced removal of dead leaves from the beaches is **illegal**.

Many **indirect** measures also protect the Posidonia meadows: Marine Protected Areas, measures intended to curb pollutant discharge and to restrict certain fishing techniques such as towed gear, and the obligation to carry out an **impact assessment** before making any request for a permit for a project that could harm the environment.

18.6. DEAD POSIDONIA LEAVES, BEACHES AND BEACH NOURISHMENT

Dead Posidonia leaves are driven by storms and currents either out to the great depths or onto the beaches. There they form **banquettes** that can be 2 metres thick (in exceptional cases). These

137 Macrophytes=multicellular plant, i.e. big ones.

138 This is the case for most types of pollution, including oil spills, which are (rightly) very much subject of mediatisation but are reversible in a few years.



banquettes **protect the beaches from erosion**. But ill-informed tourists do not appreciate this, so much so that many regional authorities (administrative districts, *collectivités territoriales*) see these dead leaves as waste and get rid of them, in the same way that they would plastic bottles and cigarette butts.

The **reduction of beaches** is a complex phenomenon, with many causes. The regression of the meadows (which cushion the force of the waves and swell) and the elimination of dead Posidonia leaves on the beaches are part of the cause. Adding more sand on the beaches to counterbalance the fact that they are decreasing may accentuate the regression of the meadows and constitute a kind of "vicious circle". The decreasing beach problem is an example of the economic superiority of **integrated management** of a problem (decreasing beaches, dead leaves on beaches, beach nourishment and protection of meadows) compared to a case-by-case approach (the beach is decreasing so it is necessary to add more sand).

As well as the necessary protection (or reconstitution) of the **dunes** behind the beach, which does not specifically concern Posidonia, and the checking of the causes of the meadows' regression, the most sustainable solution is **not to remove the dead Posidonia leaves from the beaches**. Contrary to what one might think, bathers accept this perfectly well as long as it is explained to them. The success of "**ecological beaches**", whose number is rapidly growing, is there to prove this.

As for **beach nourishment**, if this is really necessary and if it is part of overall management of the decreasing beach problem, it must be done with the appropriate sediment (see §6.3) and at a distance of at least **300 metres** from the nearest meadows.

18.7. COASTAL DEVELOPMENT IN THE MARITIME PUBLIC DOMAIN

Shallow seabeds have a major role in the Mediterranean: since light is not a restrictive factor there, plant production is maximal. Moreover, **nurseries** for many species of fish of economic interest are located there. Their destruction by covering by coastal development (land reclamation -land gained from the sea-, ports, etc.) is **irreversible** on a human scale. As well as meadows possibly being buried under a reclamation or included inside a port, these facilities have often led to the **indirect** destruction of areas of meadow that are much larger than that of the facility itself.

No coastal development must be carried out over a meadow. Meadows included in a **port** have very little chance of survival in the long term, and the few examples of patches of Posidonia surviving in ports should not hide what is generally the case. Facilities (rocky groynes, ports, land reclamation) have a great impact on the meadows, even if they are not located directly above them: they can in fact modify sedimentary flow either by a sedimentary deficit or by an excess of sedimentation. In any case, **10 metres** is the minimum distance that must be respected between rip-rap and the nearest living Posidonia.

The indirect destruction of the meadows by coastal development is partly linked to the **construction techniques**. In order to minimize this impact, companies that are the beneficiaries of tenders should be subjected to a number of constraints. The choice of company must not be

made systematically to favour the lowest bidder (the cheapest) but the **best bidder** (the most credible enterprise as regards respect for environment protection regulations). When the development work is being done, there should be checks that the contract specifications in this matter¹³⁹ are being respected.

Dumping **fine material** (diameter less than 1 mm), or slabs mixed with fine material, at sea is to be totally ruled out. **Geotextile** protection screens must be set up around the building site, to minimize the turbidity caused. Site equipment must work **from the land**, not from the sea; if it is essential to use equipment at sea, it must not anchor over or be supported by Posidonia meadows. Finally, **summer** is the season to be avoided because of Posidonia's biological characteristics.

18.8. MOORING

Here we draw a distinction between moorings using anchors (anchoring), organized mooring (using deadweight moorings or some other system of fixed mooring legally provided) and unauthorized mooring (using deadweight moorings without permission).

Organized mooring using deadweight moorings causes the Posidonia meadows much more

Fig. 151. The negative impact of a deadweight mooring and its chain on the Posidonia meadow off Aygade port, the Levant Island (Var, France). Here the deadweight mooring is not linked to a mooring buoy but to a buoy that marks the entrance to the port. The entire centre of the photo shows a vast area of "dead matte" that has been opened up in the meadow by the movement of the chain. Photo by E. Charbonnel.



harm that anchors (Fig. 151). Therefore it must systematically be avoided. When providing organized mooring, meadow areas should be avoided. If this is not possible, it is indispensable that "ecological" mooring systems be used (see § 8.2.2) that have no impact on the meadow, with an intermediate buoy that prevents the mooring cable making contact with the seabed.

The impact anchors have on the meadow depends on the **type** of anchor used (Hall-type anchors are less harmful), the **size** of the boat (cruisers have a greater impact than little leisure boats), and the **practice** of removing the anchor (it is preferable that the boat first be stationed above the anchor, which is then lifted vertically).

Be that as it may, and subject to optimizing anchoring techniques, the anchoring of little leisure boats certainly represents a lesser threat to the meadow than trawling, pollution, and of course coastal development. When it is less than 2 boats/day/hectare (on average) or 10 boats/hectare (at peak periods), it is not necessary to provide organized mooring, except in Marine Protected Areas or sectors of heritage importance.

139 It is shocking that it is often individuals or NGOs (non-governmental organisations) who warn the public departments or the regional authorites (administrative districts, collectivités territoriales) that a building site is not respecting environmental clauses.

18.9. MARKING THE 300-METRE ZONE

During the touristic season, markers are set up 300 metres from the shore to protect **bathing** areas. Yellow buoys are linked to a deadweight mooring; as in mooring, when the deadweight mooring is located in a meadow both it and the chain that links it to the buoy **erode** the meadow (Fig. 151). Moreover, at the end of the touristic season the deadweight mooring is removed, and the following year it is not sunk in exactly the same place, so that there are increasingly more patches where the meadow is degraded.

We recommend (i) not removing deadweight moorings at the end of the touristic season; the precision of the (GPS or seamark) locating systems is such that they can easily be found the following year; (ii) using an intermediate buoy that prevents the chain eroding the meadow; (iii) if possible, replacing the deadweight mooring by an "ecological" mooring system.

18.10. TRAWLING

Fishing with towed gear is one of the main causes of the Posidonia meadows' regression at depth. Also, it harms the meadow's role of nursery for many species of fish of commercial interest. Lastly, it gives rise to **user conflicts**, between artisanal fishermen (small-scale fishery) and trawlers. Indeed, the trawlers do not usually respect the national laws (minimum depth, minimum distance from the coast) that normally ban trawling over meadows.

Laying down **anti-trawl reefs** is a solution to this problem: they dissuade the trawlers, which run the risk of damaging their equipment. We make some recommendations: (i) anti-trawl reef units must be sufficiently heavy (at least 8 t) to constitute an effective physical obstacle and not be dragged off by the trawl or harmed by the trawl's side panels; (ii) they must offer a sufficient carrying surface to grip the sediment and not sink into it; (iii) it is also important that their shape should not harm the nets of artisanal fishermen, who must be able to work in the improved areas and benefit from the anti-trawl reefs; (iv) anti-trawl units must be sunk separately so that they are spaced out (50 to 200 metres between units); (v) anti-trawl reef ensembles must occupy the maximum space possible to be really dissuasive to trawlers; (vi) if the topography and surface area of the site to be protected permit this, the units should be placed in straight lines running out from the coast (most trawlers tow their gear parallel to the coast), thus forming a series of barriers.

18.11. FISH FARMS

The impact of fish farms (cages) is due to possibly uneaten **food**, fish' **excrement**, possibly the **antibiotics** and trace elements (copper, zinc) used, and lastly the **shade** shed by cages onto the seabed. The result is usually the entry into the environment of organic matter and nutrients and diminished lighting on the seabed. The impact of fish farms obviously depends on many factors, such as the type of food used, management of the food ration (minimizing or not minimizing losses), the density of fish, the size of the farm (tonnage annually produced) and obviously the local currents.



Fig. 152. "Dead matte", with some surviving *Posidonia oceanica* shoots, under a fish farm in Corsica. Photo by G. Pergent.

When fish farms have been set up above a Posidonia meadow, the meadow located under and near the cages is either greatly **degraded** or has actually disappeared, according to how long the farm has been operating (Fig. 152).

We therefore make the following recommendations: (i) no fish farming structure must be set up directly above a *Posidonia oceanica* **meadow**; (ii) if there is a meadow nearby, a **minimum distance** of 100 metres from the cages must be respected. This distance must possibly be increased, according to depth, currents and the size of the farm; (iii) generally speaking, a facility over seabeds **45-50 metres** deep must be preferred whenever possible; (iv) an **impact assessment** should accompany any request to set up a fish farm; (v) every four years, the permit to set up a fish farm should be reviewed for

possible continuation, depending on the demonstration that the *P. oceanica* meadows nearby have not regressed. This constraint, which involves a **monitoring of the meadows**, should lead fish farmers to move as far away from the meadows as possible.

18.12. DISCHARGE OF EFFLUENTS

Generally speaking, the discharge of effluents acts mainly at 3 levels on marine coastal settlements: (i) lessening the transparency of the water; (ii) increasing the nutrient content; (iii) input of chemical contaminants. It can also give rise to a local reduction in salinity that can be harmful to Posidonia, in that the species does not like low salinity water.

When considering the impact of effluent on the meadows, it is not easy to separate the **direct** effects, such as toxicity and low salinity, from the **indirect** effects, due to the input of nutrients, such as the developing of leaf epibiota¹⁴⁰ on the leaves and the increase of grazing by herbivores. In any case, throughout the Mediterranean the meadow has disappeared, sometimes at great distances, around sewage outfalls.

We therefore make the following recommendations: (i) no new sewage outfall should open out into a meadow. This holds good whatever the level of treatment of the water; (ii) a minimum distance must be kept between the point of release and the nearest meadow. This distance varies according to the volume of water released and the level of treatment; (iii) the underwater pipe that carries the waste water should not cross the meadows, or should reduce to a minimum the length of meadow crossed; (iv) the nearest meadows must be monitored (by marking, permanent quadrats) both for new sewage outfalls and for older ones, in order to check that the level of treatment of the water is sufficient; (v) in the case of old sewage outfalls, if monitoring the meadow shows that the situation has stabilized, and especially if there has been a beginning of recuperation by the meadow (due to improved treatment of the water), we do not recommend moving the pipe or extending it beyond the limits of the meadow.

140 Leaf epibiota are organisms that develop on the leaves. They intercept the light and therefore harm Posidonia, which needs light for photosynthesis.



18.13. SOLID WASTE

Discharging silt (from the dredging of a port or canal) or slabs of rock (from the removal of rocks) onto a Posidonia meadow destroys it, at the place where it happens, irreversibly. **All dumping over the meadow should therefore be prohibited**.

The problem is that companies that have won tenders do not always respect the contract specifications when these provide for discharge outside the meadow (also see § 18.7). It is indeed economically advantageous for these companies to dump solid waste as near as possible to the site where the dredging or removal of rocks is being done.

It is shocking that it is often individuals or NGOs who warn the public departments or the district authorities (*collectivités territoriales*) that the contract clauses are not being respected. We recommend that the latter show greater vigilance.

18.14. LAYING CABLES AND PIPES ON THE SEABED

To provide an island with electricity or water, it may be necessary to lay a cable or pipe on the seabed. The impact on the Posidonia meadow is modest or even nil when the cables or pipes are merely laid over the meadow. But when they are **buried** (digging a trench), the impact is very great.

The ideal would be to avoid the cables and pipes crossing the meadows, but this is neither realistic from an economic point of view nor possible in many cases. We therefore recommend: (i) that the Contracting Authority suggest a minimum 3 points of **departure** from and/or **arrival** on land; (ii) that a precise **map** be made of the nature of the seabed (rock, sand, mud, meadows); (iii) that the selected route be the best possible compromise between the total **length** of the route (as short as possible from the economic point of view) and the length of meadow crossed (as short as possible from an ecological point of view). An assessment grid of the various scenarios is proposed (see §14.3); (iv) that there be no trench burial but that the cable or pipe be simply **laid** on the seabed, with clamping when necessary; (v) that impact **monitoring** be provided for (after 2, 5 and 10 years) in order to validate (or not) the selected scenario and to improve the future management of cables and pipes.

18.15. CAN WE RESTORE MEADOWS THAT HAVE BEEN DESTROYED?

Natural recolonization by Posidonia meadows, when the causes of their destruction have ceased to operate, is very slow, so it is tempting to try to speed up this recolonization by **transplanting**, as is done in the continental environment (reafforestation).

A certain number of **techniques** have been perfected using cuttings or seeds: cement frames in the centre of which are placed cuttings held in by wire mesh, stakes or hooks that fix cuttings directly onto the bed, "matte" clumps, etc. Optimum methods of transplanting (season, origin and length of cuttings, etc.) have also been defined. However, transplanting Posidonia suffers from the same handicap as natural recolonization: the plant's extremely slow growth. It takes decades to truly judge the possible success of the transplanting experiments that have been carried out in the Mediterranean. Significant successes have been obtained in Japan and the US, but these concern another species, the quickly-growing eelgrass *Zostera marina*. There is a serious risk that the technical possibility of transplanting and seeding Posidonia will be misappropriated **to serve as an excuse for new destruction**. This destruction is immediate and irreversible, whereas the possible success of the "compensatory" transplanting can only be judged very long after it has been done. Moreover, there are several examples of "planting for planting's sake", with no overall strategy, done at the whim of requests by locally elected politicians. (i) Posidonia has been planted in sectors where it does not exist naturally and seems never to have existed. What could justify trying to replace a sandy bed (which is in no way a biological desert, though the public is often unaware of this) by a few clumps of Posidonia? (ii) It has been planted in areas where the regression of the bed is ongoing. (iii) It has been planted in a Cymodocea meadow (a protected species, like Posidonia). Trying to replace one protected species by another protected species is hardly a very consistent strategy.

Despite the very high **cost** of the transplanting, it can be envisaged. It must in this case be set within the framework of a comprehensive reflection on the integrated management of coastal environments, of a significantly wide regional scale with, particularly, a comparison between the cost/result ratio of transplanting and that of other operations to preserve or restore the quality of the coastal environments. To help in decision-making, we suggest a **decision-making strategy** (Fig. 126) and a **code of good conduct**, the main points of which are: (i) the exact site where the transplanting is done must have been formerly occupied by Posidonia; (ii) the causes resulting in Posidonia's disappearance must have ceased to operate; (iii) transplanting must not be done near very extensive meadows; (iv) transplanting cannot be done to compensate for the destruction of a meadow; (v) before any transplanting is done there must be experimental transplanting, with at least a 3-year monitoring, to verify its feasibility.

Fig. 153 . Laying a cement marker on the limit of a *Posidonia oceanica* meadow. The progression, stability or regression of the meadow compared to the cement marker position are then measured. Photo by E. Charbonnel.



18.16. MONITORING THE POSIDONIA MEADOWS

The aim of monitoring Posidonia meadows is threefold: (i) monitoring a settlement that has great **heritage value** but is vulnerable, to rapidly notice any new regression; (ii) using the meadow as a **biological indicator**¹⁴¹ of the overall quality of the coastal water; (iii) measuring the effectiveness of regional coastal environmental policies, for example starting waste water treatment plants, improving the level of water treatment, reducing the input of the domestic and industrial pollutants brought by rivers, and setting up Marine Protected Areas.

Tools for monitoring the meadows work on 3 spatial levels: the scale of the entire meadow, the local scale within a meadow, and the microscale, for example that of a shoot.

Tools that act on **the scale of the meadow** as a whole are mapping, the use of aerial photographs for some use other than mapping, measuring cover (percentage of surface area of the

141 A biological indicator is a living organism that informs us indirectly (by its presence, health or absence) about the quality of the environment. The advantage of biological indicators is that they integrate over the long term complex physicochemical parameters that act in synergy (e.g., pollutants) and that can fluctuate enormously from day to day.

seabed covered by the living meadow) and permanent transects. These are tracks in the meadow that are exactly located so that they can be revisited later to detect possible changes there.

Tools that act **on a local scale** are laying cement markers (Fig. 153) at the upper and/or lower limit of the meadow and permanent quadrats. Markers enable changes in the meadow's limit to be observed (by comparing with photos, counting shoots etc.) even if these changes are rather minor, and thus detect them early on. Permanent quadrats are little areas (a few dozen square metres) that are precisely located and mapped, and which can be revisited later to detect possible changes there.

Tools that act on a **microscale** (that of a shoot or group of shoots) are measuring shoot density, the percentage of creeping and erect shoots, baring of the rhizomes (which expresses a sedimentary deficit), granulometry of the sediment (mud, fine sand, coarse sand, etc.), lepidochronology (recording any series of events, like that of the growth rings in trees), leaf length, etc.

Combining these tools (or some of them) has enabled meadow **monitoring systems** to be set up, the oldest and best known of which is the Posidonia Monitoring Network (RSP in French) in the Provence-Alpes-Côte d'Azur region (France). These Posidonia-based monitoring systems can provide policy-makers, local authorities and all managers of coastal areas with effective tools that are relatively cheap and user-friendly to measure the state of health of the meadows as well as that of the coastal environment.

18.17. POSIDONIA AND THE WATER FRAMEWORK DIRECTIVE

The Water Framework Directive (WFD), adopted in 2000, is a major text that structures water policy in the member states of the European Union. Its aim is to **improve and recover the quality of water** and aquatic environments (rivers, lakes, groundwater, coastal water, etc.). The WFD sets out as main deadlines the development of an assessment (late 2004) and a management plan (by 2009). The aquatic environments must be "in good ecological status" by 2015. Dispensations, if they are justified, will however be possible, with less ambitious objectives than that of "good ecological status by 2015", whether this concerns deadlines (postponing objectives to 2021, 2027) or level of objectives.

The WFD confirms and enhances verified and tested principles of water management: management by drainage basin (catchment area), balanced management of the water resource, and participation by stakeholders. But it goes further, introducing 3 major innovations: setting out the **objectives of environmental results**, taking **socio-economic** considerations into account, and **public participation**.

"Good status" consists of (i) "good **chemical** condition" of the water, this being assessed in the light of active Directives (bathing, shellfish, drinking water, etc.), and (ii) "good (or high) **ecological** status", assessed according to biological criteria.

To assess the good ecological status of waterbodies, the WFD makes the implementation of **monitoring programmes** compulsory. This monitoring must be based on descriptors or "biological quality elements". For coastal waters, the descriptors recommended in the Mediterranean by the

WFD are: phytoplankton, "macroalgae", Posidonia meadows, benthos¹⁴² of soft bottom and benthos of hard bottom. For each of these elements work is being done to assess their relevance in the WFD's future monitoring system.

As for the Posidonia meadow, its use as a biological quality element requires the perfecting (now under way) of a **Comprehensive Posidonia Index**. This index will be developed using several parameters (see §18.16 and Chapter 16). The selected parameters can differ from one member state to the next, although it is desirable that some of them are held in common. The selected parameters can also differ according to the **type of monitoring** required by the Water Framework Directive: surveillance monitoring (average field) or operational monitoring (near field). Thus, in the context of an operational monitoring, recourse to a parameter will reflect the nature of the disturbance (e.g. enrichment in nutrient, mooring, etc.) while in the surveillance monitoring it will be necessary to include parameters that give information about the state of the population, while permitting comparisons to be made between sites (on a regional and national scale).

142 The benthos is made up of all the organisms that live on the seabed, in contrast to those that live in the body of water.

REFERENCES

- ABAL E.G., DENNISON W.C., 1996. Seagrass depth range and water quality in southern Moreton Bay, Queensland, Australia. Mar Freshwater Res., 47(6): 763-771.
- ABU-ZEID M., 1991. Le Nil et sa contribution à l'environnement méditerranéen. Rencontres de l Agence régionale pour l Environnement, Provence Alpes Côte d Azur, Fr., 3: 7-18.
- ABU-ZEID M., EL-MOATASSEM M., 1993. Qualité de l'eau du Nil avant et après la construction du barrage d'Assouan et les conséquences pour la Méditerranée. Rencontres de l'Agence régionale pour l'Environnement Provence Alpes Côte d'Azur, Fr., 5: 62-68.
- ACUNTO S., PIAZZI L., BALESTRI E., CINELLI F., 1996. Segnalazioni di fioriture di *Posidonia oceanica* (L.) Delile lungo le coste toscane. *Biol Mar Medit*, 3(1): 437-438.
- ADDY C.E., 1947a. Eelgrass planting guide. Maryland Conservationist, USA, 24: 16-17.
- ADDY C.E., 1947b. Germination of eelgrass seed. J. Widl. Manag., USA, 11: 279.
- ALCOVERRO T., DUARTE C.M., ROMERO J., 1997. The influence of herbivores on *Posidonia oceanica* epiphytes. Aquat Bot, 56(2): 93-104 ALCOVERRO T., MANZANERA M., ROMERO J., 1998. Seasonal and age-dependent variability of *Posidonia oceanica* (L.) Delile photosynthetic parameters. J exp. mar. Biol. Ecol., 230(1): 1-13.
- ALCOVERRO T., MANZANERA M., ROMERO J., 2001. Annual metabolic carbon balance of the seagrass *Posidonia oceanica*: the importance of carbohydrate reserves. *Mar Ecol Progr Ser*, 211: 105-116.
- ALEEM A.A., 1955. Structure and evolution of the sea grass communities *Posidonia* and *Cymodocea* in the southeastern Mediterranean. Essays in the natural Sciences in Honor of Captain Allan Hancock, Univ. California publ.: 279-298.
- ALLUÉ-PUYELO R., OLIVELLA-PRATS I., 1994. Actions pour la protection et la conservation du littoral catalan. Actes du colloque scientifique Okéanos Pour qui la Méditerranée au 21^{4me} siècle ? Villes des rivages et environnement littoral en Méditerranée Montpellier, Fr.: 196-208.
- ALVAREZ E., MARBA N., 2001. Red de monitorización de las praderas de Baleares. Guía de campo. Govern de les Illes Baleares: 1-34. AMI D., BOUDOURESQUE C.F., 2002. Valuing benefits from protecting the seagrass *Posidonia oceanica* beds in the Mediterranean Sea.
- Conference on Risk and Uncertainty in Environmental and Ressource Economics, Wagenngen University, The Netherlands: 1-7. ANONYMOUS, 1998. Protocole relatif aux aires spécialement protégées et à la diversité biologique en Méditerranée (Barcelone, 1995). RAC-SPA édit., PNUE publ.: 1-46.
- ANONYMOUS, 2000. Action plan for the conservation of marine vegetation in the Mediterranean sea. CAR-ASP édit., PNUE publ.: 1-8. ANONYMOUS, 2001a Projet de lignes directrices pour les études d impact sur les herbiers marins. PNUE. CAR/ASP édit., UNEP(DEC)/MED WG.177/8. Tunis: 1-49.
- ANONYMOUS, 2001b. Tableau de bord de l environnement en Corse. OEC & DIREN édit.
- ANONYMOUS, 2002a. Protection des ressources marines et côtières du golfe de Gabès. Rapport de Phase 1 : Diagnostic. République tunisienne Ministère de l'Environnement et de l'aménagement du territoire, SOGREAH, GIS Posidonie et UNEP PAP-CAR: 1-229.
- ANONYMOUS, 2002b. Etude de la pratique de la pêche aux ganguis et à la senne de plage en Région PACA. IDEE, CREOCEAN, Oceanic Development, CRPMEM PACA, Nice: 1-19 + Ann.
- ANONYMOUS, 2005a. Réseau de Surveillance Posidonies. Département des Bouches-du-Rhône. Année 2004. Région Provence-Alpes-Côte d'Azur, EOL, SMEL, Fr.: 1-155.
- ANONYMOUS, 2005b. Réseau de Surveillance Posidonies. Département du Var. Année 2004. Région Provence-Alpes-Côte d'Azur, EOL, SMEL, Fr.: 1-265.
- ANONYMOUS, 2005c. Réseau de Surveillance Posidonies. Département des Alpes-Maritimes. Année 2004. Région Provence-Alpes-Côte d'Azur, EOL, SMEL, Fr.: 1-142.
- ARDIZZONE G.D., MIGLIUOLO A., 1982. Modificazioni di una prateria di *Posidonia oceanica* (L.) Delile del Medio Tirreno sottoposta ad attività di pesca a strascico. *Naturalista Siciliano, S IV,* VI (suppl.), 3: 509-515.
- ARDIZZONE G.D., PELUSI P., 1984. Yield and damage evaluation of bottom trawling on *Posidonia* meadows. *In*: Boudouresque C.F., Jeudy de Grissac A., Olivier J. edits. *International Workshop on Posidonia oceanica beds*, GIS Posidonie publ., Fr., 1: 63-72.
- ARFI R., ARNOUX A., BELLAN G., BELLAN-SANTINI D., BOURCIER M., DUKAN S., DURBEC J.P., LAUBIER L., MARINOPOULOS J., MILLOT C., MOUTIN T., PATRITI G., PERGENT-MARTINI C., PETRENKO A., 2000a. Impact du grand émissaire de Marseille et de l'Huveaune détournée sur l'environnement marin de Cortiou. Etude bibliographique raisonnée 1960-2000. Rapp. COM & Ville de Marseille, COM publ.: 1-137.
- ARFI R., ARNOUX A., BELLAN-SANTINI D., BELLAN G., BOURCIER M., LAUBIER L., PERGENT-MARTINI C., DUKAN S., DURBEC J.P., MARINOPOULOS J., MILLOT C., MOUTIN T., PATRITI G., PETRENKO A., 2000b. Cortiou, Evolution d'un site marin soumis à un rejet urbain. Synthèse bibliographique 1960-2000. Rapp. COM & Ville de Marseille, COM publ.: 1-38.
- ARGYROU M., HADJICHRISTOPHOROU M., DEMETROPOULOS A., 1999. Ecological changes of softbottom macrobenthic assemblages in relation to the sewage outfall, in the Limassol bay, Cyprus (Eastern Mediterranean). *Oebalia*, 25: 61-88.
- ASTIER J.M., 1972. Régression de l'herbier de Posidonies en rade des Vignettes à Toulon. Ann Soc Sci nat Archéol Toulon Var, 24: 97-103.
- ASTIER J.M., 1984. Impact des aménagements littoraux de la rade de Toulon, liés aux techniques d'endigage, sur les herbiers à *Posidonia* oceanica. In: Boudouresque C.F., Jeudy de Grissac A., Olivier J. edits. International Workshop on Posidonia oceanica beds, GIS Posidonie publ., Fr., 1: 255-259.
- ASTIER J.M., CASTEL A., TAILLIEZ P., TINÉ J., 1980. L'aménagement des plages artificielles du Mourillon. Rev Fond océanogr Ricard, Fr., 4: 14-23.
- ASTIER J.M., TAILLIEZ P., 1984. Pour un plan d'occupation des fonds marins : inventaire des herbiers de Posidonies du littoral du Var. Ann Soc Sci nat Archéol Toulopn Var, 36: 35-46.
- AUBERT M., 1993. Impact de l'aquaculture sur les eaux côtières méditerranéennes. *In: Pour qui la Méditerranée au 21^{erre} Siècle Le système littoral méditerranéen.* Actes du colloque scientifique Okeanos, Maison de l'Environnement de Montpellier publ., Fr.: 33-47.
- AUBY P, 1998. Compte Rendu de la réunion de travail du 7/10/1998 sur le nettoyage des plages. Parc national de Port-Cros, Hyères, Fr.: 1-5. AUGIER H., 1985. L'herbier à *Posidonia oceanica*, son importance pour le littoral méditerranéen, sa valeur comme indicateur biologique de l'état de santé de la mer, son utilisation dans la surveillance du milieu, les bilans écologiques et les études d'impact. *Vie marine*, 7: 85-113.
- AUGIER H., 1969. Un pistolet de scellement "sous-marin". Bull Mus Hist nat Marseille, 29: 25-47.
- AUGIER H., BOUDOURESQUE C.F., 1967. Végétation marine de l'île de Port-Cros (Parc national). I. La baie de La Palud. Bull Mus Hist nat Marseille, 27: 93-124.

- AUGIER H., BOUDOURESQUE C.F., 1970a. Végétation marine de l'île de Port-Cros (Parc national). VI. Le récif-barrière de Posidonies. Bull Mus Hist nat Marseille, 30: 221-228 + 1 pl. h.t.
- AUGIER H., BOUDOURESQUE C.F., 1970b. Végétation marine de l'île de Port-Cros (Parc national). V. La baie de Port-Man et le problème de la régression de l'herbier de Posidonies. *Bull Mus Hist nat Marseille*, 30: 145-164 + 1 pl. h.t.
- AUGIER H., BOUDOURESQUE C.F., 1975. Dix ans de recherches dans la zone marine du Parc national de Port-Cros (France). Troisième partie. Ann Soc Sci nat Archéol Toulon Var, 27: 133-170.
- AUGIER H., BOUDOURESQUE C.F., 1979. Premières observations sur l'herbier de Posidonies et le détritique côtier de l'île du Levant (Méditerranée, France), à l'aide du "sous-marin" Griffon de la Marine Nationale. *Trav sci Parc nation Port Cros*, 5: 141-153.
- AUGIER H., GILLES G., RAMONDA G., 1984a. L'herbier de Posidonia oceanica et la pollution par le mercure sur le littoral des Bouchesdu-Rhône et du Var (France). In: Boudouresque C.F., Jeudy de Grissac A., Olivier J. edits. International Workshop on Posidonia oceanica beds, GIS Posidonie publ., Fr., 1: 399-406.
- AUGIER H., MAUDINAS B., 1979. Influence of the pollution on the photosynthetic pigments of the marine Phanerogam *Posidonia oceanica* collected from different polluted areas of the Region of Marseille (Mediterranean sea, France). *Oecologia Plantarum*, 14(2): 169-176.
- AUGIER H., MONNIER-BESOMBES G., SIGOILLOT G., 1984b. Influence des détergents sur *Posidonia oceanica* (L.) Delile. *In*: Boudouresque C.F., Jeudy de Grissac A., Olivier J. edits. *International Workshop on Posidonia oceanica beds*, GIS Posidonie publ., Fr., 1: 407-418.
- AUGIER H., ROBERT P., MAFFRE R., 1980. Etude du régime thermique annuel des eaux au niveau des peuplements de Phanérogames marines de la baie de Port-Cros (iles d'Hyères, Méditerranée, France). *Trav sci Parc nation Port Cros*, 6: 69-131.
- AUGIER H., GIGLIOY., RAMONDA G., 1987. Peintures anti-fouling et herbier à *Posidonia oceanica* GIS Posidonie & Fac. Sci. Luminy Publ.: 1-94.
- AUGIER H., NIERI M., 1988. Cartographie, balisage et dynamique du récif-barrière à *Posidonia oceanica* de la baie de Port-Cros (Parc national). Sci Rep Port Cros nat/ Park, 14: 29-40 + 1 pl. h.t.
- AUSILI A., GABELLINI M., 2000. Un caso di studio : il porto di Viareggio. *Dragaggi portuali qualità e gestione dei materiali,* Atti Giornata 18 febbraio 1998, ICRAM publ.: 49-60.
- AVON M., PERGENT G., PERGENT-MARTINI C., CIAPETTI N., 1992. Projet Barmar: atterrissement station de Martigues. Notice d'impact. Câbles Télécom et Hightech Environnement, Hightech Environnement publ., Marseille, Fr.: 1-55 + Ann.
- BACHET F, 1992. Evaluation des retombées économiques du Parc Régional Marin de la Côte Bleue. Economic impact of the Mediterranean costal protected areas, Medpan News, 3: 43-46
- BALDISSERA-NORDIO C., GALLARATI SCOTTI G., RIGONI M., 1967. Atti Conv Nazion Attività subacquee, Italie, 1: 21-28.
- BALDISSERA-NORDIO C., GALLARATI SCOTTI G., RIGONI M., SAIBENE G., TOSCHI A., 1968. Utilizzazione dei vegetali marini nell'alimentazione dei polli. *Terra Pugliese*, 17(12): 1-8.
- BALDUZZI A., BOERO F., PANSINI M., PRONZATO R., 1981. A method for semiquantitative samplings of the sessile epifauna of the *Posidonia* oceanica beds. *Rapp Commiss Internation, Explor Sci Médit*, 27(9): 247-248.
- BALDUZZI A., BAVESTRELLO G., BELLONI S., BOERO F., CATTANEO R., PANSINI M., PRONZATO R., 1984. Valutazione dello stato di alcune praterie di *Posidonia* nel mare ligure sottoposte a diverse condizioni di inquinamento. *In*: Boudouresque C.F., Jeudy de Grissac A., Olivier J. edits. *International Workshop on Posidonia oceanica beds*, GIS Posidonie publ., Fr., 1: 73-78.
- BALDUZZI A., BIANCHI C.N., CATTANEO-VIETTI R., CERRANO C., COCITO S., COTTA S., DEGL'INNOCENTI F., DIVIACCO G., MORGIGNI M., MORRI C., PANSINI M., SALVATORI L., SENES L., SGORBINI S., TUNESI L., 1994. Primi lineamenti di bionomia bentica dell'isola Gallinaria (Mar Ligure). Atti X Congr AIOL, Alassio, 4-6 novembre 1992: 603-617.
- BALESTRI E., CINELLI F., PIAZZI L., 1998. Survival and growth of transplanted and natural seedlings of *Posidonia oceanica* (L.) Delile in a damaged coastal area. J exp. mar. Biol. Ecol., 228: 209-225.
- BALESTRI E., CINELLI F., LARDICCI C., 2003. Spatial variation in *Posidonia oceanica* structural, morphological and dynamic features in a northwestern Mediterranean coastal area: a multi-scale analysis. *Mar Ecol Progr Ser*, 250: 51-60.
- BALLESTA L., PASQUALINI V., PERGENT G., PERGENT-MARTINI C., 2000. Distribution and dynamics of *Posidonia oceanica* beds along the Albères coastline. *C R Acad Sci , Life Sci ,* 323: 407-414.
- BALLESTEROS E., 1987. Estructura i dinàmica del poblament algal de les fulles de *Posidonia oceanica* (L.) Delile als herbeis de Tossa de Mar (Girona). *Bull Inst catal Hist nat*, 54(6): 13-30.
- BALLESTEROS E., GARCIA A., LOBO A., ROMERO J., 1984. L'alguer de *Posidonia oceanica* de les illes Medes. *Els sistems naturals de les illes Medes*, Ros J., Olivella J., Gili J.M. éds., IEC publ., Barcelona: 739-759.
- BALLESTEROS E., ALCOVERRO T., CEBRIAN E., FONT X., GARCIA-RUBIES A., ROMERO J., 2005. Pseudoviviparity, a new form of asexual reproduction in the seagrass *Posidonia oceanica. Bot Mar*, 48: 175-177.
- BARBERA C., 2000. La situazione in Italia. Dragaggi portuali qualità e gestione dei materiali, Atti Giornata 18 febbraio 1998, ICRAM publ.: 11-17.
- BARNABÉ G., 1989. L'élevage du loup et de la daurade. Aquaculture, G. Barnabé coord. Lavoisier Ed., Paris, 2: 628-668.
- BAROLI M., COSSU A., CRISTINI A., DE FALCO G., GAZALE V., PERGENT G., PERGENT-MARTINI C., 2001. Concentrations of trace metals (Cd, Cu, Fe, Pb) in *Posidonia oceanica* seagrass of Liscia bay, Sardinia (Italy). In "Structure and Processes in the Mediterranean Ecosystem" Faranda F.M., Guglielmo L., Spezie G. eds. Springer-Verlag publ., Proceed. Convegno Diversità e cambiamento, Ischia 10-14 novembre 1998.
- BAY D., 1978. Etude *in situ* de la production primaire d'un herbier de Posidonies (*Posidonia oceanica* (L.) Delile) de la baie de Calvi-Corse. Progr. Rép. Stn. Océanogr. Stareso, Univ. Liège, Belg., 18: 6 p non num. + 1-251.
- BAY D., 1979. Etude" in situ" de la production primaire d'un herbier de Posidonies, Posidonia oceanica (L.) Delile, dans la baie de Calvi. Corse. Rapp PV Réun Commiss internation Explor sci Médit, 25-26(4): 201-202.
- BELIAS C., DASSENAKIS M., 2002. Environmental problems in the development of marine fish-farming in the Mediterranean sea. Ocean challenge, 12(1): 11-16.
- BELL J.D., HARMELIN-VIVIEN M.L., 1982. Fish fauna of French Mediterranean Posidonia oceanica seagrass meadows. 1. Community structure. Téthys, 10(4): 337-347.
- BELLAN G., 1985. De la connaissance fondamentale des milieux naturels marins (principalement benthiques) à leur sauvegarde. Tethys, 11(3-4): 342-349.
- BELLAN G., 1993. Les indicateurs biologiques du milieu marin Remarques introductives. In: Boudouresque C.F., Avon M., Pergent-Martini C. edits. Qualité du milieu marin Indicateurs biologiques et physico chimiques Rencontres scientifiques de la Côte Bleue, GIS Posidonie publ., 3: 35-42.
- BELLAN G., 1994. L'épuration en mer : de la Californie à la Méditerranée. Actes du colloque scientifique Pour qui la Méditerranée au 21^{eme} siècle ? Villes des rivages et environnement littoral en Méditerranée Montpellier: 255-259.

- BELLAN G., KAIM-MALKA R.A., PICARD J., 1975. Evolution récente des différentes auréoles de pollution marine des substrats meubles liées au grand collecteur de Marseille-Cortiou. *Bull Ecol*, 6(2): 57-65.
- BELLAN G., BOURCIER M., SALEN-PICARD C., ARNOUX A., CASSERLEY S., 1999. Benthic ecosystem changes associated with wastewater treatment at Marseille: implications for the protection and restoration of the Mediterranean shelf ecosystem. Water Environ Res , 71(4): 483-493.
- BELLAN-SANTINI D., 1966. Influence des eaux polluées sur la faune et la flore marines benthiques dans la Région marseillaise. Techn Sci municipales, Fr., 61(7): 285-292.
- BELLAN-SANTINI D., PICARD J., 1984. Conséquences de la régression des herbiers de Posidonies pour le benthos animal des autres milieux méditerranéens. In: Boudouresque C.F., Jeudy de Grissac A., Olivier J. edits. International Workshop on Posidonia oceanica beds, GIS Posidonie publ., Fr., 1: 423-429.
- BELLAN-SANTINI D., ARNOUX A., WILLSIE A., 1986. Distribution comparée des crustacés amphipodes de la "matte" d'herbier de Posidonies mort et vivant. Rapp PV Réun Commiss internation Explor sci Médit., 30(2): 8.
- BELLAN-SANTINI D., LACAZE J. C., POIZAT C., 1994. Les biocénoses marines et littorales de Méditerranée, synthèse, menaces et perspectives. Muséum National d'Histoire Naturelle publ., Paris: 1-246.
- BELLONE E., MEINESZ A., 1995. Cartographie sous-marine du Parc national de Port-Cros (Var, France). I Elément de cartographie de l'herbier de *Posidonia oceanica* de la baie de Port-Man. *Sci Rep Port Cros natl Park*, 16: 123-128.
- BELSHER T., 1977. Analyse des répercussions des pollutions urbaines sur le macrophytobenthos de Méditerranée (Marseille, Port-Vendres, Port-Cros). Thèse Doctorat, Univ. Aix-Marseille II: 1-287.
- BELSHER T., HOULGATTE E., BOUDOURESQUE C.F., 2005. Cartographie de la prairie à Posidonia oceanica et des principaux faciès sédimentaires marins du Parc national de Port-Cros (Var, France, Méditerranée). Sci Rep Port Cros natl Park, 21: 19-28+carte h.t.
- BEN D. Van der, 1971. Les épiphytes des feuilles de Posidonia oceanica Delile sur les côtes françaises de la Méditerranée. Mém Inst roy Sci nat Belgique, 168: 1-101 + 5 tabl. h.t.
- BEN ALAYA H., 1969. Mise en place des herbiers à Phanérogames marines et des peuplements algaux dans le golfe de Tunis. Bull Inst Océanogr, Pêche Salammbô, 1(3): 113-122.
- BEN ALAYA H., 1972. Répartition et conditions d'installation de *Posidonia oceanica* Delile et *Cymodocea nodosa* Ascherson dans le golfe de Tunis. *Bull Inst Océanogr Pêche Salammbô*, 2(3): 331-416.
- BEN MUSTAPHA K., 1995. The gulf of Gabes: a case study in Mediterranean decline. 21^{eme} session du Conseil général des Pêches pour la Méditerranée, Alicante, Espagne: 8-9.
- BEN MUSTAPHA, HATTOUR A., MHETLI M., EL ABED A., TRITAR B., 1999. Etat de la bionomie benthique des étages infra et circalittoral du golfe de Gabès. Bull Inst nation scient techn Mer Salammbô, 26: 5-48.
- BERNARD G., CADIOU G., ESCOFFIER B., LE DIREACH L., BONHOMME P., CHARBONNEL E., 2000. Surveillance de l'herbier à Posidonia oceanica du golfe de Giens (Var, France). GIS Posidonie publ., Fr.:1- 61.
- BERNARD G., DENIS J., DENEUX F., BELSHER T., SAUZADE D., BOUDOURESQUE C.F., CHARBONNEL E., EMERY E., HERVÉ G., BONHOMME P., 2001. Etude et cartographie des biocénoses de la rade de Toulon. Rapport de synthèse final. Syndicat intercommunal de l'Aire toulonnaise, Ifremer et GIS Posidonie, Ifremer publ., La Seyne: 1-150.
- BERNARD G., BOUDOURESQUE C.F., CHARBONNEL E., BONHOMME P., CADIOU G., 2002. Projet d'aménagement du port du Brusc. Cartographie et état de vitalité des peuplements marins. Conseil Général du Var, Service des Ports et GIS Posidonie publ., Marseille, Fr.: 1-87.
- BERNARD G., DENIS J., BONHOMME P., EMERY E., CADIOU G., CHARBONNEL E., HERVÉ G., BOUDOURESQUE C.F., 2003. Cartographie des biocénoses marines entre la Tour Fondue et l'île de Porquerolles. Alimentation en eau potable de l'île de Porquerolles. Rapport de synthèse final. Ville de Hyères-les-Palmiers, GIS Posidonie et Ifremer, GIS Posidonie publ., Marseille, Fr.: 1-60.
- BERTRANDY M.C., 1990. Normes de rejet Cartes de vocation des zones. Suivi de la qualité des rejets. Symposium international Protection du milieu marin contre la pollution urbaine, Atelier 2Bis, 20-22 juin 90 Marseille, Fr., 2: 1-8.
- BETHOUX J.P., GENTILI B., 1998. Functioning of the Mediterranean Sea: past and present changes related to freshwater input and climate change. J mar System, 557: 1-15.
- BIANCHI C.N., CINELLI F, MORRI C., 1995. La carta bionomica dei mari toscani : introduzione, criteri informativi e note esplicative. Atti Soc toscana Sci nat, Memprie, Ser A, Ital., 102(suppl.): 255-270 + 1 carte h.t.
- BIANCHI C.N., PEIRANO A., 1995. Atlante delle Fanerogame marine della Liguria. Posidonia oceanica e Cymodocea nodosa Centro Ricerche Ambiente Marino, ENEA publ., La Spezzia, Ital.: 1-146.
- BIANCHI C.N., ARDIZZONE G.D., BELLUSCIO A., COLANTONI P., DIVIACCO G., MORRI C., TUNESI L., 2003. La cartografia del benthos. Biol mar medit 10(suppl.): 367-394.
- BIANCHI C.N., ARDIZZONE G.D., BELLUSCIO A., COLANTONI P., DIVIACCO G., MORRI C., TUNESI L., 2004. Mediterranean marine benthos a manual of methods for its sampling and study. Gambi M.C., Dappiano M. edits., Biol mar Medit. 11(suppl. 1): 347-370.
- BIANCONI C.H., BOUDOURESQUE C.F., CALTAGIRONE A., LEFÈVRE J.R., SEMROUD R., 1990. Résultats scientifiques de la mission Scandola 49 (août 1990). Parc naturel régional de la Corse et LBMEB, Univ. Aix-Marseille 2, Fr.: 1-28.
- BLANC J.J., 1956. Etudes géologiques et sédimentologiques. Ann Inst Oceano., 32 : 123-153.
- BLANC J.J. 1971. Mouvements de la mer et notes de sédimentologie littorale.
- BLANC J.J., 1974. Phénomènes d'érosion sous-marine à la presqu'île de Giens (Var). C R Acad Sci 278: 1821-1823.
- BLANC J.J., 1975. Recherches de sédimentologie appliquée au littoral rocheux de la Provence. Aménagement et protection. Centre nation. Exploit. Océans publ., Fr.: 1-164 + 32 pl. h.t.
- BLANC J.J., JEUDY DE GRISSAC A., 1978. Recherches de géologie sédimentaire sur les herbiers à Posidonies du littoral de la Provence. Contrat CNEXO 76/4043, 77/4074. Centre nation. Exploit. Océans publ., Fr.: 1-185, i-vii, 42 pl. h.t.
- BLANC J.J., JEUDY DE GRISSAC A., 1984. Erosions "sous-marines" des herbiers à *Posidonia oceanica* (Méditerranée). In: Boudouresque C.F., Jeudy de Grissac A., Olivier J. edits. International Workshop on Posidonia oceanica beds, GIS Posidonie publ., Fr., 1: 23-28.
- BLANC J.J., JEUDY DE GRISSAC A., 1989. Réflexion géologique sur la régression des herbiers à Posidonies (départements du Var et des Bouches-du-Rhône). In: Boudouresque C.F., Meinesz A., Fresi E., Gravez V. édits. Second international Workshop on Posidonia beds, GIS Posidonie publ., Fr., 2: 273-285.
- BLANPIED C., BUROLLET P.F., CLAIREFOND P., SHIMI M., 1979. Sédiments actuels et Holocènes. La mer Pélagienne, étude sédimentologique et écologique du plateau tunisien et du golfe de Gabès. Ann Univ Provence, Fr., 6(1): 61-82.
- BLANC-VERNET L., 1984. Les Foraminifères de l'herbier à Posidonia oceanica en Méditerranée : analyse des assemblages, aspects régionaux, application aux microfaunes fossiles. In: Boudouresque C.F., Jeudy de Grissac A., Olivier J. edits. International Workshop on Posidonia oceanica beds, GIS Posidonie publ., Fr., 1: 3-14.

BOMBACE G., 1995. Le barriere artificiali nella gestione razionale della fascia costiera italiana. Biol Mar Medit, 2(1): 1–14.

BOND W., 2001. Keystone species - Hunting the snark? Science 292: 63-64.

BONHOMME P, PALLUY F, 1998. Etude cartographique de l'herbier de Posidonie au pied du môle neuf du port de Cassis. GIS Posidonie publ., Fr.: 1-27.

- BONHOMME P, ROY D., BERNARD G., CHARBONNEL E., DIVETAIN N., 1999. Extension de la cartographie des fonds et de l'herbier à Posidonia oceanica entre la calanque du Mugel et l'1le Verte. ADES et GIS Posidonie, GIS Posidonie publ., Fr.: 1-90.
- BONHOMME P, CHARBONNEL E., ESCOFFIER B., CADIOU G., BERNARD G., 2000. Mise en place d'un système de surveillance de l'herbier de Posidonie sur le littoral des Maures dans le cadre de l'observatoire marin du SIVOM du littoral des Maures. Rapport final. GIS Posidonie publ., Fr.: 1-58.
- BONHOMME P, CADIOU G., BERNARD G., CHARBONNEL E., RUITTON S., 2001. Evaluation de l'impact sur l'herbier de Posidonie et les grandes nacres du creusement d'un chenal dans le cadre de l'aménagement du site militaire du Canier. Rapport Phase 1 : Mise en place d'un carré permanent. Contrat DGA-DCEE & GIS Posidonie publ., Fr.: 1-38.
- BONHOMME P, BERNARD G., CHARBONNEL E., CADIOU G., LERICHE A., ANGLES D'ORTOLI N., LE DIREAC'H L., DENIS J., DENEUX
 E., HERVÉ G., EMERY E., CLABAUT P, 2003a. Guide méthodologique pour la cartographie des biocénoses marines. Volet n°1 : l'herbier
 à Posidonia oceanica Notice technique. Agence de l'Eau Rhône Méditerranée Corse, Conseil régional PACA, DIREN PACA/Ifremer, COM & GIS Posidonie, GIS Posidonie publ., Fr.: 1-86.
- BONHOMME P, CADIOU G., BERNARD G., BOUDOURESQUE C.F., 2003b. Suivi de l'herbier de Posidonie à la pointe du Canier (St-Mandrier, Var) : premier suivi un an après les travaux d'aménagements du site militaire du Canier. Contrat DGA-DCEE & GIS Posidonie. GIS Posidonie publ., Fr.: 1-35.
- BONHOMME P, CADIOU G., BERNARD G., CHARBONNEL E., BOUDOURESQUE C.F., 2003c. Evaluation de l impact sur l'herbier et les grandes nacres du creusement d'un chenal dans le cadre de l'aménagement du site militaire du Canier. Rapport Phase 2 et 3 : constats après travaux. Contrat DGA-DCEE & GIS Posidonie. GIS Posidonie publ., Fr.: 1-54.
- BONHOMME P., CADIOU G., BERNARD G., GANTEAUME A., BOUDOURESQUE C.F., 2004. Suivi de l'herbier de Posidonie à la pointe du Canier (St-Mandrier, Var) : deuxième suivi deux ans après les travaux d'aménagements du site militaire du Canier. Contrat DGA-DCEE & GIS Posidonie. GIS Posidonie publ., Fr.: 1-35.
- BORG J.A., SCHEMBRI P.J., 1995. Preliminary data on bathymetric and temporal changes in the morphology of a maltese *Posidonia oceanica* (L.) Delile meadow. *Rapp PV Réun Commiss internation Explor sci Médit.*: 34: 20.
- BORG J.A., ATTRILL M.J., ROWDEN A..A., SCHEMBRI P.J., JONES M.B., 2005. Architectural characteristics of two types of the seagrass Posidonia oceanica over different spatial scales. Est Coast Shelf Sci 62: 667-678.
- BOROWITZKA M.A., LETHBRIDGE R.C., 1989. Seagrass epiphytes. In Larkum A.W.D., McComb A.J., Shepherd S.A., edits. Biology of seagrasses, Aquatic Plant Studies 2. Elsevier publ., Amsterdam: 458-499.
- BOUDOURESQUE C.F., AUGIER H., BELSHER T., COPPEJANS E., PERRET M., 1975. Végétation marine de l'île de Port-Cros (Parc national). X. La régression du récif-barrière de Posidonies. *Trav Sci Parc nation Port Cros*, Fr., 1: 41-46.
- BOUDOURESQUE C.F., GIRAUD G., PERRET M., 1977. Posidonia oceanica, bibliographie. CNEXO et Université d'Aix-Marseille 2, Fr.: 1-191.
- BOUDOURESQUE C.F., GIRAUD G., PERRETBOUDOURESQUE M., 1979. Bibliography on vegetation and ecosystems of *Posidonia* oceanica, Part I. *Excerpta Botanica.*, 19(B): 145-161.
- BOUDOURESQUE C.F., GIRAUD G., PANAYOTIDIS P., 1980a. Végétation marine de l'île de Port-Cros. XIX. Mise en place d'un transect permanent. *Trav sci Parc nation Port Cros*, Fr., 6: 207-221.
- BOUDOURESQUE C.F., GIRAUD G., PERRET-BOUDOURESQUE M., 1980b. Bibliography on vegetation and ecosystems of *Posidonia* oceanica, Part II. *Excerpta Botanica.*, 20(2B): 125-135.
- BOUDOURESQUE C.F., GIRAUD G., THOMMERET J., THOMMERETY., 1980c. First attempt at dating by ¹⁴C the undersea beds of dead *Posidonia oceanica* in the bay of Port-Man (Port-Cros, Var, France). *Trav sci Parc nation Port Cros*, Fr., 6: 239-242.
- BOUDOURESQUE C.F., THOMMERET J., THOMMERET Y., 1980d. Sur la découverte d'un bioconcrétionnement fossile intercallé dans l'herbier à *Posidonia oceanica* de la baie de Calvi (Corse). Journées Etude Systém. Biogéogr. médit., CIESM publ.: 139-142.
- BOUDOURESQUE C.F., MEINESZ A., PANAYOTIDIS P., 1981. Mise en place d'un carré permanent dans un herbier de Posidonies. Rapp Commiss internation mer Médit 27(9): 245-246.
- BOUDOURESQUE C.F., MEINESZ A., 1982. Découverte de l'herbier de Posidonie. Cah Parc nation Port Cros, Fr., 4: 1-79.
- BOUDOURESQUE C.F., JEUDY DE GRISSAC A., 1983. L'herbier à *Posidonia oceanica* en Méditerranée : les interactions entre la plante et le sédiment. J Rech océanogr, 8(2-3): 99-122.
- BOUDOURESQUE C.F., CROUZET A., PERGENT A., 1983. Un nouvel outil au service de l'étude des herbiers à *Posidonia oceanica* : la lépidochronologie. *Rapp P V Réun Comm internation Explor sci Médit*, 28(3): 111-112.
- BOUDOURESQUE C.F., JEUDY DE GRISSAC A., MEINESZ A., 1984. Relations entre la sédimentation et l'allongement des rhizomes orthotropes de *Posidonia oceanica* dans la baie d'Elbu (Corse). *In*: Boudouresque C.F., Jeudy de Grissac A., Olivier J. edits. *International Workshop on Posidonia oceanica beds*, GIS Posidonie publ., Fr., 1: 185-191.
- BOUDOURESQUE C.F., JEUDY DE GRISSAC A., MEINESZ A., 1985a. Un nouveau type d'herbier à *Posidonia oceanica* : l'herbier de colline. Rapp PV Réun Commiss internation Explor sci Médit , 29(5): 173-175.

BOUDOURESQUE C.F., MEINESZ A., LEFÈVRE J.R., 1985b. Cartographie des peuplements benthiques marins de Corse. I. La formation récifale à *Posidonia oceanica* de Saint-Florent. *Ann Inst océanogr*, 61(1): 27-38.

- BOUDOURESQUE C.F., BIANCONI C.H., 1986. Posidonies profondes dans le coralligène de Sulana. Trav sci Parc nat rég Rés nat Corse, 2: 36-39.
- BOUDOURESQUE C.F., JEUDY DE GRISSAC A., 1986. Biomasse dans l'herbier à *Posidonia oceanica. Trav sci Parc nat rég Rés nat* Corse, 2: 14-15.
- BOUDOURESQUE C.F., JEUDY DE GRISSAC A., MEINESZ A., 1986a. Chronologie de l'édification d'une colline de Posidonies. Trav sci Parc nat rég Rés nat Corse, 2: 3-12.
- BOUDOURESQUE C.F., LEFÈVRE J.R., MEINESZ A., 1986b. Cartographie du carré permanent de la marina d'Elbu. Trav sci Parc nat rég Rés nat Corse, 2: 24-33.
- BOUDOURESQUE C.F., GRAVEZ V., LEVEAU M., MICHEL P., ROBERT G., SOURENIAN B., VITIELLO P., 1988. Analyse de l'état initial du golfe de Giens. Synthèse et conclusions générales. Etude d'impact pour le rejet des effluents issus de la station d'épuration d'Hyères BCEOM et GIS Posidonie, GIS Posidonie publ., Fr.: 1-25.

BOUDOURESQUE C.F., 1989. Bibliographie Posidonia oceanica Parc national de Port-Cros, GIS Posidonie publ., Fr.: 1-243.

BOUDOURESQUE C.F., BALLESTEROS E., BEN MAIZ N., BOISSET F., BOULADIER E., CINELLI F., CIRIK S., CORMACI M., JEUDY DE GRISSAC A., LABOREL J., LANFRANCO E., LUNDBERG B., MAYHOUB H., MEINESZ A., PANAYOTIDIS P., SEMROUD R., SINNASSAMY


J.M., SPAN A., VUIGNIER G., 1990a. Livre rouge "Gérard Vuignier" des végétaux, peuplements et paysages marins menacés de Méditerranée. Programme des Nations Unies pour l'Environnement publ.: 1-250.

- BOUDOURESQUE C.F., BERTRANDY M.C., BOULADIER E., FORET P., MEINESZ A., PERGENT G., VITIELLO P., 1990b. Le Réseau de surveillance des herbiers de Posidonies mis en place en Région Provence-Alpes-Côte d'Azur. *Rapp Commiss internation Mer Médit* 32(1): 11.
- BOUDOURESQUE C.F., BIANCONI C.H., MEINESZ A., 1990c. Live Posidonia oceanica in a coralligenous algal bank at Sulana, Corsica. Rapp P V Réun Commiss internation Explor sci Médit 32(1): 11.

BOUDOURESQUE C.F., AVON M., GRAVEZ V., 1991. Les espèces marines à protéger en Méditerranée. GIS Posidonie publ., Fr.: 1-448.

- BOUDOURESQUE C.F., GRAVEZ V., MEINESZ A., MOLENAAR H., PERGENT G., VITIELLO P., 1994a. L'herbier à *Posidonia oceanica* en Méditerranée : protection légale et gestion. *In: Pour qui la méditerranée au 21^{eme} Siècle Villes des rivages et environnement littoral en Méditerranée*. Actes du colloque scientifique Okeanos, Maison de l'Environnement de Montpellier publ., Fr.: 209-220.
- BOUDOURESQUE C.F., MEINESZ A., LEDOYER M., VITIELLO P., 1994b. Les herbiers à Phanérogames marines. In: Bellan-Santini D., Lacaze J.C., Poizat C. eds Les biocénoses marines et littorales de Méditerranée, synthèse, menaces et perspectives. Muséum National d'Histoire naturelle publ., Paris, Fr.: 98-118.
- BOUDOURESQUE C.F., ARRIGHI F., FINELLI F., LEFEVRE J.R., 1995a. Arrachage des faisceaux de *Posidonia oceanica* par les ancres : un protocole d'étude. *Rapp Commiss internation Mer Médit*, Monaco, 34 : 21.
- BOUDOURESQUE C.F., GRAVEZ V., MEINESZ A., MOLENAAR H., PERGENT G., VITIELLO P., 1995b. L herbier à Posidonia oceanica en Méditerranée : Protection légale et gestion. In: Pour qui la méditerranée au 21^{eme} Siècle Villes des rivages et environnement littoral en Méditerranée. Actes du colloque scientifique Okeanos, Maison de l'Environnement de Montpellier publ., Fr.: 209-220.
- BOUDOURESQUE C.F., MEINESZ A., RIBERA M.A., BALLESTEROS E., 1995c. Spread of the green alga *Caulerpa taxifolia* (Caulerpales, Chlorophyta) in the Mediterranean: possible consequences of a major ecological event. *Scientia marina*, 59(suppl. 1): 21-29.
- BOUDOURESQUE C.F., 1996. Impact de l'homme et conservation du milieu marin en Méditerranée. 2^{eme} édition. GIS Posidonie publ. (ISBN 2 905-54-21-4): 1-243.
- BOUDOURESQUE C.F., VAN KLAVEREN M.C., VAN KLAVEREN P., 1996. Proposal for a list of threatened or endangered marine and brackish species (plants, invertebrates, fish, turtles and mammals) for inclusion in appendices I, II and III of the Bern Convention. *Council of Europe, Document S/TPVS96/TPVS48E, 96A*: 1-138.
- BOUDOURESQUE C.F., 2001. La restauration des écosystèmes à Phanérogames marines. In Restauration des écosystèmes côtiers, Dreves L., Chaussepied M. édits., Ifremer publ., 29: 65-85.
- BOUDOURESQUE C.F., CHARBONNEL E., MEINESZ A., PERGENT G., PERGENTMARTINI C., CADIOU G., BERTRANDY M.C., FORET P., RAGAZZI M., RICO-RAIMONDINO V., 2000. A monitoring network based on the seagrass *Posidonia oceanica* in the nortwestern Mediterranean Sea. *Biol mar medit*, 7(2): 328-331.
- BOUDOURESQUE C.F., VERLAQUE M., 2001. Ecology of *Paracentrotus lividus*. Edible sea urchins: biology and ecology, Lawrence J. édit., Elsevier publ., Amsterdam: 177-216.
- BOUDOURESQUE C.F., 2002a. La restauration des herbiers. In: Drévès L, Chaussepied M. édits. Restauration des écosystèmes côtiers. Ifremer publ., Actes de Colloques, 29.
- BOUDOURESQUE C.F., 2002b. Concilier protection et usages du milieu marin : l'expérience du Parc national de Port-Cros. Rev Soc anciens élèves Ecole polytechnique, Jaune Rouge, 575: 31-35.
- BOUDOURESQUE C.F., 2002c. Protected marine species, prevention of species introduction and the national environmental agencies of Mediterranean countries: professionalism or amateurishness ? In Actes du congrès international "Environnement et identité en Méditerranée", Corte, 3-5 July 2002, Université de Corse Pascal Paoli publ., 4 : 75-85.
- BOUDOURESQUE C.F., 2003. The erosion of Mediterranean biodiversity. In Rodriguez-Prieto C., Pardini G. edits. The Mediterranean Sea an overview of its present state and plans for future protection. Servei de Publicacions de la Universitat de Girona: 53-112.
- BOUDOURESQUE C.F., 2004. Marine biodiversity in the Mediterranean : status of species, populations and communities. Sci Rep Port Cros natl Park 20: 97-146.
- BOUDOURESQUE C.F., CADIOU G., GUERIN B., LE DIREACH L., ROBERT P., 2004. Is there a negative interaction between biodiversity conservation and artisanal fishing in a Marine Protected Area, the Port-Cros National Park (France, Mediterranean Sea). Sci Rep Port Cros natl Park 20: 147-160.
- BOUDOURESQUE C.F., CADIOU G., LE DIREACH L., 2005. Marine protected areas: a tool for coastal areas management. In Levner E., Linkov I., Proth J.M. edits. Strategic management of marine ecosystems Springer publ., Dordrecht: 29-52.
- BOUMAZA S., SEMROUD R., 1995. Variations saissonnières des biomasses (feuilles et épiphytes) de Posidonia oceanica (L.) Delile dans l'herbier de l'anse de Kouali (Tipaza) Algérie : données préliminaires. Rapp PV Réun Commiss internation Explor sci Médit 34: 21.
- BOUMAZA S., SEMROUD S., 2000. Surveillance de l'herbier à *Posidonia oceanica* d'El Djamila (Algérie). *Proceedings of the first* Mediterranean symposium on marine vegetation RAC/SPA publ., Tunis: 110-114.
- BOURCIER M., 1979. Contribution à la connaissance des sediments marins entre le cap Croisette et le cap d'Alon (Est de Marseille) et de leur évolution de 1966 à 1972. Téthys 9(1): 31-40.
- BOURCIER M., NODOT C., JEUDY DE GRISSAC A., TINE J., 1979. Répartition des biocoenoses benthiques en fonction des substrats sédimentaires de la rade de Toulon (France). *Téthys* 9(2): 103-112.
- BOURCIER M., 1989. Régression des herbiers à *Posidonia oceanica* (L.) Delile, à l'Est de Marseille, sous l'action conjuguée des activités humaines et des modifications climatiques. *In*: Boudouresque, C.F., Meinesz A., Fresi E., Gravez V.édits. *International workshop on Posidonia beds*. GIS Posidonie publ., Fr., 2: 287-293.
- BOYRA A., NASCIMENTO F.J.A., TUYA F. SÁNCHEZ-JEREZ P., HAROUN R.J., 2004. Impact of sea-cage fish farms on intertidal macrobenthic assemblages. J mar biol Ass U K 84: 665-668.

BRADLEY R., 2000. 1000 years of climate change. Science 288: 1353-1354.

- BRAUN-BLANQUET J., ROUSSINE N., NEGRE R., 1952. Les groupements végétaux de la France méditerranéenne. Macabet Frères publ., Vaison-la-Romaine: 1-297.
- BRIX H., LYNGBY J.E., SCHIERUP H.H., 1983. Eelgrass (Zostera marina L.) as an indicator organism of trace metals in the Limfjord, Denmark. Mar Environ Res 8: 165-181.
- BROWN J.R., GOWEN R.J., MCLUSKY D.S., 1987. The effect of salmon farming on the benthos of a Scottish Sea loch. J Exp Mar Biol Ecol. 109: 39-51.
- BRULARD A., 1885. Monographie de l île de Djerba. Delagrange publ., Besançon.

BUIA M.C., GAMBI M.C., DAPPIANO M., 2003. I sistemi a Fanerogame marine. Biol mar Medit 10: 145-198.

BUIA M.C., GAMBI M.C., DAPPIANO M., 2004. Seagrass systems. In: Gambi M.C., Dappiano M. edits. Mediterranean marine benthos

a manual of methods for its sampling and study. Biol mar Medit. 11(suppl. 1): 133-183.

- BUROLLET P. F., CLAIREFOND P., WINNOCK E., 1979. La mer Pélagienne (étude sédimentologique et écologique du plateau tunisien et du golfe de Gabès). Géol méditerranéenne, Ann Univ Provence 6(1): 1-345 + cartes.
- BUROLLET P. F., OUDIN J.L., 1979. La mer Pélagienne. IV. Les constituants lithologiques. G. Répartition de la matière organique. Ann Univ Provence 6(1): 139-142.
- CADIOU G., CHARBONNEL E., BOUDOURESQUE C.F., MEINESZ A., BONHOMME P., BERNARD G., COTTALORDA J.M., GARCIA F., KRUCZEK R., BERTRANDY M.C., FORET P., BRICOUT P., MARRO C., 2004. Le Réseau de Surveillance Posidonies de la Région Provence-Alpes-Côte d'Azur. Résultats du suivi 2003. Région PACA, Agence de l'Eau, DDE & GIS Posidonie, GIS Posidonie publ., Fr.: 1-146.
- CALMET D., LEFÈVRE J.R., MEINESZ A., 1986. Evolution chimique des rhizomes de *Posidonia oceanica* en fonction de leur âge. *Trav* sci Parc nat rég Rés nat Corse, Fr., 2: 15-19.
- CALMET D., BOUDOURESQUE C.F., MEINESZ A., 1988. Memorization of nuclear atmospheric tests by rhizomes and scales of the mediterranean seagrass *Posidonia oceanica* (Linnaeus) Delile. *Aquat Bot* 30(4): 279-294.
- CALMET D., CHARMASSON S., GONTIER G., A MEINESZ., BOUDOURESQUE C.F., 1991. Chernobyl radionucleides in the Mediterranean seagrass *Posidonia oceanica*, 1986-1987. J Environ Radioactivity 13: 157-173.
- CALOZ R., COLLET C., 1997. Geographic information systems (GIS) and remote sensing in aquatic botany: methodological aspects. Aquat Bot 58: 209-228.
- CALUMPONG H., FONSECA M., 2001. Seagrass transplantation and other seagrass restoration methods. In Short F.T., Coles R.G. edits. Global seagrass research mthods. Elsevier publ., Amsterdam: 425-443.
- CALVO S., FRADÀ-ORESTANO C., 1984. L'herbier à *Posidonia oceanica* des côtes siciliennes : les formations récifales du Stagnone. *In:* Boudouresque C.F., Jeudy de Grissac A., Olivier J. edits. *International Workshop on Posidonia oceanica beds*, GIS Posidonie publ., Fr., 1: 29-37.
- CAMBRIDGE M.L., BASTYAN G.R., WALKER D.I., 2000. Recovery of *Posidonia* meadows by rhizome growth and seedling recruitment in Oyster Harbour, South-Western Australia. *Biol mar Medit.* 7(2): 332-335.
- CAMPBELL M.L., 2000. A decision-based framework to increase seagrass transplantation success. Biol mar Medit. 7(2): 336-340.
- CAMPOS-VILLACA R., 1984. Données préliminaires sur l'éthologie alimentaire de l'oursin Sphaerechinus granularis dans l'herbier à Posidonia oceanica de la baie de Port-Cros. Diplôme Etudes Approfondies Océanologie Biologique, Univ. Aix-Marseille 2, Fr.: 1-47.
- CANCEMI G., DE FALCO G., PERGENT G., 2000. Impact of a fish farming facility on a *Posidonia oceanica* meadow. *Biol Mar Medit* 7(2): 341-344.
- CANCEMI G., DE FALCO G., PERGENT G., 2003. Effects of organic matter input from a fish farming facility on a *Posidonia oceanica* meadow. *Est Coast Shelf Sci* 56: 961-969.
- CANNON J.F.M., 1979. An experimental investigation of Posidonia balls. Aquat Bot 6(4): 407-410.
- CANNON J.F.M., 1985. Seaballs and lakeballs An old Mediterranean theme with a new Irish variation. Watsonia 15: 177-181.
- CAPIOMONT A., SANDMEIER M., CAYE G., MEINESZ A., 1996. Enzyme polymorphism in *Posidonia oceanica*, a seagrass endemic to the Mediterranean. *Aquat Bot* 54: 265-277.
- CAPIOMONT A., PIAZZI L., PERGENT G., 2000. Seasonal variations of total mercury levels in foliar tissues of *Posidonia oceanica*. J mar Biol Ass UK 80(6): 1119-1123.
- CAR/ASP, 2003. Etat de mise en oeuvre du protocole ASP. Plan d'action pour la Méditerranée. Sixième réunion des points focaux nationaux pour les ASP, UNEP(DEC)/MED WG. 232/3, Marseille, 17-20 juin 2003 : 1-89.
- CARLOTTI P, BOUDOURESQUE C.F., CALMET D., 1992. Mémorisation du cadmium et de radioéléments par les rhizomes et les écailles de *Posidonia oceanica* (Potamogetonaceae). *Trav Sci Parc nat Rég Rés Nat Corse* 36: 1-34.
- CARRILLO J.A., GIL RODRÍGUEZ M.C., 1980. Cymodocea nodosa (Ucria) Ascherson (Zannicheliaceae) y las praderas submarinas o "sebadales" en el archipiélago canario. Vieaea 8(2): 365-376.
- CATSIKI V.A., PANAYOTIDIS P., PAPATHANASSIOU E., 1987. Bioaccumulation of heavy metals by seagrasses in Greek coastal waters. *Posidonia Newsletter* 1(2): 21-30.
- CAYE G., 1980. Sur la morphogénèse et le cycle végétatif de Posidonia oceanica (L.) Delile. Thèse Doct., Univ. Aix-Marseille 2, Fr.: 1-121.
- CAYE G., MEINESZ A., 1984. Observations sur la floraison et la fructification de *Posidonia oceanica* dans la baie de Villefranche et en Corse du Sud *In*: Boudouresque C.F., Jeudy de Grissac A., Olivier J. edits. *International Workshop on Posidonia oceanica beds*, GIS Posidonie publ., Fr., 1: 193-201.

CAZZUOLA F., 1880. Le piante utili e nocive agli uomini e agli animali.

- CEBRIAN J., MARBÀ N., DUARTE C.M., 1994. Estimating leaf age of the seagrass *Posidonia oceanica* (L) Delile using the plastochrone interval index. *Aquat Bot* 49(1): 59-65.
- CEBRIAN J., DUARTE C.M., 2001. Detrital stocks and dynamics of the seagrass *Posidonia oceanica* (L.) Delile in the Spanish Mediterranean. *Aquat Bot* 70: 295-309.
- CECCHERELLI G., CINELLI F., 1997. Interazioni tra *Caulerpa taxifolia* e le Fanerogame *Cymodocea nodosa* e *Posidonia oceanica* In Cossu A., Meloni M.M. edits. Atti del Convegno internazionale "Introduzione di nuove specie nel Mediterraneo e compatibilità con quelli presenti", Cagliari, 28 nov. 1996, Poseidon publ., Sardinia, Ital.: 48-49.
- CECCHERELLI G., CINELLI F., 1998. Contrasting effects of *Posidonia oceanica* on *Caulerpa taxifolia* In Boudouresque C.F., Gravez V., Meinesz A., Palluy F. edits. *Third international workshop on Caulerpa taxifolia* GIS Posidonie publ., Fr.: 209-218.
- CECCHERELLI G., PIAZZI L., CINELLI F., 2000. Response of the non-indigenous *Caulerpa racemosa* (Forsskal) J. Agardh to the native seagrass *Posidonia oceanica* (L.) Delile: effect of density of shoot and orientation of edges of meadows. *J exp mar Biol Ecol* 243: 227-240.
- CHARBONNEL E., 1993. Cartographie de l'herbier de Posidonie et des fonds marins environnants de Toulon à Hyères (Var, France). Reconnaissance par sonar latéral et photographie aérienne. Notice de présentation. Ifremer & GIS Posidonie publ., Fr.: 1-36 + 3 cartes h.t.
- CHARBONNEL E., BOUDOURESQUE C.F., BERTRANDY M.C., FORET P., MEINESZ A., PERGENT G., PERGENT-MARTINI C., RICO-RAIMONDINO V., VITIELLO P., 1993. Le Réseau de Surveillance Posidonies en Région de Provence-Alpes-Côte d'Azur (Méditerranée, France): les premiers résultats. In: Symposium international sur la Protection du Milieu Marin contre la Pollution Urbaine, Marseille, Fr., 3: 1-10.
- CHARBONNEL E., FRANCOUR P., 1994. Cartographie des fonds et de l'herbier à *Posidonia oceanica* entre la calanque du Mugel et l'île Verte (La Ciotat) pour le passage d'une canalisation d'alimentation en eau de l'île Verte. ADES, Conseil Général des Bouches-du-Rhône & GIS Posidonie, GIS Posidonie publ., Fr.: 1-65.
- CHARBONNEL E., BOUDOURESQUE C.F., BERTRANDY M.C., MEINESZ A., PERGENT-MARTINI C., RICO (RAIMONDINO V., FORET P., RAGAZZI M., LECCIA G., 1994a. Le Réseau de Surveillance Posidonies de la Région Provence-Alpes-Côte d'Azur. Rapport 1993 (3eme

phase, 3the année). GIS Posidonie publ., Fr.: 1-131.

- CHARBONNEL E., SINNASSAMY J.M., GRAVEZ V., 1994b. Réimplantation de la Phanérogame marine *Posidonia oceanica* dans le golfe de Marseille (Bouches-du-Rhône). Suivi 1993. Site du Prado: état après deux ans. Site de Morgiret: état après 1 an. GIS Posidonie publ., Fr.: 1-55.
- CHARBONNEL E., BOUDOURESQUE C.F., MEINESZ A., PERGENT-MARTINI C., RICO-RAIMONDINO V., BERTRANDY M.C., FORET P., RAGAZZI M., LECCIA G., 1995a. Le Réseau de Surveillance Posidonies de la Région Provence-Alpes-Côte d'Azur. Rapport 1994 (4eme phase, 1em année). GIS Posidonie publ., Fr.: 1-167.
- CHARBONNEL E., BOUDOURESQUE C.F., MEINESZ A., PERGENT-MARTINI C., RICO-RAIMONDINO V., SAN MARTIN G., BERTRANDY M.C., FORET P., RAGAZZI M., LECCIA G., 1995b. Le Réseau de Surveillance Posidonies de la Région Provence-Alpes-Côte d'Azur. Rapport 1995 (4^{eme} phase, 2^{eme} année). GIS Posidonie publ., Fr.: 1-193.
- CHARBONNEL E., FRANCOUR P., ABELLARD O., REMONNAY L., BOUDOURESQUE C.F., 1995c. Evaluation de l impact de la protection d'un câble électrique au moyen de matériaux de ballast sur l'herbier de Posidonie entre la Tour Fondue et l'île de Porquerolles (Var, France). EDF & GIS Posidonie, GIS Posidonie publ., Fr.: 1-40.
- CHARBONNEL E., GRAVEZ V., ABELLARD O., BOUDOURESQUE C. F., REMONNAY L., 1995d. Surveillance de l'herbier à *Posidonia oceanica* du golfe de Giens (Var-France). Second suivi. Contrat Syndicat Intercommunal Hyères-Carqueiranne pour l'assainissement de la baie de Giens & GIS Posidonie, GIS Posidonie publ., Fr.: 1-106 + 8 p Ann.
- CHARBONNEL E., MOLENAAR H., GRAVEZ V., 1995e. Réimplantation de la Phanérogame marine *Posidonia oceanica* dans le golfe de Marseille (Bouches-du-Rhône). Rapport final 1991-1995. DEGT Ville de Marseille & GIS Posidonie, GIS Posidonie publ., Fr.: 1-93.
- CHARBONNEL E., VAUGELAS J. de, CHIAVERINI D., COTTALORDA J.M., GRAVEZ V., FRANCOUR P., ABELLARD O., REMONNAY L., MENAGER V., BOUDOURESQUE C.F., 1995f. Cartographie de l'herbier de Posidonie et autres types de fonds dans le secteur des îles d'Hyères (Var, France) pour le passage d'un câble Télécom à fibres optiques. Notice d'impact. France Télécom & GIS Posidonie, GIS Posidonie publ., Fr.: 1-124.
- CHARBONNEL E., 1996. Cartographie de l herbier de Posidonie et évaluation de l'impact de la plaisance dans deux calanques de l'íle de Riou (Marseille, France). Contrat Ville de Marseille, Direction de l'Environnement et des Déchets & GIS Posidonie, GIS Posidonie publ., Fr.: 1-54.
- CHARBONNEL E., BONHOMME P., BERNARD G., GRAVEZ V., 1996. Etude des herbiers de Posidonies et autres types de fonds de la quatrième anse du Mourillon au cap Brun et de la Grande Jetée au port Saint-Louis (Toulon, Var). Ville de Toulon, Direction Urbanisme & GIS Posidonie. GIS Posidonie publ., Fr.: 1-103.
- CHARBONNEL E., BERNARD G., GRAVEZ V., BONHOMME P., FRANCOUR P., BOUDOURESQUE C.F., 1997a. Surveillance de l'herbier à *Posidonia oceanica* du golfe de Giens (var-France). Rapport du troisième suivi. Contrat Syndicat intercommunal Hyères-Carqueiranne pour l'assainissement de la baie de Giens & GIS Posidonie, GIS Posidonie publ., Fr.: 1-98.
- CHARBONNEL E., BONHOMME P., BERNARD G., PALLUY F., LE DIREACH L., BOUDOURESQUE C.F., 1997b. Etude cartographique et diagnostic écologique du milieu marin en vue de la faisabilité de la restructuration des installations littorales et portuaires de la ville de Saint-Tropez (Var). Ville de Saint-Tropez, Service des Marchés et des Subventions des Affaires Maritimes et Portuaires & GIS Posidonie, GIS Posidonie publ., Fr.: 1-98.
- CHARBONNEL E., BERNARD G., BONHOMME P., GRAVEZ V., DIVETAIN N., 1998. Evaluation et suivi de l'impact de la pose d'une canalisation sur l'herbier de Posidonie. Canalisation "sous-marine" d'eau potable raccordant les îles de Ratonneau et d'If (rade de Marseille, Bouches-du-Rhône). Etat initial et mise en place des balisages. Ministère de la Culture et de la Communication, Société des Eaux de Marseille & GIS Posidonie, GIS Posidonie publ., Fr.: 1-42.
- CHARBONNEL E., BONHOMME P., VAUGELAS J. de, GRAVEZ V., COQUILLARD Y., BERNARD G., CADIOU G., BOUDOURESQUE C.F., 1999. Notice d'impact de la pose d'un câble Alcatel à fibres optiques sur le milieu marin. Cartographie des fonds et de l'herbier de Posidonie et recommandations sur les sites d'atterrage (Agde, Marseille, Lavandou et Nice). Rapport final. Contrat TRAVOCEAN & GIS Posidonie, GIS Posidonie publ., Fr.: 1-145.
- CHARBONNEL E., BERNARD G., BONHOMME P., ESCOFFIER B., 2000a. Evaluation et suivi de l'impact de la pose d'une canalisation sur l'herbier de Posidonie. Canalisation "sous-marine" d'eau potable raccordant les îles de Ratonneau et d lf (rade de Marseille, Bouchesdu-Rhône). Rapport final. Contrat Ministère de la Culture et de la Communication, Société des Eaux de Marseille et GIS Posidonie. GIS Posidonie publ., Fr.: 1-42.
- CHARBONNEL E., BOUDOURESQUE C.F., MEINESZ A., BERNARD G., BONHOMME P., PATRONE J., KRUCZEK R., COTTALORDA J.M., BERTRANDY M.C., FORET P., RAGAZZI M., LE DIREAC'H L., 2000b. Le Réseau de Surveillance Posidonies de la Région Provence-Alpes-Côte d'Azur. Première partie : présentation et guide méthodologique. Année 2000. Région PACA, Agence de l'Eau RMC, GIS Posidonie, CQEL 13, CQEL 83, Conseil Général 06, GIS Posidonie publ. : 1-76.
- CHARBONNEL E., BOUDOURESQUE C.F., MEINESZ A., BONHOMME P., BERNARD G., PATRONE J., KRUCZEK R., COTTALORDA J.M., BERTRANDY M.C., FORET P., BRICOUT P., RAGAZZI M., CADIOU G., LE DIREAC'H L., 2000c.Le Réseau de Surveillance Posidonies de la Région Provence-Alpes-Côte d'Azur. Seconde partie: résultats du suivi 1999. GIS Posidonie publ., Marseille, Fr.: 1-142.
- CHARBONNEL E., BOUDOURESQUE C.F., MEINESZ A., BONHOMME P., BERNARD G., PATRONE J., KRUCZEK R., COTTALORDA J.M., BERTRANDY M..C., FORET P., BRICOUT P., RAGAZZI M., MARRO C., SERRE C., LE DIREAC'H L., 2001a. Le Réseau de Surveillance Posidonies de la Région Provence Alpes Côte d Azur Résultats du suivi 2001 Région PACA, Agence de l'Eau, DDE & GIS Posidonie, GIS Posidonie publ., Fr.: 1-119.
- CHARBONNEL E., RUITTON S., BACHET F., MAISONNEUVE DE L., DANIEL B., GEOFFRAY C., 2001b. Les peuplements de poissons des récifs artificiels du Parc Marin de la Côte Bleue. Suivi 2000 et évolution à moyen et long terme. Contrat Parc Marin de la Côte Bleue & GIS Posidonie, GIS Posidonie publ., Fr.: 1-92.
- CHARBONNEL E., BONHOMME P., BERNARD G., CADIOU G., BOUDOURESQUE C.F., 2002. Projet d'aménagement du port de la Madrague de Giens. Cartographie et état de vitalité des herbiers et du récif-barrière de Posidonie. Conseil Général du Vat & GIS Posidonie, GIS Posidonie publ., Fr.: 1-62.
- CHARBONNEL E., BOUDOURESQUE C.F., MEINESZ A., CADIOU G., BONHOMME P., COTTALORDA J.M., PATRONE J., KRUCZEK R., BERTRANDY M.C., FORET P., BRICOUT P., MARRO C., LE DIREACH L., 2003. Le Réseau de Surveillance Posidonies de la Région Provence-Alpes-Côte d'Azur : Résultats du suivi 2002. GIS Posidonie publ., Fr.: 1-176.
- CHESSA L.A., FRESI E., SOGGIU L., 1983. Preliminary data on consumers food web in a *Posidonia oceanica* (L.) Delile bed. *Rapp P V Réun Commiss internation Explor sci Médit* 28(3): 159-160.
- CHESSA L.A., FRESI E., 1994. Conservazione e gestione delle praterie di *Posidonia. In*: Monbailliu X., Torre A. edits. La gestione degli ambienti costieri e insulari del Mediterraneo Edizione del Sole publ., Ital.: 103-127.
- CHESSA L.A., FUSTIER V., FERNANDEZ C., MURA F., PAIS A., PERGENT G., SERRA S., VITALE L., 2000. Contribution to the knowledge

of "banquettes" of Posidonia oceanica (L.) Delile in Sardinia island. Biol Mar Médit. 7(2): 35-38.

CHISHOLM J.R.M., FERNEX F.E., MATHIEU D., JAUBERT J.M., 1997. Waste-water discharge, seagrass decline and algal proliferation on the Côte d'Azur. *Mar Pollut Bull* 34: 78-84.

CINELLI F., 1980. Le Fanerogame marine: problemi di trapianto e di riforestazione. Mem Biol Mar Oceanogr., Ital., suppl. 10: 17-25.

CINELLI F., CORMACI M., FURNARI G., MAZZELLA L., 1984. Epiphytic macroflora of *Posidonia oceanica* (L.) Delile leaves around the island of Ischia (Gulf of Naples). *In*: Boudouresque C.F., Jeudy de Grissac A., Olivier J. edits. *International Workshop on Posidonia oceanica beds*, GIS Posidonie publ., Fr., 1: 91-99.

CINELLI F., 1991. La riforestazione quale mezzo di ripristino e di controllo per le praterie di Fanerogame marine. In: Parchi marini del Mediterraneo Problemi e perspective. Atti del 2^{eme} convegno internazionale, San Teodoro. Icimar publ., Sardegna, Ital.: 78-82.

CINELLI F, FRESI E., LORENZI C., MUCEDOLA A., 1995a. Posidonia oceanica a contribution to the preservation of a major Mediterranean marine ecosystem. Revista marittima publ., Roma: 1-272.

CINELLI F., PARDI G., PAPI I., 1995b. Plant biology. In Cinelli F., Fresi E., Lorenzi C., Mucedola A. edits. La Posidonia oceanica Revista Marittima Publ., Ital., 12(Suppl.): 17-27.

CLAIREFOND P., JEUDY DE GRISSAC A., 1979. Description et analyse des structures sédimentaires en milieu marin: recensement de guelques exemples dans l'herbier de Posidonies autour de l'île de Port-Cros (Parc national). Trav sci Parc nation Port Cros 5:79-104.

CLARK PU., ALLEY R.B., POLLARD D., 1999. Northern hemisphere ice-sheet influences on global climate change. Science 286: 1104-1111.

CLARK R.B., 1986. Marine pollution. Clarendon Press publ., Oxford: i-xiii + 1-215.

COCITO S., SGORBINI S., BIANCHI C.N., MORGIGNI M., 1994. Modificazioni ambientali indotte da sversamento di materiale di escavo rilevate con telecamera filoguidata. *Biol. mar. Medit.* 1(1): 275-276.

COLANTONI P, GALLIGNANI P, FRESI E., CINELLI F, 1982. Patterns of *Posidonia oceanica* (L.) Delile beds around the island of Ischia (Gulf of Naples) and adjacent waters *Mar Eco y PSZNI* 3 (1): 53-74 + 2 cartes.

COLLART D., CHARBONNEL E., 1998. Impact des récifs artificiels de Marseillan et d'Agde sur le milieu marin et la pêche professionnelle. Bilan du suivi 1996/1997. Contrat Conseil Régional Languedoc-Roussillon & Conseil Général de l'Hérault. CEGEL & GIS Posidonie, GIS Posidonie publ., Fr.: 1-168.

CONDE POYALES F., 1989. Ficogeografía del mar de Alborán en el contexto del Mediterráneo occidental. An Jard bot Madrid 46(1): 21-26.
CONNOLLY R.M., HINDELL J.S., GORMAN D., 2005. Seagrass and epiphytic algae support nutrition of a fisheries species, Sillago schomburgkii, in adjacent intertidal habitats. Mar Ecol Progr Ser 286: 69-79.

CONRAD M., 1982. Les plantes sauvages dans la vie quotidienne des Corses. Essai d'éthnobotanique; U Patrimoniu Corsu publ., Bastia, Fr.: 1-55.

COOPER G., 1976. Jardinier de la Mer. Association-Fondation G. Cooper pour la reconquête des milieux naturels détruits. *Cahier* 1: 1-57. COOPER G., 1982. Réimplantation de *Posidonia oceanica* Protection des implants. *Bull Ecol* 13(1): 65-73.

CORREL D.S., JOHNSTON M.C., 1970. Manual of the vascular plants of Texas. Texas Res. Foundation publ., Texas: xv + 1-1881.

CORREL D.S., CORREL H.B., 1975. Aquatic and wetland plants of southwestern United States. Standford Univ. Press, California: xv + 1-1777.

CORTEMIGLIA G.C., 1979. Atti del Convegno Nazionale per la Difesa del Litorale di Chiavari, Lavagna e Sestri Levante dall erosione marina. Comune di Chiavari, Comune di Lavagna, Comune di Sestri Levante, C.N.R. Progetto Finalizzato Conservazione del suolo, Genova, F.C. 1099: 1-13.

COSTANZA R., ARGE R., DE GROOT R., FARBER S., GRASSO M., HANNON B., LIMBURG K., NAEEM S., O'NEILL R.V., PARUELO J., RASKIN R.G., SUTTON P, VAN DEN BELT M., 1997. The value of the world's ecosystem services and natural capital. *Nature* 387: 253-260.

COULON P., JANGOUX M., 1992. Rate and rhythm of feeding of the Holothuroid *Holothuria tubulosa* in the seagrass beds off the Island of Ischia (bay of Naples, Italy). *In:* Keegan B F. edit *Space and time series Data analysis in coastal benthic Ecology.* Commission of the European Communities publ.: 573-580.

CREBASSA P, 1992. Evaluation des mesures de protection des herbiers à *Posidonia oceanica*. Rapp. stage Ecole Polytechnique: 1-41 + 5p. Ann.

CRISTIANI G., 1980. Biomasse et répartition de l'herbier de *Posidonia oceanica* de la Côte Bleue (B. d. Rh. France) et pollution marine par les métaux lourds. Thèse Doct., Univ. Aix-Marseille 3, Fr.: 1-150 + ix pp + 1 carte h.t.

CROUZET A., 1981. Mise en évidence de variations cycliques dans les écailles de Posidonia oceanica (Potamogetonaceae). Trav sci Parc nation Port Cros, Fr. 7: 129-135.

CROUZET A., BOUDOURESQUE C.F., MEINESZ A., PERGENT G., 1983. Evidence of the annual character of cyclic changes of *Posidonia* oceanica scale thickness (erect rhizomes). Rapp PV Réun Commiss internation Explor sci Médit 28(3): 113-114.

CROWLEY T.J., 2002. Cycles, cycles everywhere. Science 295: 1473-1474.

CUENOD A., 1954. Flore analytique et synoptique de la Tunisie. S.E.F.A.N publ., Tunis : 1-267.

CUOMO V., VANZANELLA F., FRESI E., MAZZELLA L., SCIPIONE M.B., 1982. Microflora delle Fanerogame marine dell'isola d'Ischia : Posidonia oceanica (L.) Delile e Cymodocea nodosa (Ucria) Aschers. Bol Mus Ist Biol Univ Genova 50(suppl.): 162-166.

CURIEL D., RISMONDO A., SOLAZZI A., MARZOCCHI M., SCATTOLIN M., 1994. Valutazione dello stato di qualità dei popolamenti a Fanerogame marine in Laguna di Venezia e sperimentazione di reimpianto a Cymodocea nodosa, Zostera marina e Zostera noltii Biol mar Medit. 1(1): 407-408.

DAHDOUH-GUEBAS F., COPPEJANS E., VAN SPEYBROECK D., 1999. Remote sensing and zonation of seagrasses and algae along the Kenyan coast. *Hydrobiologia* 400: 63-76.

DANCE C., SAVY S., 1987. Predation of Paracentrotus lividus by Marthasterias glacialis: an in situ experiment at Port-Cros (France, Mediterranean). Posidonia Newletter 1(2): 35-41.

DANIEL B., BACHET F., 2003. Concession d'endigage et d'utilisation des dépendances du Domaine Public Maritime. Etude d'impact. Document d'incidence au titre du code de l'Environnement. Syndicat Mixte Parc Marin de la Côte Bleue, Fr.: 1-106 + Ann.

DARMOUL B., HADJ ALI SALEM M., VITIELLO P., 1980. Effet des rejets industriels de la Région de Gabès (Tunisie) sur le milieu marin récepteur. Bull Inst nat sci techn Océanogr Pêche Salambô 7: 5-61.

DARMOUL B., 1988. Pollution dans le golfe de Gabès (Tunisie). Bilan de six années de surveillance (1976-1981). Bull Inst nat sci techn Océanogr Pêche Salambô 15: 61-84.

DARWIN C., 1859. On the origin of species by means of natural selection, or the preservation of favoured races in the struggle for life John Murray publ., London: 1-502.

DAVEAU J., 1896. La flore littorale du Portugal. Bull Herbier Boissier 4: 209-228.

DAVIS R.C., SHORT FT., 1997. Restoring eelgrass, Zostera marina L., habitat usung a new transplanting technique: the horizontal rhizome

method. Aquat Bot. 59: 1-15.

DE GAULEJAC B., 1989. Ecologie de *Pinna nobilis*, mollusque eulamellibranche, en baie de Calvi. Etude de la coquille larvaire. Etude des possibilités de réimplantation de l'espèce. Mémoire Diplôme Etudes Approfondies Environnement Marin, Univ. Aix-Marseille 3, Fr.: 1-35.

DE LA POMÉLIE C., 1991. Les productions de la filière Loup-Daurade de 1985 à 1990. Equinoxe 35: 13-16.

- DELGADO O., GRAUA., POU S., RIERA F., MASSUTI C., ZABALA M., BALLESTEROS E., 1997. Seagrass regression caused by fish cultures in Fornells Bay (Menorca, Western Mediterranean). *Oceanol Acta* 20(3): 557-563.
- DELGADO O., RUIZ J., PEREZ M., ROMERO J., BALLESTEROS E., 1999. Effects of fish farming on seagrass (*Posidonia oceanica*) in a Mediterranean bay: seagrass decline after organic loading cessation. *Oceanol Acta* 22(1): 109-117.
- DE LUMLEY H., PILLARD B., PILLARD F., 1969. L'habitat et les activités de l'homme du Lazaret. Une cabanne acheuléenne de la grotte du Lazaret (Nice). Mém Soc Préhist Fr. 7: 183-222.
- DeMENOCAL P, ORTIS J., GUILDERSONT., SARNTHEIN M., 2000. Coherent high- and low-latitude climate variability during the Holocene warm periode. *Science* 288: 2198-1202.
- DENIS J., BERNARD G., BOUDOURESQUE C.F., CHARBONNEL E., BELSHER T., EMERY E., DENEUX F., CLABAUT P., 2001. Etude et cartographie des biocénoses de la rade de Toulon. Rapport méthodologique. Syndicat Intercommunal de l'Aire toulonnaise, Ifremer & GIS Posidonie, Ifremer publ., Fr.: 1-245.
- DENIS J., HERVÉ G., DENEUX F., SAUZADE D., BONHOMME P., BERNARD G., BOUDOURESQUE C.F., LERICHE A., CHARBONNEL E., LE DIREAC'H L., 2003. Guide méthodologique pour la cartographie des biocénoses marines. Volet n°1 : I herbier à *Posidonia oceanica*. Guide méthodologique. Agence de l'Eau, Région Provence Alpes-Côte d'Azur et DIREN PACA. Ifremer, GIS Posidonie & Centre d'Océanologie de Marseille, GIS Posidonie publ.: 1-93.
- DENNISON W.C., ALBERTE R.S., 1986. Photoadaptation and growth of *Zostera marina* L. (eelgrass) transplants along a depth gradient. J Exp Mar Biol Ecol 98: 265-282.
- DESCAMP P., PERGENT G., BALLESTA L., FOULQUIÉ M., 2005. Underwater acoustic positioning systems as tool for *Posidonia oceanica* bed survey. C R Biologies 328: 75-80.
- DIMECH M., BORG J.A., SCHEMBRI P.J., 2000a. Structural changes in a *Posidonia oceanica* meadow exposed to a pollution gradient from a marine fish-farm in Malta (Central Mediterranean). *Biol Mar Médit.* 7(2): 361-364.
- DIMECH M., BORG J.A., SCHEMBRI P.J., 2000b. The effects of a marine fish-farm on the species richness and abundance of molluscs, decapods and echinoderms associated with a *Posidonia oceanica* meadow in Malta (Central Mediterranean). *Biol mar Médit.* 7(2): 357-360.
- DIVIACCO G., SPADA E., VIRNO-LAMBERTI C., 1999. Descrizione e cartografia delle praterie di *Posidonia oceanica* (L.) Delile di Terracina (Lazio Meridionale). *Atti Associaz Ital Oceanol Limnol*. 13(1): 241-249.
- DIVIACCO G., 2000. Indagine sui popolamenti bentici del tratto costiero tra Punta Vescovado e Spotorno (SV), con particolare riferimento alle Fanerogame marine. Regione Liguria, Ufficio Parchi e Aree Protette: 1-13 + 14 pl. h.t.
- DIVIACCO G., SPADA E., VIRNO-LAMBERTI C., 2001. Le Fanerogame marine del Lazio. Descrizione e cartografia delle praterie di *Posidonia* oceanica e dei prati di *Cymodocea nodosa*. ICRAM publ., Ital.: 1-113 + 14 cartes h.t.
- DOSDAT A., KEMPF M., MERCERON M., PAQUOTTE .P, 1994. L'aquaculture française en bref. Aquaculture et Environnement, Ifremer publ.., Actes de colloques 16: 17-21.
- DOUMENGE F., 1992. Option pour rendre le tourisme compatible avec la protection de l'environnement marin. Atti II Conv Internaz ICIMAR, San Teodoro (Nuoro): 51-64.
- DREW E.A., JUPP B.P., 1976. Some aspects of the growth of *Posidonia oceanica* in Malta. *In*: Drew E.A., Lythgoe, Woods edits. *Underwater Research* Academic Press publ., London: 357-367.
- DUARTE C.M., 1991. Allometric scaling of seagrass form and productivity. Mar Ecol Prog Ser 77: 289-300.
- DUARTE C.M., CHISCANO C.L., 1999. Seagrass biomass and production: a reassessment. Aquat Bot 65: 159-174.
- DUARTE C.M., KIRKMAN H., 2001. Methods fort he measurement of seagrass abundance and depth distribution. In: Short F.T., Coles R.G. edits. Global seagrass research methods. Elsevier Scientific publ., Amsterdam: 141-153.
- DUARTE C.M., KIRKMAN H., 2003. Methods for the measurement of seagrass abundance and depth distribution. In: Short F.T., Coles R.G. edits. *Global seagrass research methods*. Elsevier Scientific publ., Amsterdam: 141-154.
- DUARTE C.M., 2004. Las praderas de Fanerógamas marinas. El papel de las praderas en la dinámica costera. *In:* Luque A.A., Templado J. edits. *Praderas y bosques marinos de Andalucía* Consejería de Medio Ambiente, Junta de Andalucía publ., Sevilla: 81-85.
- DREW E.A., JUPP B.P., 1976. Some aspects of the growth of *Posidonia oceanica* in Malta. *In: Underwater Research* Academic Press publ., London: 357-367.
- DURAND C., MANUEL M., BOUDOURESQUE C.F., MEINESZ A., VERLAQUE M., LE PARCO Y., 2002. Molecular data suggest a hybrid origin for the invasive *Caulerpa racemosa* (Caulerpales, Chlorophyta) in the Mediterranean Sea. *J. evol. Biol.* 15: 122-133.
- EL ASMI-DJELLOULI Z., DJELLOULI A.S., ABDELJAOUED S., 2000. Présentation des herbiers de la baie de Monastir (Tunisie). Proceedings of the first Mediterranean symposium on marine vegetation RAC/SPA publ., Tunis: 132-135.
- ELKALAY K., 2002. Modélisation de l'écosystème de la baie de Calvi (Méditerranée Nord-occidentale). Doctorat Université de Liège: 1-195.
- ELKALAY K., FRANGOULIS C., SKLIRIS N., GOFFART A., GOBERT S., LEPOINT G., HECQ J.H., 2003. A model of seasonal dynamics of biomass and production of the seagrass *Posidonia oceanica* in the Bay of Calvi (Northwestern Mediterranean). *Ecol Model* 167: 1-18.
- ESPER J., COOK E.R., SCHWEINGRUBER F.H., 2002. Low-frequency signals in long tree-ring chronologies for reconstructing past temperature variability. *Science* 295: 2250-2253.
- EUGÈNE C., 1979. Epifaune des herbiers de Posidonies du littoral provençal dans des secteurs pollués et non pollués Rapp P V Réun Commiss internation Explor Sci Médit. 25-26(4): 213-214.

FACCIOLI F, 1996. The morphological restoration of the Venice Lagoon. Quaderni trimestriali, Consorzio Venezia Nuova, Italie, suppl. 3-4: 1-24.
FARGHALY M.S., DENIZOT M., 1984. Posidonia oceanica (L.) Delile on the eastern coast of Libya. In: Boudouresque C.F., Jeudy de Grissac
A., Olivier J. edits. International Workshop on Posidonia oceanica beds, GIS Posidonia publ., Fr., 1: 419-422.

- FERGUSSON R.L., THAYER G.W., RICE T.R., 1980. Marine primary producers. In: Functional adaptations of marine organisms. Academic Press publ.: 9-69.
- FERNÁNDEZ-CASAS J., GAMARRA-GAMARRA R., MORALES-ABAD M.J., 1992. De flora iberica index Chartograficus. Treb Just bot, Barcelona 15: 5-422.
- FERRAT L., BINGERT A., ROMEO M., GNASSIA-BARELLI M., PERGENTMARTINI C., 2002a. Mercury uptake and enzymatic response of *Posidonia oceanica* after an experimental exposure to organic and inorganic forms. *Environm Toxicol Chem* 21(11): 2365-2371.

- FERRAT L., PERGENT-MARTINI C., FERNANDEZ C., ROMÉO M., 2002b. Is glutathione transferase (GST) activity in *Posidonia oceanica* a stress response to mercury exposure?. *Bull Mar Sci* 71(3): 1183-1190.
- FLAHAULT C., 1908. Gattung Posidonia Koenig. In: Kirchner O., Loew E., Schröter C. edits. Lebengeschichte der Blutenpflanzen Mittleeuropas. Band 1, Abteilung 1. Ulmer publ., Stuttgart: 537-543.
- FLORES-MOYA A., CONDE-POYALES F., 1998. Praderas de Fanerógamas marinas en el litoral andaluz. Conserejía de Medio Ambiente, Junta de Andalucía publ.: 1-127.
- FONSECA M.S., KENWORTHY W.J., HOMZIAK J., THAYER G.W., 1979. Transplanting of eelgrass and shoalgrass as a potential means of economically mitigating a recent loss of habitat. *In*: Cole D.P. edit. *Proc* 7th ann Conf Wetlands Restor. Creation publ., USA: 279-326.
- FONSECA M.S., FISHER J.S., ZIEMAN J.C., THAYER G.W., 1982a. Influence of the seagrass Zostera marina L. on current flow Est Coast Shelf Sci 15: 351-364.
- FONSECA M.S., KENWORTHY W.J., PHILLIPS R.C., 1982b. A cost-evaluation technique for restoration of seagrass and other plant communities. *Environm Cons.* 9(3): 237-241.
- FONSECA M.S., KENWORTHY W.J., THAYER G.W., 1982c. A low-cost planting technique for eelgrass (*Zostera marina* L.). National marine Fisheries Service, Southeast Fisheries Center, Beaufort Laboratory publ., USA: 1-15.
- FONSECA M.S., KENWORTHY W.J., THAYER G.W., 1987. Transplanting of the seagrasses *Halodule wrightii, Syringodium filiforme* and *Thalassia testudinum* for sediment stabilisation and habitat development in the southeast Region of the United States. Dept. of Army, U.S. Army Corps of Engineers, Technical Report EL-87-8: 1-59.
- FONSECA M.S., KENWORTHY W.J., THAYER G.W., 1996. Draft guidelines for mitigation and restoration of seagrass in the United States and adjacent waters. National Fisheries Service, NOAA Coastal Ocean Program Decision Analysis Seris, USA: 1-222.
- FONSECA M.S., KENWORTHYW.J., THAYER G.W., 1998. Guidelines for conservation and restoration of sea grasses in the United States and adjacent waters. National Oceanic and Atmospheric Administration, Silver Spring, Maryland, Analysis Series 12.
- FONT-QUER P., 1990. Plantas medicinales El Dioscórides renovado. Labor publ., Barcelona: cxl + 1-1033.
- FOUCHER M., 1975. Herbiers de Phanérogames marines et aménagement du littoral méditerranéen. Mémoire de Diplôme Etudes Approfondies Environnement Marin, Institut d'Aménagement régional, Univ. Aix-Marseille 3, Fr.: 1-68.
- FOULQUIE M., DUPUY DE LA GRANRIVE R., 2003. Site Natura 2000 FR 910 1414 "Posidonies du cap d'Agde". Document d'objectifs. Inventaire de l'existant et analyse écologique. ADENA publ., Agde : 1-301.
- FOURQUREAN J.W., MOORE T.O., FRY B., HOLLIBAUGH J.T., 1997. Spatial and temporal variation in C:N:P ratios, delta N-15 and delta C-13 of eelgrass Zostera marina as indicators of ecosystem processes, Tomales Bay, California, USA. Mar Ecol Progr Ser 157: 147-157.
- FRANCOUR P, 1984. Biomasse de l'herbier à *Posidonia oceanica* : données préliminaires pour les compartiments "matte", échinodermes et poissons. Mémoire Diplôme Etudes Approfondies Océanologie Biologique, Université Pierre et Marie Curie, Paris: 1-72.
- FRANCOUR P, PAUL O., 1987. Densité, biomasse et relation taille-poids chez l'oursin *Psammechinus microtuberculatus* de l'herbier à *Posidonia oceanica* de Port-Cros (France, Méditerranée). *In*: Boudouresque C.F. edit. *Colloque international sur Paracentrotus lividus et les oursins comestibles* GIS Posidonie publ., Fr.: 169-181.
- FRANCOUR P., MARCHADOUR M., 1989. Les fonds marins, et en particulier l'herbier à *Posidonia oceanica*, aux alentours du port de la Pointe Rouge (Marseille). Ville de Marseille & GIS Posidonie. GIS Posidonie publ., Fr.: 1-48.
- FRANCOUR P, 1990. Dynamique de l'écosystème à *Posidonia oceanica* dans le Parc national de Port-Cros. Analyse des compartiments "matte", litière, faune vagile, échinodermes et poissons. Thèse Doct. Océanol., Univ. Paris VI, Fr.: 1-373.
- FRANCOUR P, GRAVEZ V., 1990. Les fonds marins, et en particulier l'herbier à *Posidonia oceanica*, entre les ports de la Pointe Rouge et de La Madrague (Marseille, Méditerranée). Ville de Marseille & GIS Posidonie. GIS Posidonie publ., Fr.: 1-32.
- FRANCOUR P., NIÉRI M., URSCHELER F., 1991. Les structures anti-chalut du golfe de Beauduc (Camargue). Campagne de suivi 1991. Rapport préliminaire. GIS Posidonie publ., Fr.: 1-42.
- FRANCOUR P., SINNASSAMY J.M., URSCHELER F., FLEURY M.C., 1992. Cartographie des fonds et des herbiers à *Posidonia oceanica* entre la presqu'île de Giens et l'île de Porquerolles (Var) pour le passage d'un sea-line. Ville d'Hyères & GIS Posidonie, GIS Posidonie publ., Fr.: 1-72.
- FRANCOUR P, CHARBONNEL E., ABELLARD O., BOUDOURESQUE C.F., 1995. Cartographie de la limite supérieure de l'herbier de Posidonie au droit du port de l'Aygade (Var) et état initial des fonds – expertise du site. BETEREM & GIS Posidonie, GIS Posidonie publ., Fr.: 1-27.
- FRANCOUR P., 1997. Fish assemblages of *Posidonia oceanica* beds at Port Cros (France, NW Mediterranean): Assessment of composition and long-term fluctuations by visual census. *Mar Ecol*, *PSZNI* 18(2): 157-173.
- FRANCOUR P., POULAIN M., BERNARD G., BONHOMME P., CHARBONNEL E., 1997. Impact des mouillages forains sur l'herbier à *Posidonia oceanica* dans le Parc national de Port-Cros (Méditerranée Nord-occidentale). Contrat Parc national de Port-Cros & GIS Posidonie, GIS Posidonie publ., Fr.: 1-51.
- FRANCOUR P., BERASATEGUI C., BOUDOURESQUE C.F., LE DIREAC'H L., ARRIGHI F., DOMINICI J.M., FINELLI F., 1998. Impact des mouillages forains sur l'herbier à *Posidonia oceanica* dans la Réserve naturelle de Scandola (Méditerranée Nord-occidentale, Corse, france). GIS Posidonie publ., Fr.: 1-31.
- FRANCOUR P., GANTEAUME A., POULAIN M., 1999. Effects of boat anchoring in *Posidonia oceanica* seagrass beds in the Port-Cros National Park (Northwestern Mediterranean Sea). *Aquatic Conserv mar freshw Ecosyst* 9: 391-400.
- FRANCOUR P., SOLTAN D., 2000. Suivi des ancrages de type "Harmony" dans les herbiers à *Posidonia oceanica* de la rade d'Agay et du Parc national de Port-Cros (Var, Méditerranée Nord-occidentale). SMAT & Laboratoire Environnement Marin Littoral, LEML publ., Fr.: 1-33.
- FRANCOUR P., HARMELIN J.G., POLLARD D., SARTORETTO S., 2001. A review of marine protected areas in the Northwestern Mediterranean Region: siting, usage, zonation and management. *Aquatic Conserv mar freshw Ecosyst* 11: 155-188.
- FREDJ G., BELLAN-SANTINI D., MEINARDI M., 1992. Etat des connaissances sur la faune marine méditerranéenne. Bull Inst océanogr Monaco, num. spécial 9: 133-145.
- FRESI E., CHESSA L.A., SCARDI M., IMPAGLIAZZO C., 1984. Feeding ecology of *Palaemon xiphias* Risso, from a *Posidonia oceanica* meadow near Alghero (Sardinia). In: Boudouresque C.F., Jeudy de Grissac A., Olivier J. edits. International Workshop on Posidonia oceanica beds, GIS Posidonie publ., Fr., 1: 331-334.
- GACIA E., GRANATA T.C., DUARTE C.M., 1999. An approach to measurement of particule flux and sediment retention within seagrass (*Posidonia oceanica*) meadows. *Aquat Bot* 65: 255-268.
- GACIA E., DUARTE C.M., 2001. Sediment retention by a Mediterranean *Posidonia oceanica* meadow: the balance between deposition and resuspension. *Est Coast Shelf Sci* 52: 505-514.
- GALAN C., LOPEZ-IBOR A., TEMPLADO J., 1982. Primera cita en la peninsula ibérica de Asterina pancerii (Gasco, 1870) (Asteroidea,

Asterinidae). Actas Simp ibérico Est Bentos mar 3: 267-269.

GALLARATI-SCOTTI G., 1968. Primi risultati di prove di alimentazione con farina di Posidonia oceanica. Avicoltura 3: 85-93.

GAMBI M.C., BUIA M.C., CASOLA E., SCARDI M., 1989. Estimates of water movement in *Posidonia oceanica* beds: a first approach. *In:* Boudouresque, C.F., Meinesz A., Fresi E., Gravez V. edits. *International workshop on Posidonia beds*. GIS Posidonie publ., Fr., 2: 101-112.

- GAMBI M.C., NOWELL A.R.M., JUMARS P.A., 1990. Flume observations on flow dynamics in *Zostera marina* (eelgrass) beds. *Mar Ecol* Prog Ser. 61: 159-169.
- GAMBI M.C., GUIDETTI P., 1998. Morphological observations on seedlings of *Posidonia oceanica* (L.) Delile germinated in situ Biol mar Medit. 5(1): 549-552.
- GAMULIN-BRIDA H., POZAR A., SIMUNOVIC A., SPAN A., 1973. Les conséquences biologiques de la pollution des eaux marines sur les biocénoses benthiques. Groupement des biocénoses benthiques de la mer Adriatique du point de vue de pureté des eaux. Atti Coll internazion Oceanogr medit 5: 137-159.

GAMULIN-BRIDA H., 1974. Biocénoses benthiques de la mer Adriatique. Acta Adriatica 15(9): 1-102 + 1 carte.

- GANTEAUME A., BONHOMME P., HERVÉ G., EMERY E., BOUDOURESQUE C.F., 2004. Détermination d'une zone d'amarrage pour les paquebots de croisière face au port de Portquerolles. Préconisations techniques. GIS Posidonie publ., Fr.: 1-47.
- GANTEAUME A., BONHOMME P., BERNARD G., POULAIN M., BOUDOURESQUE C.F., 2005. Impact de l'ancrage des bateaux de plaisance sur la prairie à *Posidonia oceanica* dans le Parc national de Port-Cros (Méditerranée Nord-occidentale). *Sci Rep Port Cros natl Park*, 21: 147-162.

GARDNER W.D., 1980. Sediment map dynamics and calibration: a laboratory evaluation. J. mar. Res., USA, 38: 17-39.

- GELIN A., ARIGONI S., FRANCOUR P., HARMELIN J.G., HARMELIN-VIVIEN M., 1998. Réponse des populations de certains poissons Serranidae et Labridae à la colonisation des fonds par *Caulerpa taxifolia* en Méditerranée. *In* Boudouresque C.F., Gravez V., Meinesz A., Palluy F. edits. *Third international workshop on Caulerpa taxifolia* GIS Posidonie publ., Fr.: 197-208.
- GÉNOT I., CAYE G., MEINESZ A., ORLANDINI M., 1994. Role of chlorophyll and carbohydrate contents in survival of *Posidonia oceanica* cuttings planted to different depths. *Mar Biol* 119: 23-29.
- GERMAIN DE SAIN-PIERRE E., 1857. Sur la germination et le mode de développement de *Posidonia caulini Bull Soc bot Fr* 4: 575-577. GESAMP, 1975. Scientific criteria for the selection of sites for dumping of wastes into the sea. IMCO / FAO / UNESCO / WMO / WHO / IAEA / UN Joint group of experts on the scientific aspects of marine pollution (GESAMP). *Rep Stud GESAMP* 3: 1-21.
- GESAMP, 1982. Scientific criteria for the selection of waste disposal sites at sea. IMCO / FAO / UNESCO / WMO / WHO / IAEA / UN / UNEP Joint group of experts on the scientific aspects of marine pollution (GESAMP). Rep Stud GESAMP 16: 1-60.

GIACCONE G., CALVO S., 1980. Restaurazione del manto vegetale mediante trapianto di *Posidonia oceanica* (Linneo) Delile. Risultati preliminari. *Mem Biol Mar Oceanogr*, Ital., suppl. 10: 207-211.

GIACCONE G., COSSU A., DE LORENZO R., DRAGO D., MUSCETTA P.N., 1988. Studio dell'impatto ambientale nell'area portuale antistante lo stabilimento industriale EniChem di Porto Torres. Boll Soc Sarda Sci Nat 26: 43-67.

GIGLIO Y., 1985. Action des peintures antifouling sur l'herbier de Posidonies. Mémoire Diplôme d'Etudes Approfondies en Ecologie méditerranéenne, option Ecologie Littorale, Univ. Aix-Marseille III, Fr.: 1-57.

- GIORGI J., THELIN I., 1983. Phénologie, biomasse et production primaire de *Posidonia oceanica* (feuilles et épiphytes) dans la baie de Port-Cros. Mémoire Diplôme d'Etudes Approfondies en Ecologie méditerranéenne, Univ. Aix-Marseille III, Fr.: 1-126.
- GIRAUD G., 1977a. Contribution à la description et à la phénologie des herbiers de Posidonia oceanica (L) Delile. Thèse Doct. Spécialité, Univ. Aix-Marseille II, Fr.: 1-150.
- GIRAUD G., 1977b. Essai de classement des herbiers de Posidonia oceanica (Linné) Delile. Bot Mar 20(8): 487-491.
- GIRAUD G., 1977c. Recensement des floraisons de Posidonia oceanica (Linné) Delile en Méditerranée. Rapp PV Réun Commiss internation Explor sci Médit 24(4): 126-130.
- GIRAUD G., 1979. Sur une méthode de mesure et de comptage des structures foliaires de *Posidonia oceanica* (Linnaeus) Delile. Bull Mus Hist nat Marseille, Fr., 39: 33-39.
- GIRAUD G., BOUDOURESQUE C.F., CINELLI F., FRESI E., MAZZELLA L., 1979. Observations sur l'herbier de Posidonia oceanica (L.) Delile autour de l'île d'Ischia (Italie). G bot ital 113(4): 261-274.
- GOBERT S., BELKHIRIA S., DAUBY P., HAVELANGE S., SOULLARD M., BOUQUEGNEAU J.M., 1995. Variations temporelles de la phénologie et de la composition biochimique de la Phanérogame marine *Posidonia oceanica* en baie de Calvi. *Bull Soc Royale Sci Liège* 64(4-5): 263-284.

GOBERT, S., 2002. Variations spatiales et temporelles de l'herbier à Posidonia oceanica (L.) Delile. Thèse Doctorat Université de Liège: 1-207.

- GOBERT S., LEJEUNE P., LEPOINT G., BOUQUEGNEAU J.M., 2005. C, N, P concentrations and requirements of flowering Posidonia oceanica shoots. Hydrobiologia 533: 253-259.
- GONGORA-GONZALEZ E., IMMORDINO F., PEIRANO A., STOPPELLI N., 1996. Granulometric and geomorphologic features of the bay of Monterosso al Mare (Ligurian Sea) and their relationship with the evolution of *Posidonia oceanica* meadow. in: Albertelli G., De Malo A., Picazzo M. edits. *Atti 11* Congresso dell Associazione italiana di Oceanologia e Limnologia* 26-28 Ottobre 1994, Genova: 395-404.
- GOWEN R.J., BRADBURY N.B., 1987. The ecological impact of salmonid farming in coastal waters: A review. Oceanogr Mar Biol Ann Rev. 25: 563-575.
- GRAVEZ V., MARCHADOUR M., FRANCOUR P., BOUDOURESQUE C.F., BOULADIER E., SOURENIAN B., 1988. L'herbier à *Posidonia* oceanica du golfe de Giens. Rejets des effluents urbains dans le golfe de Giens, analyse de l'état initial. BCEOM & GIS Posidonie, GIS Posidonie publ., Fr.: 1-113.
- GRAVEZ V., NIERI M., BOUDOURESQUE C.F., 1992. Surveillance de l'herbier de Posidonie de la baie du Prado (Marseille). Rapport de synthèse 1986-1992. Direction Générale des Services techniques de la Ville de Marseille & GIS Posidonie, GIS Posidonie publ., Fr. : 1-80.
- GRAVEZV., BOISSERY P, FRANCOUR P, SINNASSAMY J.M., BOUDOURESQUE C.F., 1993. Surveillance de l'herbier à Posidonia oceanica du golfe de Giens (Var, France). Premier suivi. Syndicat Intercommunal Hyères-Carqueiranne pour l'assainissement de la baie de Giens & GIS Posidonie, GIS Posidonie publ., Fr.: 1-59.
- GRAVEZ V., GÉLIN A., CHARBONNEL E., FRANCOUR P., ABELLARD O., EEMONNAY L., 1995. Surveillance de l'herbier de Posidonie de la baie du Prado (Marseille). Deuxième phase. Suivi 1995. Ville de Marseille & GIS Posidonie, GIS Posidonie publ., Fr.: 1-56.
- GRAVEZ V., BERNARD G., BONHOMME P., PALLUY F., 1997. Surveillance de l'herbier de Posidonie de la baie du Prado (Marseille). Deuxième phase. Suivi 1997. Ville de Marseille & GIS Posidonie, GIS Posidonie publ., Fr.: 1-46.
- GRAVEZ V., BERNARD G., BONHOMME P., CADIOU G., COQUILLARD Y., DIVETAIN N., 1999. Surveillance de l'herbier de Posidonie de la baie du Prado (Marseille) - Suivi 1999. Ville de Marseille, Direction de l'Environnement et des Déchets & GIS Posidonie, GIS

Posidonie publ., Fr.: 1-45.

GRENIER C., 1860. Recherches sur Posidonia caulini Koenig. Bull Soc bot Fr 7(5-6): 362-367.

GUÉNA A., THOMAS C., 1997a. Ces "macrodéchets" qui ruinent notre littoral. 1" volet, Atlantique. Mer et Littoral, Fr., 23: 65-70.

GUÉNA A., THOMAS C., 1997b. Ces "macrodéchets" qui ruinent notre littoral. 2^{ama} volet, Manche-Mer du Nord. Mer et Littoral, Fr., 24: 49-54.

GUÉROUT M., 1981. Le combat de la flûte du Roi "La Baleine" à Port-Cros le 4 août 1710. Trav Sci Parc nation Port Cros, Fr., 7: 147-175.

GUIDETTI P., 2001. Detecting environmental impacts on the Mediterranean seagrass *Posidonia oceanica* (L.) Delile: the use of reconstructive methods in combination with "beyond BACI" designs. *J exp mar Biol Ecol* 260: 27-39.

GUILLÉN J., RAMOS A., MARTINEZ L., SÁNCHEZ-LIZASO J., 1994. Antitrawling reefs and the protection of *Posidonia oceanica* (L.) Delile meadows in the Western Mediterranean Sea: demands and aims. *Bull mar Sci* 55: 645-650.

HAMOUTENE D., MATHIEU A., HOFMANN P., SALAUN J.P., LAFAURIE M., 1995. Preparation and characterization of subcellular fractions suitable for studies of xenobiotic metabolism from leaf sheaths of a marine seagrass: *Posidonia oceanica* (Linnaeus) Delile. *Mar Environ Res* 39(1-4): 249-253.

HANDY R.D., POXTON M.G., 1993. Nitrogen pollution in mariculture – toxicity and excretion of nitrogenous compounds by fish. *Reviews* in Fish Biology and Fisheries 33: 205-241.

HARMELIN J.G., 1964. Etude de l'endofaune des "mattes" d'herbiers de Posidonia oceanica Delile. Rec Trav Stat mar Endoume 35(51): 43-106.

HARMELIN J.G., TRUE M.A., 1964. Délimitation cartographique de l'extension actuelle des herbiers de Posidonia oceanica (Delile) dans le golfe de Marseille. Rec Trav Stn mar Endoume 34(50): 157-160.

HARMELIN J.G., 1976. L'herbier de Posidonies de Port-Cros régresse-t-il ? Mise en place de jalons témoins à sa limite. Trav sci Parc nation Port Cros, Fr., 2: 189-190.

HARMELIN J.G., LABOREL J., 1976. Note préliminaire sur la morphologie de l'herbier profond de Posidonies Posidonia oceanica (Linné) Delile, à Port-Cros. Trav sci Parc nation Port Cros, Fr., 2: 105-113.

HARMELIN J.G., 1977. Evolution de l'herbier de Posidonies de Port-Cros au niveau de jalons témoins. Trav sci Parc nation Port Cros, Fr., 3: 210-211.

HARMELIN J.G., BOUCHON C., HONG J.S., 1981. Impact de la pollution sur la distribution des échinodermes des substrats durs en Provence (Méditerranée Nord-occidentale). Téthys 10(1): 13-36.

HARMELIN J.G., 1993. Invitation sous l'écume. Parc national Port Cros, Cahier découverte 10: 1-83.

HARMELIN J.G., SARTORETTO S., FRANCOUR P., 1996. Patrimoine biologique marin de l'archipel de Riou: première évaluation. Ville de Marseille, Direction de l'Environnement et des Déchets, Fr.: 1-86.

HARMELIN-VIVIEN M., 1982. Ichtyofaune des herbiers de Posidonies du Parc national de Port-Cros: I. Composition et variations spatiotemporelles. Trav sci Parc nation Port Cros, Fr., 8: 69-92.

HARMELIN-VIVIEN M.L., 1983. Ichtyofaune des herbiers de Posidonies des côtes provençales françaises. Rapp PV Réun Commiss internation Explor sci Médit 28(3): 161-163.

HARMELIN-VIVIEN M.L., 1984. Ichtyofaune des herbiers de Posidonies du Parc naturel régional de Corse. In: Boudouresque C.F., Jeudy de Grissac A., Olivier J. edits. International Workshop on Posidonia oceanica beds, GIS Posidonie publ., Fr., 1: 291-301.

HARMELIN-VIVIEN M.L., KAIM-MALKA R.A., LEDOYER M., JACOB-ABRAHAM S.S., 1989. Food partitioning among scorpaenid fishes in Mediterranean seagrass beds. J Fish Biol 34: 715-734.

HARMELIN-VIVIEN M.L., FRANCOUR P., 1992. Trawling or visual censuses ? Methodological bias in the assessment of fish populations in seagrass beds. *Mar Ecol*, *PSZN* 13(1): 41-51.

HARMELIN-VIVIEN M., FRANCOUR P., HARMELIN J.G., 1999. Impact of *Caulerpa taxifolia* on Mediterranean fish assemblages: a six year study. In: Proceedings of the workshop on invasive Caulerpa in the Mediterranean Heraklion, Crete, Greece, 18-20 March 1998. UNEP publ., Athens, Greece: 127-138.

HARTOG C. den, 1970. The sea grasses of the world. North Holland publ. Co, Amsterdam, Pays-Bas: 1-275 + 63 pl. h.t.

HARTOG C. den, 2000. Procedures for the restoration of lost seagrass beds. Biol mar Medit. 7(2): 353-356.

HAVELANGE S., LEPOINT G., DAUBY P., BOUQUEGNEAU J.M., 1997. Feeding of the sparid fish Sarpa salpa in a seagrass ecosystem : Diet and carbon flux. Mar Ecol , PSZNI 18(4): 289-297.

HECK K.L., WETSTONE G.S., 1977. Habitat complexity and invertebrate species richness and abundance in tropical seagrass meadows. J Biogeogr. 4: 135-142.

HEIP C., 1998. Un océan de diversité. Biofutur 179: 12-17.

HEMMINGA M.A., 1998. The root/rhizome system of seagrasses: an asset and a burden. J Sea Res 39: 183-196.

HEVIA M., ROSENTHAL H., GOWEN R.J., 1996. Modelling benthic deposition under fish cages. J Appl Ichthyol 12: 71-74.

HURRELL J.W., KUSHNIR Y., VISBECK M., 2001. The North Atlantic Oscillation. Science 291: 603-604.

ISSEL R., 1912. Il benthos animale delle foglie di Posidonia studiato dal punto di vista bionomico. Zool Jahrb., 33: 379-420.

ISSEL R., 1918a. La vita nelle paraterie di Posidonia. Biologia marina, capitolo XIV, XV: 405-457.

ISSEL R., 1918b. Biologia marina. Hoepli publ., Milano: 1-607.

IZZO G., NICOLAI P., 1993. Etat actuel des connaissances sur l'état du fleuve Tibre. Rencontres de l Agence régionale pour l Environnement Provence Alpes Côte d Azur, Fr., 5: 34-44.

- JEUDY DE GRISSAC A., 1984. Effets des herbiers à *Posidonia oceanica* sur la dynamique marine et la sédimentologie littorale. *In:* Boudouresque C.F., Jeudy de Grissac A., Olivier J. edits. *International Workshop on Posidonia oceanica beds*, GIS Posidonie publ., Fr., 1: 437-443.
- JEUDY DE GRISSAC A., 1984. Essais d'implantations d'espèces végétales marines: les espèces pionnières, les Posidonies. In: Boudouresque C.F., Jeudy de Grissac A., Olivier J. edits. International Workshop on Posidonia oceanica beds, GIS Posidonie publ., Fr., 1: 431-436.

JEUDY DE GRISSAC A., AUDOLY G., 1985. Etude préliminaire des banquettes de feuilles mortes de *Posidonia oceanica* de la Région de Marseille, France. *Rapp P V Réun Commiss internation Explor sci Mer Médit.* 29(5): 181-182.

JEUDY DE GRISSAC A., BOUDOURESQUE C.F., 1985. Rôle des herbiers de Phanérogames marines dans les mouvements de sédiments côtiers : les herbiers à *Posidonia oceanica Colloque franco japonais d Océanographie*, Marseille, Fr., 1: 143-151.

JIMENEZ S., CANO R., BAYLE J., RAMOS A., SANCHEZ LIZASO J.L., 1996. Las praderas de *Posidonia oceanica* (L.) Delile como zona de protección de juveniles de especies de interés comercial. *Real Soc esp Hist nat*, tomo extraordinario: 375-378.

JIMENEZ S., BAYLE J.T., RAMOS ESPLA A.A., SANCHEZ LIZASO J.L., 1997. Ictiofauna de dos praderas de *Posidonia oceanica* (L.) Delile, 1813 con distinto grado de conservación. *Publ espec Inst esp Oceanogr*, 23: 255-264.

JUPP B.P., 1977. The effects of organic pollution on benthic organisms near Marseille. Internation J Environm Studies 10: 119-123.

KAISER M.J., 1998. Scraping the bottom: are towed fishing gears a threat to benthic biodiversity ? Océanis 24(4): 258-270.

KARAKASSIS I., 1998. Aquaculture and coastal marine biodiversity. Océanis 24(4): 271-286.

KARAKASSIS I., TSAPAKIS M., HATZIYANNI E., PAPADOPOULOU K.N., PLAITI W., 1999. Benthic recovery following cessation of fish farming: a series of successes and catastrophes. *Mar Ecol Prog Ser.* 184: 205-218.

KARAKASSIS I., TSAPAKIS M., HATZIYANNI E., PAPADOPOULOU K.N., PLAITI W., 2000. Impact of cage farming of fish on the seabed in three Mediterranean coastal areas. ICES J mar Sci 57: 1462-1471.

KARAKASSIS I., TSAPAKIS M., SMITH C.J., RUMOHR H., 2002. Fish farming impacts in the Mediterranean studied through sediment profiling imagery. *Mar Ecol Progr Ser* 227: 125-133.

KAWASAKIY, IITUKAT., GOTO H., TERAWAKIT., WATANABEY, KIKUTI K., 1988. Study on the technique for Zostera bed creation. Central Res Inst electric Power Industry, Japon, Rep. nº U-14: 1-231.

KENDRICK G.A., AYLWARD M.J., HEGGE B.J., CAMBRIDGE M.L., HILLMAN K., WYLLIE A., LORD D.A., 2002. Changes in seagrass coverage in Cockburn Sound, Western Australia between 1967 and 1999. Aquat Bot 73: 75-87.

KENWORTHY W.J., FONSECA M.S., HOMZIAK J., THAYER G.W., 1980. Development of a transplanted seagrass (Zostera marina L.) meadow in Back Sound, Carteret County, North Carolina. In: Cole D.P. edit. Proc. 7th ann Conf. Wetlands Restor Creation publ., USA: 175-193.

KITTING C.L., FRY B., MORGAN M.D., 1984. Detection of inconspicuous epiphytic algae supporting food webs in seagrass meadows. *Oecologia* 62: 145-149.

KNOCHE H., 1923. Flora balearica Etude phytogéographique sur les îles baléares III Partie générale Roumegous and Dehan publ., Montpellier: 1-411.

KOMATSUT, 1996. Influence of a Zostera bed on the spatial distribution of water flow over a broad geographic area. In: Kuo J., Phillips R.C., Walker D.I., Kirkman H. edits. Seagrass biology proceedings of an international workshop: 111-116.

KRIJGSMAN W., HILGEN F.J., RAFFI I., SIERRO F.J., WILSON D.S., 1999. Chronology, causes and progression of the Messinian salinity crisis. Nature 400: 652-655.

KUO J., HARTOG C. den, 2001. Seagrass taxonomy and identification key. In: Short, Coles, Short edits. Global seagrass research methods. Elsevier publ., Amsterdam: 31-58.

LABORELDEGUEN F., LABOREL J., 1977. Broutage des Posidonies à la plage du Sud. Trav Sci Parc national de Port Cros, Fr., 3: 213-214.

LAMBINON J., 1994. L'introduction et la réintroduction d'espèces vivantes: remède à la perte de biodiversité, ou bien déviation des objectifs et de l'éthique de la conservation de la nature ? Ann Gembloux 99: 71-95.

LAMI R., 1941. L'utilisation des végétaux marins des côtes de France. Rev Bot appl Agric tropic 21(243-244): 653-670.

LAMY M., 1999. La biosphere, la biodiversité et l homme. Ellipses publ., Paris: 1-191.

LARKUM A.W.D., 1976. Ecology of Botany Bay. 1. Growth of *Posidonia australis* (Brown) Hook f. in Botany Bay and other bays of the Sidney Basin. *Austr J Mar Freshwater Res.* 27: 117-127.

LARKUM A.W.D., WEST R.J., 1990. Long-term changes of seagrass meadows in Botany Bay, Australia. Aquat Bot 37: 55-70.

LAWTON J.H., 1994. What do species do in ecosystems ? Oikos 71: 367-374.

LECOINTRE G., LE GUYADER H., 2001. Classification phylogénétique du vivant. Bélin publ., Paris: 1-543 + 15 pl. h.t.

LE DIREACH L., FRANCOUR P., 2001. Recrutement des poissons dans les herbiers de Posidonie de Port-Cros. GIS Posidonie & Parc national de Port-Cros, GIS publ., Fr.: 1-40.

- LEFÉVRE J.R., VALERIO C., MEINESZA., 1984. Optimisation de la technique de la photographie aérienne pour la cartographie des herbiers de Posidonies. *In*: Boudouresque C.F., Jeudy de Grissac A., Olivier J. edits. *International Workshop on Posidonia oceanica beds*, GIS Posidonie publ., Fr., 1: 49-55.
- LE FLOCH E., 1983. Contribution à une étude ethnobotanique de la flore tunisienne. Ministère de l'Education Nationale et de la Recherche Scientifique publ., Tunis: xv + 1-402.

LEHMANN A., LACHAVANNE J.B., 1997. Geographic information systems and remote sensing in aquatic botany. Aquat Bot 58: 195-207. LEJEUNE P., 1985. Le comportement social des Labridés méditerranéens. Cah Ethol appl 5(2): I-xii + 1-208.

LEPOINT G., HAVELANGE S., GOBERT S., BOUQUEGNEAU J.M., 1999. Fauna vs flora contribution to the leaf epiphytes biomass in a *Posidonia oceanica* seagrass bed (Revellata Bay, Corsica). *Hydrobiologia* 394: 63-67.

LERICHE A., 2004. La structure à méga-, méso- et micro-échelles des herbiers à *Posidonia oceanica* (L.) Delile fournit-elle des outils pour leur gestion ? Thèse Doctorat , Univ. Aix-Marseille II, Fr.: 1-232.

LERICHE A., BOUDOURESQUE C.F., BERNARD G., BONHOMME P., DENIS J., 2004. A one century suite of seagrass bed maps: can we trust ancient maps? *Est Coast Shelf Sci* 59: 353-362.

LERICHE-GUICHARD A., 2001. Mise en place d'un outil de surveillance de l'herbier à *Posidonia oceanica* (L.) Delile: le SIG Posidonie. Application à la Côte Bleue (B.d.R., Fr.). Mémoire Diplôme d'Etudes Approfondies en Biosciences de l'Environnement, Chimie et Santé, Univ. Aix-Marseille II, Fr.: 1-45.

LEWIS R.R., 1987. The restoration and creation of seagrass meadows in the Southeast United States. Florida Mar Res Publ. 4: 153-173. LEWIS R.R., PHILLIPS R.C., 1980. Experimental seagrass mitigation in the Florida keys. In: Cole D.P. edit. Proc 7th ann Conf Wetlands Restor. Creation publ., USA: 155-173.

LIBES M., 1984. Production primaire d'un herbier à *Posidonia oceanica* mesurée *in situ* par la méthode du carbone 14. Thèse Doctorat Spécialité Ecologie, Univ. Aix-Marseille II, Fr.: 1-199.

LOQUES F., BELLONE E., CAYE G., MEINESZ A., 1989. Transplantation de boutures de *Posidonia oceanica* sur "matte morte". Laboratoire Environnement Marin Littoral, Université de Nice-Sophia Antipolis, Fr.: 1-31.

LORENTI M., FRESI E., 1983. Grazing of Idotea baltica basteri on Posidonia oceanica: preliminary observations. Rapp PV Réun Commiss internation Explor sci Médit 28(3): 147-148.

LUQUE A.A., TEMPLADO J., 2004. Praderas y bosques marinos de Andalucía. Conserejía de medio ambiente, Junta de Andalucía, Sevilla: 1-334.

Mac ARTHUR R.H., WILSON E.O., 1967. The theory of Island biogeography. Princeton Univ. Press publ., USA.

MACHIAS A., KARAKASSIS I., LABROPOULOU M., SOMARAKIS S., PAPADOPOULOU K.N., PAPACONSTANTINOU C., 2004. Changes in wild fish assemblages after the establishment of a fish farming zone in an oligotrophic marine ecosystem. *Est Coast Shelf Sci* 60: 771-779.

MACHIAS A., KARAKASSIS I., GIANNOULAKI M., PAPADOPOULOU K.N., SMITH C.J., SOMARAKIS S., 2005. Response of demersal fish communities to the presence of fish farms. *Mar Ecol Progr Ser* 288: 241-250.

MAGGI P, 1973. Le problème de la disparition des herbiers à Posidonies dans le golfe de Giens (Var) Sci Pêche, Fr., 221: 7-20.

MAGGIONCALDA D., 2002. Analisi storica dell'evoluzione dell'ambiente marino costiero in relazione agli usi antropici in un tratto di costa ligure : Noli - Spotorno (SV). Tesi di laurea.

MAKKAVEEVA E.B., 1968. Rapports entre les composants végétaux et animaux dans les biocoenoses d'herbier des mers du bassin méditerranéen. Rapp P V Réun Commiss internation Explor sci Médit 19(2): 101-102.

MARION A.F., 1883. Esquisse d'une topographie zoologique du golfe de Marseille. Ann Mus Hist nat Marseille, Fr., 1: 6-108 + 1 carte. MARIOTTI M., ARILLO A., PARISI V., NICOSIA E., DIVIACCO G., 2002. Biodiversità in Liguria. La rete Natura 2000. Assessorato Ambiente

e Territorio, Dipartimento Tutela dell'Ambiente e Territorio, Dipartimento Tutela dell'Ambiente e Edilizia, Serrore Politiche e Programmi Ambientali, Microart's S.p.A., Recco (GE): 1-300 + CD.

MARTIN M.A, SANCHEZ-LIZASO J.L, RAMOS-ESPLÀ A.A., 1997. Cuantificación del impacto de las artes de arrastre sobre la pradera de Posidonia oceanica (L.) Delile. Publ Espec Inst Esp Oceanogr 23: 243-253.

MASERTI B.E., FERRARA R., PATERNO P., 1988. Posidonia as an indicator of mercury contamination. Mar Poll Bull. 19(8): 381-382.

MASSÉ H., 1962. Cartographie bionomique de quelques fonds meubles de la partie Sud orientale du golfe de Marseille. Rec Trav Stat Mar Endoume 27(42): 221-259.

MATEO, M.A., SABATE, S., 1993. Wet digestion of vegetable tissue using a domestic microwave oven. Analytica Chimica Acta 279: 273-279.

MATEO M.A., ROMERO J., 1996. Evaluating seagrass leaf litter decomposition : An experimental comparison between litter-bag and oxygenuptake methods. J exp mar Biol Ecol 202 (2) : 97-106.

MATEO M.A., ROMERO J., PEREZ M., LITTLER MM., LITTLER D.S., 1997. Dynamics of millenary organic deposits resulting from the growth of the Mediterranean seagrass *Posidonia oceanica Est Coast Shelf Sci* 44(1): 103-110.

MATEO-MINGUEZ M.A., 1995. El compartimento detrítico en ecosistemas de Fanerógamas marinas mediterráneas. Tesis Doctoral, Univ. Barcelona, Esp.: 1-190.

MATTEUCCI G., 2000. Studio e controllo degli effett dello scarico di materiali di dragaggio portuali sui fondali di un'area marina: un caso di studio in Adriatico. In: Dragaggi portuali qualità e gestione dei materiali, Atti Giornata 18 febbraio 1998. ICRAM publ.: 75-90.

MAUVAIS J.L., 1990. Ports de plaisance. Conséquences sur le milieu marin. Outils de prévision. Equinoxe, Fr., 30: 8-13.

MAYOT N., BOUDOURESQUE C.F., LERICHE, 2005 Unexpected response of the seagrass *Posidonia oceanica* to a warm Water episode in the Northwestern Mediterranean Sea. *C R Biologies* 328: 291-296.

MAZZELLA L., GAMBI M.C., RUSSO G.F., WITTMAN J.K., 1983. Flowering in *Posidonia oceanica* (L.) Delile prairies around the Island of Ischia (Gulf of Naples). *Rapp PV Réun Commiss internation Explor sci Médit* 28(3): 117-119.

- MAZZELLA L., GAMBI M.C., RUSSO G.F., BUIA M.C., 1984. Deep flowering and fruiting of *Posidonia oceanica* beds around the Island of Ischia (Gulf of Naples). *In*: Boudouresque C.F., Jeudy de Grissac A., Olivier J. edits. *International Workshop on Posidonia oceanica* beds, GIS Posidonie publ., Fr., 1: 203-209.
- MAZZELLA L., OTT J.A., 1984. Seasonal changes in some features of *Posidonia oceanica* (L.) Delile leaves and epiphytes at different depths. *In*: Boudouresque C.F., Jeudy de Grissac A., Olivier J. edits. *International Workshop on Posidonia oceanica beds*, GIS Posidonie publ., Fr., 1: 119-127.

MAZZELLA L., SCIPIONE M.B., GAMBI M.C., FRESI E., BUIA M.C., RUSSO G.F., DE MAIO R., LORENTI M., RANDO A., 1986. Le praterie sommerse del Mediterraneo. Stazione zoologica di Napoli publ.: 1-63.

MAZZELLA L., BUIA M.C., GAMBI M.C., LORENTI M., RUSSO G., SCIPIONE M.B., ZUPO V., 1992. Plant-animal trophic relationships in the *Posidonia oceanica* ecosystem of the Mediterranean Sea: a review. *In:* Jangoux M., Mazzella L. *et al* edits. *Plant animal interactions in marine benthos* Systematic Association special volume 46, Clarendon Press publ.: 165-187.

MAZZELLA L., BUIA M.C., GAMBI M.C., LORENTI M., RUSSO G., SCIPIONE M.B., ZUPO V., 1995. A review of the trophic organization in the *Posidonia oceanica* ecosystem. *In:* Cinelli F, Fresi E., Lorenzi C., Mucedola A. edits. *La Posidonia oceanica*, Revista marittima publ., Ital., 12(suppl.): 31-47.

MAZZOLA A., MIRTO S., LA ROSAT., FABIANO M., DANOVARO R., 2000. Fish-farming effects on benthic community structure in coastal sediments: analysis of meiofaunal recovery. *ICES J mar Sci* 57: 1454-1461.

McDERMOTT F., MATTEY D.P., HAWKESWORTH C., 2001. Centennial-scale Holocene climate variability revealed by a high-resolution speleothem ¹⁸O record from SW Ireland. *Science* 294: 1328-1330.

McKENZIE J.A., 1999. From desert to deluge in the Mediterranean. Nature 400: 613-614.

McKENZIE L.J., FINKBEINER M.A., KIRKMAN H., 2001. Methods for mapping seagrass distribution. In: Short F.T., Coles R.G. edits. Global seagrass research mthods. Elsevier publ., Amsterdam: 101-121.

McMILLAN C., LIPKIN Y., BRAGG L.H., 1975. The possible origin of pecular *Thalassia testudinum* from Texas as *Posidonia oceanica* Contrib mar Sci 19: 101-106.

McREA J.E., GREENE H.G., O'CONNELL V.M., WAKEFIELD W.W., 1999. Mapping marine habitats with high resolution sidescan sonar. Oceanol Acta 22(6): 679-686.

McROY C.P., McMILLAN C., 1977. Production ecology and physiology of seagrasses. In: McRoy C.P., Helfferich C. edits. Seagrass ecosystem a scientific perspective. Marcel Dekker publ., New York: 53-87.

McROY C.P., HELFFERICH C., 1980. Applied aspects of seagrasses. In: Phillips R.C., McRoy C.P. edits. Handbook of seagrass biology an ecosystem perspective. Garland STPM Press publ., New York: 297-343.

MEINESZ A., 1976. Note préliminaire concernant le repiquage de végétaux marins, en particulier de l'algue Caulerpa prolifera Rapp P V Réun Comm internation Explor sci Médit 24(4): 169-170.

MEINESZ A., LEFÈVRE J.R., 1976a. Inventaire des restructurations et impacts sur la vie "sous-marine" littorale. Alpes-Maritimes et Principauté de Monaco. DDE-CIPALM & Université de Nice publ.: 1-67 + 17 pl. h. t.

MEINESZ A., LEFÈVRE J.R., 1976b. L'aménagement de la côte entre Menton et Théoule (Alpes Maritimes et Monaco). Inventaire des restructurations du rivage et impacts sur la vie sous marine littorale. SOS Vie Nature Environnement 1 (num. spéc.): 1-35.

MEINESZ A., 1977. Balisage de la limite inférieure de l'herbier de *Posidonia oceanica* en rade de Villefranche-sur-Mer (Alpes-Maritimes, France). Rapp Comm int Explor Mer Médit 24(6): 143-144.

MEINESZ A., LAURENT R., 1977. Compte-rendu de l'opération "Poséidon 1977". Deuxième campagne de cartographie de la limite inférieure de l'herbier de *Posidonia oceanica* dans les Alpes-Maritimes. Cellule d'Intervention contre la Pollution dans les Alpes-Maritimes: 1-9 + 1 carte h.t.

MEINESZ A., 1978. Etude expérimentale de bouturage de certains végétaux sous-marins dans les ports et les plages artificielles. 15^{eme} Journées de l'Hydraulique, Rev Soc Hydrotechnique Fr 5(9-1): 4.

MEINESZ A., LAURENT R., 1978. Cartographie et état de la limite inférieure de l'herbier de *Posidonia oceanica* dans les Alpes-Maritimes. Campagne Poséïdon 1976. *Bot mar* 21(8): 513-526.

MEINESZ A., LEFÉVRE J.R., 1978. Destruction de l'étage infralittoral des Alpes-Maritimes (France) et de Monaco par les restructurations de rivage. Bull Ecol 9(3): 259-276.

MEINESZ A., VERLAQUE M., 1979. Note préliminaire concernant quelques expériences de repiquage de Caulerpa prolifera et de Zostera

noltii dans la zone de rejet de l'effluent thermique de la centrale électrique de Martigues-Ponteau (golfe de Fos, France). Rapp PV Réun Commiss internat Explor sci Médit. 25-26: 209-212.

- MEINESZ A., LAURENT R., 1980. Cartes de la limite inférieure de l'herbier de *Posidonia oceanica* dans les Alpes-Maritimes (France). Campagne Poséidon 1976. Ann Inst Océanogr 56(1): 45-54.
- MEINESZ A., ASTIER J.M., LEFEVRE J.R., 1981a. Impact de l'aménagement du domaine maritime sur l'étage infralittoral du Var, France (Méditerranée occidentale). Ann Inst océanogr 57(2): 65-77.
- MEINESZ A., CUVELIER M., LAURENT R., 1981b. Méthodes récentes de cartographie et de surveillance des herbiers de Phanérogames marines. Leurs applications sur les côtes françaises de la Méditerranée. *Vie Milieu* 31 (1): 27-34.
- MEINESZ A., ASTIER J.M., BODOY A., CRISTIANI G., LEFEVRE J.R., 1982. Impact de l'aménagement du domaine maritime sur l'étage infralittoral des Bouches-du-Rhône (France, Méditerranée occidentale). *Vie Milieu* 32(2): 115-124.
- MEINESZ A., LEFEVRE J.R., 1984. Régénération d'un herbier à *Posidonia oceanica* quarante années après sa destruction par une bombe dans la rade de Villefranche (Alpes-Maritimes). *In*: Boudouresque C.F., Jeudy de Grissac A., Olivier J. edits. *International Workshop* on *Posidonia oceanica beds*, GIS Posidonie publ., Fr., 1: 39-44.
- MEINESZ A., BOUDOURESQUE C.F., JEUDY DE GRISSAC A., LAMARE J.P., LEFÈVRE J.R., MANCHE A., 1985. Aménagement et préservation du milieu marin littoral en Région Provence-Alpes-Côte d'Azur: bilan et perspectives. *Colloque franco japonais* d Océanographie, Marseille, Fr., 1: 133-142.
- MEINESZ A., BOUDOURESQUE C.F., BIANCONI C.H., SIMONIAN M., 1987. Cartographie de la limite inférieure de l'herbier de Posidonies de la baie d Elbo. Trav sci Parc nat rég Rés nat Corse 13: 27-37.
- MEINESZ A., BOUDOURESQUE C.F., LEFEBVRE J.R., 1988. A map of the *Posidonia oceanica* beds of Marina d'Elbu (Corsica, Mediterranean). Mar Ecol, PSZN 9(3): 243-252.
- MEINESZ A., 1989. Impact d'un aménagement portuaire sur le domaine maritime de la Réserve des îles Lavezzi. Laboratoire Environnement Marin Littoral, Université de Nice-Sophia Antipolis, Fr.: 1-15.
- MEINESZ A., BELLONE E., 1989. Localisation des herbiers à *Posidonia oceanica* sur le parcours du câble EDF à immerger dans les eaux du Parc national de Port-Cros. EDF & GIS Posidonie, GIS Posidonie publ., Fr.: 1-6 + 2 pl. h.t.
- MEINESZ A., BELLONE E., ASTIER J.M., LEFÈVRE J.R., VITIELLO P., 1990a. Impact des aménagements construits sur le domaine maritime de la Région Provence-Alpes-Côte d'Azur. Délégation régionale à l'Architecture et à l'Environnement publ., Fr.: 1-38.
- MEINESZ A., CAYE G., LOQUES F., MACAUX S., 1990b. Analyse bibliographique sur la culture des Phanérogames marines. Posidonia Newsletter, Fr., (1): 1-67.
- MEINESZ A., GENOT I., HESSE B., 1990c. Données quantitatives sur les biocénoses littorales marines de Corse et impact de l'aménagement du littoral. Laboratoire Environnement Marin, Université de Nice-Sophia Antipolis, Fr.: 1-22.
- MEINESZ A., HESSE B., 1991. Introduction et invasion de l'algue tropicale Caulerpa taxifolia en Méditerranée Nord-occidentale. Oceanologica Acta 14(4): 415-426.
- MEINESZ A., CAYE G., LOQUES F., MOLENAAR H., 1991a. Restoration of damaged areas with transplantation of seaweeds and seagrasses in the Mediterranean : review and perspectives. *Oebalia* 17(1 suppl.): 131-142.
- MEINESZ A., LEFEVRE J.R., ASTIER J.M., 1991b. Impact of coastal development on the infralittoral zone along the Southern Mediterranean shore of continental France. *Mar Poll Bull* 23: 343-347.
- MEINESZ A., MOLENAAR H., BELLONE E., LOQUES F, 1992. Vegetative reproduction in *Posidonia oceanica* I. Effects of rhizome length and transplantation season in orthotropic shoots. *Mar Ecol*, *PSZNI* 13(2): 163-174.
- MEINESZ A., CAYE G., LOQUES F., MOLENAAR H., 1993. Polymorphism and development of *Posidonia oceanica* transplanted from different parts of the Mediterranean into the national Park of Port-Cros. *Bot mar* 36: 209-216.
- MEINESZ A., BELSHER T., THIBAUT T., ANTOLIC B., BEN MUSTAPHA K., BOUDOURESQUE C.-F., CHIAVERINI D., CINELLI F., COTTALORDA J.M., DJELLOULI A., EL ABED A., ORESTANO C., GRAU A.M., IVESA L., JAKLIN A., LANGAR H., MASSUTI-PASCUAL E., PEIRANO A., TUNESI L., DE VAUGELAS J., ZAVODNIK N., ZULIEJEVIC A., 2001a. The introduced green alga *Caulerpa taxifolia* continues to spread in the Mediterranean. *Biol I Inv* 3: 201-210.
- MEINESZ A., COTTALORDA J.M., CHIAVERINI D., VAUGELAS J. de, 2001b. Représentation cartographique de l'abondance de quelques algues et invertébrés du littoral de l'îlot Bagaud (Parc national de Port-Cros). Sci Rep Port Cros nat/ Park 18: 123-141.
- MENDEZ S., PERGENT G., PERGENT-MARTINI C., 1997. Impact of fish farming facilities on coastal ecosystems in the Mediterranean. In: Ozhan E. edit. Third International Conference on the Mediterranean Coastal Environment, MEDCOAST 97. MEDCOAST Secretariat, Middle East Technical University publ., Turkey: 197-211.
- MERCERON M., KEMPF M., 1995. Environmental quality of a salmonid farm site in a flushed area (Cherbourg, France). In: Quality in aquaculture. European Aquaculture Soc. Publ. 23: 258-259.
- MINER M.C., KEMPF M., 1999. Aquaculture et Environnement : Réglementation et pratique des élevages de poissons marins. Ifremer publ., Actes de Colloques 23: 1-188.
- MILAZZO M., BADALAMENTI F., RAGGIO S., CHEMELLO R., 2002. Effetti degli ancoraggi sulla prateria a *Posidonia oceanica* della riserva marina Isola di Ustica: dati preliminari. *Biol Mar Medit* 9(1): 686-689.
- MILAZZO M., BADALAMENTI F., CECCHERELLI G., CHEMELLO R., 2004. Boat anchoring on *Posidonia oceanica* beds in a marine protected area (Italy, Western Mediterranean): effect of anchor types in different anchoring stages. *J exp mar Biol Ecol* 299: 51-62.
- MILLET B., BLANC F., MORHANGE C., 2000. Modélisation numérique de la circulation des eaux dans le Vieux-Port de Marseille vers 600 ans avant J.-C. *Méditerranée* 1(2): 61-64.
- MODIGH M., LORENTI M., MAZZELLA L., 1998. Carbon assimilation in *Posidonia oceanica*: Biotic determinants. *Bot Mar* 41(3): 249-256. MOLENAAR H., CAYE G., LOQUES F., MEINESZ A., 1989. Transplantations de *Posidonia oceanica* réalisées à Cannes en juin 1989.
- Méditerranée 2000 & Laboratoire Environnement Marin Littoral, Univ. Nice-Sophia Antipolis, Fr.: 1-23. MOLENAAR H., MEINESZ A., 1991. Transplantations de la Phanérogame marine *Posidonia oceanica* à Cannes et analyse architecturale
- des rhizomes plagiotropes. Laboratoire Environnement Marin Littoral, Univ. Nice-Sophia Antipolis, Fr.: 1-37. MOLENAAR H., 1992. Etude de la transplantation de boutures de la Phanérogame marine *Posidonia oceanica* (L.) Delile. Modélisation de l'architecture et du mode de croissance. Thèse Doct. Sci. Vie, Univ. Nice Sophia Antipolis, Fr.: 1-221.
- MOLENAAR H., MEINESZ A., 1992a. Vegetative reproduction in *Posidonia oceanica* II. Effects of depth changes on transplanted orthotropic shoots. *Mar. Ecol.*, *PSZNI* 13(2): 175-185.
- MOLENAAR H., MEINESZ A., 1992b. Transplantations de *Posidonia oceanica* (L.) Delile à Cannes. Méditerranée 2000 & Laboratoire Environnement Marin Littoral, Univ. Nice-Sophia Antipolis, Fr.: 1-27.
- MOLENAAR H., MEINESZ A., 1992c. Valorisation biologique des fonds marins des Alpes-Maritimes. 1. Transplantation de *Posidonia* oceanica à Nice. Laboratoire Environnement Marin Littoral, Univ. Nice-Sophia Antipolis, Fr.: 1-25.

- MOLENAAR H., MEINESZ A., 1992d. Valorisation biologique des fonds marins des Alpes-Maritimes. 2. Transplantation de *Posidonia* oceanica à Cannes. Laboratoire Environnement Marin Littoral, Univ. Nice-Sophia Antipolis, Fr.: 1-12.
- MOLENAAR H., MEINESZ A., 1993. (1) Suivi de 2000 boutures de Posidonia oceanica transplantées à l'Est de la baie de Cannes. (2) Transplantations de Posidonia oceanica au centre de la baie de Cannes: effets de la disposition des boutures et incidence du substrat. Modélisation de l'architecture et du mode de croissance. Méditerranée 2000 & Laboratoire Environnement Marin Littoral, Univ. Nice-Sophia Antipolis, Fr.: 1-270.
- MOLENAAR H., MEINESZ A., CAYE G., 1993. Vegetative reproduction in *Posidonia oceanica* Survival and development in different morphological types of transplanted cuttings. *Bot mar* 36: 481-488.
- MOLENAAR H., 1994. Suivi de l'évolution de l'herbier de *Posidonia oceanica* endommagé lors de la pose de conduites "sous-marines" entre la pointe de la Croisette et l'île Sainte Marguerite (baie de Cannes). Méditerranée 2000 & Laboratoire Environnemenrt Marin Littoral, Univ. Nice-Sophia Antipolis, Fr.: 1-53 + 55 Ann.
- MOLENAAR H., MEINESZ A., 1995. Vegetative reproduction in *Posidonia oceanica* : survival and development of transplanted cuttings according to different spacings, arrangements and substrates. *Bot mar* 38: 313-322.
- MOLENAAR H., 2001. Impact de Caulerpa taxifolia sur les espèces et les écosystèmes méditerranéens : la compétition avec Posidonia oceanica In Gravez V., Ruitton S., Boudouresque C.F., Le Direac'h L., Meinesz A., Scabbia G., Verlaque M. edits. Fourth international workshop on Caulerpa taxifolia GIS Posidonie publ., Fr.: 220-231.
- MOLINIER R., PICARD J., 1952. Recherches sur les herbiers de Phanérogames marines du littoral méditerranéen français. Ann Inst océanogr 27(3): 157-234.
- MOLINIER R., PICARD J., 1953. Notes biologiques à propos d'un voyage d'études sur les côtes de Sicile. Ann Inst océanogr 28(4): 163-187 + 4 pl. h.t.
- MOLINIER R., PICARD J., 1954. Eléments de bionomie marine sur les côtes de Tunisie. Bull Stat Océanogr Salammbô 48: 3-47.
- MOLINIER R., PICARD J., 1956. Aperçu bionomique sur les peuplements marins littoraux des côtes rocheuses méditerranéennes de l'Espagne. Bull Trav publ Stat Aquic Pêche Castiglione 8: 251-268.
- MOLINIER R., 1960. Etude des biocénoses marines du Cap-Corse. Vegetatio 9: 121-312.
- MOLINIER R., PELLEGRINI M., 1966. Contribution à l'étude chimique des Phanérogames marines: composition en acides aminés des feuilles de Posidonies (*Posidonia oceanica* Delile). Médecine tropicale 26(4): 1-15.
- MORENO D., AGUILERA P.A., CASTRO H., 2001. Assessment of the conservation status of seagrass (*Posidonia oceanica*) meadows: implications for monitoring strategy and the decision-making process. *Biol Conservation* 102: 325-332.
- MORRI C., 1991. Présentation d'un indice synthétique pour l'évaluation de l'épiphytisme foliaire chez *Posidonia oceanica* (L.) Delile. *Posidonia Newsletter* 4(1): 33-37.
- MUNDAY B.W., ELEFTHERIOU A., KENTOURI M., DIVANACH P., 1994. Quantitative statistical analysis of the literature concerning the interaction of the environment and aquaculture Identification of gaps and lacks. J Appl Ichthyol. 10(4): 319-325.
- NÉDÉLEC H., BEDHOMME A.L., BOUDOURESQUE C.F., THÉLIN I., 1981. Prolongation du transect permanent de la baie de Port-Cros. Trav sci Parc nation Port Cros 7: 197-201.
- NÉDÉLEC H., VERLAQUE M., 1984. Alimentation de l'oursin Paracentrotus lividus (Lamarck) dans un herbier à Posidonia oceanica (L.) Delile en Corse (Méditerranée – France). In: Boudouresque C.F., Jeudy de Grissac A., Olivier J. edits. International Workshop on Posidonia oceanica beds, GIS Posidonie publ., Fr., 1: 349-364.
- NIÉRI M., BOUDOURESQUE C.F., AUGIER H., GRAVEZ V., 1986. Mise en place du système de surveillance de l'herbier de Posidonies de la baie du Prado (Marseille). Observations initiales. SOMICA & GIS Posidonie publ., Fr.: 1-64.
- NIÉRI M., MEINESZ A., MOLENAAR H., SLOECK O., 1991. Réimplantation de la Phanérogame Posidonia oceanica dans le golfe de Marseille (Bouches-du-Rhône). Ville de Marseille & GIS Posidonie publ., Fr.: 1-35.
- NIÉRI M., CHARBONNEL E., BOUDOURESQUE C.F., BERTRANDY M.C., FORET P., MEINESZ A., PERGENT G., PERGENT-MARTINI C., RICO-RAIMONDINO V., 1993a. Les objectifs du Réseau de Surveillance Posidonies mis en place en Région Provence-Alpes-Côte d'Azur. In: Boudouresque C.F., Avon M., Pergent-Martini C. edits. Qualité du milieu marin Indicateurs biologiques et physico chimiques GIS Posidonie publ., Fr: 73-87.
- NIÉRI M., GRAVEZ V., BOUDOURESQUE C.F., SINNASSAMY J.M., 1993b. Le système de surveillance de l'herbier à Posidonia oceanica de la baie du Prado (Marseille, France). Les carrés permanents. In: Colloque international sur la pollution marine Ville de Marseille publ., Fr.: 1-10.
- NODOT C., ASTIER J.M., TAILLIEZ P., TINÉ J., 1978. Etude d impact des aménagements littoraux du Mourillon sur l'herbier de Posidonia oceanica de la rade des Vignettes (Toulon, Var). Ann Soc Sci nat Archéol Toulon Var 30: 118-133.
- NOTEN T.M.P.A., 1983. Detached shoots of Zostera noltii Hornem. as a means of dispersal: a transplantation experiment. In: Proceedings internat Symp aquat Macrophytes. Nijmegen, Netherlands: 161-164.
- OLIVER G., 1993. "Touristification" du littoral Languedoc-Roussillon. In: Pour qui la méditerranée au 21^{eme} Siècle Le système littoral méditerranéen. Actes du colloque scientifique Okeanos, Maison de l'Environnement de Montpellier publ., Fr.: 187-189.
- ORTH R.J., 2000. Protection and restoration of seagrasses: addressing global concerns from a local perspective in Chesapeake Bay, USA. Biol mar Medit. 7(2): 401-404.
- OTT J., MAURER L., 1977. Strategies of energy transfer from marine Macrophytes to consumer levels: the *Posidonia oceanica* example. *In:* Keegan B.F., O'Céidigh P., Boaden P.J.S. edits. *Biology of benthic organisms* UK: 493-502.
- OTT J.A., 1980. Growth and production in Posidonia oceanica (L.) Delile. Mar Ecol , PZSN 1(1): 47-64.
- PAILLARD M., GRAVEZ V., CLABAUT P., WALKER P., BLANC J.J., BOUDOURESQUE C.F., BELSHER T., URSCHELER F., POYDENOT F., SINNASSAMY J.M., AUGRIS C., PEYRONNET J.P., KESSLER M., AUGUSTIN J.M., LE DREZEN E., PRUDHOMME C., RAILLARD J.M., PERGENT G., HOAREAU A., CHARBONNEL E., 1993. Cartographie de l'herbier de Posidonie et des fonds marins environnants de Toulon à Hyères (Var, France). Reconnaissance par sonar latéral et photographie aérienne. Notice de présentation. Ifremer & GIS Posidonie, GIS Posidonie publ., Fr.: 1-36 + 3 cartes h.t.
- PALING E.I., VAN KEULEN M., WHEELER K., PHILLIPS J., DYHRBERG R., 2001a. Mechanical seagrass transplantation on Success Bank, Western Australia. Ecol Engineering 16: 331-339.
- PALING E.I., VAN KEULEN M.; WHEELER K., PHILLIPS J., DYHRBERG R., LORD D.A., 2001b. Improving mechanical seagrass transplantation. Ecol. Engineering 18: 107-113.
- PALING E.I., VAN KEULEN M., WHEELER K.D., 2003. The influence of spacing on mechanically transplanted seagrass survival in a high wave energy regime. *Restor Ecol* 11: 56-61.
- PANAYOTIDIS P, BOUDOURESQUE C.F., MARCOT-COQUEUGNIOT J., 1981. Microstructure de l'herbier à *Posidonia oceanica* (Linnaeus) Delile. *Bot mar* 24(3): 115-124.

PANAYOTIDIS P., GIRAUD G., 1981. Sur un cycle de renouvellement des feuilles de *Posidonia oceanica* (Linnaeus) Delile dans le golfe de Marseille. *Vie Milieu* 31(2): 129-136.

PAP/CAR, 1996. Approche pour l'aménagement de zones côtières en relation avec l'aquaculture en Méditerranée. PAP-10/EAM/GL.1, Split, Croatie: 1-38.

PASKOFF R., 1993. Côtes en danger. Masson publ., Paris, Fr.: 1-250.

PASQUALINI V., PERGENT G., 1993. Mise en place d'un câble "sous-marin" entre Piantarella et Cala di Greco (Corse du Sud). Retour sur site 1993. CODIL & GIS Posidonie, GIS Posidonie publ., Fr.: 1-23 + Ann.

PASQUALINI V., PERGENT-MARTINI C., PERGENT G., 1995. Etude de la formation récifale de Saint-Florent (Haute-Corse). Cartographie et évolution. DIREN Corse & GIS Posidonie, GIS Posidonie publ., Fr.: 1-49.

- PASQUALINI V., 1997. Caractérisation des peuplements et types de fonds le long du littoral Corse (Méditerranée, France). Thèse Doct., Univ. Corse, Fr.: 1-190.
- PASQUALINI V., PERGENT-MARTINI C., FERNANDEZ C., PERGENT G., 1997a. The use of airborne remote sensing for benthic cartography: advantages and reliability. International J. Remote Sensing 18(5): 1167-1177.
- PASQUALINI V., PERGENT-MARTINI C., PERGENT G., 1997b. Mediterranean coastal resources management : the example of the Island of Corsica. In: Proceedings of the Fourth International Conference Remote sensing for marine and coastal environments, technology and applications Orlando, Florida, USA, 1: 632-640.

PASQUALINI V., PERGENT-MARTINI C., PERGENT G., 1999. Environmental impacts identification along corsican coasts (Mediterranean Sea) using image processing. Aquat Bot 65: 311-320.

PASQUALINI V., CLABAUT P., PERGENT G., BENYOUSSE L., PERGENT-MARTINI C., 2000. Contribution of side scan sonar to the management of Mediterranean littoral ecosystems. *Internat J Remote Sensing* 21(2): 367-378.

PASQUALINI V., PERGENT-MARTINI C., CLABAUT P., MARTEEL E., PERGENT G., 2001. Integration of aerial remote sensing, photogrammetry, and GIS technologies in seagrass mapping. *Photogrammetric Engineering and Remote Sensing* 67(1): 99-105.

PASQUALINI V., PERGENT-MARTINI C., PERGENT G., AGREIL M., SKOUFAS G., SOURBES L., TSIRIKA A., 2005. Use of SPOT 5 for mapping seagrasses : an application to *Posidonia oceanica Remote Sensing Environment* 94: 39-45.

PAUL O., VERLAQUE M., BOUDOURESQUE C.F., 1984. Etude du contenu digestif de l'oursin régulier *Psammechinus microtuberculatus* dans l'herbier à *Posidonia oceanica* de la baie de Port-Cros (Var, France). *In*: Boudouresque C.F., Jeudy de Grissac A., Olivier J. edits. *International Workshop on Posidonia oceanica beds*, GIS Posidonie publ., Fr., 1: 365-371.

PEARCE F., 1996. L'enjeu de l'eau. La Tour du Valat publ., Fr.: 1-82.

PEIRANO A., BIANCHI N.C., 1995. Decline of the seagrass *Posidonia oceanica* in response to environmental disturbance: a simulationlike approach off Liguria (NW Mediterranean Sea). *In: Proc. 30th European marine biological Symposium*, Southampton: 87-95.

PELLEGRINI D., 2000. I referimenti internazionali e le linee di indirizzo per una revisione del D.M.24.01.96. In: Atti Giornata 18 febbraio 1998, Dragaggi portuali qualità e gestione dei materiali ICRAM publ.: 19-37.

PELLISSIER E., 1853. Description de la régence de Tunis. In: Exploration scientifique de l'Algérie pendant les années 1840, 1841, 1842. Imprimerie impériale, Paris.

PÉRÉS J.M., PICARD J., 1963. Aperçu sommaire sur les peuplements marins benthiques entourannt l'île de Port-Cros. Terre Vie 110(4): 336-448.

PÉRÈS J.M., PICARD J., 1964. Nouveau manuel de bionomie benthique de la mer Méditerranée. Rec Trav Stat mar Endoume 31(47): 3-137.

PÉRÉS J.M., PICARD J., 1975. Causes de la raréfaction et de la disparition des herbiers de *Posidonia oceanica* sur les côtes françaises de la Méditerranée. *Aquat Bot* 1(2): 133-139.

PÉRÈS J.M., 1984. La régression des herbiers à *Posidonia oceanica In:* Boudouresque C.F., Jeudy de Grissac A., Olivier J. edits. *International Workshop on Posidonia oceanica beds,* GIS Posidonie publ., Fr., 1: 445-454.

PERGENT G., BOUDOURESQUE C.F., CROUZET A., 1983. Variations cycliques dans les écailles des rhizomes orthotropes de *Posidonia* oceanica Trav sci Parc nation Port Cros 9: 107-148.

PERGENT G., 1985. Floraison des herbiers à Posidonia oceanica dans la Région d'Izmir (Turquie). Posidonia Newsletter 1(1): 15-21.

PERGENT G., BOUDOURESQUE C.F., VADIER B., 1985. Etude préliminaire des herbiers à *Posidonia oceanica* (L.) Delile de la côte des Albères (Pyrénées-Orientales, France). Ann Inst océanogr., N.S., 61(2): 97-114.

PERGENT G., PERGENT C., 1985. Cartographie de l'herbier à Posidonia oceanica de la baie d'Urla-Iskele (Turquie). Rapp PV Réun Commiss internation Explor sci Médit 29(6): 231-234.

PERGENT G., PERGENT C., 1988. Localisation et état de l'herbier de Posidonies sur le littoral PACA : Bouches-du-Rhône. DRAE PACA & GIS Posidonie, GIS Posidonie publ., Fr.: 1-53.

PERGENT G., PERGENT C., FRANCOUR P., 1988. Cortiou état zéro. Reconnaissance des herbiers, mise en place d'un balisage. Ville de Marseille & GIS Posidonie publ., Fr.: 1-107.

PERGENT G., BEN MAIZ N., BOUDOURESQUE C.F., MEINESZ A., 1989a. The flowering of *Posidonia oceanica* over the past fifty years : a lepidochronological study. *In*: Boudouresque, C.F., Meinesz A., Fresi E., Gravez V. edits. *International workshop on Posidonia beds*. GIS Posidonie publ., Fr., 2: 69-76.

PERGENT G., BOUDOURESQUE C.F., CROUZET A., MEINESZ A., 1989b. Cyclic changes along *Posidonia oceanica* rhizomes (lepidochronology): present state and perspectives. *Mar Ecol PSZN* 10(3): 221-230.

PERGENT G., 1990a. Lepidochronological analysis of the seagrass *Posidonia oceanica* (L.) Delile: a standardized approach. *Aquat Bot* 37: 39-54.

PERGENT G., 1990b. Evaluation of *Posidonia oceanica* primary production using lepidochronological analysis : preliminary results. *Rapp PV Réun Comm internation Explor sci Médit* 32(1) B17: 6.

- PERGENT G., PERGENT-MARTINI C., 1990. Some applications of lepidochronological analysis in the seagrass *Posidonia oceanica Bot* mar 33: 299-310.
- PERGENT G., 1991a. La protection légale de la Posidonie en France: Un outil efficace. Nécessité de son extension à d'autres pays méditerranéens. In : Boudouresque C.F., Avon M., Gravez V. edits. Les Espèces Marines à Protéger en Méditerranée, Rencontres scientifiques de la Côte Bleue. GIS Posidonie publ., Fr., 2: 29-34.

PERGENT G., 1991b. Les indicateurs écologiques de la qualité du milieu marin en Méditerranée. Oceanis 17(4): 341-350.

PERGENT G., BOUDOURESQUE C.F., THÉLIN I., MARCHADOUR M., PERGENT-MARTINI C., 1991. Map of benthic vegetation and seabottom types in the harbour at Banyuls-sur-Mer (P-O., France). *Vie Milieu* 41(2-3): 165-168.

PERGENT G., PERGENT-MARTINI C., 1991. Leaf renewal cycle and primary production of *Posidonia oceanica* in the bay of Lacco Ameno (Ischia, Italy) using lepidochronological analysis. *Aquat Bot* 42: 49-66.

- PERGENT G., PERGENTMARTINI C., RICO-RAIMONDINO V., 1992. Evaluation of past primary production of *Posidonia oceanica* using lepidochronological data. *Rapp P V Réun Commiss internation Explor sci Médit* 33: 47.
- PERGENT G., ZAOUALI J., 1992. Analyse phénologique et lépidochronologique de *Posidonia oceanica* dans une lagune hyperhaline du Sud tunisien. *Rapp P V Réun Commiss internation Explor sci Médit* 33: 48.
- PERGENT G., KEMPF M., 1993. L'environnement marin côtier en Tunisie. 1. Rapport de synthèse. 2. Etude documentaire. 3. Ann. Agence Nationale pour la Protection de l'Environnement (Tunisie), Ifremer & GIS Posidonie, Fr.: 1-55 + 1-395 + 1-173.
- PERGENT G., SEMROUD R., BABA AHMED R., DELBAL F., KHATAL Y., REMILI A., ROBERT P., BOUDOURESQUE C.F., 1993. Données préliminaires sur la répartition et l'état de l'herbier à *Posidonia oceanica* de la Région d'El Kala (Algérie). Sci Rep Port Cros nation Park 15: 253-263.
- PERGENT G., ROMERO J., PERGENT-MARTINI C., MATEO M.A., BOUDOURESQUE C.F., 1994. Primary production, stocks and fluxes in the Mediterranean seagrass *Posidonia oceanica Mar Ecol Progr Ser* 106: 139-146.
- PERGENT G., PERGENT-MARTINI C., 1995. Indicateurs biologiques : Posidonies. In: Etat des connaissances opérationnelles sur la contamination et les indicateurs de pollution chimique toxique du milieu marin Résumé des contributions IARE publ., Fr.: 123-135.
- PERGENT G., PERGENT-MARTINI C., BOUDOURESQUE C.F., 1995. Utilisation de l'herbier à *Posidonia oceanica* comme indicateur biologique de la qualité du milieu littoral en Méditerranée: état des connaissances. *Mésogée* 54: 3-29.
- PERGENT G., RICO-RAIMONDINO V., PERGENT-MARTINI C., 1997. Fate of primary production in *Posidonia oceanica* meadows of the Mediterranean. *Aquat Bot* 59: 307-321.
- PERGENT G., MENDEZ S., PERGENT-MARTINI C., PASQUALINI V., 1999. Preliminary data on the impact of fish farming facilities on *Posidonia* oceanica meadows in the Mediterranean. *Oceanol Acta* 22(1): 95-107.
- PERGENT G., PERGENT-MARTINI C., 1999. Mercury levels and fluxes in Posidonia oceanica meadows. Environ Poll 106: 33-37.
- PERGENT G., PERGENTMARTINI C., 2000. Mission d'étude en Libye. Etude des formations végétales de la lagune de Frawà. Centre d'Activités Régionales pour les Aires Spécialement Protégées et Université de Corse publ.: 1-48 + Ann.
- PERGENT G., DJELLOULI A., HAMZA A.A., ETTAYEB K.S., EL MANSOURI A.A., TALHA F.M., HAMZA M.A., PERGENT-MARTINI C., PLATINI F., 2002a. Characterization of the benthic vegetation in the Farwà lagoon (Libya). *J Coastal Conservation* 8: 119-126.
- PERGENT G., PERGENT-MARTINI C., MIMAULT B., PASQUALINI V., 2002b. Impacts et tracés potentiels d'un câble d'alimentation électrique entre la Corse et la Sardaigne. Expertise biologique. EDF & Seagrass 2000, Seagrass 2000 publ., Fr.: 1-34.
- PERGENT G., PERGENT-MARTINI C., PASQUALINI V., 2003. Mise en place d'un câble sous-marin dans la Cala Sciumara : incidence des travaux et propositions pour réduire les impacts sur le milieu naturel. EDF & Seagrass 2000, Seagrass 2000 publ., Corte, Fr.: 1-22.
- PERGENT G., PERGENTMARTINI C., ABIVEN T., BOISSERY P., FILMONT P., HAUDEN S., MIMAULT B., PASQUALINI V., PATRONE J., sous presse. Use of *Posidonia* Meadows to Monitor the Mediterranean Coastline: The Corsican Monitoring Network. *In: Medcoast* 05 International Conference, Kusadasi (Turkie).
- PERGENT-MARTINI C., PERGENT G., 1990. Suivi de l'évolution de l'herbier à *Posidonia oceanica* après la mise en service de la station d'épuration de Marseille. LBMEB & GIS Posidonie, GIS Posidonie publ., Fr.: 1-58.
- PERGENT-MARTINI C., CENTENE L., FERNANDEZ C., PERGENT G., 1992a. Etude de site entre Piantarella et Cala di Greco (Corse du Sud) en vue de la mise en place d'un câble sous-marin. CODIL & GIS Posidonie, GIS Posidonie publ., Fr.: 1-41.
- PERGENT-MARTINI C., PERGENT G., ROMERO J., 1992b. The litter in the *Posidonia oceanica* ecosystem: preliminary results. *In*: Keegan B F. edit. *Space and time series Data analysis in coastal benthic Ecology* Commission of the European Communities publ.: 541-547.
- PERGENT-MARTINI C., PERGENT G., 1993. Impact of the Marseille sewage treatment plant after four years operation In Symposium international protection du milieu marin contre les pollutions urbaines, 20-22 octobre 1993, Marseille, Fr., 3: 1-3.
- PERGENT-MARTINI C., PERGENT C., RICO-RAIMONDINO V., 1993. Posidonia oceanica beds, a biological indicator of the marine environment quality. In: Pour qui la méditerranée au 21^{eme} Siècle Le système littoral méditerranéen. Actes du colloque scientifique Okeanos, Maison de l'Environnement de Montpellier publ., Fr.:207-212.
- PERGENT-MARTINI C., 1994. Impact d'un rejet d'eaux usées urbaines sur l'herbier à *Posidonia oceanica*, avant et après la mise en service d'une station d'épuration. Thèse Doctorat, Univ. Corté, Fr.: 1-191.
- PERGENT-MARTINI C., PERGENT G., 1994. Lepidochronological analysis in the Mediterranean seagrass *Posidonia oceanica*: state of the art and future developments. *Oceanol Acta* 17(6): 673-681.
- PERGENT-MARTINI C., RICO-RAIMONDINO V., PERGENT G., 1994. Primary production of *Posidonia oceanica* in the Mediterranean basin. Mar Biol 120: 9-15.
- PERGENT-MARTINI C., PASQUALINI V., PERGENT G., 1995a. Monitoring of *Posidonia oceanica* meadows near the outfall of the sewage treatment plant at Marseilles (Mediterranean France). *EARSeL Advances in Remote Sensing* 4(1): 128-134.
- PERGENT-MARTINI C., RICO-RAIMONDINO V., PERGENT G., 1995b. Nutriment impact on *Posidonia oceanica* seagrass meadows: preliminary data. *Mar Life* 5(2): 3-9.
- PERGENT-MARTINI C., PERGENT G., 1996. Spatio-temporal dynamics of *Posidonia oceanica* beds near a sewage outfall (Mediterranean - France). *In:* Kuo J., Phillips R.C., Walker D.I., Kirkman H. edits. *Seagrass Biology Proceedings of an International Workshop*. Univ. Western Australia publ.: 299-306.
- PERGENT-MARTINI C., RICO-RAIMONDINO V., PERGENT G., 1996. Impact des nutriments sur les herbiers à *Posidonia oceanica* Données préliminaires. J Rech océanogr 21(1-2): 35-39.
- PERGENT-MARTINI C., 1998. Posidonia oceanica: a biological indicator of past and present mercury contamination in the Mediterranean Sea. Mar Environm Res 45(2): 101-111.
- PERGENT-MARTINI C., CAMUGLIO V., PERGENT G., 1998. Impact des anciennes exploitations minières sur la contamination en métauxtraces du littoral Corse. Rapp PV Réun Comm Int Explor sci Médit 35: 572-573.
- PERGENT-MARTINI C., PERGENT G., FERNANDEZ C., FERRAT L., 1999. Value and use of *Posidonia oceanica* as a biological indicator. *In:* Land ocean interactions managing coastal ecosystems. Proc. MEDCOAST 99 - EMECS 99 Joint Conference. MEDCOAST, Middle East Technical Univ. Publ, Turkey, 1: 73-90.
- PERGENT-MARTINI C., 2000. Protection des habitats d'herbiers de Phanérogames marines de Méditerranée. Les études d'impacts en milieu marin. CAR/ASPTunis & EqEL publ.: 1-49 + Ann.
- PERGENT-MARTINI C., PASQUALINI V., 2000. Seagrass population dynamics before and after the setting up of a wastewater treatment plant. *Biol mar Medit* 7(2): 405-408.
- PERGENT-MARTINI C., PERGENT G., 2000. Are marine phanerogams a valuable tool in the evaluation of marine trace-metal contamination : example of the Mediterranean Sea ? Int J environm Poll 13(1-6): 126-147.
- PERGENT-MARTINI C., PASQUALINI V., PERGENT G., 2000. Impact de la station d'épuration de la ville de Marseille sur l'herbier à Posidonia oceanica du secteur de Cortiou. Contrat Ville de Marseille & GIS Posidonie, GIS Posidonie publ., Fr.: 1-36.

- PERGENT-MARTINI C., PASQUALINI V., PERGENT G., FERRAT L., 2002. Effect of a newly set up wastewater-treatment plant on a marine Phanerogam seagrass bed – a medium term monitoring program. *Bull mar Sci* 71(3): 1227-1236.
- PERGENTMARTINI C., LEONI V., PASQUALINI V., ARDIZZONE G.D., BALESTRI E., BEDINI R., BELLUSCIO A., BELSHER T., BORG J., BOUDOURESQUE C.F., BOUMAZA S., BOUQUEGNEAU J.M., BUIA M.C., CALVO S., CEBRIAN J., CHARBONNEL E., CINELLI F., COSSU A., DI MAIDA G., DURAL B., FRANCOUR P., GOBERT S., LEPOINT G., MEINESZ A., MOLENAAR H., MANSOUR H.M., PANAYOTIDIS P., PEIRANO A., PERGENT G., PIAZZI L., PIRROTTA M., RELINI G., ROMERO J., SANCHEZ-LIZASO J.L., SEMROUD R., SHEMBRI P., SHILI A., TOMASELLO A., VELIMIROV B., 2005. Descriptors of *Posidonia oceanica* meadows: use and application. *Ecol Indicators* 5: 213-230.

PHILLIPS D.J.H., 1994. Macrophytes as biomonitors of trace metals. Biomonitoring of Coastal Waters and Estuaries: 85-103.

- PHILLIPS R.C., 1980a. Responses of transplanted and indigenous *Thalassia testudinum* Banks *ex* Königs and *Halodule wrightii* Aschers. to sediment loading and cold stress. *Contrib Mar Sci.* 23: 79-87.
- PHILLIPS R.C., 1980b. Transplanting methods. In: Phillips R.C., McRoy K.P. edits. Handbook of seagrass biology an ecosystem perspective. Garland STMP press publ., New York, 4: 41-56.
- PIAZZI L., CINELLI F., 1995. Restoration of the littoral sea bottom by means of transplantation of cuttings and sprouts. In: Cinelli F., Fresi E., Lorenzi C., Mucedola A. edits. Posidonia oceanica, a contribution to the preservation of a major Mediterranean marine ecosystem. Revista marittima publ., Italy: 69-71.
- PIAZZI L., ACUNTO S., BALESTRI E., CINELLI F., 1996. Osservazioni preliminari sulla germinazione di semi e sviluppo in situ di piantine di Posidonia oceanica (L.) Delile. Inform bot ital 28: 61-66.
- PIAZZI L., BALESTRI E., MAGRI M., CINELLI F., 1998. Experimental transplanting of Posidonia oceanica (L.) Delile into a disturbed habitat in the Mediterranean Sea. Bot mar. 41: 593-601.
- PIAZZI L., ACUNTO S., CINELLI F., 1999. In situ survival and development of Posidonia oceanica (L.) Delile seedlings. Aquat Bot 63: 103-112.
- PIAZZI L., BALESTRI E., BALATA D., CINELLI F., 2000. Pilot transplanting experiment of *Posidonia oceanica* (L.) Delile to restore a damaged coastal area in the Mediterranean Sea. *Biol. mar. Medit.* 7(2): 409-411.
- PIAZZI L., CECCHERELLI G., MEINESZ A., VERLAQUE V., AKÇALI B., ARGYROU M., BALATA D., BALLESTEROS E., CINELLI F., D'ARCHINO R., JAVEL J., MIFSUD C., PALA D., PANAYOTIDIS P., PEIRANO A., PERGENT G., PETROCELLI A., RUITTON S., ZULJEVIC A., 2005. Invasion of *Caulerpa racemosa* (Caulerpales, Chlorophyta) in the Mediterranean Sea: the balance of thirteen years of spread. *Cryptogamie, Agologie* 26: 189-202.
- PICARD J., 1965a. Importance, répartition et rôle du matériel organique végétal issu des prairies de Posidonies. Rapp PV Réun Commiss internation Explor sci Médit 18(2): 91-92.
- PICARD J., 1965b. Recherches qualitatives sur les biocoenoses marines des substrats meubles dragables de la Région marseillaise. Thèse Doct., Univ. Aix-Marseille II, Fr.: 1-160 + 1 carte.
- PIOVETTI L., SERVE L., COMBAUT G., GADEL F., 1984. Analyse des substances phénoliques des restes de *Posidonia oceanica* (L.) Delile provenant de sédiments holocènes et de dépots actuels. *In*: Boudouresque C.F., Jeudy de Grissac A., Olivier J. edits. *International Workshop on Posidonia oceanica beds*, GIS Posidonie publ., Fr., 1: 137-144.
- PIRC H., 1983. Belowground biomass of *Posidonia oceanica* (L.) Delile and its importance to the growth dynamics. *In: International Symposium aquatic macrophytes*, Nijmegen: 177-181.
- PIRC H., WOLLENWEBER B., 1988. Seasonal changes in nitrogen, free amino acids, and C/N ratio in Mediterranean seagrasses. Mar Ecol., PSZNI. 9(2): 167-179.
- PLATINI F., 2000. La protection des habitats aux herbiers en Méditerranée. Rapport PNUE, PAM, CAR/ASP publ.: 1-65.
- PNUE, 1999. Rapport de l'atelier sur le tourisme et le développement durable en Méditerranée. UNEP, MCSD Blue Plan. MAP Technical Reports 126: 1-255 + Ann.
- PONT D., 1993. Les charges polluantes véhiculées par le Rhône et ses conséquences locales: de la nécessité d'un dispositif de suivi pérenne. Rencontres de l'Agence Régionale pour l'Environnement Provence Alpes Côte d'Azur, Fr., 5: 9-12.
- POR F.D., 1978. Lessepsian migrations The influx of Red Sea biota into the Mediterranean by way of the Suez canal. Springer Verlag publ., Berlin: x + 1-228.
- PORCHER M, 1984. Impact des mouillages forains sur les herbiers à Posidonia oceanica. In: Boudouresque C.F., Jeudy de Grissac A., Olivier J. edits. International Workshop on Posidonia oceanica beds, GIS Posidonie publ., Fr., 1: 145-148.
- PORCHER M., JEUDY DE GRISSAC A., 1985. Inventaire des mouillages forains autour de l'île de Porquerolles (Var, France). Posidonia Newsletter 1(1): 23-30.
- PORCHER M., 1987. Un exemple de technique de chantier visant à limiter les apports de sédiments fins sur les fonds marins sensibles. *Posidonia Newsletter* 1(2): 31-33.
- POTTIER J., 1929. Etude sur les possibilités d'utilisation des plantes marines tunisiennes pour la nourriture du bétail. Ann Inst océanogr 6(3): 321-362 + 8 pl. h.t.
- POULAIN M., 1996. Le mouillage forain dans le Parc national de Port-Cros. Impact sur les herbiers à *Posidonia oceanica* Mémoire DESS Ecosystèmes méditerranéens, Univ. Corse, Fr.: 1-62.
- PRAT N., 1993. Ecosystèmes perturbés du littoral méditerranéen espagnol et notes sur les conséquences éventuelles des transferts d'eau prévus pour leur écologie. Rencontres de l Agence Régionale pour l Environnement Provence Alpes Côte d Azur, Fr., 5: 56-61.
- PROCACCINI G., BUIA M.C., GAMBI M.C., PEREZ M., PERGENT G., PERGENT-MARTINI C., ROMERO J., 2003. Seagrass status and extent along the Mediterranean coasts of Italy, France and Spain. In: Green E.P., Short F.T., Spalding M.D. edits. World Atlas of Seagrass Present status and future conservation. University of California Press publ.
- PRUVOT G., 1894. Essai sur la topographie et la constitution des fonds sous-marins de la Région de Banyuls-sur-Mer, de la plaine du Roussillon au golfe de Rosas. Arch Zool exp gén 3: 599-672.
- RAC/SPA, 1999. Action plan for the conservation of marine vegetation in the Mediterranean Sea. United Nations Environment Programme, Mediterranean Action Plan publ., Tunisia: 1-48.
- RAMOGE, 2002. La défense des plages contre l'érosion dans la zone RAMOGE. RAMOGE publ., Monaco: 1-43.
- RAMOS A.A., 1990. The marine reserve of Nueva Tabarca island (Alicante, Spain). Management aspects. In: Atti del 1^{er} Convegno internazionale "Parchi marini del Mediterraneo, Aspetti naturalistici e gestionali", San Teodoro, 28-30 avril 1989, Italy: 107-117.
 RAMOS-ESPLÁ A.A., 1984. Cartografía de la pradera superficial de Posidonia oceanica en la Bahía de Alicante (SE, España). In: Boudouresque
- C.F., Jeudy de Grissac A., Olivier J. edits. International Workshop on Posidonia oceanica beds, GIS Posidonie publ., Fr., 1: 57-61. RAMOS-ESPLÁ A.A., MARTINEZ-PEREZ L., ARANDA A., GUILLEN J.E., SANCHEZ-JEREZ P., SANCHEZ-LIZASO J.L., 1993. Proteccion de
- la pradera de *Posidonia oceanica* (L.) Delile mediante arrecifes artificiales disuasorios frente a la pesca de arrastre ilegal; el caso de

El Campello (SE ibérico). Publ Espec Inst Esp Oceanogr 11: 431-439.

- RAMOS-ESPLÁ A.A., ARANDA A., GRAS D., GUILLEN J.E., 1994. Impactos sobre las praderas de Posidonia oceanica (L.) Delile en el SE español: necesidad de establecer herramientas de ordenamiento y gestión del litoral. In: Pour qui la méditerranée au 21^{eme} Siècle Villes des rivages et environnement littoral en Méditerranée. Actes du colloque scientifique Okeanos, Maison de l'Environnement de Montpellier publ., Fr.: 64-69.
- RAMOS-ESPLA A.A., McNEILL S.E., 1994. The status of marine conservation in Spain. Ocean and coastal Management 24: 125-138.
- RAMOS-ESPLÁ A.A., SEVA A.M., SANCHEZ LIZASO J.L., BAYLE J.T., 1997. Megabentos asociado a dos praderas de *Posidonia oceanica* (L.) Delile, 1813 del sureste ibérico con diferente grado de conservación. *Publ espec Inst esp Oceanogr* 23: 265-271.
- RAMOS-ESPLÁ A.A., GUILLEN J.E., BAYLE J.T., SANCHEZ-JÉREZ P., 2000. Artifical anti-trawling reefs off Alicante, South-Eastern Iberian Peninsula : evolution of reef block and set designs. In: Jensen A.C., Collins K.J., Lockwood A.P.M. edits. Artificial Reefs in European Seas Kluwer Academic publ.: 195-218.
- RAMOS-MARTOS A., RAMOS-ESPLÁ A.A., 1989. Utilization of acoustic methods in the cartography of the *Posidonia oceanica* bed in the bay of Alicante (SE Spain). *Posidonia Newsletter* 2(1): 17-19.
- RANIELLO R., PROCACCINI G., 2002. Ancient DNA in the seagrass Posidonia oceanica Mar Ecol Progr Ser 227: 169-173.
- REGIONE LIGURIA, 2000. Piano Territoriale di Coordinamento della Costa.
- RELINI G., 1972. Degradazione dell'ambiente marino ad opera di scarichi di corpi solidi. Alcune conseguenze economiche, estetiche ed ecologiche. In: Atti del II Simposio Nazionale sulla conservazione della natura, Bari, 26-30 Aprile 1972: 133-143.
- RELINI G., PEIRANO A., TUNESI L., ORSI-RELINI L., 1986. The artificial reef in the Marconi Gulf (Eastern Ligurian Riviera). FAO Fish Rep. 357: 95-103.
- RELINI G., 1992. Depauperamento e protezione della fauna marina italiana. Boll Mus Ist biol Univ Genova 56-57: 9-52.
- RELINI G., RELINI M., TORCHIA G., 1995. La barriera artificiale di Loano. Biol Mar Medit. 2(1): 21-64.
- RELINI G., 2000. The Loano artificial reef. In Artificial Reefs in European Seas. In: Jensen A.C., Collins K.J., Lockwood A.P.M. edits. Artificial Reefs in European Seas. Kluwer Academic publ.: 129-149.
- REUSCH T.B.H., 2001. New markers old questions : population genetics of seagrasses. Mar Ecol Progr Ser 211: 261-274.
- RIBERA G., COLOREU M., RODRIGUEZ-PRIETO C., BALLESTEROS E., 1997. Phytobenthic assemblages of Addaia bay (Menorca, Western Mediterranean): composition and distribution. *Bot mar* 40(6): 523-532.
- RICO-RAIMONDINO V., 1995. Contribution à l'étude des stocks et flux d'éléments dans les herbiers à Posidonia oceanica Thèse Doctorat Ecologie, Univ. Aix-Marseille II, Fr.: 1-248.
- RISMONDO A., CURIEL D., SOLAZZI A., MARZOCCHI M., CHIOZZOTTO E., SCATTOLIN M., 1995. Sperimentazione di trapianto di Fanerogame marine in Laguna di Venezia: 1992-1994. In: SITE Atti, Ital., 16: 699-701.
- RIVOIRE G., CERUTI C., 1989. Les dépôts de dragages. La législation sur la Posidonie et les sites de dumping ne sont pas respectés. L'exemple de la baie de La Ciotat. Rapport inédit: 1-9.
- ROBERT P., 1983. Dégradation de l'herbier de Posidonies dans la zone de mouillage organisé de la baie de Port-Cros. *Trav sci Parc nation* Port Cros 9: 195-197.
- ROBERT P., 1988. Etude du régime thermique des principales biocénoses marines benthiques du Parc national de Port-Cros (Var, France). Diplôme de Recherche universitaire, Univ. Aix-Marseille II, Fr.: 1-209.
- ROMÉO M., GNASSIA-BARELLI M., JUHEL T., MEINESZ A., 1995. Memorization of heavy metals by scales of the seagrass *Posidonia* oceanica, collected in the NW Mediterranean. *Mar Ecol Progr Ser* 120: 211-218.
- ROMERO J., 1986. Note sur une méthode d'évaluation de la densité des faisceaux dans les herbiers de Posidonies. Rapp PV Réun Commiss internation Explor sci Médit 30(2): 1-266.
- ROMERO J., PERGENT G., PERGENT-MARTINI C., MATEO M.A., REGNIER C., 1992. The detritic compartment in a *Posidonia oceanica* meadow : litter features, decomposition rates and mineral stocks. *Mar Ecol*, *PSZN* 13(1): 69-83.
- ROMERO J., PEREZ M., ALCOVERRO T., MATEO M.A., SANCHEZ-LIZASO J.L., 1998. Production ecology of *Posidonia oceanica* (L.) Delile meadows in Nueva tabarca marine Reserve: growth, biomass and nutriment stocks along a bathymetric gradient. *Oecologia aquatica* 11: 111-121.
- ROMERO J., 1999. Els herbassars submarins de la Mediterrània. Butlletí de la Secció de Ciènces naturals del Museu de Mataró, L Atzavara 8: 5-8.
- ROMERO J., 2004a. Las praderas de Fanerógamas marinas. La producción primaria y su destino. Caractérísticas de los restos de la planta. *In:* Luque A.A., Templado J. edits. *Praderas y bosques marinos de Andalucía* Consejería de Medio Ambiente, Junta de Andalucía publ., Sevilla: 74-81.
- ROMERO J., 2004b. Posidònia: els prats del fons del mar. La mirada del biòleg a un ecosistema mediterrani. Escola del Mar, Centre d'Estudis Marins de Badalona publ., Spain:1-159.
- ROMERO J., ALCOVERRO T., MARTÍNEZ-CREGO B., PÉREZ M., 2005. The seagrass *Posidonia oceanica* as a quality element under the Water Framework Directive: POMI,a multivariate method to assess ecological status of Catalan coastal waters. Working document of the POMI group, Univ. Barcelona and Centre d'Estudis Avançats de Blanes (CSIC), Spain: 1-14.
- ROMERO-MARTINENGO J., 1985. Estudio ecológico de las Fanerógamas marinas de la costa catalana: producción primaria de *Posidonia* oceanica (L.) Delile en las islas Medes. Tesis Doct., Univ. Barcelona, Spain: 1-261.
- ROS J., 1994. La santé de la Méditerranée. Pour la Science 202: 80-88.
- ROS J., 2003. The state of health of the Mediterranean Sea : study cases and considerations from the ecological point of view. In: Rodríguez-Prieto C., Pardini G. edits. The Mediterranean Sea an overview of its present stateand plans for future protection Servei de Publicacions de la Universitat de Girona publ.: 13-39.
- ROY D., DIVETAIN N., BERNARD G., COQUILLARD Y., BONHOMME P., GRAVEZ V., 1999. Prospection de trois systèmes de mouillage en rade d'Hyères – Analyse et recommandations pour une éventuelle installation en baie de Port-Cros. Parc national de Port-Cros & GIS Posidonie, GIS Posidonie publ., Fr.: 1-32.
- RUITTON S., BOUDOURESQUE C.F., 1994. Impact de Caulerpa taxifolia sur une population de l'oursin Paracentrotus lividus à Roquebrune-Cap Martin (Alpes-Maritimes, France). In: Boudouresque C.F., Meinesz A., Gravez V. edits. First international workshop on Caulerpa taxifolia,. GIS Posidonie publ., Fr.: 371-378.
- RUITTON S., CHIAVERINI D., 1997. Cartographie des fonds et de l'herbier de Posidonies entre les îles Sainte-Marguerite et Saint-Honorat pour le passage d'une canalisation d'eau potable (Alpes-Maritimes, France). Lyonnaise des Eaux & GIS Posidonie, GIS Posidonie publ., Fr.: 1-69 + 1 carte.
- RUITTON S., BOUDOURESQUE C.F., BONHOMME P., BERNARD G., CADIOU G., ESCOFFIER B., 2001a. Compétition entre l'algue introduite *Caulerpa taxifolia* et la Phanérogame marine *Posidonia oceanica* en Méditerranée Nord-occidentale: suivi pluriannuel de

deux carrés permanents. Conseil Régional PACA & GIS Posidonie, GIS Posidonie publ., Fr.: 1-45.

RUITTON S., CHARBONNEL E., BOUDOURESQUE C.F., MEINESZ A., BONHOMME P., BERNARD G., PATRONE J., KRUCZEK R., COTTALORDA J.M., BERTRANDY M.C., FORET P., BRICOUT P., RAGAZZI M., CADIOU G., THIBAUT T., 2001b. Le Réseau de Surveillance Posidonies de la Région Provence-Alpes-Côte d'Azur. Seconde partie: résultats du suivi 2000. Région PACA, Agence de l'Eau, DDE & GIS Posidonie, GIS Posidonie publ., Fr.: 1-129.

RUIZ J.M., PEREZ M., ROMERO J., 2001. Effects of fish farm loadings on seagrass (*Posidonia oceanica*) distribution, growth and photosynthesis. *Mar Poll Bull* 42(9): 749-760.

RUIZ J.M., ROMERO J., 2001. Effects of *in situ* experimental shading on the Mediterranean seagrass *Posidonia oceanica*. Mar Ecol Progr Ser 215: 107-120.

RUIZ J.M., ROMERO J., 2003. Effects of disturbances caused by coastal constructions on spatial structure, growth dynamics and photosynthesis of the seagrass *Posidonia oceanica Mar Poll Bull* 46: 1523-1533.

RUIZ-FERNANDEZ J.M., 2000. Respuesta de la Fanerógama marina *Posidonia oceanica* (L.) Delile a perturbaciones antrópicas. Mem. Doct. Biol., Univ. Murcia, Spain: 1-212.

RUSSO G.F., VINCI D., 1991. The hand towed net method in *Posidonia oceanica* beds: 1. A preliminary study on the sample size for gatsropod taxocene in a shallow stand. *Posidonia Newsletter* 4(1): 27-31.

SAIDANE A., DE WAELE N., VAN DE VELDE R., 1979. Contribution a l'étude du compostage de plantes marines en vue de la préparation d'un amendement organique et d'un substrat horticole. Bull Inst Nation Sci Techn Océanogr Pêche Salammbô 6(1-4): 133-150.

SALAT J., PASCUAL J., 2002. The oceanographic and meteorological station at L'Estartit (NW Mediterranean). In: Tracking long term hydrological change in the Mediterranean Sea. CIESM Workshop Series, 16: 29-32.

SALEN-PICARD C., BELLAN G., BELLAN-SANTINI D., ARLHAC D., MARQUET R., 1997. Changements à long terme dans une communauté benthique d'un golfe Méditerranée (golfe de Fos). Oceanologica Acta 20(1): 299-310.

SANDULLI R., 2004. Il ruolo degli indicatori biologici nella valutazione dello stato dell'ambiente marino. *Biol mar Medit* 11 (2): 185-192. SANCHEZ-JEREZ P., 1994. Degradación de las praderas de *Posidonia oceanica* (L.) Delile por la pesca de arrastre en El Campello (SE ibérico): influencia sobre la estructura de la comunidad animal asociada. Tesis Licenciatura, Univ. Alicante, Spain: 1-123.

SANCHEZ-LIZASO J.L., GUILLEN-NIETO J.E., RAMOS-ESPLA A.A., 1990. The regression of *Posidonia oceanica* meadows in El Campello (SE Spain). Rapp P V Réun Commiss internation Explor sci Médit 32(1): 7.

SÁNCHEZ-LISAZO J.L., RAMOS-ESPLÁ A.A., 1994. Incidencia de los herbívoros sobre la Fanerógama marina Posidonia oceanica en la reserva marina de Tabarca, España. Inv mar CICIMAR, Mex., 9(2): 103-108.

SÁNCHEZ-LIZASO J.L., 2004. Las praderas de Fanerógamas marinas. Estructura de las praderas. In: Luque A.A., Templado J. edits. Praderas y bosques marinos de Andalucía Consejería de Medio Ambiente, Junta de Andalucía publ., Sevilla: 71-73.

SAUVAGEAU C., 1890. Observations sur la structure des feuilles des plantes aquatiques (suite). J Bot Paris 4(10): 181-192.

SAUVAGEAU C., 1920. L utilisation des algues marines. Doin publ., Paris, Fr.: 1-400.

SAUVAGEAU C., 1927. Sur les problèmes du Giraudya Bull Stat biol Arcachon 24: 1-74.

SAPORTA G. de, MARION A.F., 1878. Révision de la flore heersienne de Gelingen. Mém Soc roy Sci Lettres Belg 41: 1-108 + 14 pl. h.t. SAVY S., 1987. Les prédateurs de Paracentrotus lividus (Echinodermata). In: Boudouresque C.F. edit. Colloque international sur Paracentrotus lividus et les oursins comestibles. GIS Posidonie publ., Fr.: 413-423.

SCHEMBRI P.J., 1995. The state of *Posidonia oceanica* (L.) Delile meadows in the Maltese Islands (Central Mediterranean). Rapp PV Réun Commiss internation Explor sci Médit 34; 1-123.

SDAGE, 2003. Connaissance et gestion de l'érosion du littoral. Guide technique n°9. Agence de l'Eau Rhône Méditerranée Corse, Fr.: 1-53. SEMROUD R., VERLAQUE M., CROUZET A., BOUDOURESQUE C.F., 1992. On a broad-leaved form of the seagrass *Posidonia oceanica* (Posidoniaceae) from Algiers. *Aquat Bot* 43: 181-198.

SEMROUD R., BOUMAZA S., BOUKORTT R., 1998. Surveillance de l'herbier à *Posidonia oceanica* (L.) Delile au large de la grande plage d'El Djarnila (baie de Bou-Ismaïl, Algérie). Secrétariat d'Etat à l'Environnement & ISMAL, Algeria: 1-49 + 13 pl. h.t.

SERI F, DE GARA L., CARETTO S., LEO L., SANTAMARIA P., 2004. Influence of an increased NaCl concentration on yield and quality of cherry tomato grown in Posidonia *Posidonia oceanica* (L) Delile). J Sci Food Agric 84: 1885-1890.

SHEPHERD S.A., MCCOMB A.J., BULTHUIS D.A., NEVERAUSKAS V., STEFFENSEN D.A., WEST R., 1989. Decline of seagrasses. In: Larkum A.W.D., McComb A.J., Shepherd S.A. edits. Biology of Seagrasses A treatise on the biology of seagrasses with special reference to the Australian Region Elsevier, Amsterdam: 346-393.

SHERIDAN P., MCMAHAN G., HAMMERSTROM K., PULICH W., 1998. Factors affecting restoration of *Halodule wrightii* to Galveston Bay, Texas. *Restor Ecol* 6.

SHORT F.T., WOLF J., JONES G.E., 1989. Sustaining eelgrass to manage a healthy estuary. In: Proc Sixth Symp Coastal and Ocean Management / ASCE. July 11-14, 1989, Charleston, SC: 3689-3706.

SIEGEL S., 1956. Nonparametric statistics for the behavorial sciences. McGraw-Hill book company inc. publ., New York: xvii + 1-312.

SILBERSTEIN K., CHIFFINGS A.W., McCOMB A.J., 1986. The loss of seagrass in Cockburn Sound, Western Australia. III. The effect of epiphytes on productivity of *Posidonia australis* Hook f. Aquat Bot 24: 355-371.

SMAT NEPTUNE ENVIRONNEMENT, 2000. Protection des fonds marins. Ancrages spécifiques Harmony. Smat Neptune Environnement, Fr.: 1-25.

SOLIS-WEISS V., ALEFFI F., BETTOSO N., ROSSIN P., 2004. Gli indicatori biologici nel benthos del Golfo di Trieste. Biol mar Medit 11(2): 351-354.

SOLTAN D., VERLAQUE M., BOUDOURESQUE C.F., FRANCOUR P., 2001. Changes in macroalgal communities in the vicinity of a Mediterranean sewage outfall after the setting up of a treatment plant. *Mar Poll Bull* 42(1): 59-70.

SOMASCHINI A., GRAVINA M.F., ARDIZZONE G.D., 1994. Polychaete depth distribution in a *Posidonia oceanica* bed (rhizome and "matte" strata) and neighbouring soft and hard bottoms. *Mar Ecol P S Z N I*. 15(2): 133-151.

SORDINA J.B., 1951. Marine plants of Greece and their use in agriculture. Prak Hell Hydrobiol Inst. 5: 73-124.

SOUGY A., 1996. Les Posidonies face aux attaques de la mer. Mer et Littoral 15: 45-47.

SPINA E., 2003. Le CTSN et l'environnement du CETACE. Centre Technique des Systèmes Navals, Direction des Centres d'Expertise et d'Essais, Fr.: 1-6.

STANBURY K.B., STARR R.M., 1999. Applications of Geographical Information Systems (GIS) to habitat assessment and marine resource management. *Oceanol Acta* 22(6): 6999-703.

STANLEY D.J., 1993. Nile delta : recent geological evolution and human impact. In: Pour qui la méditerranée au 21^{eme} Siècle Le système littoral méditerranéen. Actes du colloque scientifique Okeanos, Maison de l'Environnement de Montpellier publ., Fr.: 231-234.

STONER A.W., 1980. The role of seagrass biomass in organization of benthic macrofaunal assemblages. Bull Mar Sci. 30(3): 531-551.

- STOPPELLI N., PEIRANO A., 1996. Continuous flowering of *Posidonia oceanica* (L.) Delile in the Bay of Monterosso al Mare (SP) (Northwestern Mediterranea Sea). *Boll Mus Ist Biol Univ Genova* 60-61: 31-40.
- TÄCKHOLM V., DRAR M., 1954. Flora of Egypt. Vol. III. Angiospermae, part Monocotyledones : Liliaceae-Musaceae. Bull Fac Sci , Egypte, 30: i-xiv + 1-648.
- TAVIANI M., 2002. The Mediterranean benthos from late Miocene up to present : ten million years of dramatic climatic and geologic vicissitudes. Biol mar Medit 9(1): 445-463.
- TAVOSO M.A., 1997. La convention de Barcelone pour la protection de la mer Méditerranée contre la pollution et le développement durable. UNEP publ., Athens. *MAP technical reports Series*, 117: 1-102.
- TCHERNIA A., POMEY P., HESNARD A., COUVERT M., GIACOBBI M.F., GIRARD M., HAMON E., LAUBENHEIMER F., LECAILLE F., 1978. L'épave romaine de la Madrague de Giens (Var) (Campagnes 1972-1975). Fouilles de l'Institut d'archéologie méditerranéenne. *Gallia*, suppl.24: 1-122 + 42 pl. h.t.
- TEINTURIER B., 1993. Aménagement du littoral : uniformité ou diversité ? Equinoxe 45: 26-32.
- TEMPLADO J., 1984. Las praderas de Posidonia oceanica en el sureste español y sus biocénosis. In: Boudouresque C.F., Jeudy de Grissac A., Olivier J. edits. International Workshop on Posidonia oceanica beds, GIS Posidonie publ., Fr., 1: 159-172.
- TEMPLADO J., 2004. Las praderas de Fanerógamas marinas. Introducción. In: Luque A.A., Templado J. edits. Praderas y bosques marinos de Andalucía Consejería de Medio Ambiente, Junta de Andalucía publ., Sevilla: 57-59.
- TEMPLADO-GONZALEZ J., 1982. Moluscos de las formaciones de Fanerogamas marinas en las cosatas de Cabo de Palos (Murcia). Tesis doctoral, Universidad Complutense Madrid, Spain: 1-351.
- TERRADOS J., DUARTE C.M., 2000. Experimental evidence of reduced particle resuspension within a sagrass (*Posidonia oceanica* L.) meadow. J exp mar Biol Ecol 243: 45-53.
- THÉBAUDIN F., CADEAU L., 1987. Une prairie sous-marine, la Posidonie. Bibliothèque de Travail, Fr., 989: 1-28.
- THÉLIN I., BEDHOMME A.L., 1983. Biomasse des épiphytes des feuilles de Posidonia oceanica dans un herbier superficiel. Rapp PV Réun Commiss internation Explor sci Médit 28(3): 125-126.
- THÉLIN I., BOUDOURESQUE C.F., 1983. Longévité des feuilles de *Posidonia oceanica* dans un herbier de la baie de Port-Cros (Var, France). Rapp PV Réun Commiss internation Explor sci Médit 28(3): 115-116.
- THÉLIN I., BOUDOURESQUE C.F., 1985.
- THOMAS C., 1997. Les "macrodéchets" du littoral. Bull Inform Cedre 9 (1): 12-14.
- THORAUG A., 1974. Transplantation of the seagrass Thalassia testudinum König. Aquaculture 4: 177-183.
- THORHAUG A., AUSTIN C.B., 1976. Restoration of seagrass with economic analysis. Environm Conserv. 3: 259-367.
- THORHAUG A., 1979. The flowering and fruiting of restaured Thalassia beds: a preliminary note. Aquat Bot 6: 189-192.
- TIMOSHENKO A., 1996. From Stockholm to post-Rio : evolving towards sustainable development. Naturopa 80: 6-7.
- TOCCI C., 1996. Champs récifaux formant obstacle aux arts traînants. J Rech océanogr 21(1-2): 42-44.
- TOMÀS-NASH F., 2004. Herbivory in seagrass ecosystems. Population dynamics and trophic interactions in a *Posidonia oceanica* (L.) Delile meadow. Doctoral thesis, Univ.t Barcelona, Spain: 1-175.
- TORCHIA G., PALANDRINI G., CIMA C., RELINI M., RELINI G., 2000. La cartografia nello studio dell'invasione dell'alga Caulerpa taxifolia Biol mar Medit 7(1): 517-527.
- TOURRE Y., 2002. The North Atlantic Oscillation and the Arctic Oscillation. Medias Newsletter 13: 8-10.
- TRAER K., 1979. The consumption of *Posidonia oceanica* Delile by Echinoids at the isle of Ischia. *In: Proc Eur Colloquium on Echinoderms,* Brussels, 3-8 Sept. 1979: 241-244.
- TUNESI L., DIVIACCO G., MO G., 2001. Observations by submersible on the biocoenosis of the deep-sea corals off Portofino Promontory (Northwestern Mediterranean Sea). In: Martin Willison J.H. et al edits. Proc of First internation Symposium on Deep sea Corals. Ecology Action Centre and Nova Scotia Museum, Halifax, Nova Scotia: 76-87.
- TUNESI L., PICCIONE M.E., AGNESI S., 2002. Progetto pilota di cartografia bionomica dell'ambiente marino costiero della Liguria. Proposta di un Systema Informativo Geografico per la gestione di cartografie bionomiche e sedimentologiche. *Quaderno ICRAM* 2: 1-112.
- UNEP, 1996. Etat du milieu marin et littoral de la Région méditerranéenne. UNEP, MAP Technical reports 101: 1-148.
- UOTILA J., 1991. Metal contents and spread of fish farming sludge in Southwestern Finland. In: Mäkinen T. edit. Main aquaculture and environment: 121-126.
- VADON C., 1981. Les Brachyoures des herbiers de Posidonies dans la Région de Villefranche-sur-Mer: biologie, écologie et variations quantitatives des populations. Thèse Doctorat, Univ. Pierre Marie Curie, Fr.: 1-227 + ixii + 4 pl. h.t.
- VADON C., 1984. Ecologie et polychromatisme de Sirpus zariquieyi Gordon, 1953 (Crustacea, Brachyura, Pirimelidae) du littoral méditerranéen. Ann Inst océanoge, N.S., 60(1): 95-99.
- VAN KEULEN M., BOROWITZKA M.A., 2002. Comparison of water velocity profiles through morphologically dissimilar seagrasses measured with a simple and inexpensive current meter. Bull mar Sci 71 (3): 1257-1267.
- VAUGELAS J. de, TRASTOUR L., 2002. Cartographie des biocénoses marines de la Réserve du Larvotto, Monaco. Laboratoire Environnement Marin Littoral, Université de Nice-Sophia Antipolis, Fr.: 1-14.
- VELIMIROV B., WALENTA-SIMON M., 1992. Seasonal changes in specific growth rates, production and biomass of a bacterial community in the water column above a Mediterranean seagrass system. *Mar Ecol Prog Ser.* 80(2-3): 237-248.
- VERLAQUE M., 1981. Preliminary data on some Posidonia feeders Rapp PV Réun Commiss internation Explor sci Médit 27(2): 201-202.
- VERLAQUE M., NÉDÉLEC H., 1983. Biologie de Paracentrotus lividus (Lamarck) sur substrat rocheux en Corse (Méditerranée, France): alimentation des adultes. Vie Milieu 33(3-4): 191-201.
- VERLAQUE M., 1987. Relations entre *Paracentrotus lividus* (Lamarck) et le phytobenthos de Méditerranée occidentale. *In*: Boudouresque C.F. edit. *Colloque international sur Paracentrotus lividus et les oursins comestibles*. GIS Posidonie publ., Fr.: 5-36.
- VERLAQUE M., 1990. Relations entre Sarpa salpa (Linnaeus, 1758) (Téléostéen, Sparidae), les autres poissons brouteurs et le phytobenthos algal méditerranéen. Oceanol Acta 13(3): 373-388.
- VERLAQUE M., BERNARD G., 1997. Inventaire de la flore marine de la Principauté de Monaco. Rapport intermédiaire. Service Environnement, Ministère d'Etat de la Principauté de Monaco & GIS Posidonie, GIS Posidonie publ., Fr.: 1-37.
- VERLAQUE M., BOUDOURESQUE C.F., MEINESZ A., GRAVEZ V., 2000. The *Caulerpa racemosa* complex (Caulerpales, Ulvophyceae) in the Mediterranean Sea. *Botanica marina* 43: 49-68.
- VERLAQUE M., DURAND C., HUISMAN J.M., BOUDOURESQUE C.F., LE PARCO Y., 2003. On the identity and origin of the Mediterranean invasive *Caulerpa racemosa* (Caulerpales, Chlorophyta). *Eur J Phycol* 38: 325-339.
- VERLAQUE M., AFONSO-CARRILLO J., GILRODRIGUEZ M.C., DURAND C., BOUDOURESQUE C.F., LE PARCO Y., 2004. Blitzkrieg in a

marine invasion : Caulerpa racemosa var. cylindracea (Bryopsidales, Chlorophyta) reaches the Canary Islands (NE Atlantic). Biol Inv 6: 269-281.

- VELIMIROV B., 1984. Grazing of Sarpa salpa L. on Posidonia oceanica ans utilization of soluble compounds. In: Boudouresque C.F., Jeudy de Grissac A., Olivier J. edits. International Workshop on Posidonia oceanica beds, GIS Posidonie publ., Fr., 1: 381-387.
- VERNEAU N., THOMASSIN B., VIALE D., 1995. Modifications de l'herbier de Posidonies et des populations de poissons dans l'ombre portée sur le fond par des radeaux aquacoles installés dans le golfe d'Ajaccio (Corse-du-Sud, Méditerranée Nord-occidentale). J Rech Océanogr. 20(1-2): 33-41.

VIDEAU C., MERCERON M., 1992. Impact de la pisciculture marine intensive sur l'environnement. Revue bibliographique. Direction de l'Environnement et de l'aménagement littoral, Ifremer publ., Fr.: 1-106.

- VIGLIOLA L., 1998. Contrôle et régulation du recrutement des Sparidae (poissons, Téléostéens) en Méditerranée: importance des processus pré- et post-installation benthique. Thèse Doctorat Océanologie, Univ. Aix-Marseille II, Fr.: i-vii + 1-268.
- VIGLIOLA L., HARMELIN-VIVIEN M.L., BIAGI F., GALZIN R., GARCIA-RUBIES A., HARMELIN J.G., JOUVENEL J.Y., LE DIREACH-BOURSIER L., MACPHERSON E., TUNESI L., 1998. Spatial and temporal patterns of settlement among sparid fishes of the genus *Diplodus* in the Northwestern Mediterranean. *Mar Ecol Progr Ser* 168: 45-56.
- VILLELE X. de, VERLAQUE M., 1995. Changes and degradation in a *Posidonia oceanica* bed invaded by the introduced tropical alga *Caulerpa* taxifolia in the Northwestern Mediterranean. *Botanica marina* 38: 79-87.
- VIRNO-LAMBERTI C., 2000. Aree di sversamento e monitoraggio : un caso di studio nel Tirreno. In: Dragaggi portuali qualità e gestione dei materiali. Atti Giornata 18 febbraio 1998, ICRAM publ.: 91-98.
- WALKER D.I., PERGENT G., FAZI S., 2001. Seagrass decomposition. In: Short F.T., Coles R G. edits. *Global seagrass research methods*. Elsevier Scientific Publishers B.V., Amsterdam: 313-324.
- WARD T.J., 1987. Temporal variation of metals in the seagrass *Posidonia australis* and its potential as a sentinel accumulator near a lead smelter. *Mar Biol.* 95: 315-321.
- WARD T.J., 1989. The accumulation and effects of metals in seagrass habitats. *In*: Larkum A.W.D., McComb A.J., Shepherd S.A. edits. *Biology of seagrasses, Aquatic Plant Studies 2*. Elsevier publ.: 797-820.
- WARD D.H., MARKON C.J., DOUGLAS D.C., 1997. Distribution and stability of eelgrass bedes at Izembek Lagoon, Alaska. Aquat Bot 58: 229-240.
- WAYCOTT M., LES D.H., 2000. Current perspectives on marine Angiosperm evolution. Biol mar Medit 7(2): 160-163.

WEDDEL, 1877. Note sur les aegagropiles de mer. Actes Congr internation Bot , Amsterdam: 58-62.

WELLMAN C.H., OSTERLOFF P.L., MOHUDDIN U., 2003. Fragments of the earliest land plants. Nature 425: 282-285

- WEST R.J., LARKUM A.W.D., KING R.J., 1989. Regional studies seagrasses of Southeastern Australia. In: Larkum A.W.D., McComb A.J., Shepherd S.A. edits. Biology of Seagrasses A treatise on the biology of seagrasses with special reference to the Australian Region Elsevier publ., Amsterdam: 230-260.
- WILLSIE A., 1987. Structure et fonctionnement de la macrofaune associée à la "matte morte" et d'herbier vivant de *Posidonia oceanica* (L.) Delile: influence des facteurs abiotiques et biotiques. Thèse Océanologie, Univ. Aix-Marseille II, Fr.: 1-387.
- WITTMANN K., SCIPIONE M.B., FRESI E., 1981. Some laboratory experiments on the activity of the macrofauna in the fragmentation of detrital leaves of *Posidonia oceanica* (L.) Delile. *Rapp P V Réun Commiss internation Explor sci Médit* 27(2): 205-206.
- WITTMANN K.J., OTT J.A., 1982. Effects of cropping on growth in the Mediterranean seagrass *Posidonia oceanica* (L.) Delile. *Mar Ecol PSZN* 3(2): 151-159

WOLFF, T., 1976. Utilization of seagrass in the deep sea. Aquat Bot 2: 161-174.

WU R.S.S., 1995. The environmental impact of marine fish culture: Towards a sustainable future. Mar Poll Bull. 31(4-12): 159-166.

- ZALOKAR M., 1942. Les associations "sous-marines" de la côte adriatique au-dessous de Velebit. Bull Soc bot Genève 33: 171-195.
- ZUPI V., FRESI E., 1984. A study of the food web of the Posidonia oceanica ecosystem : analysis of the gut contents of Echinoderms. In: Boudouresque C.F., Jeudy de Grissac A., Olivier J. edits. International Workshop on Posidonia oceanica beds, GIS Posidonie publ., Fr., 1: 373-379.

NOTES		
5		
2		





NOTES

Conception : STILE LIBERO - www.stilelibero.mc

